

A MAIN CONJECTURE IN NON-COMMUTATIVE IWASAWA THEORY

DISSERTATION

ZUR ERLANGUNG DES AKADEMISCHEN GRADES EINES Doktors der Naturwissenschaften (Dr.rer.nat.)

von

Antonio Mejías Gil geboren in Madrid, Spanien

vorgelegt beim Fachbereich Mathematik Universität Duisburg-Essen

GUTACHTER:

PD DR. ANDREAS NICKEL PROF. DR. WERNER BLEY

Essen, 2022

Datum der mündlichen Prüfung: 11. Januar 2023

Abstract

We formulate a new equivariant Main Conjecture in Iwasawa theory of number fields and study its properties. This is done for arbitrary one-dimensional *p*-adic Lie extensions L_{∞}/K containing the cyclotomic \mathbb{Z}_p -extension K_{∞} of the base field. As opposed to existing conjectures in the area, no requirement that L_{∞}/K be abelian or that L_{∞} be totally real is imposed. We prove the independence of the Main Conjecture of essentially all of its parameters and explore its functorial behaviour. It is furthermore shown that, to a large extent, this new conjecture generalises existing ones of Burns, Kurihara and Sano and Ritter and Weiss, which enables us to deduce its validity in several cases.

Zusammenfassung

Wir formulieren eine neuartige äquivariante Hauptvermutung der Iwasawa-Theorie über Zahlkörpern und studieren deren Eigenschaften. Dies wird für beliebige eindimensionale p-adische Lie-Erweiterungen L_{∞}/K durchgeführt, welche die zyklotomische \mathbb{Z}_p -Erweiterung des Grundkörpers enthalten. Im Gegensatz zu bereits existierenden Vermutungen auf diesem Gebiet, setzen wir weder voraus, dass die zugrunde liegende Erweiterung L_{∞}/K abelsch ist, noch dass der Körper L_{∞} total reell ist. Wir zeigen, dass diese Hauptvermutung im Wesentlichen unabhängig von all ihren Parametern ist, und untersuchen ihr funktorielles Verhalten. Es wird weiterhin gezeigt, dass diese neue Vermutung weitestgehend bereits bestehende von Burns, Kurihara und Sano sowie Ritter und Weiss verallgemeinert. Dies erlaubt uns die Verifikation unserer Vermutung in etlichen Fällen. To my parents, Marysia and Gonza

Acknowledgements

At the risk of sounding trite, there are a number of people I am indebted to for their role in this thesis. I would like to take a few lines to thank them.

First and foremost, my supervisor Andreas. From the overall direction of the project to the smallest of details, your mathematical insight has been invaluable - and mathematics only accounts for half of our conversations during these four years. Thank you for your immense generosity, dedication, patience and encouragement. You have been the best advisor I could have asked for, and more.

Marysia, whom I met on the week I started my PhD and got engaged to on the year I finished it. Thank you for making this journey so much more meaningful, for your appreciation of my mathematical work consisting of drawing circles and for helping me realise that Iwasawa theory is a piece of cake compared to Polish grammar.

My parents, who have always done everything in their power to allow me to pursue this goal and have created the best possible environment for that and much more. This thesis is as much your achievement as it is mine. And my *bro* Gonza, without whom my life would be incomplete in so many ways.

My mathematical brothers, Nils and Bence, I thank for all the Iwasawa theory shared. To them and and to the rest of my friends in Essen, especially Chirantan and Gürkan, thank you for making this such an enjoyable place to be.

On a more academic note, I would like to thank Alexandre Maksoud, Dominik Bullach, Otmar Venjakob and Sören Kleine for insightful conversations about the project, as well as the ESAGA for creating an ideal space to do number theory in.

I cannot forget my Master's advisor, Daniel, without whom I might not have started a doctorate in Essen in the first place.

Last but not least, many thanks goes to Werner Bley for agreeing to review such a lengthy thesis.

Contents

In	troduction		iii
0	Notation, conventions and preliminaries		1
	0.1 Iwasawa algebras and modules		. 3
	0.2 Representations of finite groups	•	. 9
	0.3 Artin L -series		. 14
	0.4 Algebraic K-theory \ldots	•	. 17
1	Construction of the complex		23
	Setting A	•	. 23
	1.1 The global complex $\mathcal{T}^{\bullet}_{S\cup T}$	•	. 27
	1.2 The local complexes \mathcal{L}_v^{\bullet}	•	. 29
	1.3 Local-to-global maps	•	. 31
	1.4 The main complex $\mathcal{C}^{\bullet}_{S,T}$	•	. 33
	1.5 Description in terms of $R\Gamma$ -complexes	•	. 39
2	Formulation of the Main Conjecture		45
	Setting B		. 46
	2.1 An integral trivialisation		. 48
	2.2 Morphisms on finite level	•	. 52
	2.3 Regulators		. 63
	2.4 Evaluation maps		. 67
	2.5 The Main Conjecture	•	. 75
3	Basic properties		80
	3.1 Stark's conjecture and the choice of β	•	. 81
	3.2 Independence of the choice of parameters		. 90
	3.2.1 The choice of L	•	. 90
	3.2.2 The choice of α		. 92
	3.2.3 The choice of S and T	•	. 97
	3.3 Functoriality		. 111
	3.3.1 Change of L_{∞}		. 111

		Setting C	111			
	3.3	3.2 Change of K	121			
		Setting D	121			
	3.3	3.3 Reduction steps	129			
4	Compa	atibility with existing conjectures and new cases	138			
	4.1 Th	he conjecture of Burns, Kurihara and Sano	138			
	4.2 Th	he conjecture of Ritter and Weiss	153			
	Se	etting E \ldots	153			
A	Functo	oriality of refined Euler characteristics	164			
В	Detern	minant functors	169			
Bi	Bibliography					
Index						

Introduction

The study of class numbers occupies a central place in modern number theory. In this regard, much credit is due to Lamé's 1847 incorrect proof of Fermat's last theorem for prime exponent p, which failed to account for the failure of unique factorisation in what we recognise today as rings of integers of cyclotomic fields. Kummer realised that the argument could be salvaged by replacing the complex numbers with what he referred to as *ideal numbers* under the assumption that p does not divide the *class number* of $\mathbb{Q}(\zeta_p)$. This and many other examples showcase the usefulness of this invariant for solving Diophantine equations and, more broadly, understanding the arithmetic of more general rings than the integers.

In many instances, the problem of determining the class number h_K of an arbitrary number field Kis intractable in isolation. One of the most successful approaches is to instead consider its behaviour along a tower of such fields. The inception of this area of research can be traced back to the work of Iwasawa in the early 1960s, which resulted in an asymptotic law for the p-part of the class number of a number field K along its cyclotomic \mathbb{Z}_p -extension $K = K_0 \subseteq K_1 \subseteq \cdots \subseteq K_\infty$. This behaviour turned out to be governed by a certain module X_{nr} over the so-called Iwasawa algebra $\Lambda(\Gamma)$, a completed group ring isomorphic to the power series ring $\mathbb{Z}_p[[T]]$. This sparked interest in the study of this and other related modules, which were eminently algebraic in nature, and in the question of whether they could be constructed using analytic methods as well. Iwasawa postulated, in what came to be known as his Main Conjecture, that the characteristic polynomial of X_{nr} for the base field $K = \mathbb{Q}$ coincides, up to a unit of $\Lambda(\Gamma)$, with a particular power series arising from the p-adic L-function defined by Kubota and Leopoldt - an analytic object interpolating values of the Riemann zeta function.

Given the depth and richness of both disciplines, the idea of relating algebraic and analytic invariants seems, with the benefit of hindsight, extremely natural. The quintessential example of such a connection is the analytic class number formula

$$\lim_{s \to 1} (s-1)\zeta_K(s) = \frac{2^{r_1}(2\pi)^{r_2}h_K R_K}{w_K \sqrt{|d_K|}}.$$

The left-hand side is the residue at 1 of the Dedekind zeta function of the number field K, which coincides with Riemann's celebrated function when $K = \mathbb{Q}$. On the right-hand side are a number of algebraic invariants describing the arithmetic of K - most conspicuously, its class number h_K , which is often the most challenging magnitude to determine directly. The class number formula therefore offers an alternative, often effective method to compute it.

A vast generalisation of said formula comes in the form of the equivariant Tamagawa Number Conjecture (eTNC) proposed by Burns and Flach in [BF01]. Much research has been devoted to this conjecture, originally formulated for arbitrary motives. However, it is only Tate motives that concern this discussion, as they amount to certain extensions of number fields. In their case, the eTNC can intuitively be thought of as a global, finite-level version of a Main Conjecture - and yet making this connection precise is the subject of active research. On one hand, the limit of the *p*-part of the eTNC along an infinite tower should imply the validity of a Main Conjecture for that tower. On the other, the Main Conjecture for a given tower should yield a particular instance of the eTNC when passing to finite level. In practice, the latter is a more difficult problem due to *descent issues*. The article [BKS17] of Burns, Kurihara and Sano, one of the main inspirations for this work, explores precisely this question. More specifically, it is shown in it that the eTNC can indeed be recovered from a specific Main Conjecture of Gross. Another example of the significance of the eTNC is the fact that the Birch and Swinnerton-Dyer conjecture also arises as a particular case of it, namely when the motive in question arises from an elliptic curve.

Although it would be futile to try and list all existing Main Conjectures in Iwasawa theory of number fields, we point out a few milestones which may help put this text into perspective. For a much more authoritative survey on the topic, we refer to [PRS11]. Iwasawa's original conjecture described above was quickly extended to the case of the cyclotomic \mathbb{Z}_p -extension L_{∞} of a totally real finite Galois extension L of K. The Kubota-Leopoldt p-adic L-function still serves as the analytic object when L is an abelian extension of $K = \mathbb{Q}$, but in the general case it must be replaced by a new type of p-adic L-function constructed, independently, by Pierrette Cassou-Noguès ([Cas79]), Deligne and Ribet ([DR80]) and Barsky ([Bar78]). The earliest conjectures in this context were character-wise. In other words, they claimed a relation between a quotient of power series coming from the L-function associated to a single irreducible character χ of Gal(L/K), and one arising from the χ -part of some Iwasawa module. The first major breakthrough in this direction came in 1984, when Mazur and Wiles settled the abelian case with $K = \mathbb{Q}$ in [MW84]. Wiles then extended these techniques and gave a proof of the general totally real case six years later (cf. [Wil90]). A completely different approach would later be put forward by Rubin, who gave a proof of the Main Conjecture with base field \mathbb{Q} making use of the notion of an Euler system introduced by Kolyvagin and ideas of Thaine. This tool also allowed him to prove a Main Conjecture for abelian extensions of *imaginary quadratic* fields in [Rub91].

In the early 2000s, the first examples of *equivariant* Main Conjectures made an appearance - due in large part to the interest in the eTNC. In this context, the term "equivariant" alludes to the fact that information about *all characters* of some Galois group is considered simultaneously, by contrast to character-wise conjectures. One of the first examples of this is Burns and Greither's equivariant conjecture from [BG03], which they proved for abelian extensions of \mathbb{Q} and used to deduce the corresponding case of the eTNC. Shortly after, Ritter and Weiss formulated a Main Conjecture for arbitrary totally real number fields in [RW04] - with no abelianity requirement. A series of follow-up papers would culminate with a proof of their conjecture in the so-called $\mu = 0$ case (cf. [RW11]). Independently, Kakde formulated and proved in [Kak13] a Main Conjecture in the totally real, $\mu = 0$ case, which additionally allowed for \mathbb{Z}_p -extensions of higher rank. Nickel (cf. [Nic13]) and Venjakob (cf. [Ven13]) independently proved Kakde's conjecture to be equivalent to that of Ritter and Weiss. The case of imaginary quadratic base field initiated by Rubin did not stay isolated from this equivariant trend, with some cases of such conjectures being settled by Bley in [Ble06] and, more recently, Bullach and Hofer in [BH21], as an intermediate step towards proving new instances of the eTNC.

Research in this field remains very active. In 2017, Burns, Kurihara and Sano proposed the aforementioned Main Conjecture for abelian extensions of arbitrary number fields which, as a distinctive feature, is formulated in the language of determinant functors. We single out recent work of Dasgupta and Kakde essentially settling the the Brumer-Stark conjecture and hence of Hilbert's twelfth problem (see [DK20] and [DK21]); and the subsequent proof of an abelian Main Conjecture in remarkable generality by Johnston and Nickel in [JN20].

The aim of this work is to formulate a Main Conjecture which generalises, and in doing so also unifies, some of the above. Namely, our two conjectures of reference are those of Ritter and Weiss, and of Burns, Kurihara and Sano. There are essentially two aspects in favour this new conjecture's generality. Firstly, unlike in the conjecture of Ritter and Weiss, the field L_{∞} is not required to be totally real. As an immediate consequence, the existence of *p*-adic *L*-functions is not guaranteed, and must therefore be stated as part of the conjecture and its implications studied. A more subtle effect of this departure from the totally real case is that some of the relevant arithmetic modules are no longer torsion over the relevant Iwasawa algebra, which complicates the definition of an algebraic object in the conjecture. This is resolved through the use of refined Euler characteristics. Secondly, unlike in the conjecture of Burns, Kurihara, and Sano, we do not assume $\mathcal{G} = \text{Gal}(L_{\infty}/K)$ to be abelian. This has a profound impact on the algebraic machinery and representation theory involved. Furthermore, it prevents the conjecture from being stated in the language of classical determinant functors.

As far as generality is concerned, a special mention is due to Fukaya and Kato's article [FK06], which outlines a very general - and hence necessarily non-explicit - framework for the formulation of Main Conjectures for motives. However, the discussion below does not seem to be fully covered by the proposed broad motivic umbrella: the article in question concerns critical motives only, which in the number field case corresponds to the non-vanishing of the relevant L-value - but the analytic objects in our conjecture interpolate leading coefficients.

This work is structured as follows: In the preparatory chapter 0, we establish some notation and lay out the basic tools needed throughout the text in four different areas: Iwasawa algebras and their modules, representation theory of finite groups, Artin *L*-series and algebraic *K*-theory of rings. The last of these is especially useful for illustrating the general outline of the conjecture, since it is in the so-called *localisation sequence* of *K*-theory that the statement of the conjecture takes place. More specifically, two fundamental rings in our endeavours will be a certain Iwasawa algebra $\Lambda(\mathcal{G})$ and its total ring of fractions $\mathcal{Q}(\mathcal{G})$, which is Artinian semisimple. Then, letting $Z(\mathcal{Q}(\mathcal{G}))$ denote the centre of the latter, K-theory provides homomorphisms of abelian groups

$$Z(\mathcal{Q}(\mathcal{G}))^* \xleftarrow{\operatorname{nr}} K_1(\mathcal{Q}(\mathcal{G})) \xrightarrow{\partial} K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$$

known as the reduced norm and the boundary (or connecting) homomorphism. The strategy is to define an analytic and an algebraic element in the first and last terms, respectively, and conjecture that they both arise as images of the same zeta element in $K_1(\mathcal{Q}(\mathcal{G}))$.

The aim of chapter 1 is to construct the main complex of $\Lambda(\mathcal{G})$ -modules $\mathcal{C}_{S,T}^{\bullet}$, which constitutes the central algebraic object of the Main Conjecture. It arises as the cone of a morphism from local to global complexes, both of which are instances of the translation functor developed by Ritter and Weiss in [RW96]. As proved in section 1.5, the resulting complex is isomorphic in the derived category of $\Lambda(\mathcal{G})$ -modules to that employed by Burns, Kurihara and Sano in [BKS17] - so this chapter gives a new, arguably more explicit description of the previously known object. In the process of defining $\mathcal{C}_{S,T}^{\bullet}$, we will also become acquainted with a multitude of Iwasawa modules relevant to the sequel.

Chapter 2 is devoted to the formulation of the Main Conjecture and therefore occupies a distinguished place in the discussion. The first section recalls the definition of refined Euler characteristics and explains how one can be defined for our main complex by means of a certain map α . It is through the use of this machinery that $\mathcal{C}_{S,T}^{\bullet}$, together with a *trivialisation* constructed from that map, can be regarded as an element of the relative K_0 group $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$. Sections 2.2 and 2.3 delve into the definition of *regulators*, complex *p*-adic numbers that the special *L*-values need to be divided by before they can stand a chance of being interpolated. These regulators, based on the work of Tate on Stark's conjectures [Tat84], measure the difference between a finite-level map induced by α and the classical Dirichlet regulator map. The next section continues this analytic trend by studying how $Z(\mathcal{Q}(\mathcal{G}))$ decomposes as a direct product of quotient fields of power series rings by virtue of structural results of Ritter and Weiss. Only in the last section is the Main Conjecture finally stated, which takes place in two steps: first an *Interpolation Conjecture* asserting the existence of power series quotients in $Z(\mathcal{Q}(\mathcal{G}))$ which interpolate regulated special L-values; and then an equivariant Main Conjecture, which claims the existence of a zeta element in $K_1(\mathcal{Q}(\mathcal{G}))$ which is mapped to these analytic objects as well as to the refined Euler characteristic of $\mathcal{C}_{S,T}^{\bullet}$. A version of this conjecture with uniqueness is also formulated, in line with existing conjectures

The properties of the Main Conjecture are the subject of chapter 3, the longest by a substantial margin. The conjecture is shown to be unconditionally independent of all of its parameters - where we do not include the extension L_{∞}/K itself - with the possible exception of an arbitrarily chosen isomorphism $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$. Independence of the latter is only proved assuming Stark's conjecture as formulated by Tate. Nonetheless, we also prove a weaker, yet unconditional, form of independence of β . The remainder of the chapter explores functoriality properties. We first show that the conjecture for L_{∞}/K implies those for L'_{∞}/K and L_{∞}/K' with $L'_{\infty} \subseteq L_{\infty}$ and $K' \supseteq K$. Then the converse problem is studied, resulting in a characterisation of the conjecture for L_{∞}/K in terms of those for two different families of subextensions. Many particular features of the formulation of the Main Conjecture find their justification in the proofs in this chapter.

The final chapter draws a rigorous connection to the aforementioned conjectures of Ritter and Weiss and of Burns, Kurihara and Sano. In the first section, we show that our conjecture is equivalent to a modified version of that in [BKS17] in the cases when both can be formulated - that is, L_{∞}/K abelian and K_{∞}/K cyclotomic. This modification amounts to the replacement of a claim "for all characters" by the same one "for almost all characters" (meaning all but finitely many), a feature of our conjecture which is necessary for some of the results in chapter 3. Section 4.2 focuses instead on the conjecture of Ritter and Weiss. We show that, when L is a CM field containing a primitive p-th root of unity, the minus part of our Main Conjecture for L_{∞}/K holds if and only if that of Ritter and Weiss does for L_{∞}^+/K . It follows from these equivalence results that our conjecture holds whenever L_{∞}/K is abelian if the base field is either \mathbb{Q} or, under some additional restrictions, imaginary quadratic; and its minus part also does for the aforementioned type of CM extensions under some assumptions.

Two small appendices are included at the end in order not to interrupt the flow of the exposition. The first one contains a proof that refined Euler characteristics behave well under extension and restriction of scalars, a fact which is crucial for the functoriality proofs of section 3.3. The second one recalls some basic properties of determinant functors and shows two simple results which are used in the comparison to the conjecture of Burns, Kurihara and Sano.

Several directions for further research present themselves as natural continuations of this work. Ideally, the cyclotomic assumption should be done away with completely. For the vast majority of our discussion, it is enough for K_{∞}/K to satisfy the *weak Leopoldt conjecture* and for no finite places of K to split completely in L_{∞} - two properties which notably hold in the cyclotomic case. Removing these two assumptions would, however, require substantial changes. If, in addition, the rank-one *p*-adic Lie group $\mathcal{G} = \operatorname{Gal}(L_{\infty}/K)$ could be replaced by one of arbitrary higher rank, the resulting conjecture would in fact rank among the most general in the Iwasawa theory of number fields. Another, perhaps more relevant question, is that of the specific relation to the eTNC as already mentioned, this is the motivation behind several of the most recent Main Conjectures. However, this problem would also necessitate some new ideas beyond what is contained below. Lastly, we would be remiss not to point out the most evident of challenges: that of a general proof of the conjecture. Although hardly within reach at the moment, that is indeed the ultimate goal. For every Main Conjecture dreams of being proved, after all.

Chapter 0

Notation, conventions and preliminaries

This preparatory chapter summarises some properties of certain classical objects which will be featured repeatedly in the sequel: Iwasawa algebras, representations of finite groups, K-theory and Artin L-functions. The notation for many of these is also established here. We will be less detailed than in subsequent chapters and defer some of the explanations to other sources. None of the results in this chapter are original.

Some conventions and notation in place throughout the entire text are the following:

- $0 \in \mathbb{N}$.
- Rings have a unit element different from 0 but are not necessarily commutative. Given a ring R, we denote its centre by Z(R). Analogously, we denote the centre of a group G by Z(G).
- The left ideal generated by an element r of a ring R is denoted by $\langle r \rangle_R = \{sr : s \in R\}$, or simply $\langle r \rangle$ if the ring is clear from context.
- Given a ring R and a positive $n \in \mathbb{N}$, we denote the ring of n-by-n matrices over R by $M_n(R)$.
- Modules are left modules unless otherwise stated. A finite module is one of finite order, rather than a finitely generated one.
- All topological groups are assumed to be Hausdorff.
- Unadorned \otimes of modules always denotes $\otimes_{\mathbb{Z}}$. However, on basic tensors $m \otimes n \in M \otimes_R N$ we always omit the ring.
- If $\varphi \colon R \to S$ is a ring homomorphism and $f \colon M \to N$ is a homomorphism of (left) *R*-modules, then $S \otimes_{\varphi} f$ denotes the extension of f to $S \otimes_{\varphi} M \to S \otimes_{\varphi} N$ by the identity on S. We may write $S \otimes_R -$ instead of $S \otimes_{\varphi} -$ if φ is clear from context.
- If M is a module over a ring R, then M[0] denotes the cochain complex consisting of M in

degree 0 and trivial modules elsewhere; and for $i \in \mathbb{Z}$, M[i] = (M[0])[i] denotes the shift of M[0] by *i*, which consists of *M* placed in degree -i (not *i*) and trivial modules elsewhere.

- Given a profinite group G, we denote by G^{ab} its abelianisation that is, the quotient of G by the closure of the subgroup generated by all commutators. We also denote by $G^{ab}(p)$ the maximal abelian pro-p quotient of G.
- If G is a topological group, then $H \leq_o G$ (resp. $H \leq_c G$) denotes that H is an open (resp. closed) subgroup of G. The corresponding notation for normal subgroups is \leq_o, \leq_c .
- Given a field E of characteristic 0, we denote by E^c an algebraic closure of E and set $G_E = \text{Gal}(E^c/E)$.
- An (algebraic) **number field** is a finite extension of \mathbb{Q} , that is, a global field of characteristic 0. Given a rational prime p, a p-adic field is a finite extension of \mathbb{Q}_p , that is, a non-archimedean local field of characteristic 0. If K is a number field or a p-adic field, then \mathcal{O}_K denotes its ring of integers.
- Given a set of places Σ of a number field K and an algebraic extension L/K, the set of places of L above places in Σ is denoted by Σ(L). If the place w is above v, we may refer to w as a prolongation of v. In particular, {v}(L) denotes the set of prolongations of v to L.
- Given a set of places Σ of a number field K and a tower of algebraic extensions L/K'/K, we say that L/K' is unramified at Σ if it is so at every place of K' above Σ (i.e. at $\Sigma(K')$), and analogously for the terminology *completely split at* Σ . The phrase Σ -ramified means unramified outside Σ .
- Given a place v of a number field K, we denote the corresponding normalised absolute value by $|-|_v$. The completion of K with respect to $|-|_v$ is denoted by K_v .
- We denote the residue field of a number field at a non-archimedean place v by $\kappa(v)$ (the number field being implicit in v). The **(absolute) norm** $\mathfrak{N}(v)$ of v is defined as $|\kappa(v)|$.
- If K is a number field, L/K is a (not necessarily finite) Galois extension and w is a place of L, then $\operatorname{Gal}(L/K)_w$ denotes the decomposition group of L/K at w. If w is non-archimedean, we denote the corresponding inertia group by I_w (the extension being implicit). Thus, in our notation, $\operatorname{Gal}(L/K)_w/I_w \cong \operatorname{Gal}(\kappa(w)/\kappa(v))$ for non-archimedean places (where v is the place of K below w).
- Given an archimedean place v of a number field K and an algebraic extension L/K, we say that a prolongation w of v to L is **ramified** if it is complex and v is real. In other words, if v does not split completely in L/K. If L/K is Galois, this is in turn equivalent to $\text{Gal}(L/K)_w$ having order 2 (as opposed to being trivial).

We write \blacksquare at the end of proofs and \Box at the end of other roman-font environments unless they consist of a bulleted list.

0.1 Iwasawa algebras and modules

We shall first recall some basic properties of Iwasawa algebras and their modules. These objects are at the heart of the algebraic machinery necessary for the construction of our complex of interest and the formulation of the Main Conjecture. All of the results presented are well known and can be found in [NSW20], although a more specific reference will be given on several occasions.

Let p be a rational prime, E a p-adic field and G a profinite group. The ring of integers \mathcal{O}_E is Noetherian and local with maximal ideal generated by a uniformiser π . Furthermore, it is compact and complete with respect to its π -adic topology and $\mathcal{O}_E/\pi^n \mathcal{O}_E$ is finite of order a power of p for all $n \in \mathbb{N}$. The **Iwasawa algebra** of G with coefficients in \mathcal{O}_E is the completed group ring

$$\Lambda^{\mathcal{O}_E}(G) = \mathcal{O}_E[[G]] = \lim_{U \leq _o G} \mathcal{O}_E\left[G/U\right]$$

with transition maps induced by the projections $G/U \to G/V$ for $U \leq V$. Note that \mathcal{O}_E embeds canonically as a subring of $\Lambda^{\mathcal{O}_E}(G)$ and G embeds canonically as a subgroup of the units $\Lambda^{\mathcal{O}_E}(G)^*$. If G is finite, this Iwasawa algebra coincides with the usual group ring $\mathcal{O}_E[G]$. The superscript \mathbb{Z}_p is often omitted in $\Lambda(G) = \Lambda^{\mathbb{Z}_p}(G)$ and related objects such as $\Delta(G) = \Delta^{\mathbb{Z}_p}(G)$ (see below).

We endow $\Lambda^{\mathcal{O}_E}(G)$ with the topology determined by the basis of open subgroups

$$\left\{\Delta_{n,U} = \ker\left(\Lambda^{\mathcal{O}_E}(G) \twoheadrightarrow \mathcal{O}_{E_{\mathcal{I}}} \mathcal{O}_E\left[G_{\mathcal{I}}\right]\right) \colon n \in \mathbb{N}, U \leq_o G\right\}.$$

By a $\Lambda^{\mathcal{O}_E}(G)$ -module we always mean a (Hausdorff left) topological module, which amounts to a topological \mathcal{O}_E -module with a continuous G-action. Similarly, by a homomorphism of $\Lambda^{\mathcal{O}_E}(G)$ -modules we always mean a continuous such homomorphism. It is easy to show that topologically finitely generated modules and finitely generated modules coincide and they are compact; and that any abstract finitely generated $\Lambda^{\mathcal{O}_E}(G)$ -module can be endowed with a unique topology which makes it into a compact topological module (cf. [NSW20] section V§II). Compact modules over Iwasawa algebras are known as **Iwasawa modules**.

Given $H \leq_c G$, the **augmentation map** of $\Lambda^{\mathcal{O}_E}(G)$ with respect to H is the canonical continuous surjection $\operatorname{aug}_H \colon \Lambda^{\mathcal{O}_E}(G) \twoheadrightarrow \Lambda^{\mathcal{O}_E}(G/H)$. Its kernel $\Delta^{\mathcal{O}_E}(G,H)$ fits into the short exact sequence of $\Lambda(G)$ -modules

$$0 \to \Delta^{\mathcal{O}_E}(G, H) \to \Lambda^{\mathcal{O}_E}(G) \xrightarrow{\operatorname{aug}_H} \Lambda^{\mathcal{O}_E}(G/H) \to 0.$$

$$(0.1)$$

and is therefore a two-sided closed ideal known as the **augmentation ideal**. When H = G, we denote it simply by $\Delta^{\mathcal{O}_E}(G)$ and we write aug for aug_G . One has $\Delta^{\mathcal{O}_E}(G, H) = \Delta^{\mathcal{O}_E}(H)\Lambda^{\mathcal{O}_E}(G)$.

With H as above, the **module of** H-invariants M^H of a $\Lambda^{\mathcal{O}_E}(G)$ -module M is defined as the largest submodule with trivial H-action. It consists of all elements of M fixed by H. The **module of** H-coinvariants of M is the largest quotient module with trivial H-action, and it is given by

as the denominator is generated by elements of the form $\sigma m - m$ with $\sigma \in H, m \in M$. Both M^H and M_H inherit a natural $\Lambda^{\mathcal{O}_E}(G/H)$ -module structure (cf. equation (0.1)). The notation

for the module of coinvariants only features H, since it is irrelevant whether M is regarded as a $\Lambda^{\mathcal{O}_E}(G)$ -module or one over $\Lambda^{\mathcal{O}_E}(H)$ in the first place.

The only modules over Iwasawa algebras which will be of interest to us are compact and discrete ones. Every compact $\Lambda^{\mathcal{O}_E}(G)$ -module M coincides with the inverse limit of its modules of coinvariants

$$M = \lim_{U \leq {}_o G} M_U$$

and is also a pro-p group (cf. [NSW20] proposition 5.2.4). Conversely, every discrete $\Lambda^{\mathcal{O}_E}(G)$ -module N coincides with the direct limit of its modules of invariants

$$N = \varinjlim_{U \trianglelefteq_o G} N^U$$

and is also a *p*-torsion group (by which we mean every element has order a power of p). This points to some manner of duality between the two, which is formalised in the notion of **Pontryagin duality**. Given a locally compact (Hausdorff) abelian group A, we define its Pontryagin dual as

$$A^{\vee} = \operatorname{Hom}_{cts}(A, \mathbb{R}/\mathbb{Z})$$

where the unit circle \mathbb{R}/\mathbb{Z} is endowed with the quotient topology of the usual topology on \mathbb{R} . The compact-open topology (cf. [NSW20] section I§1) makes A^{\vee} into a locally compact abelian group, and thus $-^{\vee}$ induces a contravariant autoduality on the category of locally compact abelian groups. It restricts to a contravariant duality between the categories of compact abelian groups and discrete abelian groups.

If M is a locally compact $\Lambda^{\mathcal{O}_E}(G)$ -module, then M^{\vee} inherits a natural $\Lambda^{\mathcal{O}_E}(G)$ -module structure with $o \in \mathcal{O}_E$ acting as (of)(m) = f(om) and $\sigma \in G$ acting as $(\sigma f)(m) = f(\sigma^{-1}m)$. This action is compatible with Pontryagin duality, which therefore induces a contravariant autoduality on the category of locally compact $\Lambda^{\mathcal{O}_E}(G)$ -modules and a contravariant duality between compact and discrete $\Lambda^{\mathcal{O}_E}(G)$ -modules. In particular, it is exact on short exact sequences of $\Lambda^{\mathcal{O}_E}(G)$ -modules.

It is easy to verify that, for $U \leq_o G$, one has natural isomorphisms $(M_U)^{\vee} \cong (M^{\vee})^U$ for M compact; and $(N^U)^{\vee} \cong (N^{\vee})_U$ for N discrete. Note that, since compact and discrete $\Lambda^{\mathcal{O}_E}(G)$ -modules are pro-p and p-torsion respectively, the functor $-^{\vee}$ coincides with $\operatorname{Hom}_{cts}(-, \mathbb{Q}_p/\mathbb{Z}_p)$ on these two classes of modules, where $\mathbb{Q}_p/\mathbb{Z}_p$ is the Prüfer group.

The usual notion of the tensor product is often not sufficient in the theory of Iwasawa modules. Suppose, for instance, that M is a compact right $\Lambda^{\mathcal{O}_E}(G)$ -module and N is a compact left one. If both modules have infinitely many open neighbourhoods of 0 and we choose two countable families of nested ones $M \supseteq U_0 \supseteq U_1 \supseteq \cdots$ and $N \supseteq V_0 \supseteq V_1 \supseteq \cdots$ and two families of elements $\{m_i \in U_i\}_{i \in \mathbb{N}}$ and $\{y_i \in V_i\}_{i \in \mathbb{N}}$, then $\sum_{i=0}^{\infty} x_i \otimes y_i$ will in general not converge to an element in $M \otimes_{\Lambda^{\mathcal{O}_E}(G)} N$ even though the terms become arbitrarily small. This is addressed by introducing the **complete tensor product**

$$M \widehat{\otimes}_{\Lambda^{\mathcal{O}_E}(G)} N = \lim_{U,V} M / U \otimes_{\Lambda^{\mathcal{O}_E}(G)} N / V,$$

where M and N are compact as above and U (resp. V) runs over the compact open submodules of M (resp. N). This is itself a compact \mathcal{O}_E -module with respect to the inverse limit topology, a basis of which is given by the images of the subgroups $M \otimes V + U \otimes N$ with U and V as above (it is in fact the completion of $M \otimes_{\Lambda^{\mathcal{O}_E(G)}} N$ with respect to that topology). There is also a natural monomorphism $M \otimes_{\Lambda^{\mathcal{O}_E(G)}} N \hookrightarrow M \otimes_{\Lambda^{\mathcal{O}_E(G)}} N$. These and other properties of the construction can be found in [NSW20] section V§2. Of particular relevance to us is the fact that, if M or N is finitely presented, then $M \otimes_{\Lambda^{\mathcal{O}_E(G)}} N$ is already complete and therefore the natural monomorphism from before is in fact an equality

$$M \otimes_{\Lambda} \mathcal{O}_{E(G)} N = M \widehat{\otimes}_{\Lambda} \mathcal{O}_{E(G)} N.$$

$$(0.3)$$

The complete tensor product allows us to define a **compact induction** for Iwasawa modules. Namely, if $H \leq_c G$ is a closed subgroup and M is a compact left $\Lambda^{\mathcal{O}_E}(H)$ -module, then we let

$$\operatorname{Ind}_{H}^{G} M = \Lambda^{\mathcal{O}_{E}}(G) \widehat{\otimes}_{\Lambda^{\mathcal{O}_{E}}(H)} M.$$

This is itself a compact left $\Lambda^{\mathcal{O}_E}(G)$ -module (the action being on the first term) with the above topology, and thus compact induction defines an additive functor from the category of compact left $\Lambda^{\mathcal{O}_E}(H)$ -modules to that of compact left $\Lambda^{\mathcal{O}_E}(G)$ -modules (with continuous homomorphisms in both cases).

Suppose N is already a compact left $\Lambda^{\mathcal{O}_E}(G)$ -module and $f: M \to N$ is a $\Lambda^{\mathcal{O}_E}(H)$ -homomorphism with M as above (where N has restricted scalars). Then f induces a natural homomorphism of $\Lambda^{\mathcal{O}_E}(G)$ -modules

$$\operatorname{Ind}_{H}^{G} f \colon \operatorname{Ind}_{H}^{G} M \to N. \tag{0.4}$$

This should not be confused with the homomorphism $\operatorname{Ind}_{H}^{G}M \to \operatorname{Ind}_{H}^{G}N$ given by functoriality, which is usually denoted by $\operatorname{Ind}_{H}^{G}f$ as well. We will clarify what $\operatorname{Ind}_{H}^{G}f$ refers to wherever ambiguity exists.

The following properties of induction will be used on occasion:

Lemma 0.1.1. Let G be a profinite group, $H \leq_c G$ a closed subgroup and M a compact left $\Lambda^{\mathcal{O}_E}(H)$ -module. Then

- $i) \operatorname{Ind}_{H}^{G} M \cong \varprojlim_{U \trianglelefteq_{o} G} \operatorname{Ind}_{H/H \cap U}^{G/U} M_{H \cap U} \text{ as } \Lambda^{\mathcal{O}_{E}}(G) \text{-modules.}$
- ii) $(\operatorname{Ind}_{H}^{G}M)^{\vee} \cong \operatorname{Map}_{cts,H}(G, M^{\vee})$ as $\Lambda^{\mathcal{O}_{E}}(G)$ -modules, where a continuous map $f: G \to M^{\vee}$ belongs to $\operatorname{Map}_{cts,H}(G, M^{\vee})$ if $f(\tau\sigma) = \tau f(\sigma)$ for all $\tau \in H, \sigma \in G$. The G-action on $\operatorname{Map}_{cts,H}(G, M^{\vee})$ is given by $(\sigma f)(\sigma') = f(\sigma'\sigma)$.

Both facts can be found in [NSW20] section XI§3. Since a compact $\Lambda^{\mathcal{O}_E}(H)$ -module M coincides with $\varprojlim_{U \leq _o H} M_U$, point i) in the lemma can be worded as: *induction commutes with inverse limits*. This will become meaningful in the introduction to chapter 1.

In the rest of this section we consider only the case $E = \mathbb{Q}_p$, although most of the following facts still hold for any E as above. Turning our attention briefly to cohomological considerations, let M be a discrete $\Lambda(G)$ -module. One can construct **cohomology groups** $H^i(G, M)$ for $i \geq 0$ in several equivalent ways, for instance as the right derived functors of the G-invariants functor $-^G \cong \operatorname{Hom}_{\Lambda(G)}(\mathbb{Z}_p, -)$ in the category of discrete $\Lambda(G)$ -modules (which has sufficiently many injectives). Various definitions can be found in [NSW20] sections I§2, II§5 and V§2. The resulting cohomology groups are canonically isomorphic to the direct limit $H^i(G, M) \cong \varinjlim_{U \leq _{o}G} H^i(G/U, M^U)$ of classical cohomology groups and have a natural discrete $\Lambda(G)$ -module structure. A short exact sequence $M' \hookrightarrow M \twoheadrightarrow M''$ of discrete $\Lambda(G)$ -modules induces the usual long exact cohomology sequence

$$0 \to (M')^G \to M^G \to (M'')^G \to H^1(G, M') \to H^1(G, M) \to H^1(G, M'') \to H^2(G, M') \to \cdots$$

$$(0.5)$$

of discrete $\Lambda(G)$ -modules.

Cohomology groups find their dual notion in the homology groups $H_i(G, N)$ given by the left derived functors of the *G*-coinvariants functor $-_G \cong \mathbb{Z}_p \widehat{\otimes}_{\Lambda(G)}$ – on a compact module *N*. These are again compact modules and satisfy the following relation:

Lemma 0.1.2. Let G be a profinite group and N a compact left $\Lambda(G)$ -module. Then there exist functorial isomorphisms of $\Lambda(G)$ -modules

$$H^i(G, N^{\vee}) \cong H_i(G, N)^{\vee}$$

for all $i \geq 0$.

For a proof, see [NSW20] theorem 2.6.9.

When studying Iwasawa modules which are arithmetic in nature, the notion of Tate twists is sometimes relevant. Let L/K be a Galois extension of fields of characteristic 0 such that L contains the group $\mu_{p^{\infty}}$ of all *p*-power roots of unity in some algebraic closure of K. Then the *p*-adic cyclotomic character of G is defined as the unique group homomorphism

$$\chi_{\text{cyc}} \colon \operatorname{Gal}(L/K) \to \mathbb{Z}_p^*$$

such that, for all $\sigma \in \text{Gal}(L/K)$, one has $\sigma(\zeta) = \zeta^{\chi_{\text{cyc}}(\sigma)}$ for all $\zeta \in \mu_{p^{\infty}}$ - which makes it automatically continuous. In other words, χ_{cyc} captures the action of Gal(L/K) on the *p*-power roots of unity. It is injective if and only if $L = K(\mu_{p^{\infty}})$. Surjectivity is equivalent to $K \cap \mu_{p^{\infty}} = \{1\}$ if *p* is odd, and to $K \cap \mu_{p^{\infty}} = \{\pm 1\}$ if p = 2.

Suppose now that G = Gal(L/K) is as above and M is a $\Lambda(G)$ -module. For $i \in \mathbb{Z}$, the *i*-th Tate twist of M, denoted by M(i), is defined as the \mathbb{Z}_p -module M endowed with the G-action

$$\sigma \cdot^{i} m = \chi^{i}_{\rm cyc}(\sigma) \sigma m,$$

with $\chi_{\text{cyc}}(\sigma)^i \sigma \in \Lambda(G)$ acting on $m \in M$ via the original action. It follows immediately that M(0) = M and M(i)(j) = M(i+j) = M(j)(i) for all $i, j \in \mathbb{Z}$. If $\mathbb{Q}_p/\mathbb{Z}_p$ and \mathbb{Z}_p are given the trivial G-action, then

$$\mathbb{Q}_{p/\mathbb{Z}_p}(1) \cong \mu_{p^{\infty}} = \varinjlim_n \mu_{p^n} \quad \text{and} \quad \mathbb{Z}_p(1) \cong \varprojlim_n \mu_{p^n}, \tag{0.6}$$

where μ_{p^n} is the p^n -torsion of $\mu_{p^{\infty}}$. Pontryagin duality changes the sign of Tate twists: if M is a locally compact $\Lambda(G)$ -module, then $M(i)^{\vee}$ is canonically isomorphic to $M^{\vee}(-i)$. This comes from inversion of the G-action on duals: $(\sigma f)(m) = f(\sigma^{-1}m)$.

We now specialise to the *classical case* $G = \Gamma \cong \mathbb{Z}_p$, $E = \mathbb{Q}_p$. There is a fundamental non-canonical isomorphism of topological \mathbb{Z}_p -algebras

$$\Lambda(\Gamma) \cong \mathbb{Z}_p[[T]],\tag{0.7}$$

where $\mathbb{Z}_p[[T]]$ (the ring of formal power series) is endowed with the $\langle p, T \rangle$ -adic topology (see [NSW20] proposition 5.3.5 for a proof). This isomorphism arises from the choice of a topological generator γ of Γ (which we shall always write multiplicatively) by mapping $\gamma \to T + 1$. The diagram

with $w_i = (T+1)^{p^i} - 1$ then commutes and yields an isomorphism on inverse limits. We denote the successive quotients of the w_i for $i \ge 1$ by $\xi_i = w_i/w_{i-1}$, all of which are irreducible in $\mathbb{Z}_p[T]$ and hence, although not obvious, in $\mathbb{Z}_p[[T]]$. We also set $\xi_0 = w_0 = T$. The ξ_i are known as **cyclotomic polynomials**. Both w_i and ξ_i are examples of Weierstrass polynomials, which are defined as monic polynomials whose non-leading coefficients lie in $p\mathbb{Z}_p$ - that is, they become a power of T after projecting to $\mathbb{F}_p[[T]]$. A recurring idea in chapter 2 will be the fact that, under the isomorphism $\Lambda(\Gamma) \cong \mathbb{Z}_p[[T]]$, the augmentation map aug = $\operatorname{aug}_{\Gamma}$ corresponds to evaluation at T = 0. In particular, the latter is independent of the non-canonical choice of γ , since so is aug.

This isomorphism with the ring of formal powers series allows one to deduce a plethora of properties of the classical Iwasawa algebra $\Lambda(\Gamma)$. It is an integrally closed, Noetherian unique factorisation domain and a two-dimensional regular local ring whose maximal ideal is $\langle p, \gamma - 1 \rangle$ (the kernel $\Delta_{1,\Gamma}$ of the projection $\Lambda(\Gamma) \twoheadrightarrow \mathbb{F}_p$) and whose height-one prime ideals are all principal, generated by either p or an irreducible Weierstrass polynomial. We denote the localisation of $\Lambda(\Gamma)$ at a prime ideal \mathfrak{p} by $\Lambda_{\mathfrak{p}}(\Gamma)$.

The **support** of a $\Lambda(\Gamma)$ -module M is

 $\operatorname{supp}(M) = \{\mathfrak{p} \text{ a prime ideal of } \Lambda(\Gamma) : M_{\mathfrak{p}} \neq 0\},\$

which is finite if M is finitely generated and torsion. M is said to be **pseudo-null** if $\operatorname{supp}(M)$ contains no height-one prime ideals, which is equivalent to M being finite (cf. [NSW20] remark after definition 5.1.4). A homomorphism $f: M \to N$ of $\Lambda(\Gamma)$ -modules is said to be a **pseudo-isomorphism** if the localisation f_p at every height-one prime ideal \mathfrak{p} is an isomorphism of $\Lambda_p(\Gamma)$ -modules, i.e. if ker(f) and coker(f) are finite. We often denote this by $f: M \xrightarrow{\approx} N$, which defines a reflexive and transitive relation on $\Lambda(\Gamma)$ -modules. It is furthermore symmetric (and thus an equivalence relation) on finitely generated *torsion* Iwasawa modules. One of the central results of classical Iwasawa theory is the following classification:

Theorem 0.1.3 (Structure theorem for Iwasawa modules). Let M be a finitely generated $\Lambda(\Gamma)$ module, where $\Gamma \cong \mathbb{Z}_p$. Then there exists a pseudo-isomorphism of (topological) $\Lambda(\Gamma)$ -modules

$$M \xrightarrow{\approx} \Lambda(\Gamma)^r \oplus \bigoplus_{i=1}^s \Lambda(\Gamma) / \langle p^{m_i} \rangle \oplus \bigoplus_{j=1}^t \Lambda(\Gamma) / \langle F_j^{l_j} \rangle$$
(0.8)

for some $r, s, t \in \mathbb{N}$, $m_i, l_j > 0$ and irreducible Weierstrass polynomials F_j , and these are all unique up to order.

This is [NSW20] theorem 5.3.8. Here $r = \operatorname{rank}_{\Lambda(\Gamma)} M$ and one defines the **Iwasawa invari**ants $\mu(M) = \sum_{i=1}^{s} m_i$ and $\lambda(M) = \sum_{j=1}^{t} l_j \operatorname{deg}(F_j)$, as well as the characteristic polynomial¹ $\operatorname{char}(M) = \prod_{j=1}^{t} F_j^{l_j}$. If M is a torsion module, then the height-one prime ideals in $\operatorname{supp}(M)$ are precisely the $\langle F_j \rangle$, and potentially $\langle p \rangle$ (if $\mu(M) \neq 0$). If the pseudo-isomorphism in (0.8) is an isomorphism, M is said to be **elementary**.

Given a short exact sequence $M' \hookrightarrow M \twoheadrightarrow M''$ of $\Lambda(\Gamma)$ -modules, an easy application of the snake lemma to the endomorphism of each module given by multiplication by $\gamma - 1$ yields the **invariants**coinvariants exact sequence

$$0 \to (M')^{\Gamma} \to M^{\Gamma} \to (M'')^{\Gamma} \to M'_{\Gamma} \to M_{\Gamma} \to M'_{\Gamma} \to 0.$$

$$(0.9)$$

Here one may replace Γ by Γ^{p^n} either by considering the endomorphism $\gamma^{p^n} - 1$ instead, or by noting that $\Gamma^{p^n} \cong \mathbb{Z}_p$ and therefore $\Lambda(\Gamma^{p^n})$ is itself a classical Iwasawa algebra.

¹In some sources, the characteristic polynomial is scaled by $p^{\mu(M)}$.

0.2 Representations of finite groups

Group representations and group algebras will play a central role in the subsequent discussion. Both finite and infinite groups will be involved in the formulation of the Main Conjecture, yet much of the necessary machinery will be concerned with finite groups alone - even for infinite groups, we shall only consider characters which factor through a finite quotient. The following definitions and properties can be found in any standard reference on the topic, for instance [CR81] (primarily subsections §9A and §9B).

Let G be a finite group and E a field. We assume $\operatorname{char}(E) = 0$, although this requirement can be weakened to $\operatorname{char}(E) \nmid |G|$ in most instances. A **representation** (V,ρ) of G over E is a non-trivial E-vector space V together with a group homomorphism $\rho: G \to \operatorname{GL}(V)$. This is equivalent to endowing V with a left E[G]-module structure with G-action via $\sigma v = \rho(\sigma)(v)$ for $\sigma \in G, v \in V$, and both descriptions will be used interchangeably. We often denote the representation simply by V or ρ , the other element being implicit. The dimension of V is $d = \dim_E V$, which will always be finite in this text. After choosing a basis, this representation corresponds uniquely to a homomorphism $\rho: G \to \operatorname{GL}_d(E)$. Two representations with chosen bases $\rho, \rho': G \to \operatorname{GL}_d(E)$ are said to be **equivalent** if they differ by a change of basis, i.e. if $\rho'(\sigma) = M\rho(\sigma)M^{-1}$ for all $\sigma \in G$ for a fixed $M \in \operatorname{GL}_d(E)$.

The character χ of (V, ρ) is the map

$$\chi \colon G \to E$$
$$\sigma \mapsto \operatorname{Tr}(\rho(\sigma))$$

which is well defined regardless of the choice of a basis for V. In particular, $\chi(1)$ is the dimension of the representation. We say V **affords** χ . Since the trace is invariant under conjugation, characters are **class functions**, i.e. they are invariant under conjugation by elements of the group. If ρ is linear (i.e. $\dim_E(V) = 1$), then χ and ρ coincide and the character becomes a group homomorphism $\chi: G \to E^*$. Two representations have the same character if and only if they are equivalent, which in turn happens if and only if the associated vector spaces are isomorphic as E[G]-modules.

Let $\rho: G \to \operatorname{GL}_d(E)$ be a representation with character χ and choose $\sigma \in G$. Since G is finite, $\rho(\sigma)$ has finite (multiplicative) order and thus all of its eigenvalues in an algebraic closure E^c of E are roots of unity. In particular, $\chi(\sigma) \in E$ is a sum of roots of unity in E^c . The **kernel** of χ is defined as the kernel of ρ as a group homomorphism, and it can be shown that

$$\ker(\chi) = \{ \sigma \in G : \chi(\sigma) = \chi(1) \}.$$

The **trivial representation** is the linear representation which takes the constant value 1, i.e. it makes every element $\sigma \in G$ act as the identity on V = E. We denote its character by $\mathbb{1}_G$, or $\mathbb{1}$ if the group is clear from context. In particular, $\ker(\mathbb{1}_G) = G$.

A representation (V, ρ) of G over E (or, equivalently, its character χ) naturally gives rise to several other representations:

• If G is a subgroup of a finite group A, then the **induction** of V to A is the E[A]-module

$$\operatorname{ind}_{G}^{A}V = E[A] \otimes_{E[G]} V,$$

where E[A] is regarded as a right E[G]-module. Its character $\operatorname{ind}_{G}^{A}\chi$ sends $\sigma \in A$ to

$$(\operatorname{ind}_{G}^{A}\chi)(\sigma) = \frac{1}{|G|} \sum_{\substack{\tau \in A \\ \tau \sigma \tau^{-1} \in G}} \chi(\tau \sigma \tau^{-1}).$$
(0.10)

• If G is a quotient of a finite group B and we denote the corresponding projection map by $\pi: B \to G$, then the **inflation** of ρ to B is the representation

$$\inf_{G}^{B} \rho = \rho \circ \pi \colon B \twoheadrightarrow G \to \mathrm{GL}(V),$$

which has character $\inf_{G}^{B} \chi = \chi \circ \pi$.

• If C is a subgroup of G, then the **restriction** of ρ to C is the representation

$$\operatorname{res}_{C}^{G} \rho = \rho|_{C} \colon C \hookrightarrow G \xrightarrow{\rho} \operatorname{GL}(V),$$

which has character $\operatorname{res}_C^G \chi = \chi|_C$.

The dual or contragredient representation (ρ*, V*) is the E-vector space V* = Hom_E(V, E) together with the G-action (σf)(v) = f(σ⁻¹v) for σ ∈ G, f ∈ V*, v ∈ V. If B is an E-basis of V and B* its dual, then the matrix of σ acting on V* with respect to B* is the transpose inverse of that of σ acting on V with respect to B. In other words, ρ* = (ρ⁻¹)^t: G → GL_{dim_E(V)}(E). We denote the character of ρ* by X, which satisfies X(σ) = X(σ⁻¹) for all σ ∈ G.

In the case $E = \mathbb{C}$, the eigenvalues of $\rho^*(\sigma) = (\rho(\sigma)^{-1})^t$ are the inverses of those of $\rho(\sigma)$, i.e. their complex conjugates (because, as argued above, these are roots of unity). In particular, $\check{\chi}(\sigma) = \overline{\chi(\sigma)} \in \mathbb{C}$ for all $\sigma \in G$.

• If H is a subgroup of G, we denote the subspace of H-invariants by

$$V^H = \{ v \in V : hv = v \text{ for all } h \in H \}.$$

This is an *E*-vector subspace of *V*, but it may fail to be closed under the action of *G*. However, it is so if *H* is normal, in which case V^H is again a representation of *G* over *E*.

- If (V', ρ') is another representation of G over E and we denote its character by χ', then the sum of ρ and ρ' is the E-vector space V ⊕ V' with G-action given by (ρ ⊕ ρ')(σ) = ρ(σ) ⊕ ρ'(σ) for σ ∈ G. After choosing bases, this amounts to the matrix block sum of the two representations. If we denote the character of ρ ⊕ ρ' by χ + χ', then one has (χ + χ')(σ) = χ(σ) + χ'(σ) for all σ ∈ G. In particular, χ + χ' = χ' + χ.
- If V', ρ' and χ' are as above, then the **tensor product** of ρ and ρ' is the *E*-vector space $V \otimes_E V'$ with *G*-action given by $(\rho \otimes \rho')(\sigma) = \rho(\sigma) \otimes \rho'(\sigma)$ (the Kronecker product of matrices after choosing bases) for $\sigma \in G$. If we denote the character of $\rho \otimes \rho'$ by $\chi \otimes \chi'$, then one has $(\chi \otimes \chi')(\sigma) = \chi(\sigma) \cdot \chi'(\sigma)$ for all $\sigma \in G$. In particular, $\chi \otimes \chi' = \chi' \otimes \chi$.

We say a representation (V, ρ) (or its associated character χ) is **irreducible** if V is a simple E[G]module, and reducible otherwise. In the latter case, V can be decomposed into a sum of several non-empty E-vector subspaces which are closed under the action of G. We denote the set of all (resp. all irreducible) E-valued characters of G by $\operatorname{Char}_E(G)$ (resp. $\operatorname{Irr}_E(G)$). Character addition turns $\operatorname{Char}_E(G)$ into a commutative semigroup generated by $\operatorname{Irr}_E(G)$, since every character decomposes uniquely as a sum of irreducible ones. This semigroup can be enlarged to an abelian group by allowing formal integer combinations of characters and identifying them according to character addition - that is, the relation $(\chi) + (\chi') \sim (\chi + \chi')$. This is the usual process of Grothendieck completion after a formal identity (which maps every $\sigma \in G$ to 0) has been added. The result coincides in fact with the free abelian group on $\operatorname{Irr}_E(G)$, many of whose elements - the so-called **virtual characters** - no longer arise as traces of representations. If we endow this additive group with the tensor product of characters, we obtain the **ring** $\mathcal{R}_E(G)$ of E-valued virtual characters of G. A central result in the complex case is the following:

Theorem 0.2.1 (Brauer's induction theorem). The ring $\mathcal{R}_{\mathbb{C}}(G)$ is additively generated by

$${\operatorname{ind}}_{H}^{G}\lambda$$
: H an elementary subgroup of G,λ a linear \mathbb{C} -valued character of H .

Recall that a finite group H is said to be *p*-elementary (with *p* a prime) if it decomposes as $H = C_n \times H_p$ with C_n a cyclic group of order *n* coprime to *p* and H_p a *p*-group; and elementary if it is *p*-elementary for at least one *p*.

The set of *E*-valued class functions on *G* has a natural *E*-vector space structure, and it can be shown that $\operatorname{Irr}_E(G)$ constitutes a basis. This space admits a **scalar product** $\langle -, - \rangle_G$ (where the subindex *G* might be omitted if clear from the context) given by

$$\langle \chi, \psi \rangle_G = \frac{1}{|G|} \sum_{\sigma \in G} \chi(\sigma) \psi(\sigma^{-1}) \in E.$$

This is symmetric, bilinear over E and satisfies **Frobenius reciprocity**: given a subgroup of $H \leq G$ and two characters (not arbitrary class functions) χ and ψ of G and H respectively, one has

$$\langle \chi, \operatorname{ind}_{H}^{G} \psi \rangle_{G} = \langle \operatorname{res}_{H}^{G} \chi, \psi \rangle_{H}.$$
 (0.11)

The following relation is a direct consequence of the definition of the scalar product: for characters χ, ψ and φ of G, one has

$$\langle \chi, \psi \otimes \varphi \rangle_G = \langle \chi \otimes \dot{\psi}, \varphi \rangle_G. \tag{0.12}$$

Characters provide a natural framework to understand the Wedderburn-Artin structure theorem. Assume for the rest of this section that E is algebraically closed (and still of characteristic 0). Then, a version of the aforementioned theorem states that the semisimple Artinian ring E[G] is in fact a finite sum of matrix rings over E. With some character theory, one can show

$$E[G] = \bigoplus_{\chi \in \operatorname{Irr}_E(G)} E[G]e(\chi) \cong \bigoplus_{\chi \in \operatorname{Irr}_E(G)} M_{\chi(1)}(E), \qquad (0.13)$$

where the isomorphism is character-wise. Here the $e(\chi)$ are (all) the **primitive central idempo**tents² of E[G], meaning that $e(\chi) \in Z(E[G])$, $e(\chi)^2 = e(\chi)$, and $e(\chi)$ cannot be expressed as a sum of two non-zero central idempotents. They are given by the formula

$$e(\chi) = \frac{\chi(1)}{|G|} \sum_{\sigma \in G} \chi(\sigma^{-1})\sigma.$$
(0.14)

In the isomorphism (0.13), $E[G]e(\chi) \cong M_{\chi(1)}(E)$ should be understood as the ring of *E*-linear endomorphisms $\operatorname{End}_E(V_{\chi})$ (with an implicit choice of basis), where V_{χ} is an irreducible E[G]module which affords χ . Given $\sigma \in G$, the element $\sigma e(\chi) \in E[G]e(\chi) \subseteq E[G]$ is sent to the matrix describing the action of σ on V_{χ} . In particular, $e(\chi)$ itself is mapped to the identity matrix. By the Wedderburn-Artin theorem, $E[G]e(\chi)$ is a ring with unity $e(\chi)$, but not a subring of E[G]in general since the unity does not coincide.

A remarkable property of these primitive central idempotents is their behaviour with respect to projection: if N is a normal subgroup of G and we let Q = G/N, the canonical ring surjection $\varepsilon \colon E[G] \twoheadrightarrow E[Q]$ satisfies the following: for an irreducible character $\chi \in \operatorname{Irr}_E(G)$,

$$\varepsilon(e(\chi)) = \begin{cases} e(\overline{\chi}), & N \subseteq \ker(\chi) \\ 0, & \text{otherwise,} \end{cases}$$
(0.15)

where $\overline{\chi}$ is the projection of χ to Q (which is necessarily irreducible), i.e. it is defined by $\chi = \inf_{N}^{G} \overline{\chi}$.

It follows from the arithmetic of the primitive central idempotents that if χ and χ' are irreducible, then

$$\langle \chi, \chi' \rangle_G = \begin{cases} 1, & \chi = \chi' \\ 0, & \text{otherwise} \end{cases}$$

This is known as **Schur's orthogonality relation**. In particular, the irreducible characters of G form an *orthonormal* basis of the *E*-vector space of class functions on G. As mentioned above, any $\psi \in \operatorname{Char}_E(G)$ admits a unique decomposition

$$\psi = \sum_{\chi \in \operatorname{Irr}_E(G)} n_{\chi} \chi$$

with $n_{\chi} \in \mathbb{N}$. By the orthogonality relation, we can compute the multiplicities as $n_{\chi} = \langle \chi, \psi \rangle_G$. We say χ divides ψ if $n_{\chi} > 0$; and ψ is **isotypic** if only one irreducible character divides it. The above is simply a restatement of the fact that every module over a semisimple Artinian ring decomposes uniquely as a direct sum of simple modules: the representations V_{χ} for $\chi \in \operatorname{Irr}_E(G)$ constitute a system of representatives of simple E[G]-modules up to isomorphism, and therefore any finitely generated E[G]-module M has a unique expression as

$$M \cong \bigoplus_{\chi \in \operatorname{Irr}_E(G)} V_{\chi}^{n_{\chi,M}} \tag{0.16}$$

²A common notation for these in the literature is e_{χ} . In this text, however, the notation e_{χ} will be reserved for a different - although related - object (cf. section 2.4) and the two should not be confused.

with $n_{\chi,M} \in \mathbb{N}$. *M* is then the unique (up to isomorphism) module with character $\psi_M = \sum_{\chi} \chi^{n_{\chi,M}}$.

Not all hope is lost for fields which are not algebraically closed: given an irreducible *E*-valued representation (V, ρ) of *G* with character χ (still assuming $E = E^c$), the entries of all matrices in the image of ρ after the choice of a basis \mathcal{B} only amount to a finite set of elements of *E*. If *F* is a subfield of *E* which contains all of those entries, we say χ has a **realisation** (or realises) over *F*. This implies the image of χ is also contained in *F*, and therefore, $e(\chi) \in F[G] \subseteq E[G]$ is a primitive central idempotent in F[G]. In this case, $F[G]e(\chi)$ is only a matrix ring over a *skew field* whose centre contains *F*.

0.3 Artin *L*-series

Artin *L*-series generalise classical objects such as the Dedekind zeta function and Dirichlet *L*-series. Their special values play a central role in modern number theory, and they will do so in our Main Conjecture as well. All properties below can be found in [Neu99] section 7§10 and [Tat84] section I§3. The definitions from section 0.2 will be particularly relevant.

Let L/K be a Galois extension of number fields with Galois group $G = \operatorname{Gal}(L/K)$. Given a prolongation w to L of a non-archimedean place v of K and a $\mathbb{C}[G]$ -module V_{χ} with character $\chi \in \operatorname{Char}_{\mathbb{C}}(G)$, there is a well-defined \mathbb{C} -linear action of the Frobenius element $\varphi_w \in \operatorname{Gal}(\kappa(w)/\kappa(v)) \cong G_w/I_w$ on the complex vector space $V_{\chi^w}^{I_w}$. We define the local **Euler factors**³

$$L_{v}(\chi, s) = \det(1 - \Re(v)^{-s}\varphi_{w} \mid V_{\chi}^{I_{w}})^{-1}$$
(0.17)

and

$$\delta_v(\chi, s) = \det(1 - \mathfrak{N}(v)^{1-s}\varphi_w \mid V_\chi^{I_w})$$
(0.18)

for $s \in \mathbb{C}$, where the first 1 is common notation for the identity matrix. The subscript $-_v$ is justified: different prolongations w of v lead to conjugate decomposition and inertia groups, and φ_w acts on $V_{\chi}^{I_w}$ via the same matrix (under a suitable basis) as $\varphi_{\sigma(w)}$ does on $V_{\chi}^{I_{\sigma(w)}} = V_{\chi}^{\sigma I_w \sigma^{-1}} = \sigma(V_{\chi}^{I_w})$. A similar argument shows that the Euler factors depend on the character χ rather than on the representation V_{χ} : two representations with the same character χ are equivalent and hence differ only by matrix conjugation, which does not affect the determinant.

If the action of φ_w on $V_{\chi}^{I_w}$ has (not necessarily distinct) eigenvalues $\lambda_1, \ldots, \lambda_n \in \mathbb{C}$, then

$$\det(1 - \mathfrak{N}(v)^{-s}\varphi_w \mid V_{\chi}^{I_w}) = \prod_{i=1}^n (1 - \mathfrak{N}(v)^{-s}\lambda_i)$$
(0.19)

is an entire complex function on s. In particular, its inverse $L_v(\chi, s)$ is meromorphic. By the same token, $\delta_v(\chi, s)$ is entire as well.

Let S and T be two finite sets of places of K such that $S \supseteq S_{\infty}$ (the archimedean places) and $S \cap T = \emptyset$. These are sometimes referred to as the **depletion** and **smoothing** set, respectively. The (S,T)-modified⁴ Artin L-series attached to χ is defined as

$$L_{K,S,T}(\chi,s) = \prod_{v \notin S} L_v(\chi,s) \cdot \prod_{v \in T} \delta_v(\chi,s)$$
(0.20)

for $s \in \mathbb{C}$, $\operatorname{Re}(s) > 1$. Note that there are infinitely many *L*-factors but only finitely many δ -ones (see the proposition below for convergence) and that primes in *T* appear in both types. We denote

 $^{{}^{3}}L_{v}$ does not denote the completion of L at v here. There will not be any ambiguity in practice.

⁴Traditionally, Artin *L*-series have been defined for a single chosen set *S*. This is the case in the two references cited at the beginning of the present section. The (S, T)-modified version is of more recent introduction, and can be found, for instance, in [Bur11]. However, the proofs of the properties in lemma 0.3.1 for the *S*-version found in the given references also apply to the (S, T)-version, since the δ -factors have the same functorial properties as the *L*-ones and do not vanish at s = 0

 $L_{S,\emptyset}(\chi,s)$ simply by $L_S(\chi,s)$. For instance, in the case L = K and $S = S_{\infty}$, the Artin *L*-series $L_{K,S_{\infty}}(\mathbb{1},s)$ coincides with the series defining the Dedekind zeta function $\zeta_K(s)$.

Before stating some basic properties of these series, we need to introduce a few classical $\mathbb{Z}[G]$ -modules. The first one is

$$\mathcal{Y}_{L,S}^{\mathbb{Z}} = \bigoplus_{w \in S(L)} \mathbb{Z} \cdot w \cong \bigoplus_{v \in S} \operatorname{Ind}_{G_w}^G \mathbb{Z},$$

where, in the last term, w is an arbitrarily chosen prolongation of v to L and $\operatorname{Ind}_{G_w}^G$ – stands for $\mathbb{Z}[G] \otimes_{\mathbb{Z}[G_w]} -$. Note that, if G_w is normal in G, then $\operatorname{Ind}_{G_w}^G \mathbb{Z}$ is simply $\mathbb{Z}[G/G_w]$. Different choices of prolongations yield canonically isomorphic modules, since they have conjugate decomposition groups. The natural G-action is given by permuting the places or, under the isomorphism, by simple multiplication in G. If we extend scalars from \mathbb{Z} to a field of characteristic 0, then $\mathcal{Y}_{L,S}^{\mathbb{Z}}$ becomes (canonically isomorphic to) the sum of inductions of the trivial representations $\mathbb{1}_{G_w}$ of each G_w in the language of section 0.2.

The second relevant Galois module is the kernel $\mathcal{X}_{LS}^{\mathbb{Z}}$ of the **augmentation map**

u

$$\mathcal{Y}_{L,S}^{\mathbb{Z}} \twoheadrightarrow \mathbb{Z}$$

$$\sum_{w \in S(L)} z_w \cdot w \mapsto \sum_{w \in S(L)} z_w,$$
(0.21)

which is a $\mathbb{Z}[G]$ -homomorphism when \mathbb{Z} is endowed with the trivial *G*-action. This kernel is closely related to the group of *S*-units

$$\mathcal{O}_{L,S}^* = \{ u \in L^* \colon |u|_w = 1 \text{ for every place } w \text{ of } L \text{ outside } S(L) \},$$
(0.22)

which contains the finite-index subgroup of (S, T)-units

$$\mathcal{O}_{L,S,T}^* = \{ u \in \mathcal{O}_{L,S}^* \colon |u - 1|_w < 1 \text{ for all } w \in S(T) \}.$$
(0.23)

In other words, the S-units are the non-zero elements of L which are integral locally at all places outside S(L), and the (S,T)-units are the S-units which are principal locally at all places in S(T). Both are equipped with the natural action of G on L^* . The connection to $\mathcal{X}_{L,S}^{\mathbb{Z}}$ comes in the form of **Dirichlet's unit theorem**, which asserts that the **Dirichlet regulator map**

$$\mathbb{R} \otimes \mathcal{O}_{L,S}^* \to \mathbb{R} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}}$$

$$r \otimes u \mapsto -\sum_{w \in S(L)} r \log(|u|_w) \otimes w$$

$$(0.24)$$

is an isomorphism of $\mathbb{R}[G]$ -modules.

We point out that the definition of $\mathcal{Y}_{L,S}^{\mathbb{Z}}$ and $\mathcal{X}_{L,S}^{\mathbb{Z}}$ does not truly require $S \supseteq S_{\infty}$ (but Dirichlet's unit theorem does). Furthermore, one can consider all four modules above over a single number field rather a Galois extension - this simply corresponds to the case L = K, G = 1.

The relevance of these objects in the discussion of Artin L-series is clear in point iv) of the following result:

Lemma 0.3.1. In the above notation, one has the following:

- i) $L_{K,S,T}(\chi, s)$ converges on $\operatorname{Re}(s) > 1$ and has a meromorphic continuation to all of \mathbb{C} , which we denote the same and refer to as the (S,T)-modified Artin L-function attached to χ .
- ii) If M is a finite Galois extension of K containing L, then $L_{K,S,T}(\chi,s) = L_{K,S,T}(\inf_{G}^{\text{Gal}(M/K)}\chi,s)$.
- iii) If H is a subgroup of G with fixed field $L' = L^H$ and $\chi = \operatorname{ind}_H^G \psi$ for some $\psi \in \operatorname{Char}_{\mathbb{C}}(H)$, then $L_{K,S,T}(\chi,s) = L_{L',S(L'),T(L')}(\psi,s)$.
- iv) If χ' is another complex character of G, then $L_{K,S,T}(\chi + \chi', s) = L_{K,S,T}(\chi, s)L_{K,S,T}(\chi', s)$.
- v) The order of vanishing $r_S(\chi)$ of $L_{K,S,T}(\chi,s)$ at s=0 coincides with

$$\left(\sum_{v\in S} \dim_{\mathbb{C}} V_{\chi}^{G_w}\right) - \dim_{\mathbb{C}} V_{\chi}^G = \langle \chi, \psi_{\mathcal{X}} \rangle = \dim_{\mathbb{C}}(\operatorname{Hom}_{\mathbb{C}[G]}(V_{\chi}^*, \mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}}))$$

where, in the second term, w denotes an arbitrarily chosen prolongation of v; and, in the third term, $\psi_{\mathcal{X}}$ is the character of $\mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}}$ as a $\mathbb{C}[G]$ -module. In particular, it is independent of T and non-negative, i.e. $L_{K,S,T}(\chi, s)$ does not have a pole at s = 0.

In properties ii) - iv), the equality means equality as series and hence equality of Artin *L*-functions. These functorial properties constitute a powerful tool when combined with representation-theoretic results such as Brauer's induction theorem 0.2.1. For instance, it follows immediately that Artin *L*-functions decompose as products of those attached to linear characters - which are easier to manipulate. Since virtual characters $R_{\mathbb{C}}(G)$ are freely generated by $\operatorname{Irr}_{\mathbb{C}}(G)$, property iv) above provides a natural way to define *L*-functions attached to virtual characters. Another observation is that property ii) justifies why the field *L* is not featured in the notation $L_{K,S,T}$ (although it is implicit in χ at any rate): enlarging or - when possible - shrinking the top field does not affect the *L*-function. In that sense, there is a minimal field over which the function can be defined: $L^{\operatorname{ker}(\chi)}$, which is sometimes referred to as the field **cut out** by χ .

The **leading coefficient** of $L_{K,S,T}(\chi, s)$ at s = 0 is defined as

$$L_{K,S,T}^{*}(\chi,0) = \lim_{s \to 0} s^{-r_{S}(\chi)} L_{K,S,T}(\chi,s) \in \mathbb{C}^{*}.$$
(0.25)

In other words, it is the first non-zero coefficient of the series expansion of $L_{K,S,T}(\chi, s)$ around 0. This is the special value featured in our Main Conjecture.

0.4 Algebraic *K*-theory

Algebraic K-theory will provide the right setting for the formulation of our Main Conjecture. The algebraic and analytic sides of the conjecture come together in a classical so-called localisation sequence. We will only be concerned with K-groups in low degrees, namely 0 and 1. Our main reference is [Swa68].

Let R be a ring and denote by $\mathbf{P}_0(R)$ the category of finitely generated projective left R-modules with R-module homomorphisms. We define the following K-groups:

• $K_0(R)$ is the quotient of the free abelian group on the set of isomorphism classes of modules in $\mathbf{P}_0(R)$ by the subgroup generated by

$$(\overline{M \oplus N}) - (\overline{M}) - (\overline{N}),$$

where $M, N \in \mathbf{P}_0(R)$ and $\overline{-}$ means isomorphism class. Here and below, (-) denotes the image of - in the corresponding free abelian group. Given $M \in \mathbf{P}_0(R)$, we denote the class of (\overline{M}) in $K_0(R)$ by [M].

For $M, N \in \mathbf{P}_0(R)$, one has [M] = [N] in $K_0(R)$ if and only if there exists a $P \in \mathbf{P}_0(R)$ such that $M \oplus P \cong N \oplus P$ (cf. [Swa68], theorem 1.10). In that case, M and N are said to be **stably isomorphic**, which is in general a weaker condition than isomorphism.

This construction can be formulated naturally in terms of a Grothendieck completion, a notion which already appeared in section 0.2.

• Given a ring homomorphism $R \xrightarrow{\varphi} S$, consider the category $\mathbf{P}_0^{\varphi}(R, S)$ whose objects are triples (M, f, N) with $M, N \in \mathbf{P}_0(R)$ and $f: S \otimes_R M \xrightarrow{\sim} S \otimes_R N$ an isomorphism of S-modules. The map f need not arise as the scalar extension of a morphism of R-modules $M \to N$. A morphism of triples $\nu: (M, f, N) \to (M', f', N')$ consists of a pair of homomorphisms of R-modules $\nu_M: M \to M', \nu_N: N \to N'$ making the diagram

$$S \otimes_{R} M \xrightarrow{S \otimes_{R} \nu_{M}} S \otimes_{R} M'$$

$$\downarrow f \qquad \qquad \qquad \downarrow f'$$

$$S \otimes_{R} N \xrightarrow{S \otimes_{R} \nu_{N}} S \otimes_{R} N'$$

commute. Note that, since the two vertical arrows are isomorphisms, either horizontal arrow determines the other uniquely. Composition and identity morphisms in this category are defined in the obvious way.

The relative K_0 group⁵ of R and S, denoted by $K_0^{\varphi}(R, S)$, is the quotient of the free abelian group on the set of isomorphism classes of objects in $\mathbf{P}_0^{\varphi}(R, S)$ (in order to avoid notational clutter, we do not distinguish between (M, f, N) and its isomorphism class) by the subgroup generated by all elements of the form

⁵Some authors define the relative K_0 group for R and a two-sided ideal $\mathfrak{a} \leq R$, in which case it is often denoted by $K_0(R,\mathfrak{a})$. This coincides with the above definition of $K_0^{\varphi}(R, R/\mathfrak{a})$ (cf. [Swa68] p. 214) if one chooses as φ the canonical projection $R \twoheadrightarrow R/\mathfrak{a}$. In particular, Swan's definition is more general (and necessary for our purposes).

- i) ((M, gf, P)) ((M, f, N)) ((N, g, P)).
- ii) ((M, f, N)) ((M', f', N')) ((M'', f'', N'')) if there exists a short exact sequence

 $0 \to (M', f', N') \to (M, f, N) \to (M'', f'', N'') \to 0$

in $\mathbf{P}_0^{\varphi}(R, S)$ in the obvious sense.

We denote the class of ((M, f, N)) by [M, f, N]. An immediate consequence of i) is that $[M, \mathrm{Id}, M] = 0$ and $[M, f, M] = -[M, f^{-1}, M]$. As a direct application of ii), we have the following: if $f_R \colon M \xrightarrow{\sim} N$ is an isomorphism of (finitely generated projective) *R*-modules, then

$$[M, S \otimes_R f_R, N] = 0. \tag{0.26}$$

The notation $K_0(R,S)$ may be used instead of $K_0^{\varphi}(R,S)$ if the morphism φ is clear from context.

• Let $\operatorname{GL}(R)$ denote the infinite general linear group over R, that is, the direct limit $\varinjlim_n \operatorname{GL}_n(R)$ with transition maps $\operatorname{GL}_n(R) \hookrightarrow \operatorname{GL}_{n+1}(R)$ given by $M \mapsto M \oplus 1$. Then the Whitehead group of R is the abelianisation

$$K_1(R) = \mathrm{GL}(R)^{ab}.$$

By a classical result, the commutator of $\operatorname{GL}(R)$ is precisely the subgroup generated by elementary matrices. We denote the class of a matrix $A \in \operatorname{GL}_n(R) \subset \operatorname{GL}(R)$ in $K_1(R)$ by [A].

We will on occasion need an alternative description of K_1 resembling that of the relative K_0 group. Let $\mathbf{P}_0^{\mathrm{Aut}}(R)$ denote the category whose objects are pairs (M, f) with $M \in \mathbf{P}_0(R)$ and $f \in \mathrm{Aut}_R(M)$; and where a morphism $\nu \colon (M, f) \to (M', f')$ is a morphism of *R*-modules $\nu \colon M \to M'$ making the diagram

$$\begin{array}{ccc} M & \stackrel{\nu}{\longrightarrow} & M' \\ & \downarrow f & & \downarrow f' \\ M & \stackrel{\nu}{\longrightarrow} & M' \end{array}$$

commute. Composition and identity morphisms are defined in the obvious way. The group $K_1^{\text{det}}(R)$ is then defined as the quotient of the free abelian group on the set of isomorphism classes of objects in $\mathbf{P}_0^{\text{Aut}}(R)$ (which, as before, we do not distinguish from the objects themselves) by the subgroup generated by all elements of the form

- i) ((M,gf)) ((M,f)) ((M,g)).
- ii) ((M, f)) ((M', f')) ((M'', f'')) if there exists a short exact sequence

$$0 \to (M', f') \to (M, f) \to (M'', f'') \to 0$$

in $\mathbf{P}_0^{\text{Aut}}(R)$ in the obvious sense.

Analogously to the relative K_0 , we denote the class of ((M, f)) by [M, f]. It follows from i) that [M, Id] = 0 and $[M, f] = -[M, f^{-1}]$.

These two descriptions of K_1 give rise to the same group: the map

$$K_1(R) \to K_1^{\det}(R)$$

 $[A] \mapsto [R^n, f_A]$

is a well-defined group isomorphism (cf. [CR87] theorem 40.6 or [Swa68] theorem 13.4). Here n is any positive integer such that $A \in \operatorname{GL}_n(R) \subseteq \operatorname{GL}(R)$ and f_A is the R-automorphism of R^n induced by multiplication by A. Note that, since R may not be commutative, multiplying elements of R^n by A is only guaranteed to yield a homomorphism of left R-modules if done on the right. In other words, $x \in R^n$ is regarded as a row vector and $f_A(x) = xA$.

The inverse of the above isomorphism can be described as follows: let $[M, f] \in K_1^{\text{det}}(R)$. There exists an $N \in \mathbf{P}_0(R)$ and an isomorphism $\iota: M \oplus N \xrightarrow{\sim} R^n$ for some $n \in \mathbb{N}$. Then $\iota \circ (f \oplus \text{Id}_N) \circ \iota^{-1}$ is an *R*-automorphism of R^n , which in particular has a matrix expression $A_f \in \text{GL}_n(R)$ - once again regarding elements of R^n as row vectors. Mapping [M, f] to $[A_f]$ is well defined and yields the inverse $K_1^{\text{det}}(R) \xrightarrow{\sim} K_1(R)$ (cf. loc. cit.).

In the sequel, we regard this canonical isomorphism as an identification $K_1(R) = K_1^{\text{det}}(R)$.

We shall write the groups $K_0(R)$, $K_0(R, S)$ and $K_1^{\text{det}}(R)$ in additive notation, since their operations are related to the direct sum of modules; and the group $K_1(R)$ in multiplicative notation, since its operation comes from matrix multiplication. Note that all three groups are abelian.

Let now $\varphi \colon R \to S$ be any ring homomorphism. Consider the sequence abelian groups

$$K_1(R) \xrightarrow{K_1(\varphi)} K_1(S) \xrightarrow{\partial} K_0(R,S) \xrightarrow{\psi} K_0(R) \xrightarrow{K_0(\varphi)} K_0(S)$$
 (0.27)

where each arrow is defined as follows:

- $K_1(\varphi)$ sends [A] to $[\varphi(A)]$, where φ is applied entry-wise.
- ∂ sends [A] to $[R^n, f_A, R^n]$ for $A \in \operatorname{GL}_n(S)$, where f_A is the automorphism of S^n given by multiplication by A on the right. This map is often called the **connecting** or **boundary** homomorphism⁶.
- ψ sends [M, f, N] to [M] [N].
- $K_0(\varphi)$ is induced by extension of scalars $S \otimes_R -: \mathbf{P}_0(R) \to \mathbf{P}_0(S)$ via φ . In other words, it sends [M] to $[S \otimes_R M]$.

These four maps are well-defined group homomorphisms which make (0.27) into an exact sequence (cf. [Swa68] theorem 15.5), known as the **exact sequence of** *K*-theory - although technically this

⁶Another common choice for ∂ the inverse of the above, i.e. $\partial([A]) = [R^n, f_A^{-1}, R^n]$. This does not affect the exactness of the sequence, but it has implications on other conventions. Our choice is consistent with the sources we will refer to, most notably [Swa68] and [BB05].

term refers to an infinite exact sequence continuing to the left with higher K-groups. If $R \xrightarrow{\varphi} S$ is the embedding of R into some localisation, the sequence is also referred to as the **localisation** sequence.

As an addendum to the definition of the boundary homomorphism, we also let $\partial : K_1^{\text{det}}(S) \to K_0(R, S)$ denote the map resulting from the identification $K_1(S) = K_1^{\text{det}}(S)$. Relation ii) in the definition of $K_0(R, S)$ then implies the following: given $M \in \mathbf{P}_0(R)$ and an S-automorphism f of $S \otimes_R M$, one has

$$\partial([S \otimes_R M, f]) = [M, f, M]. \tag{0.28}$$

 K_0 and K_1 as above define covariant functors from \mathfrak{R} (rings with unity) to \mathfrak{Ab} (abelian groups). There is a natural way to define *contravariant* functors on a certain subcategory instead. To do so, let $\varphi \colon R \to S$ be a ring homomorphism such that S is finitely generated and projective as an R-module via φ . Then restriction of scalars $-|_R$ defines a covariant functor $\mathbf{P}_0(S) \to \mathbf{P}_0(R)$. This in turn induces a **transfer** functor on K-theory (cf. [Wei13] definition IV.6.3.2), which we can easily make explicit in low degree. On K_0 , the map is clear:

$$K_0^{\mathrm{res}}(\varphi) \colon K_0(S) \to K_0(R)$$
$$[M] \mapsto [M]_R].$$

In the case of K_1 , the map has an immediate description in terms of K_1^{det} :

$$\begin{split} K_1^{\text{det,res}}(\varphi) \colon K_1^{\text{det}}(S) \to K_1^{\text{det}}(R) \\ [M,f] \mapsto [M_{\mid_R},f_{\mid_R}]. \end{split}$$

This then induces a map on the usual K_1 by imposing commutativity of the diagram

The resulting homomorphism $K_1^{\text{res}}(\varphi)$ does not have a seamless definition in terms of matrices, but it can be described as follows. For simplicity, assume that S is free as an R-module and hence there exists an R-isomorphism $\iota: \mathbb{R}^r \to S$. Let $A = (a_{i,j})_{i,j} \in \text{GL}_n(S)$. For each i, j, let $m_{i,j}$ be the homomorphism of S-(and R-)modules $S \to S$ given by multiplication by $a_{i,j}$ on the right. The composition $\iota^{-1} \circ m_{i,j} \circ \iota$ is an endomorphism of the left R-module \mathbb{R}^r and hence can be represented by a matrix $M_{i,j} \in M_r(R)$. Then $K_1^{\text{res}}(\varphi)([A])$ is the class in $K_1(R)$ of the rn-by-rn matrix resulting from replacing each $a_{i,j}$ by $M_{i,j}$ in A.

Having defined change-of-ring maps for K_0 and K_1 , it remains to extend these to the relative Kgroup $K_0(R, S)$ in a compatible manner. This will be essential to prove the good functorial behaviour of refined Euler characteristics (appendix A) and hence of the Main Conjecture. The construction naturally involves four rings R, S, R' and S' and homomorphisms making the diagram

$$\begin{array}{ccc} R & \stackrel{\varphi}{\longrightarrow} & S \\ \downarrow^{\rho} & \qquad \downarrow^{\sigma} \\ R' & \stackrel{\varphi'}{\longrightarrow} & S' \end{array} \tag{0.29}$$

commute. Extension of scalars is then given by

$$K_0(\rho, \sigma) \colon K_0(R, S) \to K_0(R', S')$$
$$[M, f, N] \mapsto [R' \otimes_R M, S' \otimes_S f, R' \otimes_R N]$$

The fact that this is well defined relies crucially on $S' \otimes_S (S \otimes_R -)$ being the same as $S' \otimes_{R'} (R' \otimes_R -)$, which is ensured by (0.29). The exact K-theory sequences (0.27) induced by φ and φ' are then connected into a commutative diagram

In the case of restriction of scalars, we place the same additional requirement on diagram (0.29) as before, namely that R' be a finitely generated projective R-module via ρ and S' a finitely generated projective S-module via σ . Then the map

$$\begin{aligned} K_0^{\mathrm{res}}(\rho,\sigma) \colon K_0(R',S') &\to K_0(R,S) \\ [M,f,N] \mapsto [M_{|_R},f_{|_S},N_{|_R}] \end{aligned}$$

is well defined and makes the diagram

commute.

We end this section by briefly introducing the **reduced norm** nr: $K_1(R) \to Z(R)^*$ for a semisimple Artinian ring R. This map will be instrumental in formulating the analytic side of the Main Conjecture, as it connects the localisation sequence above to certain (fraction fields of) rings of power series where the analogues of *p*-adic *L*-series are expected to live. The specifics of the construction will not play an important role. More details can be found in [Wei13] III.1.2.4.

Consider first a division algebra D of finite dimension d over its centre F = Z(D), which is necessarily a field. Choose a splitting field E, that is, an extension of F such that $E \otimes_F D \cong M_n(E)$ - in particular, $d = n^2$ (n is called the **Schur index** of D). Splitting fields always exist - any maximal field inside D is one. The reduced norm map

$$\operatorname{nr} \colon D \hookrightarrow E \otimes_F D \cong M_n(E) \xrightarrow{\operatorname{det}} E$$

is known to have image contained in F and to be independent of the choice of E and the isomorphism in the middle. It restricts to a group homomorphism $D^* \to F^*$. The same construction can be used to define maps $M_r(D) \to F$ which restrict to homomorphisms $\operatorname{GL}_r(D) \to F^*$ for all $r \ge 1$. These are compatible with the embeddings $\operatorname{GL}_r(D) \hookrightarrow \operatorname{GL}_{r+1}(D)$ and hence extend to a homomorphism

$$\operatorname{nr} \colon K_1(D) \to F^*$$

This is easily generalised to the case where R is a semisimple Artinian ring in the form of

$$\operatorname{nr} \colon K_1(R) \cong \prod_{i=1}^s K_1(M_{n_i}(D_i)) = \prod_{i=1}^s K_1(D_i) \xrightarrow{\prod \operatorname{nr}} \prod_{i=1}^s F_i^* = Z(R)^*,$$

where each D_i is a division ring of finite dimension d_i over its centre F_i . Here the first isomorphism is given by the Wedderburn-Artin theorem $R \cong \prod M_{n_i}(D_i)$ together with the fact that K_1 sends products to products. The second equality is known as **Morita invariance** (cf. [Wei13] example III.1.1.4 and proposition III.1.6.4) and comes from the natural identifications $\operatorname{GL}_a(M_b(D)) = \operatorname{GL}_{ab}(D)$. As before, the resulting homomorphism nr is independent of the choice of an isomorphism in the Wedderburn-Artin theorem.

When R is a *commutative* semisimple Artinian ring, the same theorem implies it decomposes as a finite product of fields. In particular, the reduced norm becomes an isomorphism

$$\operatorname{nr}: K_1(R) \xrightarrow{\sim} R^*. \tag{0.32}$$

Chapter 1

Construction of the complex

This chapter is devoted to the construction of a certain complex $\mathcal{C}_{S,T}^{\bullet}$ in explicit Galois-theoretic and cohomological terms. This complex is the central algebraic object in our Main Conjecture, and it coincides in the relevant derived category with that employed by Burns, Kurihara and Sano in their recent conjecture (cf. [BKS17]). Although not strictly necessary for the formulation of the Main Conjecture, having this explicit description of $\mathcal{C}_{S,T}^{\bullet}$ is valuable in itself, and it will simplify subsequent tasks such as the study of how changes in the parameters (most notably the sets of places S and T) affect the complex.

The first two sections deal with the definition of some global and local complexes, respectively, which are based on the **translation functor** of Ritter and Weiss (cf. [RW96]). In section 1.3, we define a natural morphism from the local complexes to the global one. The definition of $C_{S,T}^{\bullet}$ as the cone of that morphism, along with some of its key properties, are the object of section 1.4. The remaining section proves the relation to the complex of Burns, Kurihara and Sano.

The reader who is familiar with the language of *Weil-étale cohomology complexes* or is otherwise not interested in the construction may skip this chapter and refer only to proposition 1.4.3 and theorem 1.4.6 for the essential properties of the complex. However, the next few pages (until the beginning of section 1.1) lay out some important notation for subsequent chapters - alongside that introduced at the beginning of chapter 0, which we assume now. Specifically, the following setting will apply to most of our considerations⁷:

Setting A. Let K be a number field. We consider:

- A rational prime p, different from 2 if K is not totally imaginary.
- A fixed algebraic closure \mathbb{Q}^c of \mathbb{Q} . For each place v of K, we choose and fix a distinguished prolongation v^c of v to \mathbb{Q}^c . We denote the place of a number field K' below v^c by v(K'), which is simply $K' \cap v^c$ if v is non-archimedean. This should not be confused with $\{v\}(K')$,

⁷For the sake of clarity, we will reference this and subsequent settings in the definitions and results where it is in place.

the set of prolongations of v to $K' \supseteq K$ in the notation introduced in chapter 0.

• L_{∞} , a one-dimensional p-adic Lie extension of K containing its cyclotomic \mathbb{Z}_p -extension K_{∞} - which simply amounts to $H = \operatorname{Gal}(L_{\infty}/K_{\infty})$ being finite. We denote $\operatorname{Gal}(K_{\infty}/K)$ by Γ_K , $\operatorname{Gal}(L_{\infty}/K)$ by \mathcal{G} and choose and fix an open central subgroup $\Gamma \leq_o \mathcal{G}$ such that $\Gamma \cong \mathbb{Z}_p$ as topological groups. Then $\Gamma \cap H = \{1\}$ and therefore, letting $L = L_{\infty}^{\Gamma}$, one has $L_{\infty} = LK_{\infty}$. In particular, L_{∞} is itself the cyclotomic \mathbb{Z}_p -extension of L. Lemma 1.0.2 below shows that such a Γ exists.

 K_n denotes the *n*-th layer of K_{∞}/K , and thus $\operatorname{Gal}(K_{\infty}/K_n) = \Gamma_K^{p^n}$ and $\operatorname{Gal}(K_n/K)$ is cyclic of order p^n . The same notation applies to L_n in L_{∞}/L . We define $N \in \mathbb{N}$ by $L \cap K_{\infty} = K_N$, so the image of the projection $\Gamma \hookrightarrow \mathcal{G} \twoheadrightarrow \Gamma_K$ is precisely $\Gamma_K^{p^N}$. Since $\Gamma \subseteq Z(\mathcal{G})$, the extension L_n/K is Galois (and finite) for all $n \in \mathbb{N}$ and we denote the corresponding Galois group by $\mathcal{G}_n = \mathcal{G}/\Gamma^{p^n}$.

• $S \supseteq S_{\infty} \cup S_{\text{ram}}(L_{\infty}/K)$, a finite set of places of K containing the archimedean places and the ramified places in L_{∞}/K . We denote the set of non-archimedean places in S by S_f . Since the places which ramify in the cyclotomic \mathbb{Z}_p -extension of any number field are precisely the p-adic ones, we have $S_f \supseteq S_p$ (the set of p-adic places of K).

Non-archimedean places are finitely decomposed in the cyclotomic \mathbb{Z}_p -extension, and therefore the subgroup

$$\Gamma \cap \bigcap_{w_{\infty} \in S_f(L_{\infty})} \mathcal{G}_{w_{\infty}}$$

is open in Γ , i.e. of the form $\Gamma^{p^{n(S)}}$ for some $n(S) \in \mathbb{N}$. In particular, $L_{n(S)}$ is the smallest layer L_n of L_{∞}/L such that every place in $S_f(L_n)$ is non-split in L_{∞}/L_n .

• T, a finite set of places of K disjoint from S. We let

 $T^p = \{v \in T : L_w \text{ contains a primitive } p\text{-th root of unity for some (any) place } w \in \{v\}(L)\},\$

where the equivalence between *some* and *any* comes from the fact that the elements of $\operatorname{Gal}(L/K)$ induce isomorphisms among the residue fields $\kappa(w)$ of the various prolongations $w \mid v$. By Hensel's lemma, L_w contains a primitive *p*-th root of unity if and only if $p \mid \mathfrak{N}(w) - 1$.

We do not assume

- \mathcal{G} to be abelian.
- L or K to be totally real or imaginary, with the aforementioned caveat if p = 2. Furthermore, we do not prescribe the behaviour of archimedean places in L_{∞}/K .
- |H| (i.e. $[L_{\infty}: K_{\infty}]$) to be prime to p.

Remark 1.0.1. An alternative approach to the above setting is to start with an extension M/K of number fields and assume that the cyclotomic \mathbb{Z}_p -extension M_{∞} of M is Galois over K - which is always the case, for instance, if M/K is Galois. It follows that $[M_{\infty} : K_{\infty}]$ is finite (with K_{∞} the
cyclotomic \mathbb{Z}_p -extension of K) and one is thus in the same situation as above with $L_{\infty} = M_{\infty}$. In fact, L can then be chosen to be one of the layers M_n by simply forming the compositum of M and an arbitrarily chosen L as in the setting. Since our Main Conjecture will not depend on the choice of L (as shown in subsection 3.2.1), both approaches are equivalent.

A simple argument (see for instance [RW04] section 1) shows that a Γ as in setting A exists:

Lemma 1.0.2. Let K be a number field and L_{∞}/K a Galois extension such that $\mathcal{G} = \operatorname{Gal}(L_{\infty}/K)$ contains a finite (and hence closed) subgroup H with quotient $\Gamma_K = \mathcal{G}/H \cong \mathbb{Z}_p$. Then, the short exact sequence

$$1 \to H \to \mathcal{G} \to \Gamma_K \to 1$$

of profinite groups splits and \mathcal{G} contains an open central subgroup $\Gamma \cong \mathbb{Z}_p$. For any such Γ , the field $L = L_{\infty}^{\Gamma}$ is a finite Galois extension of K and one has $\Gamma \cap H = \{1\}$ and $L_{\infty}^{H}L = L_{\infty}$.

Proof. Let π denote the projection $\mathcal{G} \to \Gamma_K$ and choose a topological generator γ_K of Γ_K . Choose a preimage g of γ_K under π . If n is the index of a p-Sylow subgroup \mathcal{G}_p of \mathcal{G} (i.e. the prime-to-ppart of |H|), then $\pi(g^n) = \gamma_K^n$ is also a topological generator of Γ_K and g^n therefore generates a procyclic, and hence pro-p, subgroup of \mathcal{G}_p . The assignment $\gamma_K^n \mapsto g^n$ uniquely determines a continuous embedding $\sigma: \Gamma_K \hookrightarrow \mathcal{G}$ which splits the short exact sequence from the proposition.

Let $\Gamma_{\mathcal{G}} = \sigma(\Gamma_K) \cong \mathbb{Z}_p$. The splitting σ yields a semidirect product decomposition $\mathcal{G} = H \rtimes \Gamma_{\mathcal{G}}$ via the conjugation homomorphism $\Gamma_{\mathcal{G}} \to \operatorname{Aut}(H)$. But $\operatorname{Aut}(H)$ is finite and hence there exists an open subgroup $\Gamma = \Gamma_{\mathcal{G}}^{p^m} \trianglelefteq_o \Gamma_{\mathcal{G}}$ which is mapped to the identity automorphism, i.e. whose elements commute with those of H and therefore with all of \mathcal{G} . Since H is finite and Γ has no elements of finite order, one has $\Gamma \cap H = \{1\}$, from which the last statement follows.

It should be noted that the choice of Γ is far from unique. In fact, if Γ is as in the setting, then Γ^{p^n} is another valid choice for any $n \in \mathbb{N}$. In other words, we can move freely along the cyclotomic tower L_{∞}/L . This will be addressed in section 3.2.

We now introduce notation for some distinguished fields:

- M_S is defined as the maximal S-ramified (i.e. unramified outside S) extension of L_{∞} , and analogously for $M_{S\cup T}$.
- $M_{T,S}^{cs}$ is defined as the maximal T-ramified extension of L_{∞} which is completely split at S.
- $M_S^{cs} = M_{\emptyset,S}^{cs}$ is defined as the maximal unramified extension of L_{∞} which is completely split at S.

By maximality, the above fields are all Galois over L_{∞} , and so is $M_{S\cup T}$ over K (because L_{∞}/K is unramified outside $S \cup T$). We set $H_{S\cup T} = \text{Gal}(M_{S\cup T}/L_{\infty}) \leq_c G_{S\cup T} = \text{Gal}(M_{S\cup T}/K)$. The following diagram illustrates some of the relevant global fields and Galois groups:



One of the reasons we restrict ourselves to the cyclotomic \mathbb{Z}_p -extension is the fact that nonarchimedean places of the bottom field are *finitely split* in it, that is, they split only into finitely many places in the top field. The other main reason is the **weak Leopoldt conjecture**, which is known in the cyclotomic case. It asserts the boundedness of the so-called *Leopoldt defect* along a \mathbb{Z}_p -tower, although we shall make use of a cohomological formulation:

Theorem 1.0.3. Setting A. One has $H_2(H_{S\cup T}, \mathbb{Z}_p) = 0$.

The homology group is defined as in lemma 0.1.2 (or immediately before it) by regarding \mathbb{Z}_p as a $\Lambda(H_{S\cup T})$ -module, where the $H_{S\cup T}$ -action is trivial. A proof can be found in [NSW20] theorems 10.3.22 and 10.3.25.

The Iwasawa algebra at the centre of our construction is $\Lambda(\mathcal{G})$. Some general properties of such rings are discussed in section 0.1. $\Lambda(\mathcal{G})$ contains the integral domain $\Lambda(\Gamma) \cong \mathbb{Z}_p[[T]]$, over which it is a free module of finite rank equal to $[\mathcal{G}:\Gamma] = [L:K]$. In particular, a $\Lambda(\mathcal{G})$ -module is finitely generated if and only if it is so over $\Lambda(\Gamma)$. As a consequence, since $\Lambda(\Gamma)$ is Noetherian, so is $\Lambda(\mathcal{G})$. Unlike $\Lambda(\mathcal{G})$, the classical Iwasawa algebra $\Lambda(\Gamma)$ has many desirable properties, and restriction of scalars to it will be an essential algebraic tool when studying the $\Lambda(\mathcal{G})$ -modules of interest to us.

In setting A, we have fixed a prolongation v^c to \mathbb{Q}^c of each place v of K. If K'/K is Galois with group G, we may denote the decomposition group $G_{v(K')}$ simply by G_v - and we will often do so for $K' = L_{\infty}$. This is a slight abuse of notation as the choice of v(K') (that is, of v^c) is not canonical. However, the resulting objects will always be canonically isomorphic. The clearest example is the compact induction $\operatorname{Ind}_{\mathcal{G}_{w_{\infty}}}^{\mathcal{G}} M = \Lambda(\mathcal{G}) \widehat{\otimes}_{\Lambda(\mathcal{G}_{w_{\infty}})} M$ (section 0.1) of a compact $\Lambda(\mathcal{G}_{w_{\infty}})$ module M, where w_{∞} is an arbitrary prolongation of v to L_{∞} . If w'_{∞} is a different prolongation and $\sigma \in \mathcal{G}$ is any Galois automorphism sending w_{∞} to w'_{∞} , one has $\mathcal{G}_{w_{\infty}} = \sigma^{-1}\mathcal{G}_{w'_{\infty}}\sigma$. Consider the $\Lambda(\mathcal{G}_{w'_{\infty}})$ -module $\sigma^{-1}M$, which is identical to M as a \mathbb{Z}_p -module but has $\mathcal{G}_{w'_{\infty}}$ -action given by $\tau \cdot \sigma^{-1} m = (\sigma^{-1} \tau \sigma) m$. Then the continuous \mathbb{Z}_p -linear map given by

$$\operatorname{Ind}_{\mathcal{G}_{w_{\infty}}}^{\mathcal{G}} M \to \operatorname{Ind}_{\mathcal{G}_{w_{\infty}}}^{\mathcal{G}} \sigma^{-1} M$$
$$\lambda \otimes m \mapsto \sigma \lambda \sigma^{-1} \otimes m$$

for $\lambda \in \Lambda(\mathcal{G}), m \in M$ is a well-defined isomorphism of $\Lambda(\mathcal{G})$ -modules.

We are thus justified in using the notation \mathcal{G}_v for $\mathcal{G}_{v(L_\infty)}$, and in particular $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}$ – for $\operatorname{Ind}_{\mathcal{G}_{v(L_\infty)}}^{\mathcal{G}}$ –. What does play an important role in some formal arguments is the fact that the chosen set of prolongations is compatible, that is, $v^c | \cdots | v(L_\infty) | v(L_n) | v(L) | v$. A natural way to think of compact induction is as the sum of one copy of the local (i.e. $\Lambda(\mathcal{G}_v)$ -)module M for each place of L_∞ above v, endowed with a natural $\Lambda(\mathcal{G})$ -action by having \mathcal{G} permute these local components. This is especially clear if \mathcal{G}_v is open in \mathcal{G} , which is the case for all non-archimedean v.

A situation we will face repeatedly in this chapter is the following: suppose given an inverse system $(\{M_n\}, \{\varphi_n \colon M_{n+1} \to M_n\}, n \in \mathbb{N})$, where each M_n is a compact $\Lambda((\mathcal{G}_n)_v)$ -module and φ_n is a $\Lambda((\mathcal{G}_{n+1})_v)$ -homomorphism. Formally, one can regard this as an inverse system in the category of left $\Lambda(\mathcal{G})$ -modules via the canonical augmentation maps $\Lambda(\mathcal{G}) \twoheadrightarrow \Lambda(\mathcal{G}_n)$. The inductions $\operatorname{Ind}_{(\mathcal{G}_n)_v}^{\mathcal{G}_n} M_n$ then have natural transition maps from level n + 1 to level n, induced by the φ_n and the projection of the local component at a prime of L_{n+1} to the prime of L_n below it. The crucial idea now is the fact that the inverse limit of this new system coincides with the outcome of first taking the inverse limit $\varprojlim_n M_n$, which yields a $\Lambda(\mathcal{G}_v)$ -module, and only then inducing this local module to \mathcal{G} . More rigorously, there is a canonical $\Lambda(\mathcal{G})$ -isomorphism

$$\lim_{n} \operatorname{Ind}_{(\mathcal{G}_n)_v}^{\mathcal{G}_n} M_n \cong \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \lim_{n} M_n,$$
(1.1)

or in other words, *induction commutes with inverse limits*. Note that this is essentially part i) of lemma 0.1.1 together with the fact that a compact module coincides with the inverse limit of its modules of coinvariants (section 0.1).

This concludes the necessary preparations for the construction of the complex $\mathcal{C}^{\bullet}_{S,T}$.

1.1 The global complex $\mathcal{T}^{\bullet}_{S \cup T}$

Our global complex is given by a four-term exact sequence of $\Lambda(\mathcal{G})$ -modules, which shall be regarded as a complex concentrated in degrees 0 and 1. It is essentially an application of the *translation* functor defined by Ritter and Weiss (cf. [RW96]), so the aim of this brief section is to show what explicit form it takes in our setting.

For the sake of generality, we consider a finite set Σ of places of K containing $S_{\infty} \cup S_{\text{ram}}(L_{\infty}/K)$ - the choice of interest to us is $\Sigma = S \cup T$. As discussed above, the maximal Σ -ramified extension M_{Σ} of L_{∞} is Galois over both L_{∞} and K. We use the same notation $G_{\Sigma} = \text{Gal}(M_{\Sigma}/K)$ and $H_{\Sigma} = \text{Gal}(M_{\Sigma}/L_{\infty})$. In particular, $H_{\Sigma} \leq_{c} G_{\Sigma}$ and $\mathcal{G} \cong G_{\Sigma}/H_{\Sigma}$.

The starting point of the construction is the canonical short exact sequence of $\Lambda(G_{\Sigma})$ -modules

$$0 \to \Delta(G_{\Sigma}) \to \Lambda(G_{\Sigma}) \xrightarrow{\text{aug}} \mathbb{Z}_p \to 0$$
(1.2)

in the notation of section 0.1. Recall that taking H_{Σ} -coinvariants of a $\Lambda(G_{\Sigma})$ -module amounts to taking its quotient by $\Delta(G_{\Sigma}, H_{\Sigma})$, which is defined by the short exact sequence

$$0 \to \Delta(G_{\Sigma}, H_{\Sigma}) \to \Lambda(G_{\Sigma}) \to \Lambda(\mathcal{G}) \to 0$$
(1.3)

Applying the H_{Σ} -coinvariants functor to (1.2) yields a long exact sequence of $\Lambda(\mathcal{G})$ -modules

$$\dots \to H_1(H_{\Sigma}, \Lambda(G_{\Sigma})) \to H_1(H_{\Sigma}, \mathbb{Z}_p) \to \Delta(G_{\Sigma})_{H_{\Sigma}} \to \Lambda(G_{\Sigma})_{H_{\Sigma}} \to (\mathbb{Z}_p)_{H_{\Sigma}} \to 0,$$
(1.4)

the dual of (0.5). Looking at each term more closely, we have:

• $H_1(H_{\Sigma}, \Lambda(G_{\Sigma})) = 0$. Note that $H_1(H_{\Sigma}, \Lambda(G_{\Sigma})) \cong H^1(H_{\Sigma}, \Lambda(G_{\Sigma})^{\vee})^{\vee}$ by lemma 0.1.2 and

$$\begin{split} \Lambda(G_{\Sigma})^{\vee} &= \operatorname{Hom}_{cts} \left(\Lambda(G_{\Sigma}), \mathbb{Q}_{p} / \mathbb{Z}_{p} \right) \\ &\cong \varinjlim_{\substack{n \in \mathbb{N} \\ U \leq_{o} G_{\Sigma}}} \operatorname{Hom}_{cts} \left(\mathbb{Z} / p^{n} \mathbb{Z} \left[G_{\Sigma} / U \right], \mathbb{Q}_{p} / \mathbb{Z}_{p} \right) \\ &\cong \varinjlim_{\substack{U \leq_{o} G_{\Sigma}}} \operatorname{Map} \left(G_{\Sigma} / U, \mathbb{Q}_{p} / \mathbb{Z}_{p} \right) \\ &= \operatorname{Map}_{cts} \left(G_{\Sigma}, \mathbb{Q}_{p} / \mathbb{Z}_{p} \right) \end{aligned}$$

is G_{Σ} -induced. Therefore, it is also H_{Σ} -induced and hence cohomologically trivial by [NSW20] propositions 1.3.6 (ii) and 1.3.7.

- $H_1(H_{\Sigma}, \mathbb{Z}_p) \cong H_{\Sigma}^{ab}(p)$, the maximal abelian pro-*p* quotient of H_{Σ} , or the Galois group over L_{∞} of its maximal Σ -ramified abelian pro-*p* extension. This is the classical Σ -ramified Iwasawa module, which we denote by X_{Σ} .
- A middle term $\Delta(G_{\Sigma})_{H_{\Sigma}}$, which we denote by Y_{Σ} .
- $\Lambda(G_{\Sigma})_{H_{\Sigma}} \cong \Lambda(\mathcal{G})$ by (1.3).
- $(\mathbb{Z}_p)_{H_{\Sigma}} = \mathbb{Z}_p$, since the H_{Σ} -action on \mathbb{Z}_p is trivial.

Thus, (1.4) becomes a four-term exact sequence

$$0 \to X_{\Sigma} \to Y_{\Sigma} \to \Lambda(\mathcal{G}) \to \mathbb{Z}_p \to 0,$$

which proves the following:

Proposition 1.1.1. Setting A. Let Σ be a finite set of places of K containing $S_{\infty} \cup S_{ram}(L_{\infty}/K)$. Then there exists a cochain complex of $\Lambda(\mathcal{G})$ -modules concentrated in degrees 0 and 1

$$\mathcal{T}_{\Sigma}^{\bullet} = [\stackrel{0}{Y_{\Sigma}} \rightarrow \Lambda(\mathcal{G})]$$

such that $H^0(\mathcal{T}_{\Sigma}^{\bullet}) = X_{\Sigma}$ and $H^1(\mathcal{T}_{\Sigma}^{\bullet}) = \mathbb{Z}_p$. It is referred to as the global complex.

1.2 The local complexes \mathcal{L}_v^{\bullet}

We now use a similar process to construct a local complex \mathcal{L}_v at any place v of K (whether archimedean or non-archimedean). Several such complexes will be added together and induced to the global Galois group \mathcal{G} in order to define the main local-to-global morphism in the next section.

Consider first the case of a **non-archimedean place** v of K. Let us explore the connection between the global and local settings. By *local* we mean we consider the tower of fields

$$(\mathbb{Q}^c)_{v^c} / (L_{\infty})_{v(L_{\infty})} / L_{v(L)} / K_v,$$

where $(\mathbb{Q}^c)_{v^c}$ denotes the union $\varinjlim_{K'/\mathbb{Q}} K'_{v(K')}$ of the completions of all number fields K' at the compatible set of places determined by v^c . This field identifies naturally with the algebraic closure K_v^c of K_v (cf. [NSW20] proposition 8.1.5), which determines an isomorphism $(G_K)_{v^c} \cong G_{K_v}$. Analogous notation will be used for other infinite algebraic extensions of K. For instance, $(L_{\infty})_{v(L_{\infty})}$ is the union of the completions of all number fields contained in L_{∞} , and one has $\operatorname{Gal}((L_{\infty})_{v(L_{\infty})}/K_v) \cong \mathcal{G}_v$. We regard these isomorphisms, and similar ones below, as identifications. Their dependence on the choice of v^c is rendered irrelevant after taking induction, as explained in the introduction to this chapter.

The extension $(L_{\infty})_{v(L_{\infty})}/L_{v(L)}$ has Galois group $\Gamma_{v(L_{\infty})}$, which is open in Γ and hence has the form Γ^{p^n} for some $n \in \mathbb{N}$. This means that $L_{v(L)} = (L_n)_{v(L_n)}$, that is, localising may cause the first few (finitely many) layers of the cyclotomic \mathbb{Z}_p -extension to *collapse*. Note that the field $(L_{\infty})_{v(L_{\infty})}$ is still the cyclotomic \mathbb{Z}_p -extension of $L_{v(L)}$. In the non-*p*-adic case $v \notin S_p$, it must necessarily be the so-called *unramified* \mathbb{Z}_p -extension of $L_{v(L)}$, since that is the only \mathbb{Z}_p -extension of an *l*-adic field for $l \neq p$. To avoid further notational clutter, we denote the absolute Galois group of $(L_{\infty})_{v(L_{\infty})}$ by $G_{L_{\infty},v} \leq_c G_{K_v}$, instead of $G_{(L_{\infty})_{v(L_{\infty})}}$. As before, this coincides with the decomposition group of v^c in $G_{L_{\infty}}$.

As for the extension L/K, it is locally trivial $(L_{v(L)} = K_v)$ if v splits completely in it, and the converse is true if the extension is Galois.

Remark 1.2.1. The field $M_{S\cup T}$, which played a prominent role in the construction of the global complex, does not make an appearance in the local case. As the maximal $(S\cup T)$ -ramified extension of K, one could expect it to locally coincide with the entire algebraic closure of K_v for $v \in S \cup T$. However, this is a subtle question related to the global realisation of local extensions, and we shall only briefly touch on it in the next section. The correct top field to consider in the local case is K_v^c .

We now proceed along the same lines as in section 1.1, starting at the short exact sequence of $\Lambda(G_{K_v})$ -modules

$$0 \to \Delta(G_{K_v}) \to \Lambda(G_{K_v}) \to \mathbb{Z}_p \to 0$$

and taking $G_{L_{\infty,v}}$ -coinvariants. This yields a long exact sequence of compact $\Lambda(\mathcal{G}_v)$ -modules

$$\dots \to H_1(G_{L_{\infty,v}}, \Lambda(G_{K_v})) \to H_1(G_{L_{\infty,v}}, \mathbb{Z}_p) \to \Delta(G_{K_v})_{G_{L_{\infty,v}}} \to \Lambda(G_{K_v})_{G_{L_{\infty,v}}} \to (\mathbb{Z}_p)_{G_{L_{\infty,v}}} \to 0.$$
(1.5)

By the same argument as for the global sequence (1.4), this reduces to

$$0 \to G^{ab}_{L_{\infty,v}}(p) \to Y_v \to \Lambda(\mathcal{G}_v) \to \mathbb{Z}_p \to 0,$$

where $Y_v = \Delta(G_{K_v})_{G_{L_{\infty,v}}}$. Note that $G^{ab}_{L_{\infty,v}}(p)$ is the Galois group over $(L_{\infty})_{v(L_{\infty})}$ of its maximal abelian pro-*p* extension.

We now move on to the case of an **archimedean place** v of K. The completions at places above v then have the usual interpretation as \mathbb{R} or \mathbb{C} according on whether those places are real or complex, but this will not play a part in the construction. Decomposition groups of archimedean places are *very small*, either trivial or of order 2 generated by complex conjugation. This simplifies the argument considerably, with the caveat that inducing from \mathcal{G}_v to \mathcal{G} , as we will do later on, is no longer a finite process - now decomposition groups are not open.

This time we directly consider the short exact sequence

$$0 \to \Delta(\mathcal{G}_v) \to \Lambda(\mathcal{G}_v) \to \mathbb{Z}_p \to 0 \tag{1.6}$$

and distinguish two cases:

- i) If $v(L_{\infty})$ is unramified in L_{∞}/K , then $\Delta(\mathcal{G}_v)$ is trivial and the above sequence is an equality $\mathbb{Z}_p = \mathbb{Z}_p$. This is always the case if p = 2 by assumption.
- ii) If $v(L_{\infty})$ is ramified in L_{∞}/K , let $\tau_{v(L_{\infty})}$ denote the complex conjugation induced by $v(L_{\infty})$ on L_{∞} . We then have a canonical decomposition

$$\Lambda(\mathcal{G}_v) \cong \left(rac{1+ au_{v(L_\infty)}}{2}
ight)\Lambda(\mathcal{G}_v)\oplus \left(rac{1- au_{v(L_\infty)}}{2}
ight)\Lambda(\mathcal{G}_v)$$

(because $2 \neq p$ is invertible in \mathbb{Z}_p), the two coefficients being precisely the primitive (central) idempotents of the semisimple algebra $\mathbb{Q}_p^c[\mathcal{G}_v]$ (cf. section 0.2). The two summands on the righthand side are \mathbb{Z}_p -free of rank 1 (generated by these idempotents), sometimes denoted by \mathbb{Z}_p^+ and \mathbb{Z}_p^- . They are the maximal submodules of $\Lambda(\mathcal{G}_v)$ on which $\tau_{v(L_\infty)}$ acts as the identity and as -1, respectively. The augmentation ideal $\Delta(\mathcal{G}_v)$ coincides with \mathbb{Z}_p^- , and mapping $z \in \mathbb{Z}_p \subseteq \Lambda(\mathcal{G}_v)$ to $z(1 + \tau_{v(L_\infty)})/2$ makes the diagram

commute.

Remark 1.2.2. i) Instead of starting directly at (1.6), we could have considered the analogous sequence for G_{K_v} and taken $G_{L_{\infty},v}$ -coinvariants as we did in the non-archimedean case. Note

that the term $H_1(G_{L_{\infty},v}, \mathbb{Z}_p)$ of the analogous sequence to (1.5) would vanish, since $G_{L_{\infty},v}$ (of order 2) has no non-trivial pro-*p* quotient. This approach leads to the very same complexes.

ii) The ramification (or splitting) behaviour of an archimedean place in L_{∞}/K is already determined at L/K, since these places are never ramified in \mathbb{Z}_p -extensions: their decomposition groups are finite, but \mathbb{Z}_p has no non-trivial finite subgroups.

In order to uniformise the notation, we set $Y_v = \Delta(G_{K_v})_{G_{L_{\infty,v}}}$ as before. By the previous remark, this module is trivial if v is unramified in L_{∞}/K , and \mathbb{Z}_p^- otherwise. We can now collect the constructions from this section:

Proposition 1.2.3. Setting A. For every place v of K there exists a cochain complex of $\Lambda(\mathcal{G}_v)$ -modules

$$\mathcal{L}_v^{\bullet} = [\overset{0}{Y_v} \to \Lambda(\overset{1}{\mathcal{G}_v})]$$

such that $H^0(\mathcal{L}_v^{\bullet}) = G^{ab}_{L_{\infty},v}(p)$ and $H^1(\mathcal{L}_v^{\bullet}) = \mathbb{Z}_p$. If v is archimedean, then $H^0(\mathcal{L}_v^{\bullet}) = 0$ and

$$\mathcal{L}_{v}^{\bullet} = \begin{cases} [0 \to \mathbb{Z}_{p}], & v \text{ is unramified in } L_{\infty}/K\\ [\Delta(\mathcal{G}_{v}) \to \Lambda(\mathcal{G}_{v})], & otherwise. \end{cases}$$

It is referred to as the local complex at v.

Although this construction works for arbitrary v, the definition of our main complex only relies on the local complexes \mathcal{L}_v^{\bullet} for places $v \in S$.

1.3 Local-to-global maps

At this point, we have defined a global complex $\mathcal{T}^{\bullet}_{S\cup T}$ and local complexes $\mathcal{L}^{\bullet}_{v}$. It is time to define morphisms from the latter to the former, which we shall then induce from \mathcal{G}_{v} to \mathcal{G} . As mentioned at the beginning of this chapter, conceptually, induction amounts to adding a copy of $\mathcal{L}^{\bullet}_{v}$ for each prolongation of v to L_{∞} and justifies the notation $\mathcal{L}^{\bullet}_{v}$ for an object which is actually defined at $v(L_{\infty})$. Before applying induction, the goal is therefore to construct morphisms of complexes of $\Lambda(\mathcal{G}_{v})$ -modules $\mathcal{L}^{\bullet}_{v} \to \mathcal{T}^{\bullet}_{S\cup T}$.

We first address degree 0 - that is, the modules Y_v and $Y_{S\cup T}$. For a place $v \in S$, consider the continuous group homomorphism f_v given by either composition in the commutative diagram

in the notation of the previous sections. Recall here that the injection $G_{K_v} \hookrightarrow G_K$ relies on the choice of a prolongation v^c of v to \mathbb{Q}^c made in setting A. The map f_v induces a continuous ring homomorphism $\Lambda(G_{K_v}) \to \Lambda(G_{S \cup T})$ which commutes with the augmentation and hence restricts to

 $\Delta(G_{K_v}) \to \Delta(G_{S \cup T})$. Since $f_v(G_{L_{\infty,v}}) \subseteq H_{S \cup T}$, the image of $\Delta(G_{L_{\infty,v}})$ under this homomorphism is contained in $\Delta(H_{S \cup T})$, which allows us to define a homomorphism of $\Lambda(\mathcal{G}_v)$ -modules

$$\alpha_v^0: \Delta(G_{K_v}) / \Delta(G_{L_{\infty,v}}) \Delta(G_{K_v}) \to \Delta(G_{S \cup T}) / \Delta(H_{S \cup T}) \Delta(G_{S \cup T}),$$

that is, $Y_v \to Y_{S\cup T}$ (cf. (0.2)). Note that this argument does not depend on whether v is archimedean or not: as shown in remark 1.2.2, the description of the degree-zero term Y_v of \mathcal{L}_v^{\bullet} as $\Delta(G_{K_v})_{G_{L_{\infty,v}}}$ applies to both archimedean and non-archimedean v.

Before continuing with the construction, we briefly touch on the natural question: for v nonarchimedean, should we expect $f_v: G_{K_v} \to G_{S \cup T}$ to be injective or surjective? The answer is, in general, no to both. The injectivity of f_v is equivalent to that of the left vertical arrow in (1.7), i.e. to whether $(M_{S \cup T})_{v(M_{S \cup T})}$ coincides with K_v^c . This is known to be the case for very large (of density 1, in particular infinite) sets $S \cup T$ (see for instance [NSW20] theorem 9.4.3), but that is far from our situation. As to the failure of surjectivity in general, [NSW20] theorem 10.8.1 contains the following example: if K is not totally real and $v \notin S_p$, then the image of f_v is far from all of $G_{S \cup T}$. Note that the cited theorem refers to the maximal pro-p quotients of these groups, but this implies f_v itself cannot be surjective.

The maps in degree 1 are substantially simpler: we let

$$\alpha_v^1 \colon \Lambda(\mathcal{G}_v) \hookrightarrow \Lambda(\mathcal{G})$$

be the canonical embedding. Together with α_v^0 , this induces a morphism of complexes of $\Lambda(\mathcal{G}_v)$ modules $\alpha_v \colon \mathcal{L}_v^{\bullet} \to \mathcal{T}_{S \cup T}^{\bullet}$, which is to say that the diagram

$$Y_v \longrightarrow \Lambda(\mathcal{G}_v)$$
$$\downarrow^{\alpha_v^0} \qquad \qquad \downarrow^{\alpha_v^1}$$
$$Y_{S \cup T} \longrightarrow \Lambda(\mathcal{G})$$

commutes. Indeed, all the arrows are homomorphisms of topological \mathbb{Z}_p -algebras, so it suffices to verify this for the classes in $Y_v = \Delta(G_{K_v})_{G_{L_{\infty,v}}}$ of the generators $\{\sigma - 1 : \sigma \in G_{K_v}\}$ of $\Delta(G_{K_v})$. But on these elements, commutativity is clear.

Now that we have the desired maps, we induce the local complexes from \mathcal{G}_v to all of \mathcal{G} . The first important realisation is that, on the exact sequence

$$0 \to H^0(\mathcal{L}_v^{\bullet}) \to [Y_v \to \Lambda(\mathcal{G}_v)] \to H^1(\mathcal{L}_v^{\bullet}) \to 0$$

defining $\mathcal{L}_{v}^{\bullet}$, the compact induction functor $\operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} - = \Lambda(\mathcal{G}) \widehat{\otimes}_{\Lambda(\mathcal{G}_{v})} -$ introduced in section 0.1 coincides with $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}_{v})} -$ and is exact:

- If v is non-archimedean, then \mathcal{G}_v is open in \mathcal{G} and therefore $\Lambda(\mathcal{G})$ is a free right $\Lambda(\mathcal{G}_v)$ -module of finite rank the generators being any set of left coset representatives of $\mathcal{G}/\mathcal{G}_v$.
- If v is archimedean, all $\Lambda(\mathcal{G}_v)$ -modules in \mathcal{L}_v^{\bullet} and its cohomology $(\Delta(\mathcal{G}_v), \Lambda(\mathcal{G}_v))$ and $\mathbb{Z}_p)$ are finitely generated, and hence finitely presented. Note here that $\Lambda(\mathcal{G}_v)$ is Noetherian as it is finitely generated as a \mathbb{Z}_p -module.

In both cases, equation (0.3) shows the desired equality of functors. Since $\Lambda(\mathcal{G})$ is a flat right $\Lambda(\mathcal{G}_v)$ -module (this is even true after replacing \mathcal{G}_v by any closed subgroup of \mathcal{G} by [Ven] lemma B.1), exactness follows. In other words, the complex

$$\operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}}\mathcal{L}_{v}^{\bullet} = [\operatorname{Ind}_{\mathcal{G}_{v}}^{0}Y_{v} \rightarrow \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}}\Lambda(\mathcal{G}_{v})]$$

has cohomology groups

$$H^{i}(\operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}}\mathcal{L}_{v}^{\bullet}) = \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}}H^{i}(\mathcal{L}_{v}^{\bullet}).$$

$$(1.8)$$

 $\mathcal{T}^{\bullet}_{S \cup T}$ has the structure of a complex of $\Lambda(\mathcal{G})$ -(and not only $\Lambda(\mathcal{G}_v)$ -)modules, so induction yields a morphism of complexes of $\Lambda(\mathcal{G})$ -modules

$$\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \alpha_v \colon \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathcal{L}_v^{\bullet} \to \mathcal{T}_{S \cup T}^{\bullet}$$

as mentioned in (0.4). Some of the resulting maps will be studied more closely when determining the cohomology of the main complex in the next section.

1.4 The main complex $C_{S,T}^{\bullet}$

We may now bring together the construction of the local and global complexes, as well as the morphisms between them, into the definition of our complex of interest. On the local side, the complexes at all places $v \in S$ are combined into

$$\mathcal{L}_{S}^{\bullet} = \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathcal{L}_{v}^{\bullet}.$$

The maps from the last section then produce a morphism $\alpha_{S,T} = \sum_{v \in S} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \alpha_v \colon \mathcal{L}_S^{\bullet} \to \mathcal{T}_{S \cup T}^{\bullet}$.

Definition 1.4.1. Setting A. The main complex $\mathcal{C}_{S,T}^{\bullet}$ of $\Lambda(\mathcal{G})$ -modules is defined as

$$\mathcal{C}_{S,T}^{\bullet} = \operatorname{Cone}(\mathcal{L}_{S}^{\bullet} \xrightarrow{\alpha_{S,T}} \mathcal{T}_{S\cup T}^{\bullet})[-1].$$

The fact that this complex is perfect will be essential for the formulation of the Main Conjecture. We recall that a **strictly perfect** complex is a bounded complex consisting of finitely generated projective modules, and a **perfect** complex is one which is isomorphic to a strictly perfect complex in the relevant derived category - in this case, $\mathcal{D}(\Lambda(\mathcal{G}))$. The perfectness of $\mathcal{C}_{S,T}^{\bullet}$ boils down to the following technical result from the theory of presentations of Iwasawa modules (cf. [NSW20] section V§6):

Lemma 1.4.2. Let $N \hookrightarrow G \twoheadrightarrow Q$ be a short exact sequence of profinite groups such that G is topologically finitely generated and has cohomological p-dimension $\operatorname{cd}_p G \leq 2$, and $H_2(N, \mathbb{Z}_p) = 0$. Then the $\Lambda(Q)$ -module $Y_{G,N} = \Delta(G)_N$ has projective dimension $\operatorname{pd}_{\Lambda(Q)} Y_{G,N} \leq 1$. In particular, the complex of $\Lambda(Q)$ -modules

$$C^{\bullet} = [Y^{0}_{G,N} \xrightarrow{v} \Lambda^{1}(Q)]$$

is perfect.

Proof. By proposition [NSW20] 5.6.7, there exists an exact sequence of $\Lambda(Q)$ -modules

$$0 \to H_2(N, \mathbb{Z}_p) \to (M^{ab}(p))_N \to \Lambda(Q)^d \to Y_{G,N} \to 0,$$
(1.9)

where $M^{ab}(p)$ is the so called *p*-relation module of G with respect to a presentation

$$1 \to M \to F \to G \to 1$$

of G by some free profinite group F of rank d - which is not unique: larger d simply results in a larger kernel M. Then $M^{ab}(p)$ has a natural $\Lambda(G)$ -module structure (with G acting by lifting and conjugating), which in turn makes $(M^{ab}(p))_N$ into a $\Lambda(Q)$ -module. The latter is projective whenever $\operatorname{cd}_p G \leq 2$ by the same proposition, which together with the triviality of $H_2(N, \mathbb{Z}_p)$ shows that (1.9) is a projective resolution of $Y_{G,N}$ of length at most 1.

The last claim in the lemma follows from the classical fact that a bounded complex consisting of modules of finite projective dimension is perfect - which can be shown by induction on the length of the complex using the fact that the cone of a morphism of perfect complexes is perfect.

The consequence of relevance to us is the following:

Proposition 1.4.3. Setting A. The global complex $\mathcal{T}^{\bullet}_{S\cup T}$ and the main complex $\mathcal{C}^{\bullet}_{S,T}$ are perfect complexes of $\Lambda(\mathcal{G})$ -modules. For any place v of K, the local complex $\mathcal{L}^{\bullet}_{v}$ is a perfect complex of $\Lambda(\mathcal{G}_{v})$ -modules.

Proof. Both the global and local complexes are of the form considered in lemma 1.4.2. In the global case $\mathcal{T}_{S\cup T}^{\bullet}$, we apply it to the short exact sequence of profinite groups $H_{S\cup T} \hookrightarrow G_{S\cup T} \twoheadrightarrow \mathcal{G}$ in the notation of section 1.1. The group $G_{S\cup T}$ is topologically finitely generated by [NSW20] corollary 10.11.15 and has $\operatorname{cd}_p G_{S\cup T} = 2$ by [NSW20] proposition 10.11.3. The hypothesis $H_2(H_{S\cup T}, \mathbb{Z}_p) = 0$ holds by the validity of the weak Leopoldt conjecture (theorem 1.0.3).

For the local complex \mathcal{L}_v^{\bullet} , we distinguish two cases. If v is archimedean, then \mathcal{L}_v^{\bullet} is in fact strictly perfect as seen directly from proposition 1.2.3. Recall at this point that, in the archimedean ramified case (where $p \neq 2$ by assumption), $\Delta(\mathcal{G}_v) \cong \mathbb{Z}_p^-$ is a direct summand of $\Lambda(\mathcal{G}_v)$ and hence projective.

For non-archimedean v, we again apply lemma 1.4.2 to $G_{L_{\infty},v} \hookrightarrow G_{K_v} \twoheadrightarrow \mathcal{G}_v$ in the notation of section 1.2. This time around, theorems 7.4.1 and 7.1.8 from [NSW20] give finite topological generatedness of G_{K_v} and $\operatorname{cd}_p G_{K_v} = 2$, respectively. The vanishing of $H_2(G_{L_{\infty},v}, \mathbb{Z}_p) \cong H^2(G_{L_{\infty},v}, \mathbb{Q}_p/\mathbb{Z}_p)^{\vee}$ (cf. lemma 0.1.2) follows from

$$H^{2}(G_{L_{\infty},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}) = \varinjlim_{n} H^{2}(G_{L_{n},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}) = 0,$$

where $G_{L_n,v}$ denotes the absolute Galois group of $(L_n)_{v(L_n)}$ and the equalities are [NSW20] propositions 1.5.1 and 7.3.10.

In order to conclude that the main complex is perfect, we first note that $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathcal{L}_v^{\bullet}$ is perfect over $\Lambda(\mathcal{G})$ for any v: both modules in $[Y_v \to \Lambda(\mathcal{G}_v)]$ have projective dimension ≤ 1 by the previous proposition, and compact induction sends projective resolutions to projective resolutions. Hence, as (a shift of) the cone of a morphism between perfect complexes, $\mathcal{C}_{S,T}^{\bullet} = \operatorname{Cone}(\mathcal{L}_S^{\bullet} \xrightarrow{\alpha_{S,T}} \mathcal{T}_{S \cup T}^{\bullet})[-1]$ is itself perfect.

We now turn our attention to the cohomology⁸ of the main complex. Its determination involves the following $\Lambda(\mathcal{G})$ -modules, all of which arise by taking classical Galois modules, extending scalars to \mathbb{Z}_p and passing to the inverse limit over the cyclotomic tower L_{∞}/L :

- $E_{S,T} = \varprojlim_n (\mathbb{Z}_p \otimes \mathcal{O}^*_{L_n,S,T})$, where $\mathcal{O}^*_{L_n,S,T}$ is the group of (S,T)-units of L_n (cf. (0.23)) and the inverse limit is taken with respect to the norm maps.
- $E_S = E_{S,\varnothing} = \varprojlim_n (\mathbb{Z}_p \otimes \mathcal{O}^*_{L_n,S})$, where $\mathcal{O}^*_{L_n,S}$ is the group of S-units of L_n (cf. (0.22)).
- $\mathcal{Y}_S = \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p$, where \mathbb{Z}_p is endowed with the trivial \mathcal{G}_v -action. This is an instance of the situation described by equation (1.1) with $M_n = \mathbb{Z}_p$ for all n and transition maps $M_{n+1} \to M_n$ equal to the identity. Therefore, \mathcal{Y}_S can be identified with the inverse limit of

$$\mathcal{Y}_{L_n,S} = \bigoplus_{v \in S} \operatorname{Ind}_{(\mathcal{G}_n)_{v(L_n)}}^{\mathcal{G}_n} \mathbb{Z}_p \cong \bigoplus_{w_n \in S(L_n)} \mathbb{Z}_p \cdot w_n,$$
(1.10)

where the last term is a free \mathbb{Z}_p -module with \mathcal{G}_n -action given by Galois conjugation of places.

By (1.8), \mathcal{Y}_S coincides with $H^0(\mathcal{L}_S^{\bullet})$

• $\mathcal{X}_S = \ker(\mathcal{Y}_S \twoheadrightarrow \mathbb{Z}_p)$ is the kernel the of canonical projection (which is essentially an augmentation map). For the same reason as above, it can be identified with the inverse limit of the kernels $\mathcal{X}_{L_n,S} = \ker(\mathcal{Y}_{L_n,S} \twoheadrightarrow \mathbb{Z}_p)$. Note that $\mathcal{X}_{L_n,S}$ is simply $\mathbb{Z}_p \otimes \mathcal{X}_{L_n,S}^{\mathbb{Z}}$ in the notation of section 0.3, and analogously for $\mathcal{Y}_{L_n,S}$.

The module \mathcal{X}_S should not be confused with X_S , the Galois group over L_∞ of its maximal *S*-ramified abelian pro-*p* extension; and neither should \mathcal{Y}_S with the $Y_S = \Delta(G_S)_{H_S}$ defined in section 1.1.

In subsequent chapters, we will use analogous notation $E_{S',T'}, E_{S'}, \mathcal{Y}_{S'}, \mathcal{X}_{S'}$ for sets of places S'and T' other than S and T. These modules will be instrumental in the formulation of the Main Conjecture (see the beginning of chapter 2, in particular the homomorphism $\alpha \colon \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S,T}$). All of them are finitely generated over $\Lambda(\mathcal{G})$ (for E_S , this follows from its structure as a $\Lambda(\Gamma)$ -module

⁸The structure of these cohomology groups will also follow from the isomorphism (in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$) between $\mathcal{C}_{S,T}^{\bullet}$ and a certain $R\Gamma$ -complex proved in section 1.5. We compute them explicitly here mainly for two reasons: first, this involves the study of some classical objects and sequences which will be used in later chapters. Second, doing so provides a self-contained argument which does not rely on known results for the $R\Gamma$ -complex. As a minor additional justification, section 1.5 requires \mathcal{G} to be abelian for compatibility with [BKS17], although this is only formal - the same isomorphism holds in the general case.

given in [NSW20] theorem 11.3.11; and for $E_{S,T} \subseteq E_S$, this is a consequence of the Noetherianity of $\Lambda(\mathcal{G})$).

In preparation for the computation of the cohomology of $\mathcal{C}^{\bullet}_{S,T}$, we determine the kernel and cokernel of

$$H^0(\alpha_{S,T})\colon H^0(\mathcal{L}_S^{\bullet})\to H^0(\mathcal{T}_{S\cup T}^{\bullet}).$$

By section 1.2, $H^0(\mathcal{L}^{\bullet}_S) = \bigoplus_{v \in S_f} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} G^{ab}_{L_{\infty},v}(p)$. Local class field theory yields an isomorphism of $\Lambda(\mathcal{G}_v)$ -modules

$$G^{ab}_{L_{\infty},v}(p) \cong \varprojlim_{n} \left((L_{n})^{*}_{v(L_{n})} \right)^{\wedge},$$

where

$$\left(\left(L_{n}\right)_{v(L_{n})}^{*}\right)^{\wedge} = \varprojlim_{m} \left(L_{n}\right)_{v(L_{n})}^{*} / \left(\left(L_{n}\right)_{v(L_{n})}^{*}\right)^{p^{m}}$$

denotes the *p*-completion of the group of units of the local field $(L_n)_{v(L_n)}$. Thus, $H^0(\mathcal{L}_S^{\bullet})$ identifies with the Iwasawa module A_S in the notation of [NSW20] section XI§3 (note that, for archimedean places $v \in S_{\infty}$, there is nothing to add as cohomology vanishes in degree 0). It should be pointed out that the definitions of $E_{S,T}$ and E_S (and also \mathcal{Y}_S and \mathcal{X}_S) boil down to a limit of *p*-completions as well, but in those instances *p*-completion amounts to $\mathbb{Z}_p \otimes -$ because the abelian groups involved are finitely generated.

Proposition 1.4.4. Setting A. There exists a short exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to E_{S,T} \to H^0(\mathcal{L}^{\bullet}_S) \xrightarrow{H^0(\alpha_{S,T})} H^0(\mathcal{T}^{\bullet}_{S\cup T}) \to X^{cs}_{T,S} \to 0,$$

where $X_{T,S}^{cs} = \text{Gal}(M_{T,S}^{cs}/L_{\infty})^{ab}(p)$ is the Galois group over L_{∞} of its maximal T-ramified abelian pro-p extension which is completely split at S.

Proof. The case $T = \emptyset$, i.e. $H^0(\alpha_{S,\emptyset}): H^0(\mathcal{L}_S^{\bullet}) \to H^0(\mathcal{T}_S^{\bullet}) = X_S$, is settled in [NSW20] theorem⁹ 11.3.10: the kernel and cokernel are isomorphic to E_S and $X_{\emptyset,S}^{cs}$, respectively, where the latter is usually simply denoted by X_S^{cs} . In applying the cited theorem, we note that $H_2(H_S, \mathbb{Z}_p) = 0$ by the weak Leopoldt conjecture (theorem 1.0.3). The fact that $H^0(\alpha_{S,\emptyset})$ coincides with the $A_S \to X_S$ therein under $H^0(\mathcal{L}_S^{\bullet}) \cong A_S$ is a consequence of our definition of the local-to-global-maps, which are induced by natural inclusions and projections of Galois groups.

We therefore need to measure the change caused by the introduction of T. Consider the commutative diagram with exact rows

$$0 \longrightarrow 0 \longrightarrow H^{0}(\mathcal{L}_{S}^{\bullet}) \Longrightarrow H^{0}(\mathcal{L}_{S}^{\bullet}) \longrightarrow 0$$

$$\downarrow \qquad \qquad \downarrow^{H^{0}(\alpha_{S,T})} \qquad \downarrow^{H^{0}(\alpha_{S,\varnothing})}$$

$$0 \longrightarrow \ker(\pi) \longrightarrow X_{S \cup T} \xrightarrow{\pi} X_{S} \longrightarrow 0$$

⁹Because of the hypothesis $p \nmid [L:K]$ in p. 739 of the reference, the cited theorem does not directly refer to the $\Lambda(\mathcal{G})$ -module structure of the objects. However, it can still be used to determine the kernel and cokernel of $H^0(\alpha_{S,\emptyset})$ as a $\Lambda(\Gamma)$ -homomorphism (i.e. choosing L rather than K as the base field), after which we simply point out that the maps $E_S \hookrightarrow A_S \cong H^0(\alpha_{S,\emptyset})$ and $X_S \twoheadrightarrow X_S^{cs}$ are \mathcal{G} -equivariant.

where π is the natural projection - that is, restriction of Galois automorphisms. The snake lemma then yields an exact sequence

$$0 \to \ker(H^0(\alpha_{S,T})) \to E_S \to \ker(\pi) \to \operatorname{coker}(H^0(\alpha_{S,T})) \to X_S^{cs} \to 0.$$
(1.11)

In order to determine the unknown terms, we construct the same sequence in a different manner, starting at the exact sequence of $\mathbb{Z}[\mathcal{G}_n]$ -modules

$$1 \to \mathcal{O}_{L_n,S,T}^* \to \mathcal{O}_{L_n,S}^* \to \bigoplus_{w_n \in T(L_n)} \kappa(w_n)^* \to Cl_{L_n,S,T} \to Cl_{L_n,S} \to 1,$$
(1.12)

where $Cl_{L_n,S}$ and $Cl_{L_n,S,T}$ denote the S-class group and (S,T)-ray class group at the layer L_n , respectively. These are the quotients of the class group and ray class group mod $\mathfrak{m}_T = \prod_{w_n \in T(L_n)} w_n$ by the subgroup generated by classes of ideals in S_f (see [Nic13] subsection 1.4 for the precise definition and the above sequence). Applying $\mathbb{Z}_p \otimes -$ is exact and yields a new sequence whose last three terms (not counting 1) are the *p*-parts of those above, since they are finite abelian groups. In particular, $\mathbb{Z}_p \otimes \kappa(w_n)^* = \kappa(w_n)^*(p)$ is isomorphic to the group of *p*-power roots of unity of $(L_n)_{w_n}$ (by definition, places above *T* cannot be *p*-adic).

All terms resulting from tensoring (1.12) with \mathbb{Z}_p are compact $\Lambda(\mathcal{G}_n)$ -modules, as they are finitely generated over \mathbb{Z}_p (for the first two, this follows from Dirichlet's unit theorem: see (0.24)). Therefore, taking inverse limits along the cyclotomic tower with respect to the norm maps produces a sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to E_{S,T} \to E_S \to \bigoplus_{v \in T^p} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1) \to X_{T,S}^{cs} \to X_S^{cs} \to 0$$
(1.13)

which is exact by [NSW20] proposition 5.2.4. In the middle term, we have used the fact that induction commutes with inverse limits (recall (1.1)). Note that the group of *p*-power roots of unity in the cyclotomic \mathbb{Z}_p -extension of $L_{v(L)}$ is trivial if $L_{v(L)}$ itself contains no primitive *p*-th root of unity, and $\mu_{p^{\infty}}$ (i.e. all possible) otherwise. Hence the only terms that survive in the limit are those corresponding to places $v \in T^p$ as defined in setting A, which yield precisely $\mathbb{Z}_p(1)$ (see equation (0.6)). Finally, the fact that the limits of $Cl_{L_n,S,T}$ and $Cl_{L_n,S}$ are precisely $X_{T,S}^{cs}$ and X_S^{cs} is a well-known consequence of global class field theory.

The last step is to connect sequences (1.11) and (1.13). The middle terms $\bigoplus_{v \in T^p} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1)$ and ker(π) are isomorphic by [NSW20] theorem 11.3.5 (the isomorphism of the summands with $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1)$ is addressed only in the proof) - this follows from applying it to $S \cup T$ and S separately, then using the snake lemma. For the fourth terms, we use the fact that the canonical projection $X_{S\cup T} \twoheadrightarrow X_{T,S}^{cs}$ is trivial on the image of $H^0(\mathcal{L}_S^{\bullet})$, as the latter is generated precisely by the projections to $X_{S\cup T}$ of the decomposition groups $G_{L_{\infty},v}^{ab}(p) = (G_{L_{\infty}})_{v^c}^{ab}(p)$ for $v \in S$. It therefore factors as ε : coker $(H^0(\alpha_{S,T})) \twoheadrightarrow X_{T,S}^{cs}$. This yields a diagram with exact rows

where the first vertical arrow is defined by commutativity of the corresponding square, and all other squares commute by the aforementioned naturality of the maps. Now a simple diagram chase shows the first and fourth vertical arrows are isomorphisms as well.

Remark 1.4.5. The argument used to define ε : coker $(H^0(\alpha_{S,T})) \rightarrow X_{T,S}^{cs}$ above is not far from showing directly that it is an isomorphism. The bulk of the proof is aimed at determining the cohomology in degree 0, which is subtler as it relies on class field theory. The proof also showcases some objects and techniques which will be relevant later on - for instance, in subsection 3.2.3.

We can now determine the cohomology of the main complex:

Theorem 1.4.6. Setting A. The perfect complex $\mathcal{C}_{S,T}^{\bullet}$ is acyclic outside degrees 0 and 1 and satisfies:

- $H^0(\mathcal{C}^{\bullet}_{S,T}) \cong E_{S,T}.$
- $H^1(\mathcal{C}^{\bullet}_{ST})$ fits into the short exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to X_{T,S}^{cs} \to H^1(\mathcal{C}_{S,T}^{\bullet}) \to \mathcal{X}_S \to 0,$$

with $X_{T,S}^{cs}$ and \mathcal{X}_S as defined in and before proposition 1.4.4, respectively.

Proof. In the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$, the complex fits in an exact triangle

$$\mathcal{C}^{\bullet}_{S,T} \to \mathcal{L}^{\bullet}_{S} \xrightarrow{\alpha_{S,T}} \mathcal{T}^{\bullet}_{S \cup T} \to$$

by definition. The cohomology of $\mathcal{L}_{S}^{\bullet}$ and $\mathcal{T}_{S\cup T}^{\bullet}$ is given by propositions 1.1.1 and 1.2.3, and it is trivial outside degrees 0 and 1 in both cases. Therefore, the long exact cohomology sequence becomes

$$0 \longrightarrow H^{0}(\mathcal{C}^{\bullet}_{S,T}) \longrightarrow H^{0}(\mathcal{L}^{\bullet}_{S}) \xrightarrow{H^{0}(\alpha_{S,T})} H^{0}(\mathcal{T}^{\bullet}_{S\cup T}) \longrightarrow$$

$$\longleftrightarrow H^{1}(\mathcal{C}^{\bullet}_{S,T}) \longrightarrow H^{1}(\mathcal{L}^{\bullet}_{S}) \xrightarrow{H^{1}(\alpha_{S,T})} H^{1}(\mathcal{T}^{\bullet}_{S\cup T}) \longrightarrow H^{2}(\mathcal{C}^{\bullet}_{S,T}) \longrightarrow 0$$

and $\mathcal{C}^{\bullet}_{S,T}$ is acyclic outside degrees 0, 1 and 2. It follows that $H^0(\mathcal{C}^{\bullet}_{S,T})$ is isomorphic to the kernel of $H^0(\alpha_{S,T})$ and $H^1(\mathcal{C}^{\bullet}_{S,T})$ fits into the short exact sequence

$$0 \to \operatorname{coker}(H^0(\alpha_{S,T})) \to H^1(\mathcal{C}^{\bullet}_{S,T}) \to \ker(H^1(\alpha_{S,T})) \to 0.$$

As already discussed, $H^1(\mathcal{L}_S^{\bullet}) \cong \mathcal{Y}_S$ and the map $H^1(\alpha_{S,T}) \colon H^1(\mathcal{L}_S^{\bullet}) \to H^1(\mathcal{T}_{S\cup T}^{\bullet}) = \mathbb{Z}_p$ identifies with the canonical projection $\mathcal{Y}_S \twoheadrightarrow \mathbb{Z}_p$, whose kernel is \mathcal{X}_S by definition. Since it is surjective, $H^2(\mathcal{C}_{S,T}^{\bullet})$ is trivial. The kernel and cokernel of $H^0(\alpha_{S,T})$ were computed in proposition 1.4.4, which concludes the proof.

1.5 Description in terms of $R\Gamma$ -complexes

This section is devoted to showing that the complex constructed in this chapter is isomorphic to that employed by Burns, Kurihara and Sano in [BKS17] (pp. 1534 and 1539-1540) in the cases where the settings coincide. This is a necessary step for section 4.1, where we prove that the Main Conjectures formulated here (section 2.5) and in said article (conjecture 3.1) are essentially the same whenever they can be compared.

The differences between the settings are as follows:

- Here L_{∞}/K is only assumed to be Galois, whereas it is required to be abelian in [BKS17]. The definition of the complex therein does not truly require commutativity (see for instance [BF98], first paragraph of section 3.2), but the formulation of the Main Conjecture (conjecture 3.1) does.
- Here K_{∞} is the cyclotomic \mathbb{Z}_p -extension of K, whereas it is an arbitrary \mathbb{Z}_p -extension in the article. Only at times do they specialise to the cyclotomic case namely in subsections 3D and 4B and, to some extent, section 5. We note that the cyclotomic \mathbb{Z}_p -extension is the only one if K is totally real and Leopoldt's conjecture holds for K and p.
- Our condition $p \neq 2$ if K is not totally imaginary does not appear in the earlier part of the cited article neither in the definition of the complex nor in the formulation of the Main Conjecture. The condition $p \neq 2$ is indeed required in their reformulation of the same conjecture ([BKS17] conjecture 3.14).

The present section, slightly more technical in its use of derived categories, $R\Gamma$ -complexes and duality properties, may be skipped without significant effect on the comprehension of the sequel the fundamental takeaway is theorem 1.5.4, i.e. the isomorphism between the complexes outlined above.

We start by introducing the complex used in [BKS17], which is essentially given by an inverse limit of finite-level complexes from [BF98]:

Definition 1.5.1. i) Let p be a prime number, F/K a Galois extension of number fields and S and T two disjoint finite sets of places of K such that $S \supseteq S_{\infty} \cup S_{ram}(F/K)$. We set

$$\mathcal{B}_{F,S,\emptyset}^{\bullet} = R \operatorname{Hom}_{\mathbb{Z}_p}(R\Gamma_c(H_{F,S},\mathbb{Z}_p),\mathbb{Z}_p)[-2],$$

where $H_{F,S}$ denotes the Galois group over F of its maximal S-ramified extension and $R\Gamma_c$ denotes the compactly supported cohomology complex (cf. [BF98] equation (3)). We then define $\mathcal{B}_{FST}^{\bullet}$ by the exact triangle

$$\mathcal{B}^{\bullet}_{F,S,T} \to \mathcal{B}^{\bullet}_{F,S,\varnothing} \to \bigoplus_{w \in T(F)} \mathbb{Z}_p \otimes \kappa(w)^*[0] \to,$$
(1.14)

where the second arrow is as in [BKS17].

ii) Setting A. We define

$$\mathcal{B}^{\bullet}_{S,T} = \varprojlim_n \mathcal{B}^{\bullet}_{L_n,S,T},$$

where the limit is taken with respect to the natural transition maps induced by restriction (note that $H_{L_{n+1},S} \leq H_{L_n,S}$). This is the complex featured in [BKS17] conjecture 3.1.

Let us examine in which category one may naturally consider these complexes. In part i), $\mathcal{B}_{F,S,\varnothing}^{\bullet}$ has a canonical structure as a complex of $\Lambda(\operatorname{Gal}(F/K))$ -modules in the usual way (cf. [NSW20] proposition 1.6.3) because the maximal S-ramified extensions of F and K coincide, and hence $H_{F,S}$ is normal in the Galois group of this extension over K. The term $\bigoplus_{w \in T(F)} \mathbb{Z}_p \otimes \kappa(w)^*$ also has a natural $G = \operatorname{Gal}(F/K)$ -action, which in particular allows us to regard it as $\bigoplus_{v \in T} \operatorname{Ind}_{G_v}^G \mathbb{Z}_p \otimes \kappa(w)^*$ (as usual, here G_v denotes G_{w_0} for any chosen prolongation w_0 of v). This endows $\mathcal{B}_{F,S,T}^{\bullet}$ itself with the structure of a complex of $\Lambda(\operatorname{Gal}(F/K))$ -modules. As for part ii), the transition maps in $\lim_{x \to \infty} \mathcal{B}_{L_n,S,T}$ are Galois-equivariant, and hence $\mathcal{B}_{S,T}^{\bullet}$ is a complex of $\Lambda(\mathcal{G})$ -modules.

We recall at this point that $R\Gamma(G, M)$ is notation for the complex which computes the cohomology of G with coefficients in M (in some suitable category, which in our case was described in section 0.1) - either the one resulting from the standard resolution, or any isomorphic one in the relevant derived category. We must first relate some fashion of $R\Gamma$ -complexes to complexes of the form considered in previous sections. This is achieved by the following proposition, the idea of whose proof is largely taken from [Nic13] (see theorem 2.4 therein):

Proposition 1.5.2. Let G be a profinite group and $N \leq_c G$ a closed normal subgroup such that $H_n(N, \mathbb{Z}_p) = 0$ for all $n \geq 2$, and set Q = G/N. Then there exists an isomorphism of complexes

$$R \operatorname{Hom}(R\Gamma(N, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p) \cong [Y_{G,N}^{-1} \xrightarrow{v} \Lambda(Q)]$$

in the derived category $\mathcal{D}(\Lambda(Q))$, where $Y_{G,N} = \Delta(G)_N = \Delta(G)/\Delta(N)\Delta(G)$ and v is the natural composition of the lift to $\Delta(G)$ with the projection to $\Lambda(Q)$.

Proof. Cohomology $H^i(N, -)$ is by definition the right derived functor of $-^N = \operatorname{Hom}_{\Lambda(N)}(\mathbb{Z}_p, -)$. Therefore, $R\Gamma(N, \mathbb{Q}_p/\mathbb{Z}_p)$ can be identified with $R \operatorname{Hom}_{\Lambda(N)}(\mathbb{Z}_p, \mathbb{Q}_p/\mathbb{Z}_p)$ in $\mathcal{D}(\Lambda(Q))$ (here and below, the $\Lambda(Q)$ -module structure is given by [NSW20] proposition 1.6.3 as before).

The contravariant functor $R \operatorname{Hom}(-, \mathbb{Q}_p/\mathbb{Z}_p)$ coincides with $R \operatorname{Hom}_{\mathbb{Z}_p}(-, \mathbb{Q}_p/\mathbb{Z}_p)$ on complexes of discrete modules, and it amounts to applying $\operatorname{Hom}_{\mathbb{Z}_p}(-, \mathbb{Q}_p/\mathbb{Z}_p) = (-)^{\vee}$ at each degree as $\mathbb{Q}_p/\mathbb{Z}_p$ is \mathbb{Z}_p -injective. Noting that $M \widehat{\otimes}_{\Lambda(N)} \mathbb{Z}_p \cong \operatorname{Hom}_{\Lambda(N)}(M, \mathbb{Q}_p/\mathbb{Z}_p)^{\vee}$ for any compact right $\Lambda(N)$ -module M (see the proof of [NSW20] corollary 5.2.9), we conclude that

$$R \operatorname{Hom}(R\Gamma(N, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p) \cong \mathbb{Z}_p \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p$$
(1.15)

in $\mathcal{D}(\Lambda(Q))$ (where $-\widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}}\mathbb{Z}_p$ denotes the left derived functor of $-\widehat{\otimes}_{\Lambda(N)}\mathbb{Z}_p$) by dualising a projective resolution of \mathbb{Z}_p into an injective one of $\mathbb{Q}_p/\mathbb{Z}_p$.

Now we consider the canonical short exact sequence $\Delta(G) \hookrightarrow \Lambda(G) \twoheadrightarrow \mathbb{Z}_p$ of compact $\Lambda(N)$ -modules, which induces an exact triangle

$$\Delta(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \to \Lambda(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \to \mathbb{Z}_p \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \to$$

in $\mathcal{D}(\Lambda(Q))$ and hence an isomorphism

$$\mathbb{Z}_p \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \cong \operatorname{Cone}(\Delta(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \to \Lambda(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p).$$
(1.16)

The groups $H_i(N, \Lambda(G))$ vanish for all $i \geq 1$, since $\Lambda(G)^{\vee} \cong \operatorname{Map}_{cts}(G, \mathbb{Q}_p/\mathbb{Z}_p)$ is G-induced and therefore cohomologically trivial (see also the first point after (1.4)). In particular, there are isomorphisms $H_{i+1}(N, \mathbb{Z}_p) \cong H_i(N, \Lambda(G))$ for all $i \geq 1$. This implies $H_i(N, \Lambda(N)) = 0$ for all $i \geq 1$ by hypothesis, and hence the complexes $\Lambda(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p$ and $\Lambda(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p$ are both acyclic outside degree 0. But a complex with a single non-trivial cohomology group is isomorphic in the derived category to the complex consisting of that group concentrated in the corresponding degree, i.e.

$$\Delta(G) \widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}} \mathbb{Z}_p \cong (\Delta(G) \widehat{\otimes}_{\Lambda(N)} \mathbb{Z}_p)[0] = Y_{G,N}[0]$$

and

$$\Lambda(G)\widehat{\otimes}_{\Lambda(N)}^{\mathbb{L}}\mathbb{Z}_p \cong (\Lambda(G)\widehat{\otimes}_{\Lambda(N)}\mathbb{Z}_p)[0] = \Lambda(Q)[0]$$

This concludes the proof by (1.15) and (1.16), since by definition of the cone we have

$$\operatorname{Cone}(Y_{G,N}[0] \to \Lambda(Q)[0]) = [Y_{G,N}^{-1} \xrightarrow{v} \Lambda(Q)]$$

and the morphisms defining both cones are compatible.

Remark 1.5.3. It follows that, as a complex of abelian groups, $[Y_{G,N} \to \Lambda(Q)]$ is actually independent of G and Q up to isomorphism in the derived category - as long as they fit into a short exact sequence $N \hookrightarrow G \twoheadrightarrow Q$. This should not be too surprising, as the cohomology groups of complexes of that form were shown to be $N^{ab}(p)$, \mathbb{Z}_p and $\{1\}$ in sections 1.1 and 1.2. Naturally, the $\Lambda(Q)$ -structure does depend on the choice of groups, and it will be fundamental in our discussion.

The above proposition can now be applied to the global and local complexes to obtain isomorphisms

$$\mathcal{T}^{\bullet}_{S \cup T} \cong R \operatorname{Hom}(R\Gamma(H_S, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p)[-1]$$
(1.17)

and

$$\mathcal{L}_{v}^{\bullet} \cong R \operatorname{Hom}(R\Gamma(G_{L_{\infty},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1],$$

where $G_{L_{\infty},v}$ denotes $G_{(L_{\infty})_{v(L_{\infty})}} = (G_{L_{\infty}})_{v^c}$ as in section 1.2. The hypothesis on the vanishing of $H_n(N, \mathbb{Z}_p)$ is satisfied in these cases by the same reasons described at the beginning of the proof of proposition 1.4.3. The only case not covered there is that of an archimedean place v, which is even simpler by classical homology of finite groups: if $G_{L_{\infty},v}$ is trivial, then its homology in degree ≥ 1 vanishes for any module; and otherwise, $|G_{L_{\infty},v}| = 2$ (so p is odd by setting A) and hence the groups $H_n(G_{L_{\infty},v}, \mathbb{Z}_p)$ are 2-torsion \mathbb{Z}_p -modules, i.e. trivial, for all $n \geq 1$.

In combination with some duality properties, this new description of the global and local complexes allows us to prove the main result of this section. As mentioned before, nothing in the construction of $\mathcal{B}^{\bullet}_{S,T}$ requires \mathcal{G} to be abelian and we can therefore state the following theorem in the generality of setting A.

Theorem 1.5.4. Setting A. There is an isomorphism of complexes $C_{S,T}^{\bullet} \cong \mathcal{B}_{S,T}^{\bullet}$ in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$.

Proof. As in some of the previous proofs, it is convenient to initially disregard the set T altogether and consider

$$\mathcal{C}^{\bullet}_{S,\varnothing} = \operatorname{Cone}(\mathcal{L}^{\bullet}_S \xrightarrow{\alpha_{S,\varnothing}} \mathcal{T}^{\bullet}_S)[-1].$$

Let us consider the local side first, letting $G_{L_n,v} = G_{(L_n)_{v(L_n)}} = (G_{L_n})_{v^c}$ by analogy with $G_{L_{\infty},v}$. We also set $\mathcal{G}_{n,v} = (\mathcal{G}_n)_{v(L_n)}$. Then there are isomorphisms

$$\mathcal{L}_{S}^{\bullet} = \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathcal{L}_{v}^{\bullet} \cong \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} R \operatorname{Hom}(R\Gamma(G_{L_{\infty},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1]$$
$$\cong \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} R \operatorname{Hom}(\varinjlim_{n} R\Gamma(G_{L_{n},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1]$$
$$\cong \varprojlim_{n} \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_{n}} R \operatorname{Hom}(R\Gamma(G_{L_{n},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1]$$
$$\cong \varprojlim_{n} R \operatorname{Hom}\left(\bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_{n}} R\Gamma(G_{L_{n},v}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1]\right)$$

in $\mathcal{D}(\Lambda(\mathcal{G}))$, where

- the first isomorphism is essentially proposition 1.5.2, which we can apply as explained after remark 1.5.3. On the left-hand side, $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}$ – denotes the functor $\Lambda(\mathcal{G}) \widehat{\otimes}_{\Lambda(\mathcal{G}_v)}$ – applied to the local complex degree-wise. The isomorphism $\mathcal{L}_v^{\bullet} \cong R \operatorname{Hom}(R\Gamma(G_{L_{\infty},v}, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p)[-1]$, however, takes place in the derived category $\mathcal{D}(\Lambda(\mathcal{G}_v))$, where induction is really *derived induction*, i.e. the derived completed tensor product $\Lambda(\mathcal{G}) \widehat{\otimes}_{\Lambda(\mathcal{G}_v)}^{\mathbb{L}}$ – applied to complexes. The reason these two notions of induction coincide in this case is that, as discussed at the end of section 1.3, $\Lambda(\mathcal{G}) \widehat{\otimes}_{\Lambda(\mathcal{G}_v)}$ – is exact on the sequence defining \mathcal{L}_v^{\bullet} for all v.
- the second isomorphism is [NSW20] proposition 1.5.1. Note that $G_{L_{\infty},v} = \bigcap_n G_{L_n,v}$ and this intersection is simply an inverse limit with respect to the canonical embeddings. Even though the cited result only refers to cohomology groups, its proof shows the isomorphism exists on the level of $R\Gamma$ -complexes.
- the last two isomorphisms are formal: R Hom takes colimits (in particular direct limits and sums) in the first component to limits (in particular inverse limits and products) and, in the cases above, $\operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n}$ commutes with \varprojlim_n in the derived category. To see this, note that if v is a non-archimedean place, then $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}_n}$ applied to a complex amounts to a finite sum of complexes isomorphic to it; and if v is archimedean, then the inverse limit coincides with the complex at the layer $L_0 = L$ (such places do not split in the cyclotomic tower).

By the same token, the global complex is isomorphic to

$$\mathcal{T}_{S}^{\bullet} \cong R \operatorname{Hom}(R\Gamma(H_{S}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1] \cong \varprojlim_{n} R \operatorname{Hom}(R\Gamma(H_{L_{n},S}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-1],$$

where $H_S = \text{Gal}(M_S/L_{\infty})$ and $H_{L_n,S} = \text{Gal}(M_S/L_n)$. Recall that M_S was defined as the the maximal S-ramified extension of L_{∞} - and hence of L_n for all n.

It follows that $\mathcal{C}^{\bullet}_{S,\varnothing}[2]$ is isomorphic to the cone of some morphism

$$\left(\varprojlim_{n} R \operatorname{Hom}\left(\bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_{n}} R\Gamma(G_{L_{n,v}}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})\right) \xrightarrow{\alpha} \left(\varprojlim_{n} R \operatorname{Hom}(R\Gamma(H_{L_{n,S}}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})\right)$$

and therefore to the inverse limit of the cones of the finite-level morphisms

$$R \operatorname{Hom}\left(\bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n} R\Gamma(G_{L_n,v}, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p\right) \xrightarrow{\alpha_n} R \operatorname{Hom}(R\Gamma(H_{L_n,S}, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p)$$

because cones are defined in terms of direct sums and shifts, both of which commute with inverse limits. The map α is induced by the map $\alpha_{S,T}$ from section 1.4 (with $T = \emptyset$) by definition and hence comes from the inflation and restriction associated to $G_{L_{\infty}} \twoheadrightarrow H_S$ and $G_{L_{\infty},v} \hookrightarrow G_{L_{\infty}}$ respectively, as these are (after changing base field from L_{∞} to K) the homomorphisms used in the definition of the local-to-global maps (equation (1.7)). Note that $R \operatorname{Hom}(-, \mathbb{Q}_p/\mathbb{Z}_p)$ causes a reversal in the direction of the resulting arrows between $R\Gamma$ -complexes. It follows that α_n is induced from $G_{L_n} \twoheadrightarrow H_{L_n,S}$ and $G_{L_n,v} \hookrightarrow G_{L_n}$.

The definition of compact-support cohomology as the shifted cone of global-to-local restriction will yield the desired isomorphism by virtue of local and global Tate duality in their derived-categorical form. Namely, [Lim12] theorem 4.2.6 (right column of the diagram therein) shows that the cone of α_n above is isomorphic in $\mathcal{D}(\Lambda(\mathcal{G}_n))$ to $R \operatorname{Hom}(R\Gamma_c(H_{L_n,S}, \mathbb{Q}_p/\mathbb{Z}_p), \mathbb{Q}_p/\mathbb{Z}_p)$. We therefore have isomorphisms

$$\mathcal{C}^{\bullet}_{S,\varnothing} \cong \varprojlim_{n} R \operatorname{Hom}(R\Gamma_{c}(H_{L_{n},S}, \mathbb{Q}_{p}/\mathbb{Z}_{p}), \mathbb{Q}_{p}/\mathbb{Z}_{p})[-2] \cong \varprojlim_{n} R\Gamma(H_{L_{n},S}, \mathbb{Z}_{p}(1))[1],$$

the last one being global Tate duality (bottom row of the diagram in the cited theorem). Lastly, we apply Nekovář duality. Although introduced in [Nek06], a closer formulation to our setting is given by Lim and Sharifi in [LS13] (theorem in p. 623). We choose $R = \mathbb{Z}_p$, dualising complex $\omega_R = \mathbb{Z}_p[0]$ and complex $T = \mathbb{Z}_p(1)[0]$. Here the Tate twist refers to the $H_{L_n,S}$ -action, which is not relevant in the category $\mathcal{D}(\mathbb{Z}_p)$ where we shall apply the cited theorem, but it is inherited by the cohomology groups of complexes in that category. It is then easy to see that $R \operatorname{Hom}_{\mathbb{Z}_p}(T, \omega_R)$ is represented by $T^* = \mathbb{Z}_p(-1)[0]$, and thus the aforementioned theorem states that

$$R\Gamma(H_{L_n,S},\mathbb{Z}_p(1)) \cong R \operatorname{Hom}_{\mathbb{Z}_p}(R\Gamma_c(H_{L_n,S},\mathbb{Z}_p),\mathbb{Z}_p)[-3]$$
(1.18)

and hence

$$\mathcal{C}^{\bullet}_{S,\varnothing} \cong \varprojlim_{n} R\Gamma(H_{L_{n},S}, \mathbb{Z}_{p}(1))[1] \cong \varprojlim_{n} R \operatorname{Hom}_{\mathbb{Z}_{p}}(R\Gamma_{c}(H_{L_{n},S}, \mathbb{Z}_{p}), \mathbb{Z}_{p})[-2] = \mathcal{B}^{\bullet}_{S,\varnothing}.$$
(1.19)

It remains to bring T into the discussion. The difference between $\mathcal{C}_{S,T}^{\bullet}$ and $\mathcal{C}_{S,\emptyset}^{\bullet}$ is given by the following exact triangle in $\mathcal{D}(\Lambda(\mathcal{G}))$:

$$\mathcal{C}^{\bullet}_{S,T} \to \mathcal{C}^{\bullet}_{S,\varnothing} \to \bigoplus_{v \in T^p} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1)[0] \to \mathcal{C}^{\bullet}_{S,T}$$

This is not difficult to show using sequence (1.13), and we shall give a formal proof much later on, namely in proposition 3.2.5. We stress that said proof does not use any result from this section.

As to the complex of Burns, Kurihara and Sano, the limit of (1.14) along the cyclotomic tower yields an exact triangle

$$\mathcal{B}^{\bullet}_{S,T} \to \mathcal{B}^{\bullet}_{S,\varnothing} \to \varprojlim_{n} \bigoplus_{w \in T(L_n)} \mathbb{Z}_p \otimes \kappa(w)^*[0] \to .$$

In the last term, the limit is taken with respect to the norm maps (cf. [BKS17] p. 1540) and it was already determined in the proof of proposition 1.4.4: it is precisely $\bigoplus_{v \in T^p} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1)[0]$. The result now follows by comparing the last two exact triangles and noting that the isomorphisms between their second and third terms are compatible because they come from the natural dualities and transition maps.

Chapter 2

Formulation of the Main Conjecture

At the core of our discussion lies the Main Conjecture, where a multitude of objects and techniques come together: number theory, homological algebra, representation theory, some *p*-adic analysis and, gluing it all, the localisation sequence of *K*-theory. This chapter develops the necessary tools in those areas and culminates in the formulation of the conjecture in section 2.5. Whereas the preceding chapter delved into an explicit construction of the arithmetic complex $C_{S,T}^{\bullet}$, and subsequent ones will study some fundamental properties of the Main Conjecture, it is this chapter that plays the central role in our pursuits.

On its most essential level, the conjecture is an existence claim in a certain group $K_1(\mathcal{Q}(\mathcal{G}))$. As is often the case in Iwasawa theory, the predicted *zeta element* $\zeta_{S,T}^{\alpha,\beta}$ should have some *analytic* properties, in terms of interpolation of special *L*-values; and some *arithmetic-algebraic* properties, in the form of a specific relation to $\mathcal{C}_{S,T}^{\bullet}$. These two conditions are captured by the vertical and horizontal arrows in the key diagram

$$\begin{array}{ccc} K_1(\mathcal{Q}(\mathcal{G})) & \stackrel{\partial}{\longrightarrow} K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G})) \\ & & \downarrow^{\psi_{\chi}} \\ \mathcal{Q}^c(\Gamma_{\chi})^* \\ & \downarrow^{ev_{\gamma_{\chi\otimes\rho}}} \\ \mathbb{Q}_p^c \cup \{\infty\} \end{array}$$

respectively. Some of these objects have been introduced in a general setting in the preliminary chapter: Artin *L*-series (section 0.3), the connecting homomorphism $\partial \colon K_1(\mathcal{Q}(\mathcal{G})) \to K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ (section 0.4) and, although concealed behind the map ψ_{χ} , reduced norms (ibid).

In the first section, we address the concept of trivialised complexes and refined Euler characteristics. This homological tool will allow us to regard $\mathcal{C}_{S,T}^{\bullet}$ as an element in the relative K-group $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ after a suitable homomorphism $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ has been chosen, paving the way for the arithmetic side of the conjecture. It is here that we will first become acquainted with some algebraic properties of $\mathcal{Q}(\mathcal{G})$, the total ring of fractions of $\Lambda(\mathcal{G})$. Much of the necessary machinery was developed in [RW04].

Section 2.2 is devoted to showing that certain maps φ_n^{α} induced by α on the layer L_n have desirable properties - namely, we prove they are isomorphisms on χ -parts for almost all Artin characters χ of \mathcal{G} . This allows us to define regulators (for the special L-values featured in the Main Conjecture) in section 2.3. Regulators are a common occurrence at the intersection between algebraic and analytic number theory, and some of the rationale behind them in our context is explained at the beginning of the corresponding section. Our definition of the regulator is based on the work of Stark and Tate, especially [Tat84].

We then move on to study certain evaluation maps in section 2.4. The term $\mathcal{Q}^c(\Gamma_{\chi})^*$ in the above diagram is the group of units of the fraction field of a power series ring with *p*-adic coefficients. The analytic side of the conjecture claims that, when evaluated at 0, the series quotient $\psi_{\chi}(\zeta_{S,T}^{\alpha,\beta})$ takes the (regulated) special *L*-value for the character χ . However, series associated to so-called *W*-equivalent characters turn out to have closely related evaluation-at-0 maps, which leads us to define *twisted* evaluation maps $ev_{\gamma_{\chi\otimes\rho}}$. These correspond to evaluation at (infinitely many) points in the open unit ball of \mathbb{Q}_p^c centred at 0. Requiring interpolation of *L*-values at almost all of these points determines a unique conjectural element $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$, which plays the role of *p*-adic *L*-functions. This Interpolation Conjecture, formulated in the final section of the chapter, constitutes the first half of the Main Conjecture. The second half is the equivariant Main Conjecture, which combines the analytic and algebraic sides by requiring that the zeta element be mapped to the interpolating series quotient $F_{S,T,\chi}^{\alpha,\beta}$ via ψ_{χ} for all χ , and to the inverse of the refined Euler characteristic of $\mathcal{C}_{S,T}^{\bullet}$ via ∂ .

The division of the Main Conjecture into these two halves is only conceptual: one may instead postulate the existence of a $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\Gamma))$ which is sent to the regulated special *L*-value directly by $ev_{\gamma_{\chi}}\psi_{\chi}$ for almost all χ (and with the same arithmetic property regarding $\mathcal{C}_{S,T}^{\bullet}$) - which yields a completely equivalent conjecture. This latter approach is used, for instance, in the Main Conjecture in [BKS17] (although in the rather different language of *determinant functors*). In our setting, however, the definition of twisted evaluation maps and a separate Interpolation Conjecture will facilitate the study of its properties, as will become apparent in the next chapter. As already mentioned, the consideration of $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ also brings about a direct connection to the *p*-adic *L*-functions featured in Main Conjectures for totally real fields.

Now that the path has been outlined, we proceed to incorporate two new elements into the setting of the previous chapter which will be necessary for the formulation of the conjecture:

Setting B. All objects and assumptions from setting A carry over to this one. Additionally, we fix the following:

- A homomorphism of $\Lambda(\mathcal{G})$ -modules $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ such that ker (α) and coker (α) are $\Lambda(\Gamma)$ torsion, where $\mathcal{Y}_{S_{\infty}}$ and $E_{S,T}$ are as in section 1.4. Proposition 2.0.1 below shows that such
 an α exists and is necessarily injective.
- An isomorphism of abstract fields $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$, where \mathbb{C}_p denotes the field of *p*-adic complex

numbers. See for instance the remark after [NSW20] proposition 10.3.2 for existence.

Neither α nor β is uniquely determined by the properties above. In the next chapter, we will study how the choice of these parameters affects the validity of the Main Conjecture. Note that setting A (sometimes with the addition of α as above) still suffices for a few of the results in this chapter which shall be made clear each time.

The local structure of $E_{S,T}$ determined by Nickel allows us to prove the existence of α :

Proposition 2.0.1. Setting A. There exists a homomorphism of $\Lambda(\mathcal{G})$ -modules $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ such that ker(α) and coker(α) are $\Lambda(\Gamma)$ -torsion. Furthermore, any such α is injective.

Proof. Recall that $\Lambda_{\mathfrak{p}}(\Gamma)$ denotes the localisation of the integral domain $\Lambda(\Gamma)$ at a prime ideal \mathfrak{p} . Since $\Lambda(\Gamma)$ is central in $\Lambda(\mathcal{G})$, the localisation $\Lambda_{\mathfrak{p}}(\mathcal{G}) = \Lambda_{\mathfrak{p}}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})$ has a natural ring structure. In particular, any $\Lambda(\mathcal{G})$ -module M can be localised to a $\Lambda_{\mathfrak{p}}(\mathcal{G})$ -module

$$M_{\mathfrak{p}} = \Lambda_{\mathfrak{p}}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} M = \Lambda_{\mathfrak{p}}(\Gamma) \otimes_{\Lambda(\Gamma)} M.$$

By [Nic20] theorem 5.5, there exists an isomorphism of $\Lambda_{\mathfrak{p}}(\mathcal{G})$ -modules

$$\varphi \colon T_{\mathfrak{p}} \oplus \Lambda_{\mathfrak{p}}(\mathcal{G})^{r_1 - r'_1 + r_2} \oplus \bigoplus_{v \in S'_{\infty}} (\mathrm{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p)_{\mathfrak{p}} \xrightarrow{\sim} (E_S)_{\mathfrak{p}}$$

for almost any (and conjecturally any) height-one prime ideal \mathfrak{p} of $\Lambda(\Gamma)$, where T is either the trivial module or $\mathbb{Z}_p(1)$ (the inverse limit of p-power roots of unity), S'_{∞} is the set of archimedean places of K which ramify in L_{∞} , and r_1, r'_1 and r_2 are the number of real, real ramified (in L_{∞}) and complex places of K, respectively.

Fix a prime ideal \mathfrak{p} for which such φ exists and set

$$\mathcal{Y}' = T \oplus \Lambda(\mathcal{G})^{r_1 - r'_1 + r_2} \oplus \bigoplus_{v \in S'_{\infty}} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p,$$

so that $\varphi \colon (\mathcal{Y}')_{\mathfrak{p}} \to (E_S)_{\mathfrak{p}}$. There is a canonical embedding of

$$\mathcal{Y}_{S_{\infty}} = \bigoplus_{v \in S_{\infty}} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathbb{Z}_{p} = \bigoplus_{v \in S_{\infty} \setminus S_{\infty}'} \Lambda(\mathcal{G}) \oplus \bigoplus_{v \in S_{\infty}'} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathbb{Z}_{p}$$

into \mathcal{Y}' with cokernel T. Let $\mathcal{Q}(\Gamma)$ denote the field of fractions of $\Lambda(\Gamma)$. Consider the composition φ' given by

$$\begin{array}{cccc} \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \mathcal{Y}_{S_{\infty}} \to \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \mathcal{Y}' = & \mathcal{Q}(\Gamma) \otimes_{\Lambda_{\mathfrak{p}}(\Gamma)} (\mathcal{Y}')_{\mathfrak{p}} \\ & & \downarrow \\ & & \downarrow \\ \mathcal{Q}(\Gamma) \otimes_{\varphi} \\ & \mathcal{Q}(\Gamma) \otimes_{\Lambda_{\mathfrak{p}}(\Gamma)} (E_{S})_{\mathfrak{p}} = & \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} E_{S} \to \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} E_{S,T} \end{array}$$

where the last map is the inverse of the isomorphism induced by the embedding $E_{S,T} \hookrightarrow E_S$ (which has $\Lambda(\Gamma)$ -torsion cokernel, as can be deduced from sequence (1.13) and the structure theorem 0.1.3). Since T is torsion as well, φ' is a $\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})$ -isomorphism. We can now clear denominators by observing that the natural map

$$\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \operatorname{Hom}_{\Lambda(\mathcal{G})}(M, N) \to \operatorname{Hom}_{\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})}(\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} M, \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} N)$$
(2.1)

is an isomorphism whenever M is finitely generated over $\Lambda(\mathcal{G})$ (cf. [CR81] corollary 8.18). Since $\mathcal{Y}_{S_{\infty}}$ is so (by $|S_{\infty}|$ elements), there exist a non-zero $d \in \Lambda(\Gamma)$ and a homomorphism of $\Lambda(\mathcal{G})$ -modules $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ such that $\varphi' = d^{-1}\alpha$. In particular,

$$\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \alpha \colon \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \mathcal{Y}_{S_{\infty}} \to \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} E_{S,T}$$

is an isomorphism (because multiplication by d is itself an isomorphism after localising), which implies that ker(α) and coker(α) are $\Lambda(\Gamma)$ -torsion.

We now show that $\mathcal{Y}_{S_{\infty}}$ is $\Lambda(\Gamma)$ -torsion free and consequently ker (α) is trivial. As shown after (1.6), \mathbb{Z}_p is a projective $\Lambda(\mathcal{G}_v)$ -module for all $v \in S_{\infty}$, and hence $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p$ is projective over $\Lambda(\mathcal{G})$. But $\Lambda(\mathcal{G})$ is itself free as a $\Lambda(\Gamma)$ -module, so $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p$ is also projective over $\Lambda(\Gamma)$. Since projective modules over integral domains are torsion-free, the claim follows.

- Remark 2.0.2. i) It follows from the last part of the proof that $\mathcal{Y}_{S_{\infty}}$ is in fact free as $\Lambda(\Gamma)$ -module, as the classical Iwasawa algebra is a local ring. This can be seen from the definition too: as a $\Lambda(\Gamma)$ -module, $\mathcal{Y}_{S_{\infty}}$ is simply $\bigoplus_{w \in S_{\infty}(L)} \operatorname{Ind}_{\Gamma_{w_{\infty}}}^{\Gamma} \mathbb{Z}_p$ (use lemma 0.1.1 i)), where w_{∞} is any chosen prolongation of w to L_{∞} . Since archimedean places do not split in L_{∞}/L , this sum of inductions amounts to $\Lambda(\Gamma)^{|S_{\infty}(L)|}$.
- ii) Once we understand the structure of $\Lambda(\mathcal{G})$ in some detail, it will become clear that whether a $\Lambda(\mathcal{G})$ -module is $\Lambda(\Gamma)$ -torsion or not is in fact independent of the choice of Γ (with the conditions from setting A). This is unnecessary for the moment and will be explained at the beginning of subsection 3.2.1.

2.1 An integral trivialisation

The aim of this section is to construct a trivialisation for the complex $\mathcal{C}_{S,T}^{\bullet}$ and define the corresponding refined Euler characteristic¹⁰ $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet},t^{\alpha})$ in the relative K-group $K_0(\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G}))$, which plays a major role in the formulation of the Main Conjecture. After introducing refined Euler characteristics in an abstract setting, we specialise to our objects of interest, present the fundamental ring $\mathcal{Q}(\mathcal{G})$ and use the homomorphism α from setting B to obtain $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet},t^{\alpha})$.

Let R be a ring. Recall the notion of perfect complexes introduced in section 1.4: a (cochain) complex of (left) R-modules is called strictly perfect if it is bounded and consists of finitely generated projective modules, and perfect if it isomorphic to a strictly perfect complex in the derived

¹⁰In our treatment of refined Euler characteristics, we follow [BB05] sections 5 and 6. Our conventions match those of the object χ^{old} therein, which is related to an alternative definition via theorem 6.2 from the article.

category $\mathcal{D}(R)$. Given a complex C^{\bullet} of *R*-modules, we denote its modules of *i*-th cocycles and *i*-th coboundaries by $Z^{i}(C^{\bullet})$ and $B^{i}(C^{\bullet})$ respectively, both inside C^{i} . We set

$$H^{\operatorname{even}}(C^{\bullet}) = \bigoplus_{i \in \mathbb{Z}} H^{2i}(C^{\bullet}) \quad \text{and} \quad H^{\operatorname{odd}}(C^{\bullet}) = \bigoplus_{i \in \mathbb{Z}} H^{2i+1}(C^{\bullet})$$

and define C^{even} , C^{odd} , $Z^{\text{even}}(C^{\bullet})$, $Z^{\text{odd}}(C^{\bullet})$, $B^{\text{even}}(C^{\bullet})$ and $B^{\text{odd}}(C^{\bullet})$ analogously. If C^{\bullet} is bounded (for instance, strictly perfect), all of these are finite sums. If C^{\bullet} is perfect, at least $H^{\text{even}}(C^{\bullet})$ and $H^{\text{odd}}(C^{\bullet})$ are.

Let $\varphi \colon R \to S$ be a homomorphism of rings such that S is flat as a right R-module via φ and semisimple Artinian as a ring. Given an R-module M, we denote (in this section only) the Smodule $S \otimes_R M$ by M_S . Analogously, C_S^{\bullet} denotes the result of applying $S \otimes_R -$ to the complex C^{\bullet} degree-wise.

A trivialisation (over S) for a perfect complex C^{\bullet} of R-modules is an isomorphism of S-modules

$$t: H^{\text{odd}}(C_S^{\bullet}) \xrightarrow{\sim} H^{\text{even}}(C_S^{\bullet}).$$

Note that t need not come from extension of scalars of an R-homomorphism $H^{\text{odd}}(C^{\bullet}) \to H^{\text{even}}(C^{\bullet})$. One may then refer to the pair (C^{\bullet}, t) as a **trivialised complex**. Such a trivialisation gives rise to a well-defined **refined Euler characteristic** $\chi_{R,S}(C^{\bullet}, t) \in K_0(R, S)$ as follows (cf. section 0.4 for the definition and properties of the relative K_0). Choose a strictly perfect representative P^{\bullet} of C^{\bullet} (i.e. $P^{\bullet} \cong C^{\bullet}$ in $\mathcal{D}(R)$). The flatness of S implies $H^i(P^{\bullet}_S) = H^i(P^{\bullet})_S$ for all *i*, and analogously for B^i and Z^i . Therefore, the sequences

$$0 \to B^i(P_S^{\bullet}) \to Z^i(P_S^{\bullet}) \to H^i(P_S^{\bullet}) \to 0$$

are exact for all i and the same is true of

$$0 \to Z^i(P_S^{\bullet}) \to P_S^i \to B^{i+1}(P_S^{\bullet}) \to 0.$$

Furthermore, by semisimplicity of S, both kinds of sequences are split. A choice of splittings then gives rise to an isomorphism

$$\varphi_t \colon P_S^{\text{odd}} \cong B^{\text{even}}(P_S^{\bullet}) \oplus Z^{\text{odd}}(P_S^{\bullet})$$

$$\cong B^{\text{even}}(P_S^{\bullet}) \oplus B^{\text{odd}}(P_S^{\bullet}) \oplus H^{\text{odd}}(P_S^{\bullet})$$

$$\cong B^{\text{even}}(P_S^{\bullet}) \oplus B^{\text{odd}}(P_S^{\bullet}) \oplus H^{\text{even}}(P_S^{\bullet})$$

$$\cong Z^{\text{even}}(P_S^{\bullet}) \oplus B^{\text{odd}}(P_S^{\bullet})$$

$$\cong P_S^{\text{even}}$$

$$(2.2)$$

of S-modules, where the third isomorphism is induced by t. This trivialisation was defined on the cohomology of C^{\bullet} , but isomorphic complexes in $\mathcal{D}(R)$ have isomorphic cohomology and therefore t induces a trivialisation for P^{\bullet} as well. The **refined Euler characteristic** of the trivialised complex (C^{\bullet}, t) is defined as

$$\chi_{R,S}(C^{\bullet},t) = [P^{\text{odd}},\varphi_t, P^{\text{even}}] \in K_0(R,S).$$
(2.3)

Crucially, $\chi_{R,S}(C^{\bullet}, t)$ is independent of the choices of a strictly perfect representative P^{\bullet} of C^{\bullet} and splittings for the short exact sequences - only the complex and the trivialisation matter. See [BB05], sections 5 and 6 for a proof of this¹¹. As an immediate consequence, if D^{\bullet} is another complex and $q: C^{\bullet} \xrightarrow{\sim} D^{\bullet}$ is an isomorphism in $\mathcal{D}(R)$ under which t becomes $t_D: H^{\text{odd}}(D^{\bullet}_S) \xrightarrow{\sim} H^{\text{even}}(D^{\bullet}_S)$, then

$$\chi_{R,S}(C^{\bullet}, t) = \chi_{R,S}(D^{\bullet}, t_D).$$
(2.4)

In order to apply this construction to our context, we introduce the rings $\mathcal{Q}(\Gamma)$ and $\mathcal{Q}(\mathcal{G})$, which will in fact be essential for most of the sequel. While both were already present in the proof of proposition 2.0.1 (one explicitly and one implicitly), a closer look is in order now. In general, given a profinite group G, we denote by $\mathcal{Q}(G)$ the total ring of fractions of the Iwasawa algebra $\Lambda(G) = \mathbb{Z}_p[[G]]$ (section 0.1) obtained by inverting all regular elements. For an *p*-adic field E (i.e. a finite extension of \mathbb{Q}_p), we set $\mathcal{Q}^E(G) = E \otimes_{\mathbb{Q}_p} \mathcal{Q}(G)$. We also let

$$\mathcal{Q}^{c}(G) = \mathbb{Q}_{p}^{c} \otimes_{\mathbb{Q}_{p}} \mathcal{Q}(G) = \varinjlim_{E} \mathcal{Q}^{E}(G), \qquad (2.5)$$

with E running over all p-adic fields and transition maps given by the natural inclusions.

In the notation from setting A, $\Lambda(\Gamma)$ is an integral domain which lies in the centre of $\Lambda(\mathcal{G})$ and such that $[\Lambda(\mathcal{G}) : \Lambda(\Gamma)] = [L : K]$ is finite. Therefore, for a $\Lambda(\mathcal{G})$ -module, being finitely generated is equivalent to being so over $\Lambda(\Gamma)$. An essential property of $\mathcal{Q}(\mathcal{G})$ is that it can in fact be obtained by adding inverses of all non-zero elements of $\Lambda(\Gamma)$ to $\Lambda(\mathcal{G})$. In other words, there is a canonical isomorphism

$$\mathcal{Q}(\mathcal{G}) \cong \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})$$
(2.6)

(see [RW04] p. 551 for a proof¹²). In particular a $\Lambda(\mathcal{G})$ -module is torsion if and only if it is $\Lambda(\Gamma)$ torsion (which we may even take as a definition). It also follows that $\mathcal{Q}(\mathcal{G})$ is a finite-dimensional $\mathcal{Q}(\Gamma)$ -algebra and $\Lambda(\mathcal{G})$ is a $\Lambda(\Gamma)$ -order in it. Given the desirable properties of the ring $\Lambda(\Gamma) \cong \mathbb{Z}[[T]]$ (Noetherian integral domain, regular local ring, easily characterised prime ideals, etc.), some questions concerning $\Lambda(\mathcal{G})$ -modules will be tackled by restricting scalars to $\Lambda(\Gamma)$. As explained in section 0.1, the non-canonical isomorphism $\Lambda(\Gamma) \cong \mathbb{Z}_p[[T]]$ is the continuous \mathbb{Z}_p -linear map induced by

$$\gamma \mapsto T+1$$

for any topological generator γ of Γ , which in our case was fixed in setting A. We sometimes refer to the elements of $\mathcal{Q}(\Gamma) \cong \operatorname{Frac}(\mathbb{Z}_p[[T]])$ as **series quotients**. It is worth noting that $\mathcal{Q}(\Gamma)$ does not contain something like $\mathbb{Q}_p[[T]]$ - we say the elements in $\mathcal{Q}(\Gamma)$ have bounded denominators. It also follows from definition (2.5) that every series quotient in $\mathcal{Q}^c(\Gamma)$ has coefficients contained in a

¹¹As mentioned above, the Euler characteristic defined here corresponds to the χ^{old} in section 6 of the cited paper. Independence is really shown for a different construction in a more general setting in section 5. However, theorem 6.2 therein shows the difference between both constructions is measured by a certain $\delta^{1}_{\Lambda,\Sigma}(B^{\text{odd}}(P_{\Sigma}), -\text{Id})$ which is indeed independent of the above choices.

¹²We will resort to this article for multiple algebraic results. The setting of the article differs from ours in that L_{∞} is assumed to be totally real there, and p odd. However, the algebraic results we cite are not affected by this difference.

finite extension of \mathbb{Q}_p (in both the numerator and the denominator): it cannot, for instance, be of the form $\sum_{i=0}^{\infty} x_i T^i$ with $\mathbb{Q}_p(x_i: i \in \mathbb{N})$ of infinite degree over \mathbb{Q}_p .

In order to construct a refined Euler characteristic for the perfect complex $\mathcal{C}_{S,T}^{\bullet}$ of $\Lambda(\mathcal{G})$ -modules from chapter 1, we consider the embedding $\Lambda(\mathcal{G}) \hookrightarrow \mathcal{Q}(\mathcal{G})$. The semisimplicity of $\mathcal{Q}(\mathcal{G})$ is shown in the proof of [RW04] proposition 5. The flatness of $\mathcal{Q}(\mathcal{G})$ as a right $\Lambda(\mathcal{G})$ -module is clear: on $\Lambda(\mathcal{G})$ -modules, $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} -$ coincides with $\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} -$ by (2.6), and fields of fractions of integral domains are flat.

The cohomology of $\mathcal{C}^{\bullet}_{S,T}$ was established in theorem 1.4.6: one has $H^{\text{even}}(\mathcal{C}^{\bullet}_{S,T}) = H^0(\mathcal{C}^{\bullet}_{S,T}) \cong E_{S,T}$, and $H^{\text{odd}}(\mathcal{C}^{\bullet}_{S,T}) = H^1(\mathcal{C}^{\bullet}_{S,T})$ fits in a short exact sequence $X^{cs}_{T,S} \hookrightarrow H^1(\mathcal{C}^{\bullet}_{S,T}) \twoheadrightarrow \mathcal{X}_S$. Therefore, a trivialisation amounts to an isomorphism

$$t\colon \mathcal{Q}(\mathcal{G})\otimes_{\Lambda(\mathcal{G})}H^1(\mathcal{C}^{\bullet}_{S,T})\xrightarrow{\sim}\mathcal{Q}(\mathcal{G})\otimes_{\Lambda(\mathcal{G})}E_{S,T}.$$

Note here that, even though $H^{\text{even}}(\mathcal{C}_{S,T}^{\bullet})$ is not technically $E_{S,T}$ but rather isomorphic to it, we can indeed identify them by (2.4). Our chosen trivialisation is given by extension of scalars $t^{\alpha} = \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} t^{\alpha}_{\iota}$ of a map $t^{\alpha}_{\iota} \colon H^{1}(\mathcal{C}_{S,T}^{\bullet}) \to E_{S,T}$ on integral level. This $\Lambda(\mathcal{G})$ -homomorphism is defined as the composition

$$t_{\iota}^{\alpha} = \alpha \varphi \pi \colon H^{1}(\mathcal{C}_{S,T}^{\bullet}) \xrightarrow{\pi} \mathcal{X}_{S} \xrightarrow{\varphi} \mathcal{Y}_{S_{\infty}} \xrightarrow{\alpha} E_{S,T},$$
(2.7)

which we shall refer to as the **integral trivialisation**. The notation t_{ι}^{α} reflects the dependence on the choice of α , which will be studied in subsection 3.2.2. Let us consider each of the arrows separately:

- π is surjective with kernel $X_{T,S}^{cs}$, which is torsion: by sequence (1.13), it is enough to show that $\bigoplus_{v \in T^p} \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p(1)$ and X_S^{cs} are. The former has finite \mathbb{Z}_p -rank, which implies torsionness by the structure theorem 0.1.3; and the latter admits a surjection from the unramified Iwasawa module X_{nr} (the limit of the *p*-Hilbert class groups), which is torsion by [NSW20] proposition 11.1.4.
- φ is the inverse limit of the canonical projections $\mathcal{X}_{L_n,S} \twoheadrightarrow \mathcal{Y}_{L_n,S_{\infty}}$ which simply discard the components at finite places along the cyclotomic tower (cf. (1.10)). By the right-exactness of \varprojlim_n , it is surjective. Its kernel is \mathcal{X}_{S_f} , which has finite \mathbb{Z}_p -rank (because so does \mathcal{Y}_{S_f}) and is therefore torsion.
- α as in setting B is injective and has torsion cokernel by definition.

Since all maps have torsion kernels and cokernels, t_{ι}^{α} becomes indeed an isomorphism after tensoring with $\mathcal{Q}(\mathcal{G})$: our desired trivialisation $t^{\alpha} = \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} t_{\iota}^{\alpha}$. This induces a refined Euler characteristic

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T}, t^{\alpha}) \in K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$$
(2.8)

as explained above.

We conclude on the brief mention that $\mathcal{Y}_{S_{\infty}}$ and $E_{S,T}$ are known to be isomorphic in some cases. The classical example comes from Jannsen's work [Jan89], although the exact formulation that suits our setting appears in [NSW20] theorem 11.3.11 ii): if $p \nmid [L:K]$ and L contains no primitive p-th root of unity, then $\mathcal{Y}_{S_{\infty}} \cong E_{S,\emptyset}$ as $\Lambda(\mathcal{G})$ -modules. In particular, the α in setting B can be chosen to be this isomorphism. Note that t_{ι}^{α} will rarely be an isomorphism itself, since π is always surjective and φ is never injective unless $S = S_p \cup S_{\infty}$ and $|S_p| = 1$.

The same result provides an example where $\mathcal{Y}_{S_{\infty}}$ and $E_{S,T}$ are certainly not isomorphic: if we instead require L to contain a primitive p-th root of unity, then $E_{S,\emptyset}$ has $\mathbb{Z}_p(1)$ as a direct summand (the inverse limit of all p-power roots of unity), which could never occur in the torsion-free module $\mathcal{Y}_{S_{\infty}}$. In practice, one often imposes light conditions on the set T precisely to do away with this torsion part.

2.2 Morphisms on finite level

As mentioned in the introduction to the present chapter, special *L*-values need to be regulated before they can stand a chance of being interpolated by quotients of *p*-adic power series. This regulation comes in the form of division by certain *regulators*, (*p*-adic) complex numbers constructed as determinants of morphisms φ_n^{α} on finite level. While the definition of regulators is addressed in the next section, this one paves the way by defining the finite-level maps and proving the essential fact that they are isomorphisms on χ -parts for almost all Artin characters χ - which will guarantee the resulting regulators to be non-zero. These maps should contain information about the homomorphism α from setting B, and choosing them in a suitable way will in fact ensure the independence of the Main Conjecture from the choice of α (subsection 3.2.2). The passage to finite level is achieved by taking coinvariants of some of the $\Lambda(\mathcal{G})$ -modules we have already encountered. However, some care has to be put into accounting for the difference between those coinvariants and their natural finite-level counterparts.

We start by introducing Artin characters and proving some technical properties of modules of coinvariants. This is followed by a careful study of the kernel and cokernel of φ_n^{α} , leading to the proof that they have the desired behaviour on χ -parts (proposition 2.2.8).

By a (p-adic) **Artin character** χ of a profinite group G we always mean the trace of a representation $\rho_{\chi}: G \to \operatorname{GL}_n(\mathbb{Q}_p^c)$ with open kernel, where $n \geq 1$. Such a ρ_{χ} factors through the finite quotient $G/\ker(\rho_{\chi})$ and takes values in $\operatorname{GL}_n(E)$ for some p-adic field E, over which we say it *realises*. We denote the set of irreducible Artin characters of G by $\operatorname{Irr}_p(G)$. The usual operations on characters of finite groups (inflation, induction, restriction, sum, product, dual, etc.) can be performed on Artin characters as well in a natural way (conceptually, one can factor the characters through a common finite group first, then inflate the result of the operation back to G).

Our main interest lies in the Artin characters of \mathcal{G} . In this case, since any open subgroup of \mathcal{G} contains Γ^{p^n} for some n (because $L_{\infty} = \bigcup_n L_n$), ρ_{χ} and χ are, respectively, a representation and a character of $\mathcal{G}_n = \mathcal{G}/\Gamma^{p^n} = \operatorname{Gal}(L_n/K)$ in the sense of section 0.2. We will in general not distinguish χ from its projection to \mathcal{G}_n in the notation, but it will always be clear which layer L_n we are projecting onto. The notation $\ker(\chi) = \ker(\rho_{\chi})$ and terminology such as *irreducible* and *linear* carry over from the finite-group case.

An Artin character of \mathcal{G} is said to be of **type** W if it is linear and its kernel contains $H = \text{Gal}(L_{\infty}/K_{\infty})$. These characters identify bijectively with the group homomorphisms

$$\rho \colon \Gamma_K \to (\mathbb{Q}_p^c)^*$$

with open kernel. Any character¹³ ρ of type W factors through a finite layer K_n/K and its image is the group of p^m -th roots of unity in \mathbb{Q}_p^c for some m. Indeed, there is a group isomorphism

{Characters of
$$\mathcal{G}$$
 of type W } $\to \mu_{p^{\infty}}$ (2.9)
 $\rho \mapsto \rho(\gamma_K)$

where $\mu_{p^{\infty}}$ denotes the group of *p*-power roots of unity of \mathbb{Q}_p^c and the operation on the domain is character product. We point out that the multiplicative inverse ρ^{-1} of any linear character ρ is its dual $\check{\rho}$.

Given two Artin characters χ and χ' of \mathcal{G} , we write $\chi \sim_W \chi'$ and say they are *W*-equivalent if $\chi' = \chi \otimes \rho$ for some character ρ of type *W*. This can easily be verified to be an equivalence relation. Note that χ is irreducible if and only $\chi \otimes \rho$ is, as \oplus distributes over \otimes and linear characters are invertible.

The first necessary result concerning Artin characters is the following lemma. Here and in the sequel, expressions of the form $e(\chi)S \otimes_R -$ denote $e(\chi)(S \otimes_R -)$ rather than $(e(\chi)S) \otimes_R -$.

Lemma 2.2.1. Let $\Lambda(\Gamma)$ be the Iwasawa algebra of a profinite group $\Gamma \cong \mathbb{Z}_p$.

i) Choose a topological generator γ of Γ , which in particular induces an isomorphism $\Lambda(\Gamma) \cong \mathbb{Z}_p[[T]]$ as in section 0.1. Given a finitely generated torsion $\Lambda(\Gamma)$ -module M, set

$$n_0 = \max\{n \in \mathbb{N} \text{ such that } \langle \xi_n \rangle \in \operatorname{supp}(M)\}$$

with ξ_n as defined in the same section, or $n_0 = 0$ if that set is empty. Then $n_0 \in \mathbb{N}$ and, for all $n \ge n_0$, one has

$$\operatorname{rank}_{\mathbb{Z}_p} M^{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^{n_0}}}.$$
(2.10)

ii) Let G be a profinite group which has Γ as an open central subgroup, and m < n two natural numbers. Furthermore, let χ be an irreducible Artin character of G such that ker(χ) contains Γ^{p^n} but not Γ^{p^m} . If rank_{\mathbb{Z}_p} $M_{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^m}}$ and E is any extension of \mathbb{Q}_p over which χ has a realisation, then

$$e(\chi)E\otimes_{\mathbb{Z}_p}M_{\Gamma^{p^n}}=e(\chi)E\otimes_{\mathbb{Z}_p}M^{\Gamma^{p^n}}=0,$$

where $e(\chi) \in E[G/\Gamma^{p^n}]$ is the primitive central idempotent given by (0.14).

¹³We reserve the notation ρ for type-W characters, which will in practice not cause any confusion with the representation $\rho_{\chi} \colon \mathcal{G} \to \mathrm{GL}_n(\mathbb{Q}_p^c)$ associated to an arbitrary Artin character χ of \mathcal{G} .

Proof. We start with part i). Finitely generated torsion $\Lambda(\Gamma)$ -modules have finite support (which shows $n_0 \in \mathbb{N}$) and finite \mathbb{Z}_p -rank by the structure theorem 0.1.3, and hence the exact sequence

$$0 \to M^{{\Gamma^p}^n} \to M \xrightarrow{\gamma^{p^n} - 1} M \to M_{{\Gamma^p}^n} \to 0$$

shows that $\operatorname{rank}_{\mathbb{Z}_p} M^{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^n}}$ for all $n \in \mathbb{N}$. We therefore only need to show (2.10) for one these, and we do so for coinvariants. Let

$$f \colon M \xrightarrow{\approx} E_M = \bigoplus_{i=1}^s \Lambda(\Gamma) / \langle p^{m_i} \rangle \oplus \bigoplus_{j=1}^t \Lambda(\Gamma) / \langle F_j^{l_j} \rangle$$

be a pseudo-isomorphism as in the structure theorem for some $m_i, l_j > 0$ and irreducible Weierstrass polynomials F_j - all of which are unique up to order. Note that the height-one prime ideals in $\operatorname{supp}(M)$ are precisely the $\langle F_j \rangle$ and, if s > 0, $\langle p \rangle$ as well.

We now show that it suffices to prove (2.10) for E_M rather than M. Indeed, decomposing the exact sequence

$$0 \to \ker(f) \to M \xrightarrow{f} E_M \to \operatorname{coker}(f) \to 0$$

$$(2.11)$$

$$\operatorname{im}(f)$$

into two short ones and taking invariants-coinvariants (cf. (0.9)) on each of them yields new exact sequences

$$\cdots \to \ker(f)_{\Gamma^{p^n}} \to M_{\Gamma^{p^n}} \to \operatorname{im}(f)_{\Gamma^{p^n}} \to 0$$

and

$$\cdots \to \operatorname{coker}(f)^{\Gamma^{p^n}} \to \operatorname{im}(f)_{\Gamma^{p^n}} \to (E_M)_{\Gamma^{p^n}} \to \operatorname{coker}(f)_{\Gamma^{p^n}} \to 0$$

for all $n \in \mathbb{N}$. Since ker(f) and coker(f) are finite, so are their invariants and coinvariants, which in particular have \mathbb{Z}_p -rank equal to 0. It therefore follows from the last two sequences that rank $\mathbb{Z}_p M_{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} (E_M)_{\Gamma^{p^n}}$. The right-hand side is easily determined, since taking Γ^{p^n} coinvariants amounts to forming the quotient by the submodule generated by $w_n = (T+1)^{p^n} - 1$, and hence

$$(E_M)_{\Gamma^{p^n}} = \bigoplus_{i=1}^s \mathbb{Z}_{p^{m_i}\mathbb{Z}}[1, T, \dots, T^{p^n-1}] \oplus \bigoplus_{j=1}^t \Lambda(\Gamma)_{\langle F_j^{l_j}, w_n \rangle}$$

In writing the first sum (the μ -part), we are tacitly using [NSW20] corollary 5.3.3. All terms of that sum are finite and therefore do not contribute to the \mathbb{Z}_p -rank. For those in the second sum (the λ -part), we resort to the fact that, given any two non-zero elements $x, y \in \Lambda(\Gamma)$, the quotients $\Lambda(\Gamma)/\langle x, y \rangle$ and $\Lambda(\Gamma)/\langle \gcd(x, y) \rangle$ are pseudo-isomorphic (recall here that $\Lambda(\Gamma)$ is a unique factorisation domain) and in particular have the same \mathbb{Z}_p -rank.

The polynomial w_n decomposes into irreducibles as $w_n = \xi_n \cdot \ldots \cdot \xi_0$, with $\xi_i \neq \xi_{i'}$ whenever $i \neq i'$. The F_i are irreducible by definition. Furthermore, by the Weierstrass preparation theorem [NSW20] theorem 5.3.4, two Weierstrass polynomials which differ by a unit in $\Lambda(\Gamma)$ must in fact coincide. It follows that $gcd(F_i^{l_j}, w_n)$ is ξ_i if $F_j = \xi_i$ for some $0 \le i \le n$, and 1 otherwise. Therefore,

$$\operatorname{rank}_{\mathbb{Z}_p} \Lambda(\Gamma) / \langle F_j^{l_j}, w_n \rangle = \begin{cases} 0, & F_j \neq \xi_i \text{ for all } i \\ 0, & F_j = \xi_i \text{ and } n < i \\ \deg(F_j), & F_j = \xi_i \text{ and } n \ge i. \end{cases}$$

After fixing F_j , this rank is either identically 0 (first case) or independent of n for $n \ge i$ (second and third cases). This shows i).

Let us now prove part ii). The canonical surjection $M_{\Gamma^{p^n}} \twoheadrightarrow M_{\Gamma^{p^m}}$ (m < n) induces an epimorphism

$$q \colon E \otimes_{\mathbb{Z}_p} M_{\Gamma^{p^n}} \twoheadrightarrow E \otimes_{\mathbb{Z}_p} M_{\Gamma^{p^m}}$$

of $E[G/\Gamma^{p^n}]$ -modules, where the codomain is regarded as such via the projection

$$\varepsilon \colon E[G/\Gamma^{p^n}] \twoheadrightarrow E[G/\Gamma^{p^m}]$$

But q is in fact an isomorphism (even after tensoring with \mathbb{Q}_p only) as both modules of coinvariants have the same \mathbb{Z}_p -rank. Since χ does not factor through G/Γ^{p^m} by hypothesis, $\varepsilon(e(\chi)) = 0$ by (0.15) (together with the argument in the last paragraph of that section) and hence

$$e(\chi)E\otimes_{\mathbb{Z}_p} M_{\Gamma^{p^n}} \cong \varepsilon(e(\chi))E\otimes_{\mathbb{Z}_p} M_{\Gamma^{p^m}} = 0.$$

As for the invariants, we have

$$\operatorname{rank}_{\mathbb{Z}_p} M^{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} M_{\Gamma^{p^m}} = \operatorname{rank}_{\mathbb{Z}_p} M^{\Gamma^{p^n}}$$

by the assumptions in ii) and the same argument as at the beginning of part i). Therefore, the canonical injection $M^{{\Gamma p}^m} \hookrightarrow M^{{\Gamma p}^n}$ induces an embedding

$$q\colon E\otimes_{\mathbb{Z}_p} M^{{\Gamma^p}^m} \hookrightarrow E\otimes_{\mathbb{Z}_p} M^{{\Gamma^p}^n}$$

of $E[G/\Gamma^{p^n}]$ -modules, which turns out to be an isomorphism because of the above equality of $\mathbb{Z}_{p^{-1}}$ -ranks. This again implies

$$0 = \varepsilon(e(\chi))E \otimes_{\mathbb{Z}_p} M^{{\Gamma^p}^m} \cong e(\chi)E \otimes_{\mathbb{Z}_p} M^{{\Gamma^p}^n}.$$

- Remark 2.2.2. i) For a finitely generated $\Lambda(\Gamma)$ -module, not only does the condition that it is torsion imply that its modules of Γ^{p^n} -coinvariants and Γ^{p^n} -invariants have bounded \mathbb{Z}_p -rank as $n \to \infty$ (as stated in part i) of the lemma), but the two are in fact equivalent. The converse can easily be proved by contradiction using the structure theorem for Iwasawa modules.
- ii) The proof shows that the n_0 in part i) is in fact minimal with property (2.10). This invariant is the d(M) in [NSW20] definition 5.3.12 in disguise, except for our convention that $n_0 = 0$ (as opposed to -1) when the support of M contains no $\langle \xi_i \rangle$.

The previous lemma will be useful when studying the vanishing of the kernel and cokernel of the finite-level maps φ_n^{α} , which we shall construct by essentially taking coinvariants of the integral trivialisation $t_{\iota}^{\alpha} = \alpha \varphi \pi$ defined in (2.7). Our aim is to compare these morphisms to the Dirichlet regulator map $\mathbb{R} \otimes \mathcal{O}_{L_n,S}^* \to \mathbb{R} \otimes \mathcal{X}_{L_n,S}^{\mathbb{Z}}$ (0.24), so only the last two arrows $\alpha \varphi \colon \mathcal{X}_S \to E_{S,T}$ play a role. The difference between the coinvariants of \mathcal{X}_S and its finite-level counterpart $\mathcal{X}_{L_n,S}$ (defined after (1.10)) is given by the following result:

Proposition 2.2.3. Setting A. Let $S' \supseteq S_{\infty}$ be a finite set of places of K containing all archimedean places and at least one non-archimedean place, and denote by $S'_f \neq \emptyset$ the set of non-archimedean places in S'. Then, for all $n \in \mathbb{N}$, there exists a canonical isomorphism of $\Lambda(\mathcal{G}_n)$ -modules

$$(\mathcal{Y}_{S'})_{\Gamma^{p^n}} \cong \mathcal{Y}_{L_n,S'}$$

and a canonical epimorphism of $\Lambda(\mathcal{G}_n)$ -modules

$$(\mathcal{X}_{S'})_{\Gamma^{p^n}} \twoheadrightarrow \mathcal{X}_{L_n,S'}$$

with finite kernel of order

$$\min\{[\Gamma^{p^n}:\Gamma^{p^n}\cap\mathcal{G}_v]:v\in S'_f\}$$

In particular, the above epimorphism is an isomorphism if and only if $\Gamma^{p^n} \subseteq \mathcal{G}_v$ for some $v \in S'_f$, or in other words, if and only if some $w_n \in S'_f(L_n)$ is non-split in L_∞/L_n .

Proof. As usual, we set $\mathcal{G}_{n,v} = (\mathcal{G}_n)_{v(L_n)}$. The first isomorphism is:

$$(\mathcal{Y}_{S'})_{\Gamma^{p^n}} \cong \bigoplus_{v \in S'} (\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p)_{\Gamma^{p^n}}$$
$$\cong \bigoplus_{v \in S'} \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})} (\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}_v)} \mathbb{Z}_p)$$
$$\cong \bigoplus_{v \in S'} \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G}_v)} \mathbb{Z}_p$$
$$= \bigoplus_{v \in S'} \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G}_{n,v})} \mathbb{Z}_p$$
$$= \bigoplus_{v \in S'} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}} \mathbb{Z}_p$$
$$= \mathcal{Y}_{L_n,S'}.$$

In the second isomorphism, we have used (0.3) - the conditions to apply it are met as explained several times in the previous chapter (cf. for instance section 1.3).

Consider now the Γ^{p^n} -invariants-coinvariants exact sequence (0.9) induced by $\mathcal{X}_{S'} \hookrightarrow \mathcal{Y}_{S'} \twoheadrightarrow \mathbb{Z}_p$:

$$0 \to \mathcal{X}_{S'}{}^{\Gamma^{p^n}} \to \mathcal{Y}_{S'}{}^{\Gamma^{p^n}} \xrightarrow{\varepsilon} \mathbb{Z}_p \to (\mathcal{X}_{S'})_{\Gamma^{p^n}} \to (\mathcal{Y}_{S'})_{\Gamma^{p^n}} \to \mathbb{Z}_p \to 0$$

Since Γ is central in \mathcal{G} , all arrows are $\Lambda(\mathcal{G}_n)$ -homomorphisms. By exactness, there exists a canonical surjection

$$(\mathcal{X}_{S'})_{\Gamma^{p^n}} \twoheadrightarrow \ker((\mathcal{Y}_{S'})_{\Gamma^{p^n}} \twoheadrightarrow \mathbb{Z}_p) \cong \mathcal{X}_{L_n,S'},$$

where the isomorphism follows from $(\mathcal{Y}_{S'})_{\Gamma^{p^n}} \cong \mathcal{Y}_{L_n,S'}$. In order to determine its kernel, which is isomorphic to $\operatorname{coker}(\mathcal{Y}_{S'})_{\Gamma^{p^n}} \xrightarrow{\varepsilon} \mathbb{Z}_p)$, we study the action of Γ^{p^n} on $\mathcal{Y}_{S'}$ more closely. This module decomposes as $\mathcal{Y}_{S'} = \mathcal{Y}_{S'_{\infty}} \oplus \mathcal{Y}_{S'_{f}}$. We have already seen that $\mathcal{Y}_{S'_{\infty}}$ is free as a $\Lambda(\Gamma)$ -module (cf. remark 2.0.2 i)), so $\mathcal{Y}_{S'_{\infty}} \stackrel{\Gamma^{p^n}}{\to}$ is trivial. For each non-archimedean $v \in S'_{f}$, we treat the direct summand $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p$ of $\mathcal{Y}_{S'_{t}}$ separately.

As explained in (1.10), there is an isomorphism

$$\operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}}\mathbb{Z}_{p} \cong \varprojlim_{m} \Lambda(\mathcal{G}_{m}) \otimes_{\Lambda(\mathcal{G}_{m,v})} \mathbb{Z}_{p} \cong \varprojlim_{m} \bigoplus_{w_{m} \in \{v\}(L_{m})} \mathbb{Z}_{p} \cdot w_{m}.$$

This map is \mathcal{G} -equivariant, with \mathcal{G} acting on the right-hand side by permuting the places w_m . Consider an arbitrary element $x = \lim_{m \to \infty} x_m \in \operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p$, where

$$x_m = \sum_{w_m} z_{w_m} w_m \in \bigoplus_{w_m} \mathbb{Z}_p \cdot w_m$$

for all $m \in \mathbb{N}$. Suppose x is invariant under Γ^{p^n} , and therefore so is x_m for all m. Let v_n be any prolongation of v to L_n . The group Γ^{p^n} acts transitively on the set of places of L_{∞} above v_n and thus, for any fixed $m \ge n$, the coefficient z_{w_m} must coincide for all $w_m \mid v_n$. Since v_n splits into p^{m-n} places in L_m (because it is archimedean), the p-adic integer z_{v_n} is infinitely divisible by p and hence 0. This holds for any $v_n \mid v$, so x must in fact be trivial itself. This shows $(\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p)^{\Gamma^{p^n}} = 0$. Let now $v \in S'_f$. Then \mathcal{G}_v is open in \mathcal{G} and

$$\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \mathbb{Z}_p \cong \bigoplus_{w_\infty \in \{v\}(L_\infty)} \mathbb{Z}_p \cdot w_\infty$$

has finitely many summands. Suppose $x = \sum_{w_{\infty}} z_{w_{\infty}} w_{\infty} \in \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathbb{Z}_{p}$ is $\Gamma^{p^{n}}$ -invariant. By the same transitivity argument as in the archimedean case, $z_{w_{\infty}}$ coincides with $z_{w'_{\infty}}$ whenever w_{∞} and w'_{∞} lie above the same place v_{n} of L_{n} . That is, x is of the form

$$x = \sum_{v_n|v} z_{v_n} \sum_{w_\infty|v_n} w_\infty.$$

Conversely, any element of that form is Γ^{p^n} -invariant. Such an element has augmentation (i.e. sum of coefficients) given by

$$\varepsilon(x) = \sum_{v_n \mid v} z_{v_n} [\Gamma^{p^n} : \Gamma^{p^n} \cap \mathcal{G}_v] \in \mathbb{Z}_p$$

Here $[\Gamma^{p^n}:\Gamma^{p^n}\cap\mathcal{G}_v]$ is the number of places of L_∞ above $v(L_n)$, and therefore above v_n (because L_∞/K is Galois). It is necessarily a power of p.

All in all, the image of

$$\mathcal{Y}_{S'}{}^{\Gamma^{p^n}} = \mathcal{Y}_{S'_f}{}^{\Gamma^{p^n}} \xrightarrow{\varepsilon} \mathbb{Z}_p$$

is generated (as an ideal in \mathbb{Z}_p) by $\{[\Gamma^{p^n} : \Gamma^{p^n} \cap \mathcal{G}_v] : v \in S'_f\}$. Since these are all *p*-powers, the same ideal is generated by their minimum, and it has index equal to that minimum. The result follows.

In our case of interest, S' will be the set S from setting A. The proposition shows that a sufficient condition for $(\mathcal{X}_S)_{\Gamma^{p^n}} \xrightarrow{\sim} \mathcal{X}_{L_n,S}$ is $n \ge n(S)$.

Example 2.2.4. It can indeed happen that the map $(\mathcal{X}_S)_{\Gamma P^n} \twoheadrightarrow \mathcal{X}_{L_n,S}$ fails to be injective. Let

$$f_0(x) = x^3 - 30x - 1$$

and

$$f_1(x) = x^9 - 3x^8 - 27x^7 + 90x^6 + 87x^5 - 336x^4 - 73x^3 + 288x^2 - 27x - 27x^2 - 27x^2$$

For $i \in \{0,1\}$, let L_i be the number field formed by adjoining a root of f_i to \mathbb{Q} . These are [LMFDB, Number field 3.3.107973.1] and [LMFDB, Number field 9.9.1258767452176317.1], respectively. As can be seen in the LMFDB, L_1 contains L_0 (for a suitable choice of roots) as well as $\mathbb{Q}(\zeta_9)^+ = \mathbb{Q}(\zeta_9 + \zeta_9^{-1})$. The field $\mathbb{Q}(\zeta_9)^+$ is the only subfield of $\mathbb{Q}(\zeta_9)$ of relative degree 2, and therefore necessarily the first layer of the cyclotomic \mathbb{Z}_3 -extension of \mathbb{Q} . Since L_1 is an extension of L_0 of degree 3 containing $\mathbb{Q}(\zeta_9)^+$ (and $L_0 \neq \mathbb{Q}(\zeta_9)^+$), it is precisely the first layer of the cyclotomic \mathbb{Z}_3 -extension of L_0 .

Set $L = K = L_0$ and let L_{∞} be the cyclotomic \mathbb{Z}_3 -extension of L. Let $\mathcal{G} = \Gamma = \text{Gal}(L_{\infty}/L) \cong \mathbb{Z}_3$. Consider the finite set of places $S = S_{\infty} \cup S_3$ of K consisting of the archimedean places and the 3-adic ones, and set $T = \emptyset$. This is an example of setting A for p = 3.

The referenced LMFDB pages show that the prime 3 is totally ramified in L but splits into three primes in L_1 . Therefore, the only prime v in $S_3(L)$ splits in in L_1/L , which implies

$$\left|\ker((\mathcal{X}_S)_{\Gamma} \twoheadrightarrow \mathcal{X}_{L,S})\right| = [\Gamma : \Gamma_{w_{\infty}}] \ge 3$$

by the previous proposition (with w_{∞} denoting any prolongation of v to L_{∞}). The next layer L_2 can be verified to contain exactly three places above 3, which means all 3-adic places of L_1 are non-split in L_2/L_1 (and hence in L_{∞}/L_1 by the structure of Γ). It follows that $\Gamma_{w_{\infty}} = \Gamma^3$ and the above inequality is in fact an equality.

As an informal remark, random sampling using *SageMath* seems to hint at this situation being rather infrequent. Specifically, the prime $p \in \{3, 5\}$ was already non-split in the first layer of the cyclotomic \mathbb{Z}_p -extension of most tested number fields (all of low degree). This would imply that $(\mathcal{X}_S)_{\Gamma^{p^n}} \twoheadrightarrow \mathcal{X}_{L_n,S}$ is often an isomorphism for all n. However, the testing was not systematic enough to make anything other than a passing comment out of it.

Now that a relation between $(\mathcal{X}_S)_{\Gamma^{p^n}}$ and $\mathcal{X}_{L_n,S}$ has been established, we address the same descent question for $E_{S,T}$. The key concept now is that of universal norms:

Definition 2.2.5. Let F be an algebraic extension of a number field E. Consider two disjoint finite sets of places $S' \supseteq S_{\infty}$ and T' of E. The group of p-adic **universal norms** of (S', T')-units in E is defined as

$$\mathcal{E}_{E,S',T'}^F = \bigcap_{E'} N_{E'/E}(\mathbb{Z}_p \otimes \mathcal{O}_{E',S',T'}^*) \subseteq \mathbb{Z}_p \otimes \mathcal{O}_{E,S',T'}^*,$$

where E' runs over the finite extensions of E contained in F. Given a finite extension E' of E contained in F, $\mathcal{E}^F_{E',S',T'}$ denotes $\mathcal{E}^F_{E',S'(E'),T'(E')}$.

If $E \subseteq F_0 \subseteq F_1 \subseteq \cdots$ is a family of finite extensions of E such that $F = \bigcup_n F_n$, then one has

$$\mathcal{E}_{E,S',T'}^F = \bigcap_{n \in \mathbb{N}} N_{F_n/E}(\mathbb{Z}_p \otimes \mathcal{O}_{F_n,S',T'}^*)$$

by the transitivity of the norm maps. More generally, this works for any cofinal family in the set of all finite extensions of E inside F. Also relevant to our purposes is the fact that $\mathcal{E}_{E,S',T'}^F$ coincides with the image of the canonical projection

$$\left(\varprojlim_{E'} \mathbb{Z}_p \otimes \mathcal{O}_{E',S',T'}^*\right) \to \mathbb{Z}_p \otimes \mathcal{O}_{E,S',T'}^*$$

where E' is as before and the inverse limit is taken with respect to the norm maps. This is not difficult to show using the classical result that the inverse limit of non-empty compact Hausdorff spaces is non-empty.

Let us now specialise to our Iwasawa-theoretic setting A. We omit the superscript F from the notation of the universal norms whenever it is L_{∞} . Since taking Γ^{p^n} -coinvariants yields the maximal quotient with trivial Γ^{p^n} -action, there is a canonical epimorphism of $\Lambda(\mathcal{G}_n)$ -modules

$$(E_{S,T})_{\Gamma^{p^n}} \twoheadrightarrow \mathcal{E}_{L_n,S,T}.$$

by the previous displayed equation. This map can be shown to be injective, and therefore an isomorphism, using work of Fukaya and Kato. Namely, the short exact sequence $\Lambda(\mathcal{G}) \xrightarrow{1-\gamma^{p^n}} \Lambda(\mathcal{G}) \twoheadrightarrow \Lambda(\mathcal{G}_n)$ induces an exact triangle

$$\mathcal{C}^{\bullet}_{S,T} \to \mathcal{C}^{\bullet}_{S,T} \to \Lambda(\mathcal{G}_n) \otimes^{\mathbb{L}}_{\Lambda(\mathcal{G})} \mathcal{C}^{\bullet}_{S,T} \to$$

in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$, and therefore a long exact sequence

$$\cdots \to H^{-1}(\Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{C}_{S,T}^{\bullet}) \to H^0(\mathcal{C}_{S,T}^{\bullet}) \xrightarrow{1-\gamma^{p^n}} H^0(\mathcal{C}_{S,T}^{\bullet}) \to H^0(\Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{C}_{S,T}^{\bullet}) \to \cdots$$

In order to determine the last term, we resort to a result whose proof we defer to chapter 3 for the sake of exposition: by lemma 3.3.5, there exist $\Lambda(\mathcal{G}_n)$ -isomorphisms

$$H^{i}(\Lambda(\mathcal{G}_{n})\otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}}\mathcal{C}_{S,T}^{\bullet}) \cong H^{i}(\mathcal{B}_{L_{n},S,T}^{\bullet})$$

for all *i*, where $\mathcal{B}^{\bullet}_{L_n,S,T}$ is the complex of Burns, Kurihara and Sano on finite level (see definition 1.5.1). Its cohomology is trivial outside degrees 0 and 1, and $\mathbb{Z}_p \otimes \mathcal{O}^*_{L_n,S,T}$ in degree 0 by [BKS17] p. 1535. The exactness of the above sequence in cohomology yields an embedding

 $(E_{S,T})_{\Gamma^{p^n}} \cong \operatorname{coker}(H^0(\mathcal{C}_{S,T}^{\bullet}) \xrightarrow{1-\gamma^{p^n}} H^0(\mathcal{C}_{S,T}^{\bullet})) \hookrightarrow \mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^*$, which proves our injectivity claim. We point out that essentially the same question is treated using more direct methods in existing literature (see for instance theorem 7.3 from [Kuz72] or section 1 of [MN15]), from which an alternative proof is likely to follow.

We denote by ι_n the composition

$$\iota_n \colon (E_{S,T})_{\Gamma^{p^n}} \xrightarrow{\sim} \mathcal{E}_{L_n,S,T} \hookrightarrow \mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^*.$$

This concludes the preparations for the following definition:

Definition 2.2.6. Setting A, α as in setting B. For $n \ge n(S)$, the **finite-level map** φ_n^{α} is the $\Lambda(\mathcal{G}_n)$ -homomorphism given by the composition

$$\begin{array}{ccc} (\mathcal{X}_S)_{\Gamma^{p^n}} & \xrightarrow{(\alpha\varphi)_{\Gamma^{p^n}}} & (E_{S,T})_{\Gamma^{p^n}} \\ & \stackrel{\uparrow}{\swarrow} & & \stackrel{\downarrow_{l_n}}{\longrightarrow} \\ \mathcal{X}_{L_n,S} & \xrightarrow{-\overset{\varphi_n^{\alpha}}{\longrightarrow}} & \mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^* \end{array}$$

where the left vertical arrow is the inverse of the canonical isomorphism from proposition 2.2.3, $(\alpha \varphi)_{\Gamma^{p^n}}$ is the map induced by $\alpha \varphi \colon \mathcal{X}_S \to E_{S,T}$ (cf. (2.7)) on coinvariants and ι_n is as above. \Box

Note that φ_n^{α} can be defined for lower *n* than n(S): it is enough that one (as opposed to each) place above S_f is non-split in L_{∞}/L_n by proposition 2.2.3. However, restricting to $n \ge n(S)$ will be necessary soon (cf. (KC) and proposition 2.2.8) and causes no harm here. A feature of this critical layer n(S) is that, for all $n \ge n(S)$, one has a natural bijection between $S_f(L_n)$ and $S_f(L_{n(S)})$, and hence canonical $\Lambda(\mathcal{G}_n)$ -(and even $\Lambda(\mathcal{G}_{n(S)})$ -)isomorphisms

$$\mathcal{Y}_{L_n,S_f} \cong \mathcal{Y}_{L_{n(S)},S_f}$$
 and $\mathcal{X}_{L_n,S_f} \cong \mathcal{X}_{L_{n(S)},S_f}$, (2.12)

which we regard as identifications.

The property of the finite-level maps which will ensure the regulators defined in the next section are non-zero is that they are isomorphisms on χ -parts for almost all Artin characters χ . The proof hinges on the following lemma, which allows us to control their kernels and cokernels:

Lemma 2.2.7. Setting A, α as in setting B. For $n \ge n(S)$, the kernel and cokernel of φ_n^{α} fit into the short exact sequences of $\Lambda(\mathcal{G}_n)$ -modules

$$0 \to \mathcal{X}_{L_{n(S)},S_f} \to \ker(\varphi_n^{\alpha}) \to \operatorname{coker}(\alpha)^{\Gamma^{p^n}} \to 0$$
(2.13)

and

$$0 \to \operatorname{coker}(\alpha)_{\Gamma^{p^n}} \to \operatorname{coker}(\varphi_n^{\alpha}) \to \operatorname{coker}(\iota_n) \to 0$$
(2.14)

respectively. In particular,

$$\operatorname{rank}_{\mathbb{Z}_p} \operatorname{ker}(\varphi_n^{\alpha}) = \operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\varphi_n^{\alpha}) = \operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^n}} + |S_f(L_{n(S)})| - 1$$
Proof. The homomorphism φ_n^{α} is given by a composition of three maps, the first of which is an isomorphism and therefore plays no role: $\ker(\varphi_n^{\alpha}) \cong \ker(\iota_n(\alpha\varphi)_{\Gamma^{p^n}})$ and $\operatorname{coker}(\varphi_n^{\alpha}) = \operatorname{coker}(\iota_n(\alpha\varphi)_{\Gamma^{p^n}})$. Using the injectivity of ι_n , an easy application of the snake lemma yields a six-term exact sequence

$$\ker((\alpha\varphi)_{\Gamma^{p^n}}) \hookrightarrow \ker(\iota_n(\alpha\varphi)_{\Gamma^{p^n}}) \to 0 \to \operatorname{coker}((\alpha\varphi)_{\Gamma^{p^n}}) \to \operatorname{coker}(\iota_n(\alpha\varphi)_{\Gamma^{p^n}}) \twoheadrightarrow \operatorname{coker}(\iota_n).$$
(2.15)

Let us consider $(\alpha \varphi)_{\Gamma^{p^n}}$ first. Since $\alpha \varphi$ is neither injective nor surjective in general, it is convenient to define the two short exact sequences

The map φ is surjective with kernel \mathcal{X}_{S_f} , so in particular $\operatorname{im}(\alpha \varphi) = \operatorname{im}(\alpha)$ and $\operatorname{coker}(\alpha \varphi) = \operatorname{coker}(\alpha)$. Since α is injective, we have $\operatorname{im}(\alpha) \cong \mathcal{Y}_{S_{\infty}}$ and $\operatorname{ker}(\alpha \varphi) = \operatorname{ker}(\varphi) = \mathcal{X}_{S_f}$. We now take invariantscoinvariants of both short exact sequences above, the first of which yields

$$\mathcal{X}_{S_f}{}^{\Gamma^{p^n}} \hookrightarrow \mathcal{X}_{S}{}^{\Gamma^{p^n}} \to \operatorname{im}(\alpha){}^{\Gamma^{p^n}} \to (\mathcal{X}_{S_f})_{\Gamma^{p^n}} \to (\mathcal{X}_{S})_{\Gamma^{p^n}} \twoheadrightarrow \operatorname{im}(\alpha)_{\Gamma^{p^n}}.$$
(2.16)

The Γ^{p^n} -coinvariants of \mathcal{X}_{S_f} and \mathcal{X}_S coincide with \mathcal{X}_{L_n,S_f} and $\mathcal{X}_{L_n,S}$, respectively, by proposition 2.2.3. The map between them is the canonical inclusion, which is injective and has cokernel $\operatorname{im}(\alpha)_{\Gamma^{p^n}} \cong (\mathcal{Y}_{S_{\infty}})_{\Gamma^{p^n}} \cong \mathcal{Y}_{L_n,S_{\infty}}$ by the same proposition.

The second sequence induces

$$\operatorname{im}(\alpha)^{\Gamma^{p^{n}}} \hookrightarrow E_{S,T}^{\Gamma^{p^{n}}} \to \operatorname{coker}(\alpha)^{\Gamma^{p^{n}}} \to \operatorname{im}(\alpha)_{\Gamma^{p^{n}}} \to (E_{S,T})_{\Gamma^{p^{n}}} \twoheadrightarrow \operatorname{coker}(\alpha)_{\Gamma^{p^{n}}}.$$
 (2.17)

The second term admits the description

$$E_{S,T}^{\Gamma^{p^{n}}} = \left(\lim_{m} \mathbb{Z}_{p} \otimes \mathcal{O}_{L_{m},S,T}^{*}\right)^{\Gamma^{p^{n}}} = \lim_{m} \mathbb{Z}_{p} \otimes \left(\mathcal{O}_{L_{m},S,T}^{*}\right)^{\Gamma^{p^{n}}} = \lim_{m} \mathbb{Z}_{p} \otimes \left(\mathcal{O}_{L_{m},S,T}^{*} \cap L_{n}^{*}\right)$$

where the inverse limit is taken with respect to the norm maps (for the second equality, use the left exactness of \varprojlim_m , then apply the exact functor $\mathbb{Z}_p \otimes -$ to $\mathcal{O}_{L_m,S,T}^* \stackrel{\Gamma^{p^n}}{\hookrightarrow} \mathcal{O}_{L_m,S,T}^* \stackrel{\gamma^{p^n}-1}{\longrightarrow} \mathcal{O}_{L_m,S,T}^*$). But for $m \geq n$, a unit $x \in \mathcal{O}_{L_m,S,T}^* \cap L_n^*$ is sent to $x^{p^{m-n}} \in \mathcal{O}_{L_n,S,T}^*$ via the norm map N_{L_m/L_n} , which means that the *n*-th-layer component of any element of $E_{S,T}^{\Gamma^{p^n}}$ must be infinitely *p*-divisible in L_n^* , i.e. trivial. The same argument applies to the component at any layer above L_n , and hence $E_{S,T}^{\Gamma^{p^n}}$ is trivial and sequence (2.17) amounts to a four-term exact sequence.

In order to recover some information about $(\alpha \varphi)_{\Gamma^{p^n}}$, we note that this map is the composition

$$(\alpha\varphi)_{\Gamma^{p^n}} \colon (\mathcal{X}_S)_{\Gamma^{p^n}} \twoheadrightarrow (\operatorname{im} \alpha)_{\Gamma^{p^n}} \to (E_{S,T})_{\Gamma^{p^n}}$$

and apply the snake lemma incorporating the information from sequences (2.16) and (2.17):



Together with (2.15), this shows both short exact sequences in the proposition (recall also (2.12)). The equality $\operatorname{rank}_{\mathbb{Z}_p} \ker(\varphi_n^{\alpha}) = \operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\varphi_n^{\alpha})$ follows from the exact sequence

$$0 \to \ker(\varphi_n^{\alpha}) \to \mathcal{X}_{L_n,S} \xrightarrow{\varphi_n^{\alpha}} \mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^* \to \operatorname{coker}(\varphi_n^{\alpha}) \to 0,$$

where the two middle terms have the same \mathbb{Z}_p -rank $|S(L_n)| - 1$. The last equality in the proposition is then an immediate consequence of (2.13).

Part ii) of lemma 2.2.1 hints at the importance of the \mathbb{Z}_p -rank of the Γ^{p^n} -coinvariants of a torsion Iwasawa module becoming stable as n grows. Let $n(\alpha) \in \mathbb{N}$ be minimal with the property:

$$\operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^{n(\alpha)}}} \quad \text{for all } n \ge n(\alpha).$$
(2.18)

Since $\operatorname{coker}(\alpha)$ is $\Lambda(\Gamma)$ -torsion by definition, part i) of the same lemma shows $n(\alpha)$ is well defined. The other stability condition we need is satisfied by all layers starting at n(S), which motivates the definition of

$$n(S, \alpha) = \max(n(S), n(\alpha)) \in \mathbb{N}$$

and the following kernel condition:

Condition (KC). Setting A, α as in setting B. We say an irreducible Artin character $\chi \in \operatorname{Irr}_p(\mathcal{G})$ of \mathcal{G} satisfies condition (KC) if $\Gamma^{p^{n(S,\alpha)}} \nsubseteq \operatorname{ker}(\chi)$.

The fundamental property of the finite-level maps now follows immediately from lemmas 2.2.1 and 2.2.7:

Proposition 2.2.8. Setting A, α as in setting B. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfy (KC) and choose $n \in \mathbb{N}$ such that $\Gamma^{p^n} \subseteq \operatorname{ker}(\chi)$ (in particular, $n > n(S, \alpha)$). Then φ_n^{α} induces an isomorphism

$$e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha} \colon e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S} \xrightarrow{\sim} e(\chi)\mathbb{Q}_p^c \otimes \mathcal{O}_{L_n,S,T}^*$$

of $\mathbb{Q}_p^c[\mathcal{G}_n]$ -modules on χ -parts.

Proof. We start by noting that \mathbb{Q}_p^c is flat as a (right) \mathbb{Z}_p -module, so $\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} -$ takes exact sequences of (left) $\Lambda(\mathcal{G}_n) = \mathbb{Z}_p[\mathcal{G}_n]$ -modules into ones of $\mathbb{Q}_p^c[\mathcal{G}_n]$ -modules. Furthermore, since $e(\chi) \in \mathbb{Q}_p^c[\mathcal{G}_n]$ is a

central idempotent, multiplication by it is an exact endofunctor on the category of $\mathbb{Q}_p^c[\mathcal{G}_n]$ -modules. Sequence (2.13) therefore induces a short exact sequence

$$0 \to e(\chi) \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_{n(S)}, S_f} \to e(\chi) \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \ker(\varphi_n^{\alpha}) \to e(\chi) \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^{\alpha}}} \to 0.$$

By definition of $n(S, \alpha)$, we have $\operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^{n(S,\alpha)}}}$ (and hence the same equality on ranks of coinvariants) and

$$\operatorname{rank}_{\mathbb{Z}_p} \left(\mathcal{X}_{S_f} \right)_{\Gamma^{p^n}} = \operatorname{rank}_{\mathbb{Z}_p} \left(\mathcal{X}_{S_f} \right)_{\Gamma^{p^n(S,\alpha)}} = |S_f(L_{n(S)}) - 1|$$

(recall here proposition 2.2.3). In conjunction with lemma 2.2.1 ii), this shows

$$e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S_f} = e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \operatorname{coker}(\alpha)^{\Gamma^{p^-}} = 0,$$

which in turn yields $e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \ker(\varphi_n^\alpha) = 0$ by the above exact sequence.

In order to conclude the triviality of the cokernel, we use the fact that the two middle terms of the exact sequence

$$e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \ker(\varphi_n^{\alpha}) \hookrightarrow e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S} \xrightarrow{e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}} e(\chi)\mathbb{Q}_p^c \otimes \mathcal{O}_{L_n,S,T}^* \twoheadrightarrow e(\chi)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \operatorname{coker}(\varphi_n^{\alpha})$$

have the same \mathbb{Q}_p^c -dimension by virtue of the classical Dirichlet regulator map (0.24) (extending scalars to \mathbb{C}_p in both cases).

The non-vanishing of the regulator at a character χ , to be defined in the next section, will only be guaranteed if φ_n^{α} is an isomorphism on χ parts. The previous proposition therefore highlights the importance of kernel condition (KC) for the analytic side of the Main Conjecture. Fortunately, *almost all* $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfy it: the ones which do not are in bijection with the finitely many irreducible characters of $\mathcal{G}_{n(S,\alpha)}$. Since $\operatorname{Irr}_p(\mathcal{G})$ is infinite, this leaves an ample supply of characters to consider.

2.3 Regulators

One of the few missing pieces for the formulation of the Interpolation Conjecture is the *Stark-Tate* regulator. Although definition 2.3.4 below is tailored to our specific needs, namely to include the map α from setting B, it is largely influenced by (and indeed a particular case of) Tate's work on Stark's conjectures (cf. [Tat84]) - which justifies the above terminology. Before delving into the technicalities, we point out several factors which motivate the introduction of these objects. On its analytic side, the Main Conjecture will assert the existence of certain series quotients in $Q^c(\Gamma_{\chi}) \cong \mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T_{\chi}]])$ which interpolate leading coefficients at 0 of Artin *L*-functions (cf. (0.25)). The latter we shall regard as *p*-adic complex via the (inverse of the) isomorphism $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ chosen in setting B. The **regulated special value** is simply the quotient

$$\frac{\beta^{-1}(L^*_{K,S,T}(\beta\check{\chi},0))}{R^\beta_S(\alpha,\chi)}$$

by a non-zero $R_S^{\beta}(\alpha, \chi) \in \mathbb{C}_p^*$: the *regulator*. Some reasons for its necessity are as follows:

- Series quotients in $\mathcal{Q}^c(\Gamma_{\chi})$, when evaluated at the points we will, converge to values in \mathbb{Q}_p^c . However, the leading coefficients $\beta^{-1}(L^*_{K,S,T}(\beta(\check{\chi}), 0))$ are not known to be *p*-adic algebraic in general - only *p*-adic complex.
- Leading coefficients are not independent of the choice of β , which would be a desirable feature for the Main Conjecture to have. Stark's conjecture claims (essentially) that the above quotient is indeed so. This also has implications for the field where the regulated special *L*-value should be expected to live. These topics are the subject of section 3.1.
- The Main Conjecture postulates a relation between the interpolating series quotients and the refined Euler characteristic $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet},t^{\alpha})$ of the main complex. However, the latter depends fundamentally on the map α , whereas the former are characterised by an interpolation property which is oblivious to it. The introduction of regulators, which contain information about α , corrects this disparity. This will become apparent when we prove the Main Conjecture is independent of the choice of α in subsection 3.2.2.

As discussed a few lines ago, the kernel condition is essential for the definition of non-zero regulators. To that end, we introduce the following set:

Definition 2.3.1. Setting A, α as in setting B. For an irreducible Artin character χ of \mathcal{G} , we define

$$\mathcal{K}_{S}^{\alpha}(\chi) = \{ \rho \in \operatorname{Irr}_{p}(\mathcal{G}) \text{ of type } W \text{ such that } \chi \otimes \rho \text{ satisfies } (\operatorname{KC}) \}.$$

Lemma 2.3.2. Setting A, α as in setting B. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$. Then:

- i) $\mathcal{K}^{\alpha}_{S}(\chi)$ contains almost all characters of \mathcal{G} of type W. In particular, it is infinite.
- ii) For any character ρ of \mathcal{G} of type W one has $\mathcal{K}^{\alpha}_{S}(\chi \otimes \rho) = \rho^{-1} \otimes \mathcal{K}^{\alpha}_{S}(\chi)$, where the right-hand side denotes $\{\rho^{-1} \otimes \rho' : \rho' \in \mathcal{K}^{\alpha}_{S}(\chi)\}$.
- iii) One has $\mathcal{K}^{\alpha}_{S}(\check{\chi}) = \mathcal{K}^{\alpha}_{S}(\chi)^{-1}$, where the right-hand side denotes $\{\rho^{-1} : \rho \in \mathcal{K}^{\alpha}_{S}(\chi)\}$.

Proof. The only subtle point in i) is that multiplication of χ by type-W characters is not an injective operation. In other words, it may happen that $\chi \otimes \rho = \chi$ with $\rho \neq 1$ (cf. remark 2.4.6 i)). However, we claim that every fibre of the map

$$t_{\chi} \colon \{ \text{Characters of } \mathcal{G} \text{ of type } W \} \to \operatorname{Irr}_{p}(\mathcal{G})$$

 $\rho \mapsto \chi \otimes \rho$

is finite. To see this, let $\chi' \in \operatorname{Irr}_p(\mathcal{G})$. If $\chi' \not\sim_W \chi$, the fibre of χ' is empty and we are done. Otherwise, choose $n \in \mathbb{N}$ such that $\Gamma^{p^n} \subseteq \ker(\chi) \cap \ker(\chi')$. Then, for every ρ in the fibre of χ' , one has

$$\chi(1)\rho(\gamma^{p^n}) = \chi(\gamma^{p^n})\rho(\gamma^{p^n}) = \chi'(\gamma^{p^n}) = \chi'(1) = \chi(1).$$

Since $\chi(1) \neq 0$, this implies ρ is trivial on Γ^{p^n} . But only finitely many linear characters of \mathcal{G} have that property.

Having shown the claim, part i) is trivial: as mentioned at the end of the preceding section, the set of irreducible Artin characters of \mathcal{G} which do not satisfy (KC) is finite. Therefore, so is its preimage under t_{χ} , which consists precisely of the type-W characters which do not belong to $\mathcal{K}_{S}^{\alpha}(\chi)$.

Parts ii) and iii) are immediate by

$$\widetilde{\rho} \in \mathcal{K}_{S}^{\alpha}(\chi \otimes \rho) \iff (\chi \otimes \rho) \otimes \widetilde{\rho} \text{ satisfies (KC)} \iff \chi \otimes (\rho \otimes \widetilde{\rho}) \text{ satisfies (KC)} \iff \rho \otimes \widetilde{\rho} \in \mathcal{K}_{S}^{\alpha}(\chi)$$

and

$$\rho \in \mathcal{K}_{S}^{\alpha}(\check{\chi}) \iff \check{\chi} \otimes \rho \text{ satisfies (KC)} \iff \chi \otimes \rho^{-1} \text{ satisfies (KC)} \iff \rho^{-1} \in \mathcal{K}_{S}^{\alpha}(\chi),$$

where in the next-to-last equality we have used the fact that the kernel of a character coincides with that of its dual.

The aim of the Stark-Tate regulator is to capture the difference between the finite-level maps φ_n^{α} and the Dirichlet regulator map. In order to compare the two, we bring them together to a common scalar field:

Definition 2.3.3. Setting A, β as in setting B. For $n \in \mathbb{N}$, we define the *p*-adic Dirichlet regulator map $\lambda_{n,S}^{\beta}$ via the commutative diagram

$$\mathbb{C}_{p} \otimes \mathcal{O}_{L_{n},S}^{*} \xrightarrow{\lambda_{n,S}^{p}} \mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n},S}$$

$$\downarrow^{\wr} \qquad \qquad \uparrow^{\uparrow}$$

$$\mathbb{C}_{p} \otimes_{\beta^{-1}} \mathbb{C} \otimes_{\mathbb{R}} \mathbb{R} \otimes \mathcal{O}_{L_{n},S}^{*} \xrightarrow{\sim} \mathbb{C}_{p} \otimes_{\beta^{-1}} \mathbb{C} \otimes_{\mathbb{R}} \mathbb{R} \otimes \mathcal{X}_{L_{n},S}^{\mathbb{Z}}$$

where the bottom arrow is the scalar extension of the classical regulator map (0.24).

Since the classical regulator map is Galois-equivariant, $\lambda_{n,S}^{\beta}$ is an isomorphism of $\mathbb{C}_p[\mathcal{G}_n]$ -modules. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ be an irreducible Artin character and choose $n \ge n(S)$ such that χ factors through \mathcal{G}_n . We consider the $\mathbb{C}_p[\mathcal{G}_n]$ -endomorphism

$$\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}) \colon \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S} \to \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}.$$

Note that, although the codomain of φ_n^{α} is $\mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^*$, the index $[\mathcal{O}_{L_n,S}^* : \mathcal{O}_{L_n,S,T}^*]$ is finite and thus $\mathbb{C}_p \otimes \mathcal{O}_{L_n,S,T}^*$ coincides with $\mathbb{C}_p \otimes \mathcal{O}_{L_n,S}^*$. If V_{χ} is the simple $\mathbb{C}_p[\mathcal{G}_n]$ -module with character χ , the above map induces a \mathbb{C}_p -linear endomorphism $(\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}))_*$ of $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})$ by post-composition. By analogy with existing literature, we denote the determinant of this last endomorphism by

$$\det(\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}) \mid \operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})) \in \mathbb{C}_p.$$
(2.19)

Suppose now that χ satisfies (KC). Then $\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha})$ restricts to an isomorphism

$$e(\chi)\mathbb{C}_p\otimes_{\mathbb{Z}_p}\mathcal{X}_{L_n,S}\to e(\chi)\mathbb{C}_p\otimes_{\mathbb{Z}_p}\mathcal{X}_{L_n,S}$$

on χ -parts by proposition 2.2.8 (after extending scalars from \mathbb{Q}_p^c to \mathbb{C}_p). Since

$$\operatorname{Hom}_{\mathbb{C}_{n}[\mathcal{G}_{n}]}(V_{\chi}, M) = \operatorname{Hom}_{\mathbb{C}_{n}[\mathcal{G}_{n}]}(V_{\chi}, e(\chi)M)$$
(2.20)

for any $\mathbb{C}_p[\mathcal{G}_n]$ -module M (because $e(\chi)$ acts trivially on V_{χ}), the map $(\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}))_*$ is in fact an automorphism. This justifies the non-vanishing of the regulator:

Definition 2.3.4. Setting B. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ and choose $n \ge n(S)$ such that χ factors through \mathcal{G}_n . The **Stark-Tate regulator** of α on χ -parts¹⁴ is defined as

$$R_{S}^{\beta}(\alpha,\chi) = \det(\lambda_{n,S}^{\beta} \circ (\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \varphi_{n}^{\alpha}) \mid \operatorname{Hom}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(V_{\chi},\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n},S})) \in \mathbb{C}_{p},$$

which is non-zero if χ satisfies (KC).

This object behaves well with respect to character inflation:

Lemma 2.3.5. In definition 2.3.4, $R_S^{\beta}(\alpha, \chi)$ is independent of the choice of n.

Proof. Let n be as in the definition and choose m > n. Denote the projections of χ to \mathcal{G}_n and \mathcal{G}_m by χ_n and χ_m , respectively As usual, we write $\mathcal{G}_{n,v}$ for the decomposition group of $v(L_n)$ in L_n/K and analogously for $\mathcal{G}_{m,v}$. We first compare the \mathbb{C}_p -dimension of $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi_n}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})$ and $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_m]}(V_{\chi_m}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_m,S})$. One could apply lemma 0.3.1 ii) and v), but we give a self-contained proof. The basic fact at play here is that, for an irreducible character $\psi \in \operatorname{Irr}_{\mathbb{C}_p}(\mathcal{G}_n)$, one has

$$\dim_{\mathbb{C}_p}(\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi_n}, V_{\psi})) = \begin{cases} 1, & \psi = \chi_n, \\ 0, & \text{otherwise}, \end{cases}$$

the first case being a consequence of Wedderburn's theorem (0.13) (regarding V_{χ} as a column vector). It follows that, if ψ_n is the character of $\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}$ in the sense of (0.16), the \mathbb{C}_p -dimension of $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi_n}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})$ is $\langle \chi_n, \psi_n \rangle$. The short exact sequence

$$0 \to \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S} \to \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n,S} \to \mathbb{C}_p \to 0$$

together with the fact that $\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n,S} = \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n} \mathbb{C}_p$, implies $\psi_n = (\sum_{v \in S} \operatorname{ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n} \mathbb{1}_{\mathcal{G}_{n,v}}) - \mathbb{1}_{\mathcal{G}_n}$. Therefore,

$$\langle \chi_n, \psi_n \rangle_{\mathcal{G}_n} = \left(\sum_{v \in S} \langle \chi_n, \operatorname{ind}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n} \mathbb{1}_{\mathcal{G}_{n,v}} \rangle_{\mathcal{G}_n} \right) - \langle \chi_n, \mathbb{1}_{\mathcal{G}_n} \rangle_{\mathcal{G}_n} = \left(\sum_{v \in S} \langle \operatorname{res}_{\mathcal{G}_{n,v}}^{\mathcal{G}_n} \chi_n, \mathbb{1}_{\mathcal{G}_{n,v}} \rangle_{\mathcal{G}_{n,v}} \right) - \langle \chi_n, \mathbb{1}_{\mathcal{G}_n} \rangle_{\mathcal{G}_n}$$

by Frobenius reciprocity (0.11). An analogous argument yields

$$\dim_{\mathbb{C}_p}(\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_m]}(V_{\chi_m},\mathbb{C}_p\otimes_{\mathbb{Z}_p}\mathcal{X}_{L_m,S})) = \left(\sum_{v\in S} \langle \operatorname{res}_{\mathcal{G}_{m,v}}^{\mathcal{G}_m}\chi_m,\mathbb{1}_{\mathcal{G}_{m,v}}\rangle_{\mathcal{G}_{m,v}}\right) - \langle \chi_m,\mathbb{1}_{\mathcal{G}_m}\rangle_{\mathcal{G}_m},$$

¹⁴Some authors, most notably Tate, denote by $R(\chi,...)$ a similar regulator constructed by looking at $\tilde{\chi}$ -parts instead (cf. [Tat84], p. 26). The reader should be aware of the different notational choice in this text, where $R_S^\beta(\alpha, \chi)$ really refers to the regulator on χ -parts.

after which a simple computation (for instance, directly using the definition of the scalar product of characters) shows that the two coincide.

We may regard the canonical projection $\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_m,S} \twoheadrightarrow \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}$ as a $\mathbb{C}_p[\mathcal{G}_m]$ -homomorphism via $\mathbb{C}_p[\mathcal{G}_m] \twoheadrightarrow \mathbb{C}_p[\mathcal{G}_n]$. By the same token, V_{χ_m} and V_{χ_n} are isomorphic $\mathbb{C}_p[\mathcal{G}_m]$ -modules (observe here that $\chi_m = \inf_{\mathcal{G}_m}^{\mathcal{G}_m} \chi_n$). We therefore have a \mathbb{C}_p -linear surjection

 $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_m]}(V_{\chi_m}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_m, S}) \twoheadrightarrow \operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_m]}(V_{\chi_n}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n, S}) = \operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi_n}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n, S})$

by the semisimplicity of $\mathbb{C}_p[\mathcal{G}_m]$. But we have shown the first and last terms have the same \mathbb{C}_p dimensions, and hence this surjection is an isomorphism. The lemma now follows from the commutativity of the diagram

As mentioned in (2.20), applying $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi}, -)$ to a $\mathbb{C}_p[\mathcal{G}_n]$ -module is simply a way of taking χ -parts. A more immediate method we have also encountered is to multiply the module by $e(\chi)$. Since $\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha})$ is a $\mathbb{C}_p[\mathcal{G}_n]$ -endomorphism of $e(\chi)\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}$ (and hence \mathbb{C}_p -linear), one could define the regulator as the determinant of this transformation instead. This approach is closely related to the above, but it yields a coarser magnitude: it can be shown that

$$\det(\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}) \mid e(\chi)\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}) = R_S^{\beta}(\alpha,\chi)^{\chi(1)}.$$

Subsequent sections (most notably 3.2) will illustrate why $R_S^{\beta}(\alpha, \chi)$ is the right choice for our purposes.

2.4 Evaluation maps

As explained in the introduction to the present chapter, the Interpolation Conjecture will assert the existence of certain quotients of power series which interpolate regulated special *L*-values. More specifically, the series quotient associated to a character $\chi \in \operatorname{Irr}_p(\mathcal{G})$ is expected to interpolate the special values corresponding to almost all of its ρ -twists. By a ρ -twist of χ we mean its product with a character of type W - in other words, any character which is W-equivalent to χ in the terminology of section 2.2.

In this section, we present the relation between evaluation at 0 of series quotients associated to Wequivalent characters. This is motivated by the structure of the centre of $Q^c(\mathcal{G})$ studied by Ritter and Weiss in [RW04] and leads us to define *twisted evaluation maps*, which amount to evaluation at certain *p*-adic algebraic points in the open unit ball. Requiring a single series quotient to interpolate *L*-values under infinitely many twisted evaluation maps therefore ensures its uniqueness - if it exists - which is important for the Main Conjecture.

For the first definition, let $\tilde{\Gamma} \cong \mathbb{Z}_p$ be a profinite group and fix a topological generator $\tilde{\gamma}$ (the notation here has been chosen to avoid confusion with the specific Γ and γ from setting A). The crucial isomorphism (0.7) holds for more general rings of coefficients than \mathbb{Z}_p : given a *p*-adic field E, the homomorphism of topological \mathcal{O}_E -algebras

$$\Lambda^{\mathcal{O}_E}(\tilde{\Gamma}) \to \mathcal{O}_E[[\tilde{T}]]$$
$$\tilde{\gamma} \mapsto \tilde{T} + 1$$

is an isomorphism, where $\mathcal{O}_E[[\tilde{T}]]$ is equipped with the $\langle \pi, \tilde{T} \rangle$ -adic topology for any uniformiser π of E. Under this identification, the augmentation map from section 0.1 corresponds to evaluation at 0, i.e. the diagram

$$\begin{array}{ccc} \Lambda^{\mathcal{O}_E}(\tilde{\Gamma}) & \xrightarrow{\tilde{\gamma} \mapsto \tilde{T}+1} & \mathcal{O}_E[[\tilde{T}]] \\ & \downarrow^{\operatorname{aug}_{\tilde{\Gamma}}} & \downarrow^{\tilde{T}=0} \\ & \mathcal{O}_E & = & \mathcal{O}_E \end{array}$$

commutes. Recall that its kernel, the augmentation ideal $\Delta^{\mathcal{O}_E}(\tilde{\Gamma}) = \ker(\operatorname{aug}_{\tilde{\Gamma}})$, is a height-one prime ideal of $\Lambda^{\mathcal{O}_E}(\tilde{\Gamma})$ generated by $\tilde{\gamma} - 1$. The map $\operatorname{aug}_{\tilde{\Gamma}}$ has a natural extension to the field of fractions $\mathcal{Q}^E(\tilde{\Gamma}) = \operatorname{Frac}(\Lambda^{\mathcal{O}_E}(\tilde{\Gamma})) = E \otimes_{\mathbb{Q}_p} \mathcal{Q}(\tilde{\Gamma})$:

Definition 2.4.1. Let E be a p-adic field and $\tilde{\gamma}$ a topological generator of a profinite group $\tilde{\Gamma} \cong \mathbb{Z}_p$. We define the evaluation-at-0 map with respect to $\tilde{\gamma}$ as

$$ev_{\tilde{\gamma}} \colon \mathcal{Q}^{E}(\tilde{\Gamma}) \to E \cup \{\infty\}$$
$$q = \frac{f}{g} \mapsto \begin{cases} \frac{\operatorname{aug}_{\tilde{\Gamma}}(f)}{\operatorname{aug}_{\tilde{\Gamma}}(g)}, & q \in \Lambda^{\mathcal{O}_{E}}(\tilde{\Gamma})_{\Delta^{\mathcal{O}_{E}}(\tilde{\Gamma})}\\ \infty, & \text{otherwise} \end{cases}$$

where the subscript in $\Lambda^{\mathcal{O}_E}(\tilde{\Gamma})_{\Delta^{\mathcal{O}_E}(\tilde{\Gamma})}$ denotes localisation and, in the first case, g is chosen outside $\Delta^{\mathcal{O}_E}(\tilde{\Gamma})$. Furthermore, we define

$$ev_{\tilde{\gamma}} \colon \mathcal{Q}^c(\tilde{\Gamma}) = \mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \mathcal{Q}(\tilde{\Gamma}) = \varinjlim_E \mathcal{Q}^E(\tilde{\Gamma}) \to \mathbb{Q}_p^c \cup \{\infty\}$$

as the direct limit of the above maps.

Remark 2.4.2. i) The map $ev_{\tilde{\gamma}} \colon \mathcal{Q}^{E}(\tilde{\Gamma}) \to E \cup \{\infty\}$ is well defined, i.e. independent of the choice of an expression q = f/g for $q \in \Lambda^{\mathcal{O}_{E}}(\tilde{\Gamma})_{\Delta^{\mathcal{O}_{E}}(\tilde{\Gamma})}$, because $\operatorname{aug}_{\tilde{\Gamma}}$ is a ring homomorphism on $\Lambda^{\mathcal{O}_{E}}(\tilde{\Gamma})$. The well-definedness of $ev_{\tilde{\gamma}} \colon \mathcal{Q}^{c}(\tilde{\Gamma}) \to \mathbb{Q}_{p}^{c} \cup \{\infty\}$ follows immediately from the compatibility of $\operatorname{aug}_{\tilde{\Gamma}}$ with extension of scalars - which also shows that, if E'/E is an extension of *p*-adic fields, then $\Delta^{\mathcal{O}_{E'}}(\tilde{\Gamma}) = \mathcal{O}_{E'} \otimes_{\mathcal{O}_{E}} \Delta^{\mathcal{O}_{E}}(\tilde{\Gamma})$.

ii) $E \cup \{\infty\}$ has no natural ring or *E*-vector-space structure, so $ev_{\tilde{\gamma}}$ is not a ring homomorphism and it is not *E*-linear. However, its restriction to the $\Lambda^{\mathcal{O}_E}(\tilde{\Gamma})_{\Delta^{\mathcal{O}_E}(\tilde{\Gamma})}$ is indeed a homomorphism of *E*-algebras for the same reasons as above. In other words, $ev_{\tilde{\gamma}}$ is compatible with sums, products and scaling as long as none of the involved series quotients evaluate to ∞ .

The definition of $ev_{\tilde{\gamma}}$ as the evaluation-at-0 map with respect to $\tilde{\gamma}$ may appear superfluous, since it is independent of the choice of a topological generator $\tilde{\gamma}$ of $\tilde{\Gamma}$ (because so is $aug_{\tilde{\Gamma}}$ by definition). Its relevance will become clear in a few lines, where $\tilde{\gamma}$ will be replaced not by another topological generator of the same group, but by a scaled version of itself. The inclusion of $\tilde{\gamma}$ in the notation will essentially allow us to track changes of variables in power series rings later on.

The next two results are taken from [RW04] (see the footnote¹² on page 50) and stated here without proof. More specifically, they are a rearrangement of proposition 5 and the corollary to proposition 6 in the article. They concern structural properties of $Q^c(\mathcal{G})$ (cf. (2.5) and (2.6)) which will be essential in the sequel. The first one characterises the primitive central idempotents of the ring:

Proposition 2.4.3. Setting A. The following hold:

- i) The ring $\mathcal{Q}^{c}(\mathcal{G})$ is semisimple Artinian.
- ii) Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$. Then

$$e_{\chi} = \sum_{\substack{\eta \in \operatorname{Irr}_{\mathbb{Q}_{p}^{c}}(H)\\\eta \mid \operatorname{res}_{H}^{\mathcal{G}}\chi}} e(\eta)$$

is a primitive central idempotent of $\mathcal{Q}^{c}(\mathcal{G})$.

- iii) Every primitive central idempotent of $\mathcal{Q}^{c}(\mathcal{G})$ is of the form e_{χ} as above for some $\chi \in \operatorname{Irr}_{p}(\mathcal{G})$.
- iv) Let $\chi, \chi' \in \operatorname{Irr}_p(\mathcal{G})$. Then the following are equivalent:
 - $e_{\chi} = e_{\chi'}$.
 - $\chi \sim_W \chi'$.
 - $\operatorname{res}_{H}^{\mathcal{G}}\chi = \operatorname{res}_{H}^{\mathcal{G}}\chi'.$

Consider χ and η as in part ii). For $\sigma \in \mathcal{G}$, let η^{σ} be the conjugated character given by

$$\eta^{\sigma}(\tau) = \eta(\sigma\tau\sigma^{-1})$$

on $\tau \in H$. This is again an irreducible character of H which divides $\operatorname{res}_{H}^{\mathcal{G}}\chi$, as $\chi^{\sigma} = \chi$. Thus, the elements of \mathcal{G} permute the irreducible constituents of $\operatorname{res}_{H}^{\mathcal{G}}\chi$, and this action is in fact transitive by Clifford theory. We define the stabiliser of η as the group

$$St(\eta) = \{ \sigma \in \mathcal{G} \text{ such that } \eta^{\sigma} = \eta \} \leq \mathcal{G},$$

so that when τ runs over a set of coset representatives of $St(\eta) \setminus \mathcal{G}$, the character η^{τ} runs over the irreducible constituents of $\operatorname{res}_{H}^{\mathcal{G}} \chi$ exactly once. $St(\eta)$ is an open subgroup (as can be seen by projecting χ to a finite quotient \mathcal{G}_n of \mathcal{G}) containing H, and hence of the form $\operatorname{Gal}(L_{\infty}/K_n)$ for some n. This shows that

$$w_{\chi} = [\mathcal{G} : St(\eta)] \tag{2.21}$$

is a power of p. Since all such subgroups are normal, Clifford theory shows that $St(\eta)$ does not depend of the choice of $\eta \mid \operatorname{res}_{H}^{\mathcal{G}} \chi$ and the notation of w_{χ} is thus justified. This index plays an important role in the next result, again taken from [RW04]:

Proposition 2.4.4. Setting A. Let $\chi \in Irr_p(\mathcal{G})$. Then the following hold:

- i) There exists a unique $\gamma_{\chi} \in Z(\mathcal{Q}^c(\mathcal{G})e_{\chi})$ satisfying:
 - $\gamma_{\chi} = gc$ for some $g \in \mathcal{G}$ with projection $\gamma_{K}^{w_{\chi}} \in \Gamma_{K}$ (where γ_{K} is the topological generator chosen in the setting) and some $c \in (\mathbb{Q}_{p}^{c}[H]e_{\chi})^{*}$.
 - γ_{χ} acts trivially on the \mathbb{Q}_p^c -vector space V_{χ} which affords χ .

This γ_{χ} also satisfies $\gamma_{\chi} = cg$, and it topologically generates a subgroup $\Gamma_{\chi} \cong \mathbb{Z}_p$ of $(\mathcal{Q}^c(\mathcal{G})e_{\chi})^*$.

- ii) Let E be a p-adic field such that χ has a realisation over E. Then the group Γ_{χ} from i) is contained in $Z(\mathcal{Q}^E(\mathcal{G})e_{\chi})^*$, which induces an equality of fields $\mathcal{Q}^E(\Gamma_{\chi}) = Z(\mathcal{Q}^E(\mathcal{G})e_{\chi})$.
- iii) With E as in ii), the ring homomorphism

$$j_{\chi}^E \colon Z(\mathcal{Q}^E(\mathcal{G})) \twoheadrightarrow Z(\mathcal{Q}^E(\mathcal{G})e_{\chi}) = \mathcal{Q}^E(\Gamma_{\chi}) \hookrightarrow \mathcal{Q}^E(\Gamma_K),$$

where the last arrow is induced by $\gamma_{\chi} \mapsto \gamma_{K}^{w_{\chi}}$, is independent of the choice of γ_{K} .

iv) If ρ is a type-W character of \mathcal{G} , then $\gamma_{\chi\otimes\rho} = \rho(\gamma_K)^{-w_{\chi}}\gamma_{\chi}$.

Proposition 2.4.3 implies that $\mathcal{Q}^{c}(\mathcal{G})$ decomposes as a finite product of simple rings

$$Q^{c}(\mathcal{G}) \cong \prod_{\chi/\sim_{W}} Q^{c}(\mathcal{G})e_{\chi},$$
(2.22)

where χ runs over any set of representatives of $\operatorname{Irr}_p(\mathcal{G})$ modulo *W*-equivalence. The reduced norm on K_1 introduced in section 0.4 then factors over these χ -parts by definition, i.e. there is a commutative diagram

of abelian groups. By part ii) of the last proposition, each term in the bottom-right corner is in fact the group of units of a field $\mathcal{Q}^c(\Gamma_{\chi})$ of series quotients in the variable $T_{\chi} = \gamma_{\chi} - 1$. It is precisely there that the interpolating elements of the Main Conjecture are expected to live. However, to conjecture the existence of an element therein which evaluates at 0 to the special *L*-value associated to χ would be an empty statement in isolation - and also entail an evident lack of uniqueness. Both issues can be resolved through the fundamental relation iv) above, which hints at the notion that one should consider interpolation for all ρ -twists of χ simultaneously. The following lemma lays the foundations for that approach:

Lemma 2.4.5. Setting A. Let $\chi, \rho \in \operatorname{Irr}_p(\mathcal{G})$ with ρ of type W and choose a p-adic field E such that χ and ρ (and in particular $\chi \otimes \rho$) have a realisation over it. Then $\mathcal{Q}^E(\Gamma_{\chi}) = \mathcal{Q}^E(\Gamma_{\chi \otimes \rho}) \subseteq \mathcal{Q}^E(\mathcal{G})$ and, for a series quotient $q \in \mathcal{Q}^E(\Gamma_{\chi}) \cong \operatorname{Frac}(\mathcal{O}_E[[T_{\chi}]])$, one has

$$ev_{\gamma_{\chi\otimes\rho}}(q) = q|_{T_{\chi}=\rho(\gamma_K)^{w_{\chi}}-1} \in E \cup \{\infty\},$$

where the right-hand side denotes evaluation at $T_{\chi} = \rho(\gamma_K)^{w_{\chi}} - 1 \in E$ and we set $k/0 = \infty$ for $k \in E^*$.

Proof. We first show $\mathcal{Q}^E(\Gamma_{\chi}) = \mathcal{Q}^E(\Gamma_{\chi \otimes \rho})$. This follows immediately from propositions 2.4.3 iv) and 2.4.4 ii), but we present here a more explicit proof different from that in the reference. The Iwasawa algebras $\Lambda^{\mathcal{O}_E}(\Gamma_{\chi})$ and $\Lambda^{\mathcal{O}_E}(\Gamma_{\chi \otimes \rho})$ are defined as

$$\Lambda^{\mathcal{O}_E}(\Gamma_{\chi}) = \varprojlim_{n} \mathcal{O}_E\left[\Gamma_{\chi} \mathcal{I}_{\chi}^{p^n}\right] \quad \text{and} \quad \Lambda^{\mathcal{O}_E}(\Gamma_{\chi \otimes \rho}) = \varprojlim_{n} \mathcal{O}_E\left[\Gamma_{\chi \otimes \rho} \mathcal{I}_{\chi \otimes \rho}^{p^n}\right].$$
(2.24)

Let us apply the relation $\gamma_{\chi} = \rho(\gamma_K)^{w_{\chi}} \gamma_{\chi \otimes \rho}$ from proposition 2.4.4, denoting $\rho(\gamma_K)^{w_{\chi}}$ by $\zeta \in \mathcal{O}_E$. Since ζ is a *p*-power root of unity (potentially 1), one has $\gamma_{\chi}^{p^n} = \gamma_{\chi \otimes \rho}^{p^n}$ (and hence $\Gamma_{\chi}^{p^n} = \Gamma_{\chi \otimes \rho}^{p^n}$) for all *n* larger than some fixed n_0 satisfying $\zeta^{p^{n_0}} = 1$. This yields an equality

$$\sum_{i=0}^{p^n-1} a_i \cdot \gamma_{\chi}^i \Gamma_{\chi}^{p^n} = \sum_{i=0}^{p^n-1} (a_i \zeta^i) \cdot \gamma_{\chi \otimes \rho}^i \Gamma_{\chi \otimes \rho}^{p^n}$$

in $\mathcal{O}_E[\Gamma_{\chi}/\Gamma_{\chi}^{p^n}]$ for all such n, where the left-hand term is a generic element and the right-hand one clearly lies in $\mathcal{O}_E[\Gamma_{\chi\otimes\rho}/\Gamma_{\chi\otimes\rho}^{p^n}]$. This equality commutes with the transition maps in (2.24) and therefore implies $\Lambda^{\mathcal{O}_E}(\Gamma_{\chi}) = \Lambda^{\mathcal{O}_E}(\Gamma_{\chi\otimes\rho})$, which extends to an equality of their fields of fractions $\mathcal{Q}^E(\Gamma_{\chi}) = \mathcal{Q}^E(\Gamma_{\chi\otimes\rho})$ inside the common ring $\mathcal{Q}^E(\mathcal{G})$.

The induced isomorphism

$$\mathcal{O}_E[[T_{\chi}]] \xrightarrow{T_{\chi} \mapsto \gamma_{\chi} - 1} \Lambda^{\mathcal{O}_E}(\Gamma_{\chi}) = \Lambda^{\mathcal{O}_E}(\Gamma_{\chi \otimes \rho}) \xrightarrow{\gamma_{\chi \otimes \rho} \mapsto T_{\chi \otimes \rho} + 1} \mathcal{O}_E[[T_{\chi \otimes \rho}]]$$
$$\gamma_{\chi} = \rho(\gamma_K)^{w_{\chi}} \gamma_{\chi \otimes \rho}$$

gives the relation $T_{\chi} = \rho(\gamma_K)^{w_{\chi}}(T_{\chi \otimes \rho} + 1) - 1$, which identifies evaluation at $T_{\chi \otimes \rho} = 0$ with evaluation at $T_{\chi} = \rho(\gamma_K)^{w_{\chi}} - 1$.

Passing now to the field of fractions, let $q = f/g \in \mathcal{Q}^E(\Gamma_{\chi \otimes \rho})$ be in reduced form (recall that $\mathcal{O}_E[[T_{\chi \otimes \rho}]] \cong \Lambda^{\mathcal{O}_E}(\Gamma_{\chi \otimes \rho})$ is a unique factorisation domain) with $fg \neq 0$. If $g \in \Delta^{\mathcal{O}_E}(\Gamma_{\chi \otimes \rho})$, then the above argument shows

$$g|_{T_{\chi}=\rho(\gamma_K)^{w_{\chi}}-1} = g|_{T_{\chi\otimes\rho}=0} = 0 \text{ and } f|_{T_{\chi}=\rho(\gamma_K)^{w_{\chi}}-1} = f|_{T_{\chi\otimes\rho}=0} \neq 0$$

(otherwise they would share the factor $T_{\chi\otimes\rho}$) and hence $q|_{T_{\chi}=\rho(\gamma_K)^{w_{\chi}}-1} = \infty$, which matches the definition of $ev_{\gamma_{\chi\otimes\rho}}(q)$. If $y \notin \Delta^{\mathcal{O}_E}(\Gamma_{\chi\otimes\rho})$, one has

$$ev_{\gamma_{\chi\otimes\rho}}(q) = \frac{f|_{T_{\chi\otimes\rho}=0}}{g|_{T_{\chi\otimes\rho}=0}} = \frac{f|_{T_{\chi}=\rho(\gamma_{K})^{w_{\chi}}-1}}{g|_{T_{\chi}=\rho(\gamma_{K})^{w_{\chi}}-1}} = q|_{T_{\chi}=\rho(\gamma_{K})^{w_{\chi}}-1}$$

as well.

Remark 2.4.6. i) The map $ev_{\gamma_{\chi}}$ depends on χ alone, rather than on its specific expression in terms of ρ -twists. As a sanity check, suppose that $\chi = \chi \otimes \rho$ for some ρ of type W. The lemma then says that

$$-|_{T_{\chi}=\rho(\gamma_K)^{w_{\chi}}-1} = ev_{\gamma_{\chi\otimes\rho}} = ev_{\gamma_{\chi}} = -|_{T_{\chi}=0}$$

This can only happen if $\rho(\gamma_K)^{w_{\chi}} = 1$, but by proposition 2.4.4 iv), this is precisely the case.

For an example of $\chi = \chi \otimes \rho$ with non-trivial ρ , let ρ be the inflation to \mathcal{G} of the linear character of Γ_K which sends γ_K to a primitive *p*-th root of unity. Denote by $G = \operatorname{Gal}(L_{\infty}/K_1)$ the preimage of Γ_K^p in \mathcal{G} under the canonical projection, which is open and normal. Let ψ be an Artin character of G whose induction $\chi = \operatorname{ind}_G^{\mathcal{G}} \psi$ is irreducible - such characters can be easily shown to exist in some cases. Then, for any $g \in \mathcal{G}$, one has either $g \in G$, in which case $\rho(g) = 1$; or $g \notin G$, in which case $\chi(g) = 0$. Therefore, $\chi = \chi \otimes \rho$.

ii) It follows from the lemma that a series $x \in \mathcal{O}_E[[T_{\chi}]]$ evaluated at $T_{\chi} = \zeta - 1$, where ζ is a *p*-power root of unity, yields an element in \mathcal{O}_E - in other words, it converges. This agrees with basic *p*-adic analysis: since the element $\zeta - 1$ is in the maximal ideal of $\mathbb{Z}[\zeta]$ and hence of \mathcal{O}_E , any series

$$\sum_{i=0}^{\infty} a_i (\zeta - 1)^i$$

in \mathcal{O}_E has terms tending to 0 and is therefore convergent by the completeness of E.

iii) Taking direct limits on the first statement of the lemma yields $\mathcal{Q}^c(\Gamma_{\chi}) = \mathcal{Q}^c(\Gamma_{\chi \otimes \rho})$.

It stands to reason to refer to $ev_{\gamma_{\chi\otimes\rho}}$ as a twist of $ev_{\gamma_{\chi}}$, and thus as a **twisted evaluation map**. Since type-W characters can map γ_K to roots of unity of order an arbitrary power of p, fixing a series quotient $q \in \mathcal{Q}^c(\Gamma_{\chi})$ and applying $ev_{\chi\otimes\rho}$ for all ρ corresponds to evaluating it at infinitely many points of \mathbb{Q}_p^c . These points lie in the open unit ball centred at 0 and become arbitrarily close to its boundary, as the p-adic valuation of $\zeta_{p^n} - 1$ (with ζ_{p^n} of order p^n) is $1/(p^{n-1}(p-1))$ for $n \geq 1$.

The twisted evaluation maps are closely related to the ρ^{\sharp} in the cited article (cf. [RW04], definition on page 557). This automorphism of $Q^c(\Gamma_K)$ is induced by the continuous \mathcal{O}_E -linear automorphism of $\Lambda^{\mathcal{O}_E}(\Gamma_K)$ determined by $\rho^{\sharp}(\gamma_K) = \rho(\gamma_K)\gamma_K$. The connection to our maps is given by the fact that the composition

$$\mathcal{Q}^{c}(\Gamma_{\chi}) \xrightarrow{\gamma_{\chi} \mapsto \gamma_{K}^{w_{\chi}}} \mathcal{Q}^{c}(\Gamma_{K}) \xrightarrow{\rho^{\sharp}} \mathcal{Q}^{c}(\Gamma_{K}) \xrightarrow{ev_{\gamma_{K}}} \mathbb{Q}_{p}^{c} \cup \{\infty\},$$
(2.25)

is precisely $ev_{\gamma_{\chi\otimes\rho}}$. Indeed,

$$ev_{\gamma_K}(\rho^{\sharp}(\gamma_K^{w_{\chi}})) = ev_{\gamma_K}(\rho(\gamma_K)^{w_{\chi}}\gamma_K^{w_{\chi}}) = \rho(\gamma_K)^{w_{\chi}} = ev_{\gamma_{\chi\otimes\rho}}(\gamma_{\chi}).$$

The arrows in (2.25) play an important role in the Hom-formulation of Ritter and Weiss, but we shall bypass them using the twisted evaluation maps. This will allow us to conjecture the existence of interpolating series quotients in $\mathcal{Q}^c(\Gamma_{\chi})$, rather than in the larger $\mathcal{Q}^c(\Gamma_K)$.

We now address the dependence of $ev_{\gamma_{\chi}}$ on the choice of γ_{K} :

Lemma 2.4.7. Setting A. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ and choose a finite extension E of \mathbb{Q}_p over which χ has a realisation. Fix a topological generator $\tilde{\gamma}_K$ of Γ_K in addition to the γ_K in the setting, and let $\tilde{\gamma}_{\chi}$ and γ_{χ} be the corresponding elements of $Z(\mathcal{Q}^E(\mathcal{G})e_{\chi})$ constructed in proposition 2.4.4. Denote by Γ_{χ} and $\tilde{\Gamma}_{\chi}$ the groups topologically generated by γ_{χ} and $\tilde{\gamma}_{\chi}$, respectively. Then $\mathcal{Q}^E(\Gamma_{\chi}) = \mathcal{Q}^E(\tilde{\Gamma}_{\chi}) =$ $Z(\mathcal{Q}^E(\mathcal{G})e_{\chi})$ and the maps

$$ev_{\gamma_{\chi}}, ev_{\gamma_{\tilde{\chi}}} \colon \mathcal{Q}^E(\Gamma_{\chi}) = \mathcal{Q}^E(\widetilde{\Gamma}_{\chi}) \to E \cup \{\infty\}$$

coincide.

Proof. By proposition 2.4.4 ii), we have equalities $\mathcal{Q}^E(\Gamma_{\chi}) = Z(\mathcal{Q}^E(\mathcal{G})e_{\chi}) = \mathcal{Q}^E(\widetilde{\Gamma}_{\chi})$ (not mere isomorphisms), where the middle term - most notably, e_{χ} - does not depend on the choice of γ_K . Consider the diagram



The diamond commutes by proposition 2.4.4 iii), as do clearly the three triangles. This shows the commutativity of the entire diagram, whose two horizontal arrows are precisely $ev_{\gamma_{\chi}}$ and $ev_{\gamma_{\tilde{\chi}}}$.

Remark 2.4.8. The topological generators of \mathbb{Z}_p are precisely the units \mathbb{Z}_p^* , and therefore any γ_K and $\tilde{\gamma}_K$ as in the lemma satisfy $\tilde{\gamma}_K = \gamma_K^v$ for some $v \in \mathbb{Z}_p^*$. It can then be shown (cf. [RW04] proof of proposition 6 (4)) that the resulting γ_{χ} and $\tilde{\gamma}_{\chi}$ are related by the equation $\tilde{\gamma}_{\chi} = \gamma_{\chi}^v$ as well, which implies $\Gamma_{\chi} = \tilde{\Gamma}_{\chi}$. Note the difference between this situation and that of W-equivalent characters (proposition 2.4.4, lemma 2.4.5), where Γ_{χ} and $\Gamma_{\chi\otimes\rho}$ may not in coincide even though $\mathcal{Q}^E(\Gamma_{\chi})$ and $\mathcal{Q}^E(\Gamma_{\chi\otimes\rho})$ always do.

Our last aim in this section is to show that series quotients are uniquely determined by their values at any infinite set of integral points where they converge and therefore, in our particular setting, by their values under the twisted evaluation maps $ev_{\gamma_{\chi\otimes\rho}}$. An important comment regarding evaluation of series quotients at points of \mathbb{Q}_p^c and \mathbb{C}_p is in order. We phrase it as a remark for subsequent reference: Remark 2.4.9. Consider a finite extension E/\mathbb{Q}_p and a series $f \in \mathcal{O}_E[[T]]$. Let us denote the open unit ball in \mathbb{Q}_p^c by $B_1^{\mathbb{Q}_p^c}(0) = \{x \in \mathbb{Q}_p^c : |x|_p < 1\}$ with $|-|_p$ normalised to $|p|_p = p^{-1}$, i.e. $v_p(p) = 1$. Any $x \in B_1^{\mathbb{Q}_p^c}(0)$ lies in some finite extension F/\mathbb{Q}_p , which we can choose to contain E. In fact, xbelongs to the maximal ideal of \mathcal{O}_F and hence the series f converges at x to a value $f(x) \in \mathcal{O}_F$ by completeness. The result is independent of the choice of F, and we can thus evaluate series in $\mathcal{O}_E[[T]]$ at points of $B_1^{\mathbb{Q}_p^c}(0)$.

In the case of a series quotient $q \in \operatorname{Frac}(\mathcal{O}_E[[T]])$, evaluation is well defined by expressing q = f/gwith $f, g \in \mathcal{O}_E[[T]]$ coprime and $fg \neq 0$ (if f = 0, the discussion is obvious). Namely, for $x \in B_1^{\mathbb{Q}_p^c}(0)$, $x \in F$ as above, we set $q(x) = f(x)/g(x) \in F \cup \{\infty\}$ with the usual convention that $k/0 = \infty$ for $k \neq 0$. In order to prove that this is well defined, we only need to show one cannot simultaneously have f(x) = g(x) = 0. This is clear when F = E but requires an argument otherwise. We resort to the Weierstrass preparation theorem (cf. [NSW20] theorem 5.3.4), by which f has a (unique) decomposition

$$f = \pi_E^s u P$$

with π_E a chosen uniformiser of $E, s \in \mathbb{N}, u \in \mathcal{O}_E[[T]]^*$ a unit and $P \in \mathcal{O}_E[T]$ a Weierstrass polynomial (cf. section 0.1). Note that u(x) cannot vanish, since crucially both u and u^{-1} converge at x (because $|x|_p < 1$). Hence f(x) = 0 implies P(x) = 0, and therefore the minimal polynomial $\min_E(x) \in E[T]$ of x over E divides P. An easy computation with the ultrametric inequality then shows that $\min_E(x) \in \mathcal{O}_E[T]$. It follows that the decomposition $P = \min_E(x)P'$ in E[T] takes place entirely over $\mathcal{O}_E[T]$, as can be seen by multiplying P' by a suitable power of π_E passing to $\mathcal{O}_E/\mathfrak{m}_E[T]$. This shows that $\min_E(x)$ divides the series f in $\mathcal{O}_E[[T]]$. By the same argument, g(x) = 0 implies $\min_E(x) \mid g$. Since we chose q = f/g in reduced form, it cannot happen that f(x) = g(x) = 0, as desired.

The key takeaway from this, which will be essential for the next two results, is that one can evaluate an element $h \in \mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T]]) = \varinjlim_E \operatorname{Frac}(\mathcal{O}_E[[T]])$ at all points of $B_1^{\mathbb{Q}_p^c}(0)$ by choosing a fixed representative h = f/g independent of the point.

Although it will not be relevant in the sequel, there is a natural way to extend the above discussion to $B_1^{\mathbb{C}_p}(0) = \{x \in \mathbb{C}_p : |x|_p < 1\}$: for h = f/g as above (with $fg \neq 0$) and $x \in B_1^{\mathbb{C}_p}(0) \setminus B_1^{\mathbb{Q}_p^c}(0)$, simply set h(x) = f(x)/g(x). Note that, in this case, neither f(x) nor g(x) can be zero, as x would then be a root of the polynomial appearing in the corresponding Weierstrass decomposition, and hence algebraic over \mathbb{Q}_p . If we approximate x by a Cauchy sequence $\{x_n\}_{n\in\mathbb{N}} \subseteq B_1^{\mathbb{Q}_p^c}(0)$, then

$$h(x) = \frac{f(x)}{g(x)} = \frac{\lim_{n \to \infty} f(x_n)}{\lim_{n \to \infty} g(x_n)} = \lim_{n \to \infty} \frac{f(x_n)}{g(x_n)} = \lim_{n \to \infty} h(x_n),$$

where the second and third equalities use the continuity of power series and that of a quotient of continuous functions. The last equality is immediate by definition if f and g are chosen coprime to begin with, but it holds in fact without this restriction as well: since the sequence $\{x_n\}$ does not converge in \mathbb{Q}_p^c , any p-adic field - and in particular the splitting fields of the polynomials in the Weierstrass decompositions of f and g - contains only finitely many of its terms. Therefore, $f(x_n) \neq 0 \neq g(x_n)$ for large enough n.

The following lemma extends a classical application of the Weierstrass preparation theorem to the case of series quotients:

Lemma 2.4.10. Let S be an infinite subset of $B_1^{\mathbb{Q}_p^c}(0)$. If $q, q' \in \mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T]])$ satisfy q(x) = q'(x) for all $x \in S$ (including possibly $q(x) = q'(x) = \infty$), then q = q'.

Proof. There exists a p-adic field E such that

$$q = \frac{f}{g}, \ q' = \frac{f'}{g'}$$

with $f, g, f', g' \in \mathcal{O}_E[[T]]$, $gg' \neq 0$ and both fractions irreducible in $\mathcal{O}_E[[T]]$. By remark 2.4.9, q and q' can be evaluated (potentially to ∞) at all $x \in B_1^{\mathbb{Q}_p^c}(0)$ as q(x) = f(x)/g(x) (i.e. 0/0 does not occur) and analogously for q'. Therefore, one has

$$(fg' - gf')(x) = 0 (2.27)$$

for all $x \in S$. This illustrates why $q(x) = q'(x) = \infty$ is admitted: it is tantamount to g(x) = g'(x) = 0, which makes the above equation hold all the same (it can only happen at finitely many x at any rate). By a consequence of the Weierstrass preparation theorem (see for instance [Was] corollary 7.4), the series $fg' - gf' \in \mathcal{O}_E[[T]]$ must be identically 0, as it vanishes at infinitely many points of $B_1^{\mathbb{Q}_p^c}(0)$. In other words, q = f/g = f'/g' = q'.

The following immediate consequence, which hints at the sets $\mathcal{K}_{S}^{\alpha}(\chi)$ from definition 2.3.1, will be relevant to the formulation of the Main Conjecture:

Corollary 2.4.11. Setting A. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ be an irreducible Artin character of \mathcal{G} , \mathcal{K} an infinite set of type-W characters, and $q, q' \in \mathcal{Q}^c(\Gamma_{\chi})$ two series quotients. If $ev_{\gamma_{\chi \otimes \rho}}(q) = ev_{\gamma_{\chi \otimes \rho}}(q')$ for all $\rho \in \mathcal{K}$, then q = q'.

Proof. We regard q and q' as elements of $\mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T_{\chi}]])$ via the usual identification of $\gamma_{\chi} - 1$ with T_{χ} . A character ρ of type W corresponds uniquely to the choice $\rho(\gamma_K) = \zeta$ of a p-power root of unity $\zeta \in \mu_{p^{\infty}} \subseteq \mathbb{Q}_p^c$ (cf. (2.9)). In particular, the set $S = \{\rho(\gamma_K)^{w_{\chi}} - 1 : \rho \in \mathcal{K}\} \subseteq B_1^{\mathbb{Q}_p^c}(0)$ (with w_{χ} as in (2.21)) is infinite because the homomorphism $-^{w_{\chi}} : \mu_{p^{\infty}} \to \mu_{p^{\infty}}$ has finite kernel. Since $ev_{\gamma_{\chi}\otimes\rho}$ corresponds to evaluation at $T_{\chi} = \rho(\gamma_K)^{w_{\chi}} - 1$ by lemma 2.4.5, q and q' coincide at all points of S by assumption. The result now follows from lemma 2.4.10.

2.5 The Main Conjecture

We are now in a position to formulate the equivariant Main Conjecture. As explained at the beginning of the chapter, this conjecture consists of two parts: interpolation and algebraicity. The former is of analytic nature and claims the existence of certain series quotients which interpolate regulated special values of Artin *L*-series. The latter postulates a *K*-theoretic relation between these series quotients and the refined Euler characteristic of the main complex $C_{S,T}^{\bullet}$. After stating

both conjectures, we shall discuss some immediate consequences of their formulation and introduce a refined version which incorporates a uniqueness claim.

The Interpolation Conjecture concerns the irreducible Artin characters of \mathcal{G} , which we have defined as *p*-adic in nature. Since Artin *L*-functions are constructed from complex characters a priori, we must relate the two spaces via the $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ fixed in setting B. Namely, given $\chi \in \operatorname{Irr}_p(\mathcal{G})$, the composition $\beta\chi$ is a complex character of \mathcal{G} which factors through some finite Galois extension K'of K. This gives rise to an Artin *L*-function $L_{K,S,T}(\beta\chi, s)$ for K'/K as explained in section 0.3, and one is free to enlarge K' without altering the function by lemma 0.3.1 ii) - in practice, we will always take $K = L_n$ for some large enough n. The special value of interest to us is the leading coefficient $L_{K,S,T}^*(\beta\chi, 0)$ of the power series expansion of this function at s = 0 (cf. (0.25)), which we bring back to the *p*-adic setting by considering $\beta^{-1}(L_{K,S,T}^*(\beta\chi, 0)) \in \mathbb{C}_p^*$. The resulting *p*-adic complex value is not independent of the choice of β , but it should be so when regulated - this is precisely the content of Stark's conjecture, which we explore in section 3.1. The Interpolation **Conjecture** is the following assertion about the leading coefficients:

Conjecture (IC($L_{\infty}/K, \chi, L, S, T, \alpha, \beta$)). Setting *B*. Let $\chi \in \operatorname{Irr}_{p}^{c}(\mathcal{G})$.

There exists a series quotient $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ such that

$$ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha,\beta}) = \frac{\beta^{-1}(L_{K,S,T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\chi\otimes\rho)}$$
(2.28)

for almost all $\rho \in \mathcal{K}_{S}^{\alpha}(\chi)$.

Remark 2.5.1. i) $\check{\chi} \otimes \rho^{-1}$ is the dual character $(\chi \otimes \rho)$ of $\chi \otimes \rho$.

- ii) If the Interpolation Conjecture holds, the predicted element $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ is unique by corollary 2.4.11. Note that $\rho \in \mathcal{K}_S^{\alpha}(\chi)$ contains infinitely many characters by lemma 2.3.2.
- iii) The regulator $R_S^{\beta}(\alpha, \chi \otimes \rho)$ is non-zero for $\rho \in \mathcal{K}_S^{\alpha}(\chi)$ (cf. definitions 2.3.1 and 2.3.4). Recall that, even though it relies on the implicit choice of an $n \ge n(S)$ to define the finite level maps φ_n^{α} over, it is independent of that choice by lemma 2.3.5.
- iv) If Stark's conjecture holds, the field where the coefficients of $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ lie can be narrowed down to $\mathbb{Q}_{p,\chi} = \mathbb{Q}_p(\chi(h): h \in H)$ with some principal unit adjoined to it, and the same is true if one assumes the equivariant Main Conjecture formulated below instead. Both facts are explained at the end of section 3.1.
- v) The topological generator γ_K of Γ_K is not listed as a parameter of the conjecture, since both $\mathcal{Q}^c(\Gamma_{\chi})$ and the twisted evaluation maps $ev_{\gamma_{\chi\otimes\rho}}$ are independent of its choice by lemma 2.4.7.
- vi) The specific nature of Γ_{χ} does not play a role in the Interpolation Conjecture: one could have considered $\mathcal{Q}^c(\tilde{\Gamma})^*$ for any abstract $\tilde{\Gamma} = \overline{\langle \tilde{\gamma} \rangle} \cong \mathbb{Z}_p$ and replaced the twisted evaluation maps by evaluation at some specific values of $\tilde{T} = \tilde{\gamma} - 1$ as in lemma 2.4.5. It is however convenient to expressly consider $\mathcal{Q}^c(\Gamma_{\chi})^*$ with a view towards the second part of the Main Conjecture, where

a zeta element is expected to be mapped to the $F_{S,T,\chi}^{\alpha,\beta}$ under certain maps $K_1(\mathcal{Q}(\mathcal{G})) \to \mathcal{Q}^c(\Gamma_{\chi})^*$ where the meaning of the latter becomes relevant.

vii) The left-hand side of (2.28) always lies in $\mathbb{Q}_p^c \cup \{\infty\}$ (see for instance remark 2.4.9). Therefore, if the Interpolation Conjecture holds, the regulated special values (which are *p*-adic complex by definition) must be in fact algebraic over \mathbb{Q}_p .

The interpolation property (2.28) of the predicted series quotient $F_{S,T,\chi}^{\alpha,\beta}$ is limited to ρ -twists of χ which do not factor through too low a layer, namely $n(S,\alpha)$ - and even then, only to almost all. The aim of this part of the conjecture is not to interpolate values of all *L*-functions of intermediate extensions of L_{∞}/K , but rather *enough* values that the interpolating element is uniquely determined by them, since the essence of the Main Conjecture is the fact that these elements should be algebraic in nature. Whereas the restriction to $\rho \in \mathcal{K}_{S}^{\alpha}(\chi)$ is necessary for the non-vanishing of the regulator, the reason interpolation is conjectured at almost all such ρ , rather than all, is that this will enable us to prove functoriality (that is, good behaviour under change of K or L_{∞}) in section 3.3.

By virtue of our definition of the twisted evaluation maps, the Interpolation Conjecture does not refer to a single $\chi \in \operatorname{Irr}_p(\mathcal{G})$, but rather to its entire W-equivalence class:

Proposition 2.5.2. Setting B. Let $\tilde{\rho}$ be a type-W character of \mathcal{G} and $\chi \in \operatorname{Irr}_p(\mathcal{G})$. Set $\tilde{\chi} = \chi \otimes \tilde{\rho}$. Then $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \tilde{\chi}, L, S, T, \alpha, \beta)$ does, in which case the predicted elements $F_{S,T,\chi}^{\alpha,\beta}$ and $F_{S,T,\tilde{\chi}}^{\alpha,\beta}$ coincide.

Proof. We first point out that the parameters L, S, T, α and β are independent of χ , so one can indeed formulate the two Interpolation Conjectures by changing the character alone. Recall that $Q^c(\Gamma_{\chi}) = Z(Q^c(\mathcal{G})e_{\chi}) = Q^c(\Gamma_{\tilde{\chi}})$ by propositions 2.4.3 iv) and 2.4.4 ii) - this was also shown explicitly in the proof of lemma 2.4.5.

Let $\rho \in \mathcal{K}_{S}^{\alpha}(\chi)$. The characters $\chi \otimes \rho$ and $\widetilde{\chi} \otimes (\widetilde{\rho}^{-1} \otimes \rho)$ coincide, and therefore any $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^{c}(\Gamma_{\chi})^{*}$ satisfying the interpolation property (2.28) for $\chi \otimes \rho$ will also do so for $\widetilde{\chi} \otimes (\widetilde{\rho}^{-1} \otimes \rho)$:

$$\begin{split} ev_{\gamma_{\tilde{\chi}\otimes(\tilde{\rho}^{-1}\otimes\rho)}}(F_{S,T,\chi}^{\alpha,\beta}) &= ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha,\beta}) \\ &= \frac{\beta^{-1}(L_{K,S,T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\chi\otimes\rho)} \\ &= \frac{\beta^{-1}(L_{K,S,T}^*(\beta((\tilde{\chi})\otimes(\tilde{\rho}^{-1}\otimes\rho)^{-1}),0))}{R_S^\beta(\alpha,\tilde{\chi}\otimes(\tilde{\rho}^{-1}\otimes\rho))} \end{split}$$

(cf. remark 2.4.6 i) for the first equality).

Since the map $\tilde{\rho}^{-1} \otimes -: \mathcal{K}_{S}^{\alpha}(\chi) \to \mathcal{K}_{S}^{\alpha}(\tilde{\chi})$ is a bijection by lemma 2.3.2 ii), interpolation for almost all characters in the former implies that for almost all characters in the latter. This shows one implication in the statement, with the converse following from an analogous argument or by regarding χ as $\tilde{\chi} \otimes \tilde{\rho}^{-1}$.

This proposition aligns perfectly with diagram (2.23), which shows that our algebraic objects of interest decompose into χ -parts up to W-equivalence. Indeed, the last step before the formulation of the equivariant Main Conjecture is to define maps from $K_1(\mathcal{Q}(\mathcal{G}))$ to the $\mathcal{Q}^c(\Gamma_{\chi})^*$:

Definition 2.5.3. Setting A. For $\chi \in \operatorname{Irr}_p(\mathcal{G})$, we define the group homomorphism ψ_{χ} as the composition

$$\psi_{\chi} \colon K_1(\mathcal{Q}(\mathcal{G})) \xrightarrow{\operatorname{nr}} Z(\mathcal{Q}(\mathcal{G}))^* \hookrightarrow Z(\mathcal{Q}^c(\mathcal{G}))^* \twoheadrightarrow Z(\mathcal{Q}^c(\mathcal{G})e_{\chi})^* = \mathcal{Q}^c(\Gamma_{\chi})^*.$$

where the first arrow is the reduced norm introduced in section 0.4 and the equality is proposition 2.4.4 ii).

It should be noted that, even though ring homomorphisms have poor behaviour with respect to taking centres in general, the second and third arrow defining ψ_{χ} are well defined because they come from extension of scalars (from one field to another) and multiplication by a central element, respectively. An immediate consequence of the above definition is that ψ_{χ} and $\psi_{\chi'}$ coincide whenever $\psi_{\chi} \sim_W \psi_{\chi'}$. These maps are closely related to the j_{χ} of Ritter and Weiss (cf. proposition 2.4.4 iii)), which plays a prominent role in their paper [RW04]. Namely, let us extend scalars by setting $j_{\chi}^c = \mathbb{Q}_p^c \otimes_E j_{\chi}^E$. Then j_{χ}^c and ψ_{χ} fit into the commutative diagram



Note that the map $\gamma_{\chi} \mapsto \gamma_{K}^{w_{\chi}}$ is injective and commutes with the evaluation maps (see for instance the proof of lemma 2.4.7).

We now state the **equivariant Main Conjecture**, which relates the refined Euler characteristic (2.8) to the series quotients predicted in the Interpolation Conjecture by means of the localisation sequence of K-theory (0.27):

Conjecture (eMC($L_{\infty}/K, L, S, T, \alpha, \beta$)). Setting B.

Conjecture $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in Irr_p(\mathcal{G})$ and there exists a $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ such that $\partial(\zeta_{S,T}^{\alpha,\beta}) = -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}) \in K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ and

$$\psi_{\chi}(\zeta_{S,T}^{\alpha,\beta}) = F_{S,T,\chi}^{\alpha,\beta}$$

for all χ as above, where $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ is the unique series quotient satisfying the Interpolation Conjecture at χ .

A natural refinement is the equivariant Main Conjecture with uniqueness:

Conjecture (eMC^u($L_{\infty}/K, L, S, T, \alpha, \beta$)). Setting B.

Conjecture $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in Irr_p(\mathcal{G})$ and there exists a unique element $\zeta_{ST}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ such that

$$\psi_{\chi}(\zeta_{S,T}^{\alpha,\beta}) = F_{S,T,\chi}^{\alpha,\beta}$$

for all χ as above, where $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ is the unique series quotient satisfying the Interpolation Conjecture at χ . Furthermore, one has $\partial(\zeta_{S,T}^{\alpha,\beta}) = -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}) \in K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G})).$

Remark 2.5.4. i) $eMC^{u}(L_{\infty}/K, L, S, T, \alpha, \beta)$ can be reformulated as: $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds and the reduced norm nr: $K_{1}(\mathcal{Q}(\mathcal{G})) \to Z(\mathcal{Q}(\mathcal{G}))^{*}$ is injective.

The object $SK_1(R) = \ker(\operatorname{nr}: K_1(R) \to Z(R)^*)$ for a general semisimple Artinian ring R has considerable algebraic interest in itself and has been the subject of research for the last few decades. In our case of interest, Merkurjev reports (cf. [Mer94] p. 250) that Suslin conjectured the vanishing of $SK_1(R)$ for a class of rings which $\mathcal{Q}(\mathcal{G})$ is an instance of (see [RW04] p. 565 for more details). If this is the case, the equivariant Main Conjectures with and without uniqueness are equivalent.

ii) The connecting homomorphism ∂ is known to be surjective in our setting by work of Witte (cf. [Wit13] corollary 3.8), and thus $-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T}, t^{\alpha})$ always has a preimage under ∂ .

Chapter 3

Basic properties

The construction presented in chapter 2, culminating in the formulation of the Interpolation Conjecture and equivariant Main Conjecture, gives rise to some natural questions. Arguably the most immediate one is that of the dependence on the many parameters which have made an appearance so far. In one form or another, that is the subject of most results in this chapter, which has three main aims distributed across its three sections.

The first one is to study the dependence of the conjectures on the choice of $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$. As already mentioned, this question is closely linked to Stark's conjecture on leading coefficients of Artin *L*-functions. This conjecture and its various refinements have been extensively studied by many authors - among them Tate in [Tat84], who shaped its modern formulation. It will be necessary for us to incorporate the set *T* into the conjecture, which a priori only features one set of places *S*. After showing the two variants to be equivalent, we shall prove that our Main Conjecture is independent of β if Stark's one holds, and unconditionally independent of β as long as its restriction to \mathbb{Q}_p is fixed. Another consequence of Stark's conjecture will be explored, namely where the coefficients of the series quotients predicted by the Interpolation Conjecture should be expected to lie. This provides some measure of support for, and is also implied by, the Main Conjecture.

The second aim is to establish that, once L_{∞}/K has been fixed, the Main Conjecture is unconditionally independent of the choices of all parameters other than β . Section 3.2 (as well as 3.3) has been exceptionally divided into subsections for the convenience of the reader. The following theorem encapsulates the main results in the former:

Theorem 3.0.1. Setting *B*. Let $\tilde{L}, \tilde{S}, \tilde{T}$ and $\tilde{\alpha}$ be another valid set of choices of the corresponding parameters in the setting. Then:

- i) For any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, \tilde{L}, \tilde{S}, \tilde{T}, \tilde{\alpha}, \beta)$ does.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, \tilde{L}, \tilde{S}, \tilde{T}, \tilde{\alpha}, \beta)$ does.
- iii) $eMC^{u}(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC^{u}(L_{\infty}/K, \tilde{L}, \tilde{S}, \tilde{T}, \tilde{\alpha}, \beta)$ does.

As a consequence, one is justified in writing $IC(L_{\infty}/K, \chi, \beta)$, $eMC(L_{\infty}/K, \beta)$ and $eMC^{u}(L_{\infty}/K, \beta)$ which we nonetheless avoid in almost all formal mathematical claims. The proof of the independence of L will be completely straightforward, unlike that of α , S and T. By remark 2.5.4 i), it will suffice to prove i) and ii): once the equivariant Main Conjectures are shown to be equivalent, uniqueness is a purely K-theoretic question. Note that we can only argue this way because L_{∞}/K , and hence $\Lambda(\mathcal{G})$ and $\mathcal{Q}(\mathcal{G})$, are the same for all conjectures in the theorem.

The third and last goal of this chapter is to study the functoriality of the conjectures, that is, how modifying L_{∞} and K affects their validity. In the case of a finite extension L'_{∞}/L_{∞} such that L'_{∞}/K is Galois, we will prove that all three conjectures for L'_{∞}/K imply their counterparts for L_{∞}/K . If one instead replaces K by a finite extension K' contained in L_{∞} , it is necessary to assume the Interpolation Conjecture for several characters of $\operatorname{Gal}(L_{\infty}/K)$ in order to conclude that for one of $\operatorname{Gal}(L_{\infty}/K')$. As for the equivariant Main Conjecture, only the version without uniqueness will be treated. As an immediate consequence, we deduce that the general case of those two conjectures for p odd follows from the case $K = \mathbb{Q}$. We end the chapter by addressing the converse problem: whether one can deduce the conjectures for L_{∞}/K from their counterparts for a suitable family of subextensions. Two families S and \mathcal{E}_p will be described where this is the case, consisting of the subextensions determined by certain elementary subgroups and certain p-elementary subquotients of \mathcal{G} , respectively. In both instances, the discussion will be limited to the Interpolation Conjecture and the equivariant Main Conjecture without uniqueness. For the latter, we essentially use the argument from [JN20] section 10 together with some additional verifications which are not necessary in the totally real case.

The justification for many specific features of the formulation of the Main Conjecture will become apparent in the following sections, such as the duality between the characters in the numerator and denominator of the regulated *L*-value (2.28), the *almost all* quantifier in the Interpolation Conjecture, and the element $\zeta_{S,T}^{\alpha,\beta}$ being mapped to the *inverse* of the refined Euler characteristic $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}).$

3.1 Stark's conjecture and the choice of β

Stark's conjecture, as formulated by Tate in [Tat84], asserts a certain Galois-invariance property of the quotient of the leading coefficient of an Artin L-series by a regulator constructed using the classical Dirichlet map. The Stark-Tate regulator from section 2.3 is a particular instance of such objects. The classical formulation only considers S-depleted Artin L-functions, so the purpose of this section is to introduce a T-smoothed version of it - that is, a conjecture for (S, T)-modified Artin L-functions - which will be equivalent to the original S-version; and to study its implications for the Main Conjecture.

We work in the generality in which Artin L-functions were introduced in section 0.3, which is essentially the same as in our applications save for the fact that S is not required to contain all ramified places here. Tate defines the following regulators: suppose given a Galois extension of number fields L/K with Galois group G and two finite sets of places S and T of K such that $S \supseteq S_{\infty}$ and $S \cap T = \emptyset$. Consider an embedding¹⁵ $\beta \colon \mathbb{C}_p \hookrightarrow \mathbb{C}$ and a $\mathbb{Z}[G]$ -homomorphism

$$f: \mathcal{X}_{L,S}^{\mathbb{Z}} \to \mathbb{C}_p \otimes \mathcal{O}_{L,S}^*$$

in the notation of section 0.3. These maps induce a $\mathbb{C}[G]$ homomorphism

$$f^{\beta} \colon \mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}} \to \mathbb{C} \otimes \mathcal{O}_{L,S}^{*}$$

$$(3.1)$$

by extending scalars of $\mathcal{X}_{L,S}^{\mathbb{Z}} \xrightarrow{f} \mathbb{C}_p \otimes \mathcal{O}_{L,S}^* \xrightarrow{\beta \otimes \mathrm{Id}} \mathbb{C} \otimes \mathcal{O}_{L,S}^*$. Finally, let $\lambda_{L,S}^{\mathbb{C}} \colon \mathbb{C} \otimes \mathcal{O}_{L,S}^* \to \mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}}$ be the extension of scalars of the Dirichlet regulator map (0.24) from \mathbb{R} to \mathbb{C} .

We are interested in the character-wise determinant of $\lambda_{L,S}^{\mathbb{C}} \circ f^{\beta}$. Specifically, if $\chi \in \operatorname{Char}_{\mathbb{C}_p}(G)$ and V_{χ} is a $\mathbb{C}_p[G]$ -module which affords χ , then $V_{\beta\chi} = \mathbb{C} \otimes_{\beta} V_{\chi}$ is a $\mathbb{C}[G]$ -module which affords $\beta\chi$ and one may therefore define¹⁶

$$R_{S}(\beta\chi, f^{\beta}) = \det(\lambda_{L,S}^{\mathbb{C}} \circ f^{\beta} \mid \operatorname{Hom}_{\mathbb{C}[G]}(V_{\beta\chi}, \mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}})) \in \mathbb{C}$$
(3.2)

in the notation of (2.19). In other words, $R_S(\beta\chi, f^\beta)$ is the determinant of the \mathbb{C} -linear endomorphism $(\lambda_{L,S}^{\mathbb{C}} \circ f^\beta)_*$ of $\operatorname{Hom}_{\mathbb{C}[G]}(V_{\beta\chi}, \mathbb{C} \otimes \mathcal{X}_{L,S}^{\mathbb{Z}})$ given by postcomposition with $\lambda_{L,S}^{\mathbb{C}} \circ f^\beta$.

Our T-smoothed version of Stark's conjecture is the following assertion:

Conjecture (Stark^T($L/K, \chi, f, S, T$)). Consider:

- A Galois extension of number fields L/K.
- A character $\chi \in \operatorname{Char}_{\mathbb{C}_p}(\operatorname{Gal}(L/K)).$
- Two finite sets of places S and T of K such that $S \supseteq S_{\infty}$ and $S \cap T = \emptyset$.
- A homomorphism of $\mathbb{Z}[G]$ -modules $f: \mathcal{X}_{L,S}^{\mathbb{Z}} \to \mathbb{C}_p \otimes \mathcal{O}_{L,S}^*$.

There exists an $\mathcal{L}_{S,T}(\chi, f) \in \mathbb{C}_p$ such that, for any ring homomorphism $\beta \colon \mathbb{C}_p \hookrightarrow \mathbb{C}$, one has

$$\beta(\mathcal{L}_{S,T}(\chi, f)) = \frac{R_S(\beta \check{\chi}, f^\beta)}{L^*_{K,S,T}(\beta \chi, 0)},$$
(3.3)

The original conjecture is the above claim for $T = \emptyset$. However, our *T*-smoothed version is a *reformulation*, rather than a strengthening:

Proposition 3.1.1. Stark^T($L/K, \chi, f, S, T$) holds if and only if Stark's conjecture as formulated¹⁷ in [Tat84] 5.4 does for the same set of parameters minus T.

¹⁵It is well known that $\mathbb{C}_p \cong \mathbb{C}$ as abstract fields by a transcendence-degree argument. However, this isomorphism is far from unique.

¹⁶Tate uses the notation $R(\chi^{\beta}, f^{\beta})$ for the determinant on $\check{\chi}$ -parts instead (cf. [Tat84] p.28).

¹⁷It should be pointed out that Tate formulates this conjecture for any field E which can be embedded into \mathbb{C} , whereas we specialise to the case $E = \mathbb{C}_p$. However, the choice of E is irrelevant (cf. section I§6 in the reference).

Proof. Conjecture [Tat84] 5.4 asserts the existence of some $\mathcal{L}_S(\chi, f) \in \mathbb{C}_p$ such that

$$\beta(\mathcal{L}_S(\chi, f)) = \frac{R_S(\beta \check{\chi}, f^\beta)}{L^*_{K,S,\emptyset}(\beta \chi, 0)}$$
(3.4)

for any $\beta \colon \mathbb{C}_p \hookrightarrow \mathbb{C}$. The S-depleted and (S,T)-modified L-functions differ by the δ -factors (cf. (0.18)) at T

$$L_{K,S,T}(\beta\chi,s) = L_{K,S,\varnothing}(\beta\chi,s) \cdot \prod_{v \in T} \delta_v(\beta\chi,s)$$
(3.5)

in the half-plane $\operatorname{Re}(s) > 1$. This relation extends to all of \mathbb{C} , since Artin *L*-functions are by definition the meromorphic continuation of their Euler-product expression and two meromorphic functions which coincide on an open subset of \mathbb{C} (where they are analytic) are identical. Recall that δ -factors are entire functions by (0.19).

We now show that δ -factors do not vanish at s = 0. They obey the same formalism as *L*-factors (cf. lemma 0.3.1), since $\delta_v(\beta\chi, s) = L_v(\beta\chi, 1-s)^{-1}$. Thus, Brauer's induction theorem 0.2.1 allows one to express $\delta_v(\beta\chi, s)$ as a (finite) product of factors of the form $\delta_w(\lambda, s)$, where λ is a linear *complex* character of some $U \leq \text{Gal}(L/K)$ and w is a prolongation of v to L^U . In the expression

$$\delta_w(\lambda, s) = \det(1 - \mathfrak{N}(w)^{1-s}\varphi_w \mid V_\lambda^{I_w}), \qquad (3.6)$$

 $V_{\lambda}^{I_w}$ is either trivial (if λ is not trivial on I_w), in which case the above determinant takes the value 1 at all s by definition; or a 1-dimensional \mathbb{C} -vector space (if λ is trivial on I_w) upon which φ_w acts as $\lambda(\varphi_w) \in \mathbb{C}^*$. In this latter case,

$$\delta_w(\lambda, 0) = 1 - \mathfrak{N}(w)\lambda(\varphi_w)$$

is different from 0, as $\mathfrak{N}(w) > 1$ and $\lambda(\varphi_w)$ is a complex root of unity. Since the original $\delta_v(\beta\chi, 0)$ is a product of such linear factors, it does not vanish either.

Equation (3.5) now yields the relation

$$L^*_{K,S,T}(\beta\chi,0) = L^*_{K,S,\varnothing}(\beta\chi,0) \cdot \prod_{v \in T} \delta_v(\beta\chi,0)$$

on leading coefficients. It therefore remains to show that δ -factors behave well with respect to β . For $v \in T$, consider the *p*-adic δ -factor

$$\delta_v^{\mathbb{C}_p}(\chi,0) = \det(1 - \mathfrak{N}(v)\varphi_w \mid V_\chi^{I_w}) \in \mathbb{C}_p.$$

Then

$$\beta(\delta_v^{\mathbb{C}_p}(\chi, 0)) = \det(1 - \mathfrak{N}(v)\beta(\varphi_w \mid V_\chi^{I_w})), \qquad (3.7)$$

where $\beta(\varphi_w \mid V_{\chi}^{I_w})$ denotes β applied entry-wise to the matrix describing the action of φ_w on $V_{\chi}^{I_w}$. This coincides with the matrix describing the action of φ_w on $V_{\beta\chi}^{I_w} = (\mathbb{C} \otimes_{\beta} V_{\chi})^{I_w} = \mathbb{C} \otimes_{\beta} (V_{\chi}^{I_w})$ by definition, and hence

$$\beta(\delta_v^{\mathbb{C}_p}(\chi,0)) = \det(1 - \mathfrak{N}(v)\varphi_w \mid V_{\beta\chi}^{I_w}) = \delta_v(\beta\chi,0).$$

In particular, $\delta_v^{\mathbb{C}_p}(\chi, 0) \neq 0$. Therefore, assuming the existence of $\mathcal{L}_S(\chi, f) \in \mathbb{C}_p$ as in the beginning of the proof and setting $\mathcal{L}_{S,T}(\chi, f) = \mathcal{L}_S(\chi, f) \prod_{v \in T} \delta_v^{\mathbb{C}_p}(\chi, 0)^{-1} \in \mathbb{C}_p$, one has

$$\beta(\mathcal{L}_{S,T}(\chi,f)) = \frac{R_S(\beta\check{\chi}, f^\beta)}{L_{K,S,\emptyset}^*(\beta\chi, 0)} \cdot \frac{1}{\prod_{v \in T} \beta(\delta_v^{\mathbb{C}_p}(\chi, 0))}$$
$$= \frac{R_S(\beta\check{\chi}, f^\beta)}{L_{K,S,\emptyset}^*(\beta\chi, 0) \cdot \prod_{v \in T} \delta_v(\beta\chi, 0)}$$
$$= \frac{R_S(\beta\check{\chi}, f^\beta)}{L_{K,S,T}^*(\beta\chi, 0)}$$

for any $\beta \colon \mathbb{C}_p \to \mathbb{C}$, as claimed by $\mathrm{Stark}^{\mathrm{T}}(L/K, \chi, f, S, T)$.

The converse is proved analogously by multiplying by $\prod_{v \in T} \delta_v^{\mathbb{C}_p}(\chi, 0)$ at the end, rather than dividing by it.

In view of the proposition, the known properties of Stark's conjecture carry over to the smoothed version. [Tat84] constitutes a systematic study of these - here we limit ourselves to citing a few basic results from sections I§6 and I§7:

- i) If Stark's conjecture (or its T-smoothed version) holds for all characters of $\operatorname{Gal}(K/\mathbb{Q})$ for any finite Galois extension K/\mathbb{Q} (and all S, T and f), then it holds in general. The proof relies on the invariance of the conjecture under inflation and induction.
- ii) If Stark's conjecture (or its T-smoothed version) holds for all linear characters of $\operatorname{Gal}(L/K)$ for any Galois extension L/K of number fields (and all S, T and f), then it holds in general. The proof relies on Brauer's induction theorem 0.2.1 and the behaviour of the conjecture with respect to character addition and induction.
- iii) If Stark's conjecture (or its *T*-smoothed version) holds for a choice of L/K, χ and *S* and an $f_0: \mathcal{X}_{L,S}^{\mathbb{Z}} \to \mathbb{C}_p \otimes \mathcal{O}_{L,S}^*$ such that $\mathbb{C}_p \otimes f_0: \mathbb{C}_p \otimes \mathcal{X}_{L,S}^{\mathbb{Z}} \xrightarrow{\sim} \mathbb{C}_p \otimes \mathcal{O}_{L,S}^*$ is an isomorphism, then it holds for any $f: \mathcal{X}_{L,S}^{\mathbb{Z}} \to \mathbb{C}_p \otimes \mathcal{O}_{L,S}^*$ and the same remaining parameters. Note that the introduction of \mathbb{C}_p as the coefficient field is unnecessary here one can work over \mathbb{C} instead (see also the footnote¹⁷ on page 82).
- iv) Stark's conjecture (and its *T*-smoothed version) is independent of the choice of the set $S \supseteq S_{\infty}$. The proof is similar to that of proposition 3.1.1, but one needs to treat the case where the local *L*-factors vanish (which, as we have seen, cannot occur to the δ -factors).

Most cases of Stark's conjecture are unknown. Exceptions to this are:

If L/K is a Galois extension of number fields with Galois group G and χ is a character of G such that χ(σ) ∈ Q for all σ ∈ G, then Stark's conjecture is known for all choices of S and f. This is due to Tate and Artin (cf. [Tat84] chapter II) and its proof can be reduced to the case of permutation characters, which is settled by virtue of the analytic class number formula. Under the above assumption on χ, the element L_{S,∅}(χ, f) from equation (3.3) lies in Q.

• If L/K is a finite abelian extension and K is either \mathbb{Q} or an imaginary quadratic field, then Stark's conjecture is known for all choices of χ , S and f. These cases date back to Stark.

We now study the implications of Stark's conjecture for the Interpolation Conjecture, which stem from the following fact:

Lemma 3.1.2. Setting *B*. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfy (KC) and choose $n \in \mathbb{N}$ large enough that $\Gamma^{p^n} \subseteq \operatorname{ker}(\chi)$. Define f as the composition

$$f: \mathcal{X}_{L_n,S}^{\mathbb{Z}} \hookrightarrow \mathcal{X}_{L_n,S} \xrightarrow{\varphi_n^{\alpha}} \mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^* \hookrightarrow \mathbb{C}_p \otimes \mathcal{O}_{L_n,S,T}^* = \mathbb{C}_p \otimes \mathcal{O}_{L_n,S}^*$$

with φ_n^{α} as in definition 2.2.6. If $Stark^T(L_n/K, \check{\chi}, f, S, T)$ holds, then

$$\frac{\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0))}{R_S^{\beta}(\alpha,\chi)} \in \mathbb{C}_p^*$$
(3.8)

is independent of the choice of $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ and lies in $\mathbb{Q}_p(\chi) = \mathbb{Q}_p(\chi(g) \colon g \in \mathcal{G})$. In particular, it is algebraic over \mathbb{Q}_p .

Proof. Let β be as in the statement. Since $\mathcal{X}_{L_n,S} = \mathbb{Z}_p \otimes \mathcal{X}_{L_n,S}^{\mathbb{Z}}$, an immediate computation shows that the map f^{β} from (3.1) coincides with $\mathbb{C} \otimes_{\beta} (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha})$. As for the Dirichlet regulator, we have

$$\mathbb{C} \otimes_{\beta} \lambda_{n,S}^{\beta} = \lambda_{L_n,S}^{\mathbb{C}}$$

by definition 2.3.3, where $\lambda_{L_n,S}^{\mathbb{C}}$ is again the scalar extension to \mathbb{C} of the classical map.

Let M be the matrix of the \mathbb{C}_p -linear action of $\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha})$ on $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})$ (by postcomposition) with respect to an arbitrary basis $\{f_1, \ldots, f_n\}$. Then $\{1 \otimes f_1, \ldots, 1 \otimes f_n\}$ is a \mathbb{C} -basis of

$$\mathbb{C} \otimes_{\beta} \operatorname{Hom}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(V_{\chi}, \mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n}, S}) \cong \operatorname{Hom}_{\mathbb{C}[\mathcal{G}_{n}]}(V_{\beta\chi}, \mathbb{C} \otimes \mathcal{X}_{L_{n}, S}^{\mathbb{Z}})$$

(canonically) with respect to which $\lambda_{L_n,S}^{\mathbb{C}} \circ f^{\beta} = \mathbb{C} \otimes_{\beta} (\lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_n^{\alpha}))$ acts by $\beta(M)$. Hence

$$R_S(\beta\chi, f^\beta) = \det(\beta(M)) = \beta(\det(M)) = \beta(R_S^\beta(\alpha, \chi)).$$
(3.9)

Assume $\operatorname{Stark}^{\mathrm{T}}(L_n/K, \check{\chi}, f, S, T)$ and let $\mathcal{L}_{S,T}(\check{\chi}, f) \in \mathbb{C}_p$ be the element predicted therein. Then

$$\frac{R_S^{\beta}(\alpha,\chi)}{\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0))} = \beta^{-1}\left(\frac{R_S(\beta\chi,f^{\beta})}{L_{K,S,T}^*(\beta\check{\chi},0)}\right) = \mathcal{L}_{S,T}(\check{\chi},f),$$

which is independent of the choice of β . We can then take inverses since $R_S^{\beta}(\alpha, \chi) \neq 0$ (because χ satisfies (KC)), which proves the first part of the lemma.

The field $\mathbb{Q}_p(\chi)$, which coincides with $\mathbb{Q}_p(\tilde{\chi})$, is a finite (and in fact abelian) extension of \mathbb{Q}_p . The action of the absolute Galois group $G_{\mathbb{Q}_p(\chi)}$ on \mathbb{Q}_p^c extends uniquely to a continuous action on \mathbb{C}_p (recall that every element of \mathbb{C}_p is by definition a limit of elements of \mathbb{Q}_p^c) and one thus has $G_{\mathbb{Q}_p(\chi)} \hookrightarrow \operatorname{Aut}_{\mathbb{Q}_p(\chi)}(\mathbb{C}_p)$. Let $\sigma \in G_{\mathbb{Q}_p(\chi)}$, so in particular $\sigma\chi = \chi$ and $\sigma\check{\chi} = \check{\chi}$. Then $\check{\beta} = \beta\sigma^{-1}$ is another isomorphism $\mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$, and the first part of the lemma now yields

$$\begin{split} \frac{\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0))}{R_S^\beta(\alpha,\chi)} &= \frac{\check{\beta}^{-1}(L_{K,S,T}^*(\check{\beta}\check{\chi},0))}{R_S^{\check{\beta}}(\alpha,\chi)} \\ &= \frac{\sigma(\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0)))}{R_S^{\check{\beta}}(\alpha,\chi)} \\ &= \frac{\sigma(\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0)))}{\sigma(R_S^\beta(\alpha,\sigma^{-1}\chi))} = \sigma\left(\frac{\beta^{-1}(L_{K,S,T}^*(\beta\check{\chi},0)))}{R_S^\beta(\alpha,\chi)}\right), \end{split}$$

where the third equality follows easily from (3.9) noting that $f^{\beta} = f^{\tilde{\beta}}$ (they are scalar extensions of a map φ_n^{α} with coefficients in \mathbb{Z}_p , which is fixed by $G_{\mathbb{Q}_p}$). But this shows that the regulated *L*-value lies in $(\mathbb{C}_p)^{G_{\mathbb{Q}_p}(\chi)}$, which is known to coincide $\mathbb{Q}_p(\chi)$ (see for instance [Tat67] theorem 1 on p. 176).

Requiring Stark's conjecture might seem exceedingly restrictive in that it concerns all ring homomorphisms $\beta \colon \mathbb{C}_p \hookrightarrow \mathbb{C}$, whereas only isomorphisms β are relevant to us. However, it is not difficult to see that both claims - for all embeddings and for isomorphisms only - are equivalent.

The main result of this section is the independence of the Main Conjecture of β contingent on Stark's conjecture:

Proposition 3.1.3. Setting *B*. Let $\tilde{\beta} \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$. Then:

- i) Given $\chi \in \operatorname{Irr}_p(\mathcal{G})$, assume that $\operatorname{Stark}^T(L_n/K, \check{\chi} \otimes \rho^{-1}, f, S, T)$ holds for almost all $\rho \in \mathcal{K}^{\alpha}_S(\chi)$, where n is large enough that $\Gamma^{p^n} \subseteq \operatorname{ker}(\check{\chi} \otimes \rho^{-1})$ and f is constructed as in lemma 3.1.2 for each character. Then $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \tilde{\beta})$ does.
- ii) Assume that $Stark^{T}(L_{n}/K, \chi, f, S, T)$ holds for almost all $\chi \in Irr_{p}(\mathcal{G})$, where n is large enough that $\Gamma^{p^{n}} \subseteq ker(\chi)$ and f is constructed as in lemma 3.1.2 for each character. Then the conjecture $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$ does.
- iii) Under the same assumption as in ii), the conjecture $eMC^u(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC^u(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$ does.

Proof. The only aspect of the Interpolation Conjecture which references β is the regulated *L*-value $\beta^{-1}(L^*_{K,S,T}(\beta(\check{\chi} \otimes \rho^{-1}), 0))/R^{\beta}_{S}(\alpha, \chi \otimes \rho)$. Under the hypothesis in i), it is independent of β for almost all $\rho \in \mathcal{K}^{\alpha}_{S}(\chi)$ by lemma 3.1.2. Hence, an $F^{\alpha,\beta}_{S,T,\chi} \in \mathcal{Q}^{c}(\Gamma_{\chi})^{*}$ satisfying the interpolation property (2.28) for almost all such ρ will also satisfy the corresponding property for IC $(L_{\infty}/K, \chi, L, S, T, \alpha, \tilde{\beta})$.

As for ii) and iii), the only element of the equivariant Main Conjecture (with or without uniqueness) involving β is $F_{S,T,\chi}^{\alpha,\beta}$, which has just been shown to be independent of it if Stark's conjecture holds for enough characters.

We now illustrate a way in which Stark's conjecture supports the equivariant Main Conjecture by bringing attention to an issue we have tacitly ignored so far: while the interpolating series quotient $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ has coefficients in some *p*-adic field, conjecture $\operatorname{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$ claims it is the image of a zeta element $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ - which seemingly has coefficients in \mathbb{Q}_p .

In order to understand this discrepancy, we endow $\mathcal{Q}^{c}(\mathcal{G}) = \mathbb{Q}_{p}^{c} \otimes_{\mathbb{Q}_{p}} \mathcal{Q}(\mathcal{G})$ with the natural $G_{\mathbb{Q}_{p}}$ action on the first component. It is known (cf. for instance [Nic14] equation (4)) that, given $\sigma \in G_{\mathbb{Q}_{p}}$ and an irreducible Artin character χ of \mathcal{G} , one has $\sigma(e_{\chi}) = e_{\sigma\chi}$ and $\sigma(\gamma_{\chi}) = \gamma_{\sigma\chi}$ in the notation of section 2.4. In particular,

$$\sigma(T_{\chi}) = \sigma(\gamma_{\chi} - 1) = \sigma(\gamma_{\chi}) - 1 = \gamma_{\sigma\chi} - 1 = T_{\sigma\chi}$$
(3.10)

under the usual identification. These relations imply that $G_{\mathbb{Q}_p}$ not only acts on each individual component of

$$Z(\mathcal{Q}^c(\mathcal{G})) \cong \prod_{\chi/\sim_W} \mathcal{Q}^c(\Gamma_\chi)$$

(cf. propositions 2.4.3 and 2.4.4), but it might also permute them. This translates into the following relation between interpolating series quotients associated to Galois-conjugated characters:

Proposition 3.1.4. Setting B. Let $\sigma \in G_{\mathbb{Q}_p}$ and $\chi \in \operatorname{Irr}_p(\mathcal{G})$. Suppose $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Assume furthermore that $Stark^T(L_n/K, \chi \otimes \rho^{-1}, f, S, T)$ holds for almost all $\rho \in \mathcal{K}_S^{\alpha}(\chi)$ in the notation of proposition 3.1.3 i). Then $\sigma(F_{S,T,\chi}^{\alpha,\beta}) \in \mathcal{Q}^c(\Gamma_{\sigma\chi})^*$ satisfies $IC(L_{\infty}/K, \sigma\chi, L, S, T, \alpha, \beta)$.

Proof. Let us abbreviate $F_{S,T,\chi}^{\alpha,\beta}$ to F_{χ} . We first note that, for $x \in B_1^{\mathbb{Q}_p^c}(0)$ in the notation of remark 2.4.9, one has

$$\sigma(F_{\chi})|_{T_{\sigma\chi}=x} = \sigma(F_{\chi}|_{T_{\chi}=\sigma^{-1}(x)}),$$

where σ acts on F_{χ} by regarding the latter as an element of $\mathcal{Q}^c(\mathcal{G})$ rather than an abstract series quotient. In other words, if we express F_{χ} as a quotient of power series in T_{χ} , then σ is applied to both the coefficients and the variable - which explains why the result is a series quotient in $T_{\sigma\chi}$ by (3.10). The above equality then follows from the continuity of the $G_{\mathbb{Q}_p}$ -action on any *p*-adic field and implies

$$ev_{\gamma_{\sigma\chi\otimes\rho}}(\sigma(F_{\chi})) = \sigma(F_{\chi})|_{T_{\sigma\chi}=\rho(\gamma_{K})^{w_{\sigma\chi}}-1} = \sigma\left(F_{\chi}|_{T_{\chi}=\sigma^{-1}(\rho(\gamma_{K})^{w_{\chi}}-1)}\right) = \sigma\left(ev_{\gamma_{\chi\otimes\sigma^{-1}\rho}}(F_{\chi})\right) \quad (3.11)$$

for any ρ of type W. Here we have used the fact that $w_{\chi} = w_{\sigma\chi}$, as $St(\eta) = St(\sigma\eta)$ for any \mathbb{Q}_p^c -valued character η of H.

Let \doteq denote equality for almost all $\rho \in \mathcal{K}^{\alpha}_{S}(\chi)$. We have

$$ev_{\gamma_{\sigma\chi\otimes\rho}}(\sigma(F_{\chi})) = \sigma\left(ev_{\gamma_{\chi\otimes\sigma^{-1}\rho}}(F_{\chi})\right)$$
$$\stackrel{\cdot}{=} \sigma\left(\frac{\beta^{-1}(L^*_{K,S,T}(\beta(\check{\chi}\otimes\sigma^{-1}\rho^{-1}),0))}{R^{\beta}_{S}(\alpha,\chi\otimes\sigma^{-1}\rho)}\right)$$

$$\stackrel{\cdot}{=} \sigma \left(\frac{(\beta \sigma)^{-1} (L_{K,S,T}^* ((\beta \sigma)(\check{\chi} \otimes \sigma^{-1} \rho^{-1}), 0))}{R_S^{\beta \sigma}(\alpha, \chi \otimes \sigma^{-1} \rho)} \right)$$
$$= \frac{\beta^{-1} (L_{K,S,T}^* (\beta (\sigma \check{\chi} \otimes \rho^{-1}), 0))}{\sigma (R_S^{\beta \sigma}(\alpha, \chi \otimes \sigma^{-1} \rho))}$$

by, in order: equation (3.11); the Interpolation Conjecture for F_{χ} ; lemma 3.1.2 (note that composition with σ induces a bijection $\mathcal{K}^{\alpha}_{S}(\chi) \to \mathcal{K}^{\alpha}_{S}(\sigma\chi)$); and a simple manipulation. The argument used at the end of the proof of the same lemma also yields $\sigma(R_{S}^{\beta\sigma}(\alpha,\chi\otimes\sigma^{-1}\rho)) = R_{S}^{\beta}(\alpha,\sigma\chi\otimes\rho)$. Hence,

$$ev_{\gamma\sigma\chi\otimes\rho}(\sigma(F_{\chi})) \doteq \frac{\beta^{-1}(L^*_{K,S,T}(\beta(\sigma\check{\chi}\otimes\rho^{-1}),0))}{R^{\beta}_S(\alpha,\sigma\chi\otimes\rho)}$$

and $\sigma(F_{\chi})$ satisfies IC $(L_{\infty}/K, \sigma\chi, L, S, T, \alpha, \beta)$.

This has the following immediate Galois-invariance consequence for the Main Conjecture:

Corollary 3.1.5. Setting B. Suppose $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in Irr_p(\mathcal{G})$ and $Stark^T(L_n/K, \chi, f, S, T)$ (in the notation of proposition 3.1.3) does for almost all $\chi \in Irr_p(\mathcal{G})$. Then

$$\prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta} \in \left(\prod_{\chi/\sim_W} \mathcal{Q}^c(\Gamma_\chi)^*\right)^{G_{\mathbb{Q}_p}} \subseteq \prod_{\chi/\sim_W} \mathcal{Q}^c(\Gamma_\chi)^*.$$

Proof. Given $\sigma \in G_{\mathbb{Q}_p}$, one has

$$\sigma\bigg(\prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta}\bigg) = \prod_{\chi/\sim_W} \sigma(F_{S,T,\sigma^{-1}\chi}^{\alpha,\beta}) = \prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta},$$

where the first equality reflects how σ permutes the χ -parts of $Z(\mathcal{Q}^c(\mathcal{G}))$ and the second one is proposition 3.1.4 together with uniqueness of the interpolating elements (cf. remark 2.5.1 ii)).

The Galois invariants $(\prod_{\chi/\sim_W} \mathcal{Q}^c(\Gamma_{\chi})^*)^{G_{\mathbb{Q}_p}} \cong (Z(\mathcal{Q}^c(\mathcal{G}))^*)^{G_{\mathbb{Q}_p}}$ are precisely $Z(\mathcal{Q}(\mathcal{G}))^*$ (see the proof of [RW04] theorem 8) and therefore fit into the commutative diagram

(cf. (2.23)). Both isomorphisms are canonical, and we regard them as identifications. Suppose now that $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for every $\chi \in \mathrm{Irr}_p(\mathcal{G})$ and that Stark's conjecture does

for almost all χ as in proposition 3.1.3 ii). Then corollary 3.1.5 shows that the tuple $\prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta}$ lies in fact in the image of the rightmost vertical arrow above. This supports the existence of a $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ which is mapped to it under $\prod_{\chi/\sim_W} \psi_{\chi}$ (which is precisely the analytic part of the equivariant Main Conjecture) despite the considerable difference between $\mathcal{Q}(\mathcal{G})$ and $\mathcal{Q}^c(\mathcal{G})$.

The same reasoning shows that the equivariant Main Conjecture is *weakly* independent of β . By this we mean that it is well defined on equivalence classes of isomorphisms $\mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ under the relation: $\beta \sim \tilde{\beta}$ if $\beta|_{\mathbb{Q}_p} = \tilde{\beta}|_{\mathbb{Q}_p}$.

Corollary 3.1.6. Setting B. Let $\tilde{\beta} \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ be a ring isomorphism whose restriction to \mathbb{Q}_p coincides with that of β . Then $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$ does.

Proof. The equivariant Main Conjecture for β asserts that $\prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta} \in \prod_{\chi/\sim_W} \mathcal{Q}^c(\Gamma_{\chi})^*$ lives in the image of $K_1(\mathcal{Q}(\mathcal{G}))$ under $\prod_{\chi/\sim_W} \psi_{\chi}$ and is therefore $G_{\mathbb{Q}_p}$ -invariant by diagram (3.12). In particular, (3.10) implies $F_{S,T,\chi}^{\alpha,\beta} = \sigma(F_{S,T,\sigma^{-1}\chi}^{\alpha,\beta})$ for any $\sigma \in G_{\mathbb{Q}_p}$. The composition $\sigma = \tilde{\beta}^{-1}\beta$ is an automorphism of \mathbb{C}_p which fixes \mathbb{Q}_p and thus restricts to an element of $G_{\mathbb{Q}_p}$. Hence the same argument as in (3.11) together with the relation $R_S^{\tilde{\beta}}(\alpha,\chi\otimes\rho) = \sigma(R_S^{\beta}(\alpha,\sigma^{-1}(\chi\otimes\rho)))$ (see the proof of lemma 3.1.2) yields

$$\begin{split} ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha,\beta}) &= ev_{\gamma_{\chi\otimes\rho}}(\sigma(F_{S,T,\sigma^{-1}\chi}^{\alpha,\beta})) = \sigma(ev_{\gamma_{\sigma^{-1}(\chi\otimes\rho)}}(F_{S,T,\sigma^{-1}\chi}^{\alpha,\beta})) \\ &= \sigma\left(\frac{\beta^{-1}(L_{K,S,T}^*(\beta\sigma^{-1}(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\sigma^{-1}(\chi\otimes\rho))}\right) \\ &= \frac{\tilde{\beta}^{-1}(L_{K,S,T}^*(\tilde{\beta}(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\chi\otimes\rho)} \end{split}$$

for almost all $\sigma^{-1}\rho \in \mathcal{K}_{S}^{\alpha}(\sigma^{-1}\chi)$. Since composition with σ defines a bijection $\mathcal{K}_{S}^{\alpha}(\sigma^{-1}\chi) \to \mathcal{K}_{S}^{\alpha}(\chi)$, $F_{S,T,\chi}^{\alpha,\beta}$ satisfies $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \tilde{\beta})$ as well. It follows that any $\zeta_{S,T}^{\alpha,\beta} \in K_{1}(\mathcal{Q}(\mathcal{G}))$ satisfying $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$ will also satisfy $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$, as only the analytic part of the Main Conjecture depends on β . A symmetric argument proves the converse.

- Remark 3.1.7. i) By remark 2.5.4 i), this corollary automatically implies the equivalence of the Main Conjectures with uniqueness $eMC^{u}(L_{\infty}/K, L, S, T, \alpha, \beta)$ and $eMC^{u}(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$.
- ii) It follows from the proof (essentially, from the displayed equation) that $eMC(L_{\infty}/K, L, S, T, \alpha, \tilde{\beta})$ implies a weaker version of $Stark^{T}(L_{n}/K, \chi, f, S, T)$ (in the notation of proposition 3.1.3) for almost all $\chi \in Irr_{p}(\mathcal{G})$, where one fixes a $\tilde{\beta} \colon \mathbb{C}_{p} \xrightarrow{\sim} \mathbb{C}$ and replaces "for any ring homomorphism $\beta \colon \mathbb{C}_{p} \hookrightarrow \mathbb{C}$ " by "for any isomorphism $\beta \colon \mathbb{C}_{p} \xrightarrow{\sim} \mathbb{C}$ such that $\beta|_{\mathbb{Q}_{p}} = \tilde{\beta}|_{\mathbb{Q}_{p}}$ ".

We end on a brief discussion of how Stark's conjecture and the equivariant Main Conjecture narrow down the field where the coefficients of $F_{S,T,\chi}^{\alpha,\beta}$ should be expected to lie. In order to understand this, we introduce the following notation from [Nic14]: for an Artin character χ of \mathcal{G} , set

$$\mathbb{Q}_{p,\chi} = \mathbb{Q}_p(\chi(h) : h \in H).$$

This is a finite abelian extension of \mathbb{Q}_p , as it is contained in some cyclotomic field $\mathbb{Q}_p(\zeta_n)$. Note that $\chi \sim_W \chi'$ implies $\mathbb{Q}_{p,\chi} = \mathbb{Q}_{p,\chi'}$. Given an automorphism $\sigma \in G_{\mathbb{Q}_{p,\chi}}$, one has $\operatorname{res}_H^{\mathcal{G}}\chi = \operatorname{res}_H^{\mathcal{G}}\sigma\chi$ and hence $\sigma(e_{\chi}) = e_{\sigma\chi} = e_{\chi}$. Therefore, the $G_{\mathbb{Q}_p}$ -action on $\mathcal{Q}^c(\mathcal{G})$ restricts to a $G_{\mathbb{Q}_{p,\chi}}$ -action on $\mathcal{Q}^c(\mathcal{G})e_{\chi} = \mathcal{Q}^c(\Gamma_{\chi})$. It then follows from proposition 3.1.4 that, assuming the Interpolation Conjecture for χ and Stark's for suitable characters, one has $\sigma(F_{S,T,\chi}^{\alpha,\beta}) = F_{S,T,\sigma\chi}^{\alpha,\beta} = F_{S,T,\chi}^{\alpha,\beta}$ for all $\sigma \in G_{\mathbb{Q}_{p,\chi}}$. In other words,

$$F_{S,T,\chi}^{\alpha,\beta} \in \left(\left(\mathcal{Q}^c(\Gamma_{\chi}) \right)^* \right)^{G_{\mathbb{Q}_{p,\chi}}}.$$
(3.13)

The invariants in question have been explicitly determined in [Nic14]. Namely, there is a field isomorphism

$$(\mathcal{Q}^{c}(\Gamma_{\chi}))^{G_{\mathbb{Q}_{p,\chi}}} = \mathcal{Q}^{\mathbb{Q}_{p,\chi}}(\Gamma_{\chi}'),$$

where $\Gamma'_{\chi} \cong \mathbb{Z}_p$ is a subgroup of $(\mathcal{Q}^c(\Gamma_{\chi}))^*$ which is topologically generated by $\gamma'_{\chi} = u\gamma_{\chi}$ for a certain principal unit u in some p-adic field. In particular, $F^{\alpha,\beta}_{S,T,\chi}$ is a quotient of power series in T_{χ} with coefficients in $\mathbb{Q}_{p,\chi}(u)$.

By diagram (3.12), another way to ensure $\sigma(F_{S,T,\chi}^{\alpha,\beta}) = F_{S,T,\sigma\chi}^{\alpha,\beta} = F_{S,T,\chi}^{\alpha,\beta}$ for all $\sigma \in G_{\mathbb{Q}_{p,\chi}}$, and therefore (3.13), is to assume $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$.

3.2 Independence of the choice of parameters

Unlike in the case of β , where the assumption of Stark's conjecture is necessary to show the independence of the Main Conjecture, the rest of the parameters admit unconditional proofs. This concerns L, α , S and T - the change of L_{∞} or K is treated in the next section. We divide these results into three subsections devoted to L, α , and S and T, respectively. Although the idea is to modify these parameters one at a time, a change in one of them may have implications for some of the remaining ones, which will be accounted for in each subsection.

3.2.1 The choice of L

We first prove the independence of a choice of L (or equivalently, Γ). All other parameters of $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ - aside from L_{∞}/K , which is fixed beforehand - are independent of this choice:

- $\chi \in \operatorname{Irr}_p(\mathcal{G})$ is an Artin character whose definition does not involve Γ . However, we will need to be mindful of related objects which do depend on Γ , most notably $\mathcal{K}_S^{\alpha}(\chi)$ (see the point on α below).
- S and T are defined over K in terms of ramification in L_{∞} . The set $T^p \subseteq T$, which a priori could depend on L, is not an arbitrary parameter (and it is in fact independent, as will become apparent).
- $\alpha: \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S,T}$ is \mathcal{G} -equivariant. It should be noted that, although many objects have been defined as limits of Galois modules along the cyclotomic tower L_{∞}/L , they do not truly depend on the choice of L: the same can be obtained as the inverse limit over all finite (Galois) extensions of K contained in L_{∞} instead. This applies, for instance, to $\mathcal{Y}_{S_{\infty}}$ and $E_{S,T}$.

Another relevant fact, which already made an appearance earlier, is that the $\Lambda(\Gamma)$ -torsionness of $\Lambda(\mathcal{G})$ -modules (which is required of $\operatorname{coker}(\alpha)$) is independent of the choice of Γ . This is a direct consequence of (2.6): if $\widetilde{\Gamma}$ is another such choice and M is a $\Lambda(\mathcal{G})$ -module, then

$$\mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} M = 0 \iff \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} M = 0 \iff \mathcal{Q}(\widetilde{\Gamma}) \otimes_{\Lambda(\widetilde{\Gamma})} M = 0.$$
(3.14)

Explicitly, suppose $m \in M$ satisfies $\lambda m = 0$ for some non-zero $\lambda \in \Lambda(\tilde{\Gamma})$, which is in particular central and regular in $\Lambda(\mathcal{G})$. By the cited equation, $\lambda^{-1} = l/\lambda$ in $\mathcal{Q}(\mathcal{G})$ for some $l \in \Lambda(\mathcal{G})$ and $\lambda \in \Lambda(\Gamma)$. Therefore, $\lambda m = l\lambda m = 0$.

Although this shows that α is indeed independent of the choice of L, the related $n(S, \alpha) \in \mathbb{N}$ (introduced immediately before (KC)) is not so.

• $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ is unrelated to L.

Proposition 3.2.1. Setting *B*. Let \tilde{L} be another valid choice for the parameter *L*. Then:

- i) For any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, \tilde{L}, S, T, \alpha, \beta)$ does, in which case the interpolating elements $F_{S,T,\chi}^{\alpha,\beta}$ coincide.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, \tilde{L}, S, T, \alpha, \beta)$ does, in which case the zeta elements $\zeta_{ST}^{\alpha, \beta}$ coincide.

Proof. Let us consider the Interpolation Conjecture first. The compositum $L\widetilde{L}$ is an intermediate extension of L_{∞}/L and L_{∞}/\widetilde{L} , and hence equal to L_m and $\widetilde{L}_{\widetilde{m}}$ for some $m, \widetilde{m} \in \mathbb{N}$. Therefore, it suffices to show that the conjecture is independent of the choice of a layer along the cyclotomic tower, and for this, in turn, that

$$IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta) \iff IC(L_{\infty}/K, \chi, L_1, S, T, \alpha, \beta)$$

(note that L_m is the first layer of the cyclotomic \mathbb{Z}_p -extension of L_{m-1}), from which the same assertion for \widetilde{L} and \widetilde{L}_1 will follow by analogy.

The elements involved in the definition of $\mathcal{Q}^c(\Gamma_{\chi})$ (cf. proposition 2.4.4), most notably γ_{χ} , are oblivious to Γ , as are the evaluation maps $ev_{\gamma_{\chi\otimes\rho}}: \mathcal{Q}^c(\Gamma_{\chi}) \to \mathbb{Q}_p^c \cup \{\infty\}$ (see for instance lemma 2.4.5 and use $T_{\chi} = \gamma_{\chi} - 1$). In other words, they are the same in the formulation of $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ and $\mathrm{IC}(L_{\infty}/K, \chi, L_1, S, T, \alpha, \beta)$.

Let $\mathcal{K}_{S,0}^{\alpha}(\chi)$ be the set $\mathcal{K}_{S}^{\alpha}(\chi)$ defined as in 2.3.1 for $L = L_0$, and analogously for $\mathcal{K}_{S,1}^{\alpha}(\chi)$. Then $\mathcal{K}_{S,0}^{\alpha}(\chi) \cap \mathcal{K}_{S,1}^{\alpha}(\chi)$ contains almost all type-W characters of \mathcal{G} , since each set does (it is in fact easy to see that $\mathcal{K}_{S,1}^{\alpha}(\chi) \subseteq \mathcal{K}_{S,0}^{\alpha}(\chi)$ and they very frequently coincide).

Suppose $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K,\chi,L,S,T,\alpha,\beta)$. Then, for almost all characters $\rho \in \mathcal{K}_{S,0}^{\alpha}(\chi) \cap \mathcal{K}_{S,1}^{\alpha}(\chi)$ (and therefore almost all $\rho \in \mathcal{K}_{S,1}^{\alpha}(\chi)$), one has

$$ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha,\beta}) = \frac{\beta^{-1}(L_{K,S,T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\chi\otimes\rho)},$$
(3.15)

where the numerator and the denominator are the leading coefficient and Stark-Tate regulator computed at some large enough layer L_n , which we can choose to contain L_1 by lemmas 0.3.1 ii) and 2.3.5. It follows that $F_{S,T,\chi}^{\alpha,\beta}$ satisfies $\text{IC}(L_{\infty}/K, \chi, L_1, S, T, \alpha, \beta)$, which concludes the proof of i).

In order to prove ii), we first point out that the K-groups $K_1(\mathcal{Q}(\mathcal{G}))$ and $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ are independent of the choice of Γ by definition (recall (3.14) regarding $\Lambda(\Gamma)$ -torsionness), as is the connecting homomorphism ∂ between them. The main complex $\mathcal{C}_{S,T}^{\bullet}$ was shown to be isomorphic in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$ to an inverse limit of complexes $\mathcal{B}_{L_n,S,T}^{\bullet}$ along the cyclotomic tower (cf. definition 1.5.1 and theorem 1.5.4), but it has been established above that such limits do not depend on L. As a consequence, neither does the refined Euler characteristic $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha})$.

We are done now, as the field L does not appear in (or otherwise affect) the definition 2.5.3 of $\psi_{\chi} \colon K_1(\mathcal{Q}(\mathcal{G})) \to \mathcal{Q}^c(\Gamma_{\chi})^*$, and part i) showed that the Interpolation Conjectures for L and \widetilde{L} are equivalent and the series quotients $F_{S,T,\chi}^{\alpha,\beta}$ coincide whenever they hold. Therefore, any $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ satisfying $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$ will also satisfy $\mathrm{eMC}(L_{\infty}/K, \widetilde{L}, S, T, \alpha, \beta)$ and vice versa.

3.2.2 The choice of α

The next goal is to show the independence of the Main Conjecture of the choice of $\alpha \colon \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S,T}$. Although this requires a more technical proof than that in the previous subsection, we have chosen not to split it due to how intertwined the impact of a change in α on the analytic and algebraic sides of the conjecture is. Indeed, the way to adjust the interpolating elements $F_{S,T,\chi}^{\alpha,\beta}$ to a new α' will be to construct a certain element in the group $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ and transfer it to $\mathcal{Q}^c(\Gamma_{\chi})^*$.

As in the case of L, the homomorphism α does not affect any of the other parameters of the conjectures $(\chi, S, T \text{ and } \beta)$. It does have an influence on $n(S, \alpha)$, and therefore on $\mathcal{K}_{S}^{\alpha}(\chi)$ (cf. definition 2.3.1), but this will again not be of any consequence. An object which is fundamentally linked to α is the trivialisation t^{α} ((2.7) and preceding lines), and consequently the refined Euler characteristic $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}) \in K_{0}(\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G}))$. Precisely this will inform how to prove the equivalence of the Interpolation Conjectures.

Proposition 3.2.2. Setting *B*. Let α' be another valid choice for the parameter α . Then:

- i) For any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, L, S, T, \alpha', \beta)$ does.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S, T, \alpha', \beta)$ does.

Proof. Assume that $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha', \beta)$ holds and let $F_{S,T,\chi}^{\alpha',\beta} \in \mathcal{Q}^{c}(\Gamma_{\chi})^{*}$ be the element predicted therein. It suffices to show the existence of a series quotient $f \in \mathcal{Q}^{c}(\Gamma_{\chi})^{*}$ such that, for almost all type-W characters ρ , one has

$$ev_{\gamma_{\chi\otimes\rho}}(f) = \frac{R_S^\beta(\alpha',\chi\otimes\rho)}{R_S^\beta(\alpha,\chi\otimes\rho)},\tag{3.16}$$

since then the element $F_{S,T,\chi}^{\alpha',\beta} \cdot f$ satisfies the interpolation property for $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Recall the multiplicativity of evaluation (under the logical restrictions) explained in remark 2.4.2 ii) and the fact that $\mathcal{K}_{S}^{\alpha}(\chi) \cap \mathcal{K}_{S}^{\alpha'}(\chi)$ contains almost all ρ .

For a $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$ satisfying the kernel condition (KC) for both α and α' (for instance, $\varsigma = \chi \otimes \rho$ as above) and any n such that $\Gamma^{p^n} \subseteq \ker(\varsigma)$ (so $n \ge n(S, \alpha)$), consider the commutative diagram of $\mathbb{C}_p[\mathcal{G}_n]$ -homomorphisms



(cf. definition 2.2.6). By proposition 2.2.8, all arrows become \mathbb{C}_p -isomorphisms after applying the covariant functor $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\varsigma}, -)$, where V_{ς} is any simple $\mathbb{C}_p[\mathcal{G}_n]$ -module affording ς . The regulators $R_S^{\beta}(\alpha, \varsigma)$ and $R_S^{\beta}(\alpha', \varsigma)$ are defined as the determinants of the two resulting compositions of maps on Hom-spaces. But all arrows involved coincide except for the first vertical one, and therefore

$$\frac{R_{S}^{\beta}(\alpha',\varsigma)}{R_{S}^{\beta}(\alpha,\varsigma)} = \det(\operatorname{Hom}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(V_{\varsigma},(e(\varsigma)\mathbb{C}_{p}\otimes_{\mathbb{Z}_{p}}\alpha_{\Gamma^{p^{n}}})^{-1}\circ(e(\varsigma)\mathbb{C}_{p}\otimes_{\mathbb{Z}_{p}}\alpha'_{\Gamma^{p^{n}}}))),$$
(3.17)

which denotes the determinant of the \mathbb{C}_p -linear automorphism of $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\varsigma}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n, S_{\infty}})$ (recall proposition 2.2.3) given by postcomposition with $(e(\varsigma)\mathbb{C}_p \otimes_{\mathbb{Z}_p} \alpha_{\Gamma^{p^n}})^{-1} \circ (e(\varsigma)\mathbb{C}_p \otimes_{\mathbb{Z}_p} \alpha'_{\Gamma^{p^n}})$. Since $\alpha_{\Gamma^{p^n}}$ and $\alpha'_{\Gamma^{p^n}}$ are already isomorphisms on ς -parts after extending scalars to \mathbb{Q}_p^c , we may replace \mathbb{C}_p by \mathbb{Q}_p^c and V_{ς} by V_{ς}' in the above equation, with V_{ς}' a $\mathbb{Q}_p^c[\mathcal{G}]$ -module with character ς .

In order to look for series quotients which interpolate that difference of regulators, we turn to K-theory. This will also naturally lead to the equivalence of the equivariant Main Conjectures for α and α' . We shall make use of (an intermediate result in the proof of) [Nic10] theorem 6.4 to compare the augmentation of the reduced norm of a certain element in $K_1(\mathcal{Q}(\mathcal{G}))$ to the quotient (3.17). The obstacle is that the element in question should essentially describe $\alpha^{-1} \circ \alpha'$, but α cannot be inverted on integral level. We use the following trick: let

$$g_0 = p^{\mu(\operatorname{coker}(\alpha))} \operatorname{char}(\operatorname{coker}(\alpha))$$

where the μ -invariant and the characteristic polynomial refer to the structure of $\operatorname{coker}(\alpha)$ as a $\Lambda(\Gamma)$ -module and were defined after theorem 0.1.3. Then there exists a non-zero $l \in \mathbb{N}$ such that $g = lg_0 \in \Lambda(\Gamma)$ annihilates $\operatorname{coker}(\alpha)$, or in other words, $gx \in \operatorname{im}(\alpha)$ for all $x \in E_{S,T}$. Since α is injective, such a gx has a unique pre-image $\alpha^{-1}(gx) \in \mathcal{Y}_{S_{\infty}}$. We define a_g as the composition

$$a_g \colon \mathcal{Y}_{S_{\infty}} \xrightarrow{m_g} \mathcal{Y}_{S_{\infty}} \xrightarrow{\alpha'} E_{S,T} \xrightarrow{\alpha^{-1}} \mathcal{Y}_{S_{\infty}},$$

where m_g denotes multiplication by g and the last arrow is dashed to emphasise the fact that it does not exist in isolation, but it makes sense as part of the entire composition (note that α' commutes with multiplication by g). The map a_g is a homomorphism of $\Lambda(\mathcal{G})$ -modules because g is central in $\Lambda(\mathcal{G})$. The idea here is that multiplication by g allows us to construct an integral homomorphism containing information about the inverse of α , and the perturbations thus introduced into $K_1(\mathcal{Q}(\mathcal{G}))$ and $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\varsigma}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n, S_\infty})$ cancel out.

For the remainder of this proof, let -Q denote $Q(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} -$. Then $(a_g)_Q$ is a $Q(\mathcal{G})$ -automorphism of $(\mathcal{Y}_{S_{\infty}})_Q$ which arises as the (now honest) composition

$$(a_g)_{\mathcal{Q}} = (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}} \circ (m_g)_{\mathcal{Q}}, \qquad (3.18)$$

with the inverse of $\alpha_{\mathcal{Q}}$ mapping $\lambda^{-1}e$ to $(\lambda g)^{-1}\alpha^{-1}(ge)$ for $\lambda \in \Lambda(\mathcal{G}), e \in E_{S,T}$. Both $(\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}$ and $(m_g)_{\mathcal{Q}}$ are $\mathcal{Q}(\mathcal{G})$ -automorphisms of $(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}$.

We now turn our attention to finite level again. With ς and n as above, let \overline{x} denote the class of $x \in M$ in $M_{\Gamma^{p^n}}$ for any $\Lambda(\mathcal{G})$ -module M. Then $(m_g)_{\Gamma^{p^n}} = m_{\overline{g}}$, i.e. multiplication by $\overline{g} \in \Lambda(\mathcal{G}_n)$. The map $\alpha_{\Gamma^{p^n}}$ is no longer injective in general $(-_{\Gamma^{p^n}}$ is only right-exact), so one cannot describe $(a_g)_{\Gamma^{p^n}}$ as a composition of three maps. This is solved by passing to ς -parts, where everything is an isomorphism. Namely, let $\overline{y} \in \mathcal{Y}_{L_n, S_\infty} = (\mathcal{Y}_{S_\infty})_{\Gamma^{p^n}}$ and choose any lift $y \in \mathcal{Y}_{S_\infty}$. Then

$$\alpha_{\Gamma^{p^n}}\left(\overline{a_g(y)}\right) = \overline{\alpha(a_g(y))} = \overline{\alpha'(m_g(y))} = \alpha'_{\Gamma^{p^n}}(m_{\overline{g}}(\overline{y}))$$

and hence

$$\alpha_{\Gamma^{p^n}}\big(e(\varsigma)(\mathbb{Q}_p^c\otimes_{\mathbb{Z}_p}(a_g)_{\Gamma^{p^n}})(\overline{y})\big)=e(\varsigma)(\mathbb{Q}_p^c\otimes_{\mathbb{Z}_p}\alpha'_{\Gamma^{p^n}})(m_{\overline{g}}(\overline{y})).$$

We thus obtain a finite-level decomposition

$$e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} (a_g)_{\Gamma^{p^n}} = \left(e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \alpha_{\Gamma^{p^n}}\right)^{-1} \circ \left(e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \alpha'_{\Gamma^{p^n}}\right) \circ \left(e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} m_{\overline{g}}\right), \tag{3.19}$$

where the first map on the right-hand side is well defined because $\alpha_{\Gamma p^n}$ is an isomorphism on ς -parts (this follows from proposition 2.2.8 and (KC)).

In order to see that $m_{\overline{g}}$ also induces an isomorphism on ς -parts, consider the exact sequence of $\Lambda(\mathcal{G}_n)$ -modules

$$\operatorname{coker}(m_g)^{\Gamma^{p^n}} \to \mathcal{Y}_{L_n, S_{\infty}} \xrightarrow{m_{\overline{g}}} \mathcal{Y}_{L_n, S_{\infty}} \to \operatorname{coker}(m_g)_{\Gamma^{p^n}} \to 0, \qquad (3.20)$$

where we have simply dropped the first two terms of the invariants-coinvariants sequence (0.9)associated to the injective homomorphism m_g . It suffices to show that $e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \operatorname{coker}(m_g)_{\Gamma^{p^n}}$ and $e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \operatorname{coker}(m_g)^{\Gamma^{p^n}}$ are trivial (note that the latter surjects onto the kernel of $m_{\overline{g}}$ on ς -parts by the above sequence). Since $\mathcal{Y}_{S_{\infty}}$ is a free $\Lambda(\Gamma)$ -module of rank $|S_{\infty}(L)|$, it is enough in turn to prove $e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} (\Lambda(\Gamma)/\langle g \rangle)_{\Gamma^{p^n}} = e(\varsigma)\mathbb{Q}_p \otimes_{\mathbb{Z}_p} (\Lambda(\Gamma)/\langle g \rangle)^{\Gamma^{p^n}} = 0$. But this follows from (2.18) and lemma 2.2.1, as $\operatorname{supp}(\Lambda(\Gamma)/\langle g \rangle)$ and $\operatorname{supp}(\operatorname{coker}(\alpha))$ coincide by definition of g except possibly for the prime $\langle p \rangle$ (due to the p-power l introduced into g).

Now that the relevant properties of $(a_g)_{\Gamma^{p^n}}$ and $m_{\overline{g}}$ have been determined, a connection must be established to infinite level. The heavy lifting is done by equation (8) from [Nic10]. In our the last diagram in the proof of lemma 2.4.7), implies

notation, it states that, given a finitely generated projective $\Lambda(\mathcal{G})$ -module¹⁸ P and an endomorphism $f \in \operatorname{End}_{\Lambda(\mathcal{G})}(P)$ with torsion kernel and cokernel, one has

$$\det(\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\varsigma}', \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} f_{\Gamma^{p^n}})) = ev_{\gamma_K}(j_{\varsigma}(\operatorname{nr}([P_{\mathcal{Q}}, f_{\mathcal{Q}}]))),$$
(3.21)

where $[P_{\mathcal{Q}}, f_{\mathcal{Q}}] \in K_1^{\text{det}}(\mathcal{Q}(\mathcal{G}))$ and the reduced norm nr are as in section 0.4 and j_{ς} was introduced in proposition 2.4.4 (extending scalars from E to \mathbb{Q}_p^c now). Recall that we identify $K_1^{\text{det}}(\mathcal{Q}(\mathcal{G}))$ with $K_1(\mathcal{Q}(\mathcal{G}))$. The left-hand side of the above equation denotes, as usual, the determinant of the \mathbb{Q}_p^c linear endomorphism of $\text{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V'_{\varsigma}, \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} P_{\Gamma^{p^n}})$ induced by postcomposition with $\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} f_{\Gamma^{p^n}}$. Diagram (2.29), together with the fact that $\gamma_{\varsigma} \mapsto \gamma_K^{w_{\varsigma}}$ commutes with evaluation (see for instance

$$ev_{\gamma_K}(j_{\varsigma}(\operatorname{nr}([P_{\mathcal{Q}}, f_{\mathcal{Q}}]))) = ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([P_{\mathcal{Q}}, f_{\mathcal{Q}}])).$$
(3.22)

Consider the two choices $f = a_g$ and $f = m_g$, which are indeed endomorphisms of the finitely generated projective $\Lambda(\mathcal{G})$ -module $\mathcal{Y}_{S_{\infty}}$ with torsion kernel and cokernel. Equation (3.21) becomes

$$\det(\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\varsigma}', \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} (a_g)_{\Gamma^{p^n}})) = ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (a_g)_{\mathcal{Q}}]))$$

and

$$\det(\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\varsigma}',\mathbb{Q}_p^c\otimes_{\mathbb{Z}_p}m_{\overline{g}}))=ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}},(m_g)_{\mathcal{Q}}])).$$

We now rewrite both sides of the first equation of the two using decompositions (3.18) and (3.19), as well as the multiplicativity of determinants and the relations of $K_1^{\text{det}}(\mathcal{Q}(\mathcal{G}))$:

$$\det(\operatorname{Hom}_{\mathbb{Q}_{p}^{c}[\mathcal{G}_{n}]}(V_{\varsigma}',(e(\varsigma)\mathbb{Q}_{p}^{c}\otimes_{\mathbb{Z}_{p}}\alpha_{\Gamma^{p^{n}}})^{-1}\circ(e(\varsigma)\mathbb{Q}_{p}^{c}\otimes_{\mathbb{Z}_{p}}\alpha_{\Gamma^{p^{n}}}')))\cdot\det(\operatorname{Hom}_{\mathbb{Q}_{p}^{c}[\mathcal{G}_{n}]}(V_{\varsigma}',\mathbb{Q}_{p}^{c}\otimes_{\mathbb{Z}_{p}}m_{\overline{g}}))$$
$$=ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}},(\alpha_{\mathcal{Q}})^{-1}\circ\alpha_{\mathcal{Q}}']))\cdot ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}},(m_{g})_{\mathcal{Q}}])).$$

The second factors cancel out by the second of the two equations above, since $e(\varsigma)\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} m_{\overline{g}}$ has already been shown to be an isomorphism. This results in the fundamental relation

$$\det(\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\varsigma}',(e(\varsigma)\mathbb{Q}_p^c\otimes_{\mathbb{Z}_p}\alpha_{\Gamma^{p^n}})^{-1}\circ(e(\varsigma)\mathbb{Q}_p^c\otimes_{\mathbb{Z}_p}\alpha'_{\Gamma^{p^n}})))=ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}},(\alpha_{\mathcal{Q}})^{-1}\circ\alpha'_{\mathcal{Q}}])),$$

where the left-hand side is precisely the quotient $R_S^{\beta}(\alpha',\varsigma)/R_S^{\beta}(\alpha,\varsigma)$ by equation (3.17) (after restring scalars to \mathbb{Q}_p^c as explained immediately after it).

The upshot from the above argument is that the element $[(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}] \in K_1^{\text{det}}(\mathcal{Q}(\mathcal{G}))$, which is equivariant (i.e. independent of the choice of a particular character), has the following property: for every $\varsigma \in \text{Irr}_p(\mathcal{G})$ satisfying (KC), one has

$$ev_{\gamma_{\varsigma}}(\psi_{\varsigma}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}])) = \frac{R_{S}^{\beta}(\alpha', \varsigma)}{R_{S}^{\beta}(\alpha, \varsigma)}.$$
(3.23)

¹⁸Technically, the equation in the article only refers to the specific case where $P = \Lambda(\mathcal{G})$ and the endomorphism f is given by right multiplication by an $\alpha \in \Lambda(\mathcal{G})$. However, as the reference states, the same relation holds when those objects are replaced by general P and f as above by essentially the same proof - only with more involved indexing.

Let now $\chi \in \operatorname{Irr}_p(\mathcal{G})$ be as in the beginning of the proof. Since $\psi_{\chi \otimes \rho} = \psi_{\chi}$ for all ρ of type W, the element $f = \psi_{\chi}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}]) \in \mathcal{Q}^c(\Gamma_{\chi})$ satisfies (3.16), from which the equivalence of the Interpolation Conjectures follows.

Most of the work required to prove part ii) has already been done. Suppose $\zeta_{S,T}^{\alpha',\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ satisfies $eMC(L_{\infty}/K, L, S, T, \alpha', \beta)$ and set

$$\zeta_{S,T}^{\alpha,\beta} = \zeta_{S,T}^{\alpha',\beta} \cdot [(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}].$$

Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$. By assumption, $F_{S,T,\chi}^{\alpha',\beta} = \psi_{\chi}(\zeta_{S,T}^{\alpha',\beta})$ satisfies $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha', \beta)$, which implies that

$$\psi_{\chi}(\zeta_{S,T}^{\alpha,\beta}) = F_{S,T,\chi}^{\alpha',\beta} \cdot \psi_{\chi}([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}])$$
(3.24)

satisfies IC $(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ by part i).

It remains to study the algebraic side of the conjecture. For this we resort to [BB05], which shows that $(\alpha_Q)^{-1} \circ \alpha'_Q$ measures exactly the difference between the refined Euler characteristics defined via α and α' . As mentioned in section 2.1, the cited article features two versions of the refined Euler characteristic: one denoted by χ^{old} , which coincides with our $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}$; and one denoted by χ , which we write here as $\chi^{\text{new}}_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}$. Choosing a strictly perfect representative P^{\bullet} of $\mathcal{C}^{\bullet}_{S,T}$, one has

$$\partial([H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})_{\mathcal{Q}}, (t^{\alpha})^{-1} \circ t^{\alpha'}]) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}(\mathcal{C}_{S,T}^{\bullet}, (t^{\alpha})^{-1}) - \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}(\mathcal{C}_{S,T}^{\bullet}, (t^{\alpha'})^{-1}) = \left(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}) - \partial([B^{\text{odd}}(P^{\bullet})_{\mathcal{Q}}, -\text{Id}])\right) - \left(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha'}) - \partial([B^{\text{odd}}(P^{\bullet})_{\mathcal{Q}}, -\text{Id}])\right) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha'}) - \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}),$$

where the first two equalities are [BB05] proposition 5.6 (2) and theorem 6.2, respectively, and H^{odd} and B^{odd} are as in section 2.1. As can be seen in the definition of the integral trivialisation $t_{\iota}^{\alpha} = \alpha \varphi \pi$ (equation (2.7)), the difference between $H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})$ and $\mathcal{Y}_{S_{\infty}}$ is independent of α . Since $t^{\alpha} = (t_{\iota}^{\alpha})_{\mathcal{Q}}$ by definition, a simple computation using the group law of $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ yields

$$\partial([H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})_{\mathcal{Q}}, (t^{\alpha})^{-1} \circ t^{\alpha'}]) = [H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})_{\mathcal{Q}}, (t^{\alpha})^{-1} \circ t^{\alpha'}, H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})_{\mathcal{Q}}] = [(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}, (\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}] = \partial([(\mathcal{Y}_{S_{\infty}})_{\mathcal{Q}}, (\alpha_{\mathcal{Q}})^{-1} \circ \alpha'_{\mathcal{Q}}]),$$

where the first and last equalities are (0.28). It follows that $\zeta_{S,T}^{\alpha,\beta}$ satisfies

$$\begin{aligned} \partial(\zeta_{S,T}^{\alpha,\beta}) &= \partial(\zeta_{S,T}^{\alpha'}) + \partial([H^{\text{odd}}(\mathcal{C}_{S,T}^{\bullet})_{\mathcal{Q}}, (t^{\alpha})^{-1} \circ t^{\alpha'}]) \\ &= \left(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha'})\right) + \left(\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha'}) - \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha})\right) \\ &= -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}),\end{aligned}$$

and therefore also $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ by (3.24).
3.2.3 The choice of S and T

When dealing with L-functions, groups of units and other objects of arithmetical nature, it is often very convenient to have some flexibility in the choice of the sets of places S and T. By way of example, two common techniques are to enlarge S enough that the S-class group of some number field vanishes; and to enlarge T enough that the (S,T)-units become torsion-free as an abelian group, whether on finite or infinite level. In this subsection, we show one is free to modify these sets without affecting the validity of the Main Conjecture.

We will treat the case of S first, although most of the preparatory work will overlap with that for T. In both instances, the strategy is to enlarge the set in question one place at a time and construct an analogue of the corresponding Euler factor in $K_1(\mathcal{Q}(\mathcal{G}))$. In terms of distinguishing features, changing S requires studying the regulator, which is not affected by T; whereas modifications in T will entail changes in α (if small), unlike those in S. This is in fact the content of the first result:

Lemma 3.2.3. Setting A. Let $v_0 \notin S \cup T$ be a place of K and set $S' = S \cup \{v_0\}$. Then, the canonical embeddings and surjections of $\Lambda(\mathcal{G})$ -modules

$$E_S \hookrightarrow E_{S'}, \quad E_{S,T} \hookrightarrow E_{S',T}, \quad X_S^{cs} \twoheadrightarrow X_{S'}^{cs} \quad and \quad X_{T,S}^{cs} \twoheadrightarrow X_{T,S'}^{cs}$$

in the notation of section 1.4 (see especially proposition 1.4.4) are all equalities.

Proof. It is enough to prove the claim for the second and fourth arrows, since the rest correspond to the particular case $T = \emptyset$. We start by considering the exact sequence of \mathcal{G}_n -modules

$$1 \to \mathcal{O}_{L_n,S,T}^* \to \mathcal{O}_{L_n,S',T}^* \to \bigoplus_{w_n \in \{v_0\}(L_n)} \mathbb{Z} \cdot w_n \to Cl_{L_n,S,T} \to Cl_{L_n,S',T} \to 1,$$

where $Cl_{L_n,S,T}$ denotes the (S,T)-ray class group of L_n (see the proof of proposition 1.4.4), and analogously for $Cl_{L_n,S',T}$. Here the third arrow is given by the valuation and the next one sends $z \cdot w_n$ to the class $[w_n]^z \in Cl_{L_n,S,T}$. By the same argument which allowed us to deduce (1.13) from (1.12), the inverse limit of the previous sequence with respect to the norm maps along the cyclotomic tower L_{∞}/L results in the exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to E_{S,T} \to E_{S',T} \to \varprojlim_{n} \bigoplus_{w_n \in \{v_0\}(L_n)} \mathbb{Z}_p \cdot w_n \to X_{T,S}^{cs} \to X_{T,S'}^{cs} \to 0.$$
(3.25)

We therefore need to show the vanishing of the middle term $\varprojlim_n \bigoplus_{w_n \in \{v_0\}(L_n)} \mathbb{Z}_p \cdot w_n$, which coincides with $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}} \varprojlim_n \mathbb{Z}_p \cdot v(L_n)$ (in the notation of setting A) by the commutativity of inverse limits with induction (cf. (1.1)). Hence, it suffices to consider $\varprojlim_n \mathbb{Z}_p \cdot v(L_n)$. The transition maps, induced by the norm on units, are given by $v(L_{n+1}) \mapsto f_{n+1|n} \cdot v(L_n)$ (see for instance [Neu99] proposition III.1.2 (iv)), where $f_{n+1|n}$ is the inertia degree of $v(L_{n+1})$ in L_{n+1}/L_n . But the properties of the cyclotomic \mathbb{Z}_p -extension imply that, for a high enough layer L_{n_0} , $v(L_{n_0})$ is inert (i.e. unramified and non-split) in L_{∞}/L_{n_0} , as $v_0 \in S_f \setminus S_p$. Thus, the transition maps become multiplication by p from the layer L_{n_0} onwards, from which $\varprojlim_n \mathbb{Z}_p \cdot v(L_n) = 0$ follows.

Remark 3.2.4. Even though we always assume S to contain all ramified places in L_{∞}/K , all this proof requires is $v_0 \in S_f \setminus S_p$. In other words, the objects in the statement all coincide with their counterparts defined on the set $\Sigma = S_{\infty} \cup S_p$, which holds a distinguished place in Iwasawa theory. Another path to the same conclusion is the realisation that it suffices to consider the $\Lambda(\Gamma)$ -structure of those modules (i.e. the case $L_{\infty} = K_{\infty}, L = K$), and the ramified places in the cyclotomic \mathbb{Z}_p -extension are precisely the *p*-adic ones.

Let us explore how the parameters of the Main Conjecture are affected by the addition of a place $v_0 \notin S \cup T$ to S:

- L_{∞}/K and L are defined before introducing S and T. The condition $S \supseteq S_{\infty} \cup S_{\text{ram}}(L_{\infty}/K)$ and $S \cap T = \emptyset$ automatically implies $S \cup \{v_0\} \supseteq S_{\infty} \cup S_{\text{ram}}(L_{\infty}/K)$ and $(S \cup \{v_0\}) \cap T = \emptyset$.
- By the previous lemma, the canonical embedding $E_{S,T} \hookrightarrow E_{S \cup \{v_0\},T}$ is in fact an equality. Therefore, the choice of a $\Lambda(\mathcal{G})$ -homomorphism $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ is exactly the same as the choice of an $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S \cup \{v_0\},T}$.
- The choice of $\chi \in \operatorname{Irr}_p(\mathcal{G})$ is independent of S. Note that $n(S \cup \{v_0\}) \ge n(S)$, and therefore $n(S \cup \{v_0\}, \alpha) \ge n(S, \alpha)$ if the same α is chosen as explained in the previous point. This means $\mathcal{K}^{\alpha}_{S \cup \{v_0\}}(\chi) \subseteq \mathcal{K}^{\alpha}_{S}(\chi)$ in general, but the difference only consists of finitely many characters by lemma 2.3.2.
- $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ is unrelated to S.

The question of how the Main Conjecture varies under enlargement of S must be addressed on both the analytic side, concerning the existence of series quotients interpolating regulated special L-values; and the algebraic one, in terms of the refined Euler characteristic of the arithmetic complex $\mathcal{C}_{ST}^{\bullet}$. We start with the latter.

The key homological result in our study of refined Euler characteristics is their additivity in exact triangles of perfect complexes with compatible trivialisations. We are thus tasked with finding an exact triangle measuring the difference between $C_{S,T}^{\bullet}$ and $C_{S\cup\{v_0\},T}^{\bullet}$ for $v_0 \notin S \cup T$. The proof will involve answering the analogous question for $C_{S,T}^{\bullet}$ and $C_{S,T\cup\{v_0\}}^{\bullet}$. We note that, although the argument below relies on the explicit construction of the complex from chapter 1, it is likely an alternative (and potentially simpler) proof exists using the $R\Gamma$ -formulation of section 1.5.

Proposition 3.2.5. Setting A. Let $v_0 \notin S \cup T$ be a place of K. Then there exist exact triangles¹⁹

$$\mathcal{C}^{\bullet}_{S,T} \to \mathcal{C}^{\bullet}_{S \cup \{v_0\},T} \to \operatorname{Ind}^{\mathcal{G}}_{\mathcal{G}_{v_0}} \mathbb{Z}_p[-1] \to$$
(3.26)

and

$$\mathcal{C}^{\bullet}_{S,T\cup\{v_0\}} \to \mathcal{C}^{\bullet}_{S,T} \to \overline{\mathrm{Ind}}^{\mathcal{G}}_{\mathcal{G}_{v_0}} \mathbb{Z}_p(1)[0] \to$$
(3.27)

¹⁹The second triangle may seem obvious in light of definition 1.5.1 and theorem 1.5.4 - but this theorem was proved precisely on the promise that we would show such a relation later on.

in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$, where

$$\overline{\mathrm{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1) = \begin{cases} \mathrm{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1), & L_{v_0(L)} \text{ contains a primitive p-th root of unity} \\ 0, & \text{otherwise.} \end{cases}$$

Proof. Let us denote $S \cup \{v_0\}$ and $T \cup \{v_0\}$ by S' and T', respectively. We will make use of the fact that $S \cup T' = S' \cup T$, and in particular, $\mathcal{T}^{\bullet}_{S \cup T'} = \mathcal{T}^{\bullet}_{S' \cup T}$. There are no obvious non-trivial morphisms between $\mathcal{C}^{\bullet}_{S,T}$ and $\mathcal{C}^{\bullet}_{S',T}$ on the level of complexes. On the global side, the natural map goes in the direction $\mathcal{T}^{\bullet}_{S' \cup T} \to \mathcal{T}^{\bullet}_{S \cup T}$ (this induces, for instance, the projection $X_{S' \cup T} \to X_{S \cup T}$ on H^0). In the case of the \mathcal{C}^{\bullet} -complexes, however, the canonical maps on cohomology (cf. theorem 1.4.6) are reversed - most notably $\mathcal{X}_S \hookrightarrow \mathcal{X}_{S'}$.

We solve this issue by considering the auxiliary complex

$$\mathcal{C}^{\bullet}_{S,T'} = \operatorname{Cone}(\alpha_{S,T'} \colon \mathcal{L}^{\bullet}_{S} \to \mathcal{T}^{\bullet}_{S \cup T'})[-1]$$

in the notation of section 1.4. Recall that $\alpha_{S,T'}$ is induced by $G_{K_v} = (G_K)_{v^c} \hookrightarrow G_K \twoheadrightarrow G_{S\cup T}$ in degree 0 (equation (1.7)) and given by the natural map $\left(\bigoplus_{v \in S} \Lambda(\mathcal{G})\right) \twoheadrightarrow \Lambda(\mathcal{G})$ in degree 1. Our aim is to define *morphisms of morphisms of complexes* from $\alpha_{S,T'}$ to $\alpha_{S,T}$ and $\alpha_{S',T}$ - that is, four morphisms of complexes $\varepsilon_{\mathcal{L}}, \varepsilon_{\mathcal{T}}, \mu_{\mathcal{L}}$ and $\mu_{\mathcal{T}}$ making the two diagrams

commute. Set $\varepsilon_{\mathcal{L}} = \mu_{\mathcal{T}} = \text{Id.}$ As $\mu_{\mathcal{L}}$, we choose the natural morphism

$$\mu_{\mathcal{L}} \colon \mathcal{L}_{S}^{\bullet} = \bigoplus_{v \in S} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathcal{L}_{v}^{\bullet} \to \bigoplus_{v \in S'} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathcal{L}_{v}^{\bullet} = \mathcal{L}_{S'}^{\bullet}.$$

In the case of $\varepsilon_{\mathcal{T}}$, we specify the morphism on each degree, of which there are two non-trivial ones. In degree 0, we let $\varepsilon_{\mathcal{T}}^0: Y_{S'\cup T} \twoheadrightarrow Y_{S\cup T}$ be the canonical surjection induced by $G_{S'\cup T} \twoheadrightarrow G_{S\cup T}$ (cf. section 1.1). As for degree 1, take $\varepsilon_{\mathcal{T}}^1 = \mathrm{Id}: \Lambda(\mathcal{G}) \to \Lambda(\mathcal{G})$.

A number of verifications are in order, namely whether those maps are indeed morphisms of complexes and whether (3.28) commutes. In the case of ε , this follows from the commutativity of



The only two squares which deserve mention are the top and left ones, which hinge on the compatibility of Galois restriction $G_K \twoheadrightarrow G_{S \cup T'} \twoheadrightarrow G_{S \cup T} \twoheadrightarrow \mathcal{G}$. A similar diagram proves the analogous claims for $\mu_{\mathcal{L}}$ and $\mu_{\mathcal{T}}$. Although immediate, these checks should not be neglected - for instance, commutativity fails for the very natural choice $\nu_{\mathcal{L}} \colon \mathcal{L}_{S'}^{\bullet} \to \mathcal{L}_{S}^{\bullet}$ and $\nu_{\mathcal{T}} = \varepsilon_{\mathcal{T}}$.

A morphism of morphisms of complexes induces a morphism between their cones, which in our case implies the existence of maps

$$\varepsilon_{\mathcal{C}}$$
: Cone $(\alpha_{S,T'}) \to$ Cone $(\alpha_{S,T})$ and $\mu_{\mathcal{C}}$: Cone $(\alpha_{S,T'}) \to$ Cone $(\alpha_{S',T})$.

Our interest now lies in the cones of these new maps. Since the horizontal arrows in (3.28) are more complicated than the vertical ones, we resort to the following property of triangulated categories: the four cones resulting from a commutative square of morphisms fit into a nine-term diagram with two new exact triangles connecting them. In the case of ε , this translates into

where all arrows exist on level of complexes except possibly ι . The map $\varepsilon_{\mathcal{L}} = \text{Id}$ has trivial cone, and hence ι : $\text{Cone}(\varepsilon_{\mathcal{T}}) \xrightarrow{\sim} \text{Cone}(\varepsilon_{\mathcal{C}})$ is an isomorphism in $\mathcal{D}(\Lambda(\mathcal{G}))$. Consider the cohomology of $\text{Cone}(\varepsilon_{\mathcal{T}})$, which fits into the exact sequence

$$0 \longrightarrow H^{-1}(\operatorname{Cone}(\varepsilon_{\mathcal{T}})) \longrightarrow H^{0}(\mathcal{T}^{\bullet}_{S \cup T'}) \xrightarrow{H^{0}(\varepsilon_{\mathcal{T}})} H^{0}(\mathcal{T}^{\bullet}_{S \cup T}) \longrightarrow H^{0}(\operatorname{Cone}(\varepsilon_{\mathcal{T}})) \longrightarrow H^{0}(\operatorname{Cone}(\varepsilon_{\mathcal{T}})) \longrightarrow H^{1}(\mathcal{T}^{\bullet}_{S \cup T'}) \xrightarrow{H^{1}(\varepsilon_{\mathcal{T}})} H^{1}(\mathcal{T}^{\bullet}_{S \cup T}) \longrightarrow H^{1}(\operatorname{Cone}(\varepsilon_{\mathcal{T}})) \longrightarrow 0$$

and is trivial elsewhere because $\mathcal{T}_{S\cup T'}^{\bullet}$ and $\mathcal{T}_{S\cup T}^{\bullet}$ are acyclic outside degrees 0 and 1 (proposition 1.1.1). The maps induced by $\varepsilon_{\mathcal{T}}$ on cohomology are the canonical projection $H^0(\varepsilon_{\mathcal{T}}): X_{S\cup T'} \twoheadrightarrow X_{S\cup T}$ and the identity $H^1(\varepsilon_{\mathcal{T}}): \mathbb{Z}_p \xrightarrow{\mathrm{Id}} \mathbb{Z}_p$. Thus, $H^0(\mathrm{Cone}(\varepsilon_{\mathcal{T}})) = H^1(\mathrm{Cone}(\varepsilon_{\mathcal{T}})) = 0$ and $H^{-1}(\mathrm{Cone}(\varepsilon_{\mathcal{T}}))$ is isomorphic to ker $(X_{S\cup T'} \twoheadrightarrow X_{S\cup T})$. This kernel already appeared in the proof of proposition 1.4.4 as the term ker (π) , where it was shown to be precisely $\overline{\mathrm{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1)$ in our newly introduced notation. It follows that $\mathrm{Cone}(\varepsilon_{\mathcal{C}})$ itself has cohomology concentrated in at most one degree, which in turn implies

$$\operatorname{Cone}(\varepsilon_{\mathcal{C}}) \cong \overline{\operatorname{Ind}}_{\mathcal{G}_{\nu_0}}^{\mathcal{G}} \mathbb{Z}_p(1)[1]$$
(3.30)

in $\mathcal{D}(\Lambda(\mathcal{G}))$. Shifting the triangle $\operatorname{Cone}(\alpha_{S,T'}) \to \operatorname{Cone}(\alpha_{S,T}) \to \operatorname{Cone}(\varepsilon_{\mathcal{C}}) \to \operatorname{by} -1$ yields (3.27). We use the same technique to compute $\operatorname{Cone}(\mu_{\mathcal{C}})$. An analogous 9-term diagram shows

$$\operatorname{Cone}(\mu_{\mathcal{C}}) \cong \operatorname{Cone}(\operatorname{Cone}(\mu_{\mathcal{L}}) \to \operatorname{Cone}(\mu_{\mathcal{T}})) \cong \operatorname{Cone}(\mu_{\mathcal{L}})[1],$$

where the term $\operatorname{Cone}(\mu_{\mathcal{L}})$ can be determined directly using the obvious short exact sequence of complexes

$$0 \to \mathcal{L}_{S}^{\bullet} \xrightarrow{\mu_{\mathcal{L}}} \mathcal{L}_{S'}^{\bullet} \to \operatorname{Ind}_{\mathcal{G}_{v_{0}}}^{\mathcal{G}} \mathcal{L}_{v_{0}}^{\bullet} \to 0.$$

This results in an isomorphism $\operatorname{Cone}(\mu_{\mathcal{C}}) \cong \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathcal{L}_{v_0}^{\bullet}[1] = [\operatorname{Ind}_{\mathcal{G}_{v_0}}^{-1} Y_{v_0} \to \Lambda(\mathcal{G})]$ in $\mathcal{D}(\Lambda(\mathcal{G}))$ in the notation of section 1.2. The cohomology of the local complexes was established in proposition 1.2.3 and commutes with induction by (1.8), so we have

$$H^{i}(\operatorname{Cone}(\mu_{\mathcal{C}})) \cong \operatorname{Ind}_{\mathcal{G}_{v_{0}}}^{\mathcal{G}} H^{i+1}(\mathcal{L}_{v_{0}}^{\bullet}) = \begin{cases} \operatorname{Ind}_{\mathcal{G}_{v_{0}}}^{\mathcal{G}} G_{L_{\infty},v_{0}}^{ab}(p), & i = -1 \\ \operatorname{Ind}_{\mathcal{G}_{v_{0}}}^{\mathcal{G}} \mathbb{Z}_{p}, & i = 0 \\ 0 & \text{otherwise} \end{cases}$$

Now class field theory brings $\operatorname{Cone}(\varepsilon_{\mathcal{C}})$ and $\operatorname{Cone}(\mu_{\mathcal{C}})$ together by virtue of the $\Lambda(\mathcal{G})$ -isomorphism

$$\overline{\mathrm{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1) \cong \mathrm{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} G_{L_{\infty},v_0}^{ab}(p) = \ker(\mathrm{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} Y_{v_0} \to \Lambda(\mathcal{G}))$$

(cf. [NSW20] theorem 11.2.3 (ii)), which allows us to define a morphism

$$\varphi \colon \operatorname{Cone}(\varepsilon_{\mathcal{C}}) \xrightarrow{\sim} \overline{\operatorname{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1)[1] \to [\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\overline{\mathcal{G}}^1} Y_{v_0} \to \Lambda^0(\mathcal{G})] = \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathcal{L}_{v_0}^{\bullet}[1] \xrightarrow{\sim} \operatorname{Cone}(\mu_{\mathcal{C}})$$

in $\mathcal{D}(\Lambda(\mathcal{G}))$. The cone of φ is acyclic outside degree 0 and thus isomorphic in the derived category to

$$\operatorname{Cone}(\varphi) \cong H^0(\operatorname{Cone}(\varphi))[0] \cong H^0(\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathcal{L}_{v_0}^{\bullet})[0] = \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[0].$$

We conclude the argument by considering the commutative diagram

$$\begin{array}{ccc} \operatorname{Cone}(\varepsilon_{\mathcal{C}})[-1] & \longrightarrow & \operatorname{Cone}(\alpha_{S,T'}) & \stackrel{\varepsilon_{\mathcal{C}}}{\longrightarrow} & \operatorname{Cone}(\alpha_{S,T}) & \longrightarrow \\ & & & & \downarrow^{\varphi[-1]} & & \parallel \\ \operatorname{Cone}(\mu_{\mathcal{C}})[-1] & \longrightarrow & \operatorname{Cone}(\alpha_{S,T'}) & \stackrel{\mu_{\mathcal{C}}}{\longrightarrow} & \operatorname{Cone}(\alpha_{S',T}) & \longrightarrow \end{array}$$

where the rows are rotated versions of natural exact triangles in $\mathcal{D}(\Lambda(\mathcal{G}))$. This induces a third vertical morphism κ : Cone $(\alpha_{S,T}) \rightarrow$ Cone $(\alpha_{S',T})$ whose cone we can compute using the same technique as before:

$$\operatorname{Cone}(\kappa) \cong \operatorname{Cone}(\varphi[-1])[1] = \operatorname{Cone}(\varphi) \cong \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[0].$$

The desired triangle (3.26) is now a shift of $\operatorname{Cone}(\alpha_{S,T}) \xrightarrow{\kappa} \operatorname{Cone}(\alpha_{S',T}) \to \operatorname{Cone}(\kappa) \to$.

In order to relate the refined Euler characteristics of $\mathcal{C}_{S,T}^{\bullet}$ and $\mathcal{C}_{S\cup\{v_0\},T}^{\bullet}$ using the above proposition, we must trivialise them by choosing a map $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T} = E_{S\cup\{v_0\},T}$ as in setting B. Note that, even though the same α can be employed for both complexes, the trivialisations t^{α} themselves will differ (cf. (2.7)), since the cohomology in degree 1 does not coincide. **Corollary 3.2.6.** Setting *B*. Let $v_0 \notin S \cup T$ be a place of *K* and denote by t^{α} and t_0^{α} the trivialisations constructed in section 2.1 for the complexes $C_{S,T}^{\bullet}$ and $C_{S\cup\{v_0\},T}^{\bullet}$, respectively. Then one has

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S\cup\{v_0\},T},t_0^{\alpha}) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T},t^{\alpha}) - [\Lambda(\mathcal{G}),(1-\varphi_{v_0(L_{\infty})}^{-1})_r,\Lambda(\mathcal{G})]$$

in $K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$, where $\varphi_{v_0(L_\infty)} \in \mathcal{G}_{v_0}$ is the Frobenius automorphism at $v_0(L_\infty)$ and $(1-\varphi_{v_0(L_\infty)}^{-1})_r$ denotes right multiplication by $1-\varphi_{v_0(L_\infty)}^{-1}$.

Proof. The first step is to show that $[\Lambda(\mathcal{G}) \xrightarrow{(1-\varphi_{v_0(L_{\infty})}^{-1})r} \Lambda^1(\mathcal{G})]$ is a strictly perfect representative of $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[-1]$. Consider the endomorphism $(1-\varphi_{v_0(L_{\infty})}^{-1})r$ of the regular left $\Lambda(\mathcal{G}_{v_0})$ -module, which maps $x \in \Lambda(\mathcal{G}_{v_0})$ to $x(1-\varphi_{v_0(L_{\infty})}^{-1})$. Recall that \mathcal{G}_{v_0} is notation for $\mathcal{G}_{v_0(L_{\infty})}$, which is a procyclic group topologically generated by $\varphi_{v_0(L_{\infty})}$ (because $v_0 \notin S_{\operatorname{ram}}(L_{\infty}/K)$), or equivalently, by $\varphi_{v_0(L_{\infty})}^{-1}$. In particular, $\Lambda(\mathcal{G}_{v_0})$ is a commutative ring. The sequence of (topological left) $\Lambda(\mathcal{G}_{v_0})$ -modules

$$0 \to \Lambda(\mathcal{G}_{v_0}) \xrightarrow{(1-\varphi_{v_0(L_{\infty})}^{-1})_r} \Lambda(\mathcal{G}_{v_0}) \to \mathbb{Z}_p \to 0$$
(3.31)

is exact: the third arrow is simply the augmentation map and injectivity follows from taking inverse limits on the exact sequence

$$0 \to \mathbb{Z}_p \cdot \operatorname{Tr}_{\mathcal{G}_{n,v_0}} \to \Lambda(\mathcal{G}_{n,v_0}) \xrightarrow{(1-\varphi_{v_0(L_n)}^{-1})_r} \Lambda(\mathcal{G}_{n,v_0})$$

along the cyclotomic tower. Here \mathcal{G}_{n,v_0} is the decomposition group of $v_0(L_n)$ in L_n/K , which is a finite cyclic group, and the transition maps on the first term send $\operatorname{Tr}_{\mathcal{G}_{n+1,v_0}} \in \Lambda(\mathcal{G}_{n+1,v_0})$ to $p \cdot \operatorname{Tr}_{\mathcal{G}_{n,v_0}} \in \Lambda(\mathcal{G}_{n,v_0})$ for *n* large enough (because v_0 splits into finitely many places in L_∞). The left-exactness of the inverse limit implies $\ker((1 - \varphi_{v_0(L_\infty)}^{-1})_r) = \varprojlim_n \mathbb{Z}_p \cdot \operatorname{Tr}_{\mathcal{G}_{n,v_0}} = 0.$

This yields a quasi-isomorphism of complexes of $\Lambda(\mathcal{G}_{v_0})$ -modules

$$[\Lambda(\overset{0}{\mathcal{G}}_{v_0}) \xrightarrow{(1-\varphi_{v_0(L_{\infty})}^{-1})_r} \Lambda(\overset{1}{\mathcal{G}}_{v_0})] \to \mathbb{Z}_p[-1]$$

which can then be induced to \mathcal{G} by applying compact induction (introduced in section 0.1). We have $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \Lambda(\mathcal{G}_{v_0}) = \Lambda(\mathcal{G})$ and $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} (1 - \varphi_{v_0(L_{\infty})}^{-1})_r = (1 - \varphi_{v_0(L_{\infty})}^{-1})_r$. Note that $\Lambda(\mathcal{G})$ is no longer abelian and thus the left-right distinction becomes relevant. The functor $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}}$ – is exact on (3.31) for the reasons described in section 1.3, which shows that

$$[\Lambda(\mathcal{G}) \xrightarrow{(1-\varphi_{v_0(L_{\infty})}^{-1})_r} \Lambda(\mathcal{G})] \cong \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[-1]$$

in $\mathcal{D}(\Lambda(\mathcal{G}))$ - and, in particular, that the right-hand side is perfect. Since $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p$ is $\Lambda(\mathcal{G})$ -torsion, the zero map constitutes a trivialisation for $\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[-1]$, with associated refined Euler characteristic

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}}\mathbb{Z}_p[-1],0) = -[\Lambda(\mathcal{G}),(1-\varphi_{v_0(L_{\infty})}^{-1})_r,\Lambda(\mathcal{G})]$$
(3.32)

according to (2.3). The negative sign appears because the map $(1 - \varphi_{v_0(L_{\infty})}^{-1})_r$ goes from even to odd degree.

We must now combine the exact triangle (3.26) with the additivity of refined Euler characteristics studied in [BB05]. Although corollary 6.6 therein is specific to the semisimple case, it cannot be applied here, as it requires injectivity of the reduced norm nr: $K_1(\mathcal{Q}(\mathcal{G})) \to Z(\mathcal{Q}(\mathcal{G}))^*$ (recall at this point remark 2.5.4 i)). Instead, we rely on theorem 5.7 from the same source. As in the proof of proposition 3.2.2, two versions of refined Euler characteristics must be considered, which we again denote by $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}$ (the one in section 2.1) and $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}$. Of particular importance is the fact that trivialisations for $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}$ are maps from even to odd degree, as opposed to those for $\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}$.

Let $S' = S \cup \{v_0\}$. In order to apply theorem 5.7 to (3.26), we need to verify the commutativity hypothesis on diagram (15) from the cited article. The square brackets therein denote a certain universal determinant functor $[-]: \mathcal{D}(\mathcal{Q}(\mathcal{G})) \to \mathcal{V}(\mathcal{Q}(\mathcal{G}))$ from the derived category of left $\mathcal{Q}(\mathcal{G})$ modules to a Picard category $\mathcal{V}(\mathcal{Q}(\mathcal{G}))$. The following facts will be used:

- $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\text{odd}}(\text{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[-1]) = \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\text{even}}(\text{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p[-1]) = 0$. Therefore, the last term in each row of the diagram is [0], the unit object $\mathbb{1} \in \mathcal{V}(\mathcal{Q}(\mathcal{G}))$. The vertical arrow between them is $[0_{\text{Hom}}]$, where 0_{Hom} denotes the zero homomorphism.
- The homomorphisms induced by the arrow $\mathcal{C}_{S,T}^{\bullet} \to \mathcal{C}_{S',T}$ in cohomology are injective: in degree 0, it is simply the embedding $H^0(\mathcal{C}_{S,T}^{\bullet}) \cong E_{S,T} \hookrightarrow E_{S',T} \cong H^0(\mathcal{C}_{S',T}^{\bullet})$; and in degree 1, it fits into the diagram

where the left vertical arrow is in fact an equality by lemma 3.2.3.

This shows that the second and third terms in each row of diagram (15) are again [0] = 1, and the vertical arrows [Id] and [-Id] are both equal to $[0_{Hom}]$. Since functors preserve commutativity, it suffices to verify that of

$$\begin{aligned} \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\operatorname{even}}(\mathcal{C}_{S,T}^{\bullet}) & \longrightarrow \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\operatorname{even}}(\mathcal{C}_{S',T}^{\bullet}) \\ & \downarrow^{(t^{\alpha})^{-1}} & \downarrow^{(t^{\alpha}_{0})^{-1}} \\ \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\operatorname{odd}}(\mathcal{C}_{S,T}^{\bullet}) & \longrightarrow \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{\operatorname{odd}}(\mathcal{C}_{S',T}^{\bullet}) \end{aligned}$$

which follows immediately from the commutative diagram

where all arrows become isomorphisms after $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} - (cf. (2.7))$. We can therefore apply [BB05] theorem 5.7 and conclude that

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}(\mathcal{C}_{S',T}^{\bullet},(t_0^{\alpha})^{-1}) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}(\mathcal{C}_{S,T}^{\bullet},(t^{\alpha})^{-1}) + \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}^{\text{new}}(\text{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}}\mathbb{Z}_p[-1],0).$$
(3.33)

Now the same argument as in the proof of proposition 3.2.2 yields the corresponding relation on our usual refined Euler characteristics:

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S',T}, t^{\alpha}_{0}) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T}, t^{\alpha}) + \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathrm{Ind}_{\mathcal{G}_{v_{0}}}^{\mathcal{G}}\mathbb{Z}_{p}[-1], 0).$$

This concludes the proof by (3.32).

The above results offer a homological comparison of the Main Conjectures defined on S and $S \cup \{v_0\}$. Over on the analytic side, one needs to determine the change in leading coefficients and regulators caused by the addition of the prime v_0 . The difference between the Artin *L*-series is given by the local Euler factor L_{v_0} defined in (0.17), which *almost never* vanishes at s = 0:

Lemma 3.2.7. Setting A, β as in setting B. Let $v_0 \notin S_{ram}(L_{\infty}/K)$ be a place of K and χ an Artin character of \mathcal{G} . Then, for almost all ρ of type W, one has

$$L_{v_0}(\chi \otimes \rho, 0)^{-1} = \det(1 - \varphi_{v_0(L_n)} \mid V_{\beta(\chi \otimes \rho)}) \neq 0$$

in the notation of (0.17), where n is chosen large enough that $\Gamma^{p^n} \subseteq \ker(\chi \otimes \rho)$.

Proof. The proof is almost immediate and can be done entirely over \mathbb{C} . To see this, note that there is an equality of sets

$$\{\beta(\chi \otimes \rho)\}_{\rho} = \{(\beta \chi) \otimes \tilde{\rho}\}_{\tilde{\rho}},$$

where ρ runs over the type-W characters of \mathcal{G} (*p*-adic by definition) and $\tilde{\rho}$ runs over the \mathbb{C} -valued linear characters of \mathcal{G} with open kernel which are trivial on H.

For ρ as above, set $\tilde{\chi} = \beta \chi$ and $\tilde{\rho} = \beta \rho$ and choose $n = n(\rho) \in \mathbb{N}$ such that $\tilde{\chi}$ and $\tilde{\rho}$ factor over \mathcal{G}_n . Then

$$\det(1 - \varphi_{v_0(L_n)} \mid V_{\tilde{\chi} \otimes \tilde{\rho}}) = \det(1 - \tilde{\rho}(\varphi_{v_0(L_n)}) M_{\tilde{\chi}}(\varphi_{v_0(L_n)})),$$

where $M_{\tilde{\chi}}(\varphi_{v_0(L_n)})$ is the matrix describing the action of $\varphi_{v_0(L_n)}$ on the $\mathbb{C}[\mathcal{G}_n]$ -module $V_{\tilde{\chi}}$. We now note that the right-hand side is the value of the polynomial $f(t) = \det(1 - tM_{\tilde{\chi}}(\varphi_{v_0(L_n)}))$ at $t = \tilde{\rho}(\varphi_{v_0(L_n)}) \in \mathbb{C}^*$. This f (which is almost the characteristic polynomial of $M_{\tilde{\chi}}(\varphi_{v_0(L_n)})$) has degree $\chi(1) = \tilde{\chi}(1)$ and is independent of $\tilde{\rho}$ (even though n may change, the matrix of the action of $\varphi_{v_0(L_n)}$ on $V_{\tilde{\chi}}$ does not).

Now the result follows from the fact that f(t) has finitely many roots $x \in \mathbb{C}$ and, for any such x, only finitely many $\tilde{\rho}$ as above satisfy $\tilde{\rho}(\varphi_{v_0(L_{\infty})}) = x$. To see why, fix one such $\tilde{\rho}$ and note that every other type-W (and hence irreducible) character with that property factors through the finite group $\mathcal{G}/(\mathcal{G}_{v_0} \cap \ker(\tilde{\rho}))$.

The last preparatory step to show the independence of the conjectures of the choice of S concerns the Stark-Tate regulator:

Lemma 3.2.8. Setting *B*. Let $v_0 \notin S \cup T$ be a place of *K* and $\chi \in \operatorname{Irr}_p(\mathcal{G})$. Then one has

$$R_{S}^{\beta}(\alpha, \chi \otimes \rho) = R_{S \cup \{v_{0}\}}^{\beta}(\alpha, \chi \otimes \rho)$$
(3.34)

for almost all $\rho \in \mathcal{K}^{\alpha}_{S \cup \{v_0\}}(\chi) \subseteq \mathcal{K}^{\alpha}_{S}(\chi)$.

Proof. Set $S' = S \cup \{v_0\}$. The canonical short exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to \mathcal{X}_S \to \mathcal{X}_{S'} \to \mathcal{Y}_{\{v_0\}} \to 0,$$

which remains exact on finite level, induces an embedding

$$\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi\otimes\rho},\mathbb{C}_p\otimes_{\mathbb{Z}_p}\mathcal{X}_{L_n,S})\hookrightarrow\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi\otimes\rho},\mathbb{C}_p\otimes_{\mathbb{Z}_p}\mathcal{X}_{L_n,S'})$$

for all $\rho \in \mathcal{K}^{\alpha}_{S \cup \{v_0\}}(\chi)$ and sufficiently large *n* depending on ρ . We claim this is an equality, that is, $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi \otimes \rho}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n,\{v_0\}}) = 0$, for almost all such ρ . As explained in the proof of lemma 2.3.5, this amounts to $\langle \chi \otimes \rho, \operatorname{Ind}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \mathbb{1}_{\mathcal{G}_{n,v_0}} \rangle_{\mathcal{G}_n} = 0$, since $\operatorname{Ind}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \mathbb{1}_{\mathcal{G}_{n,v_0}}$ is the character afforded by the $\mathbb{C}_p[\mathcal{G}_n]$ -module $\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{Y}_{L_n,\{v_0\}}$. Recall at this point that \mathcal{G}_{n,v_0} denotes $(\mathcal{G}_n)_{v_0(L_n)}$. By (0.11) and (0.12), one has

$$\langle \chi \otimes \rho, \operatorname{ind}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \mathbb{1}_{\mathcal{G}_{n,v_0}} \rangle_{\mathcal{G}_n} = \langle \operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \chi \otimes \rho, \mathbb{1}_{\mathcal{G}_{n,v_0}} \rangle_{\mathcal{G}_{n,v_0}} = \langle \operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \rho, \operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \check{\chi} \rangle_{\mathcal{G}_{n,v_0}}$$

The restriction $\operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \rho$ is a linear, and therefore irreducible, character of \mathcal{G}_{n,v_0} . This implies that the last scalar product is non-zero if and only if $\operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \rho$ is one of the finitely many irreducible divisors of $\operatorname{res}_{\mathcal{G}_{n,v_0}}^{\mathcal{G}_n} \check{\chi}$ (note that these do not depend on the layer *n*). But only finitely many type-*W* characters of \mathcal{G} have a particular restriction to \mathcal{G}_{v_0} , as explained at the end of the proof of lemma 3.2.7.

Let us now turn to the finite-level maps φ_n^{α} induced by α for $n \ge n(S)$ (cf. definition 2.2.6), which we denote here by $\varphi_{n,S}^{\alpha}$ and $\varphi_{n,S'}^{\alpha}$ depending on the set of places used. These two homomorphisms arise as the composition of the rows of the commutative diagram

in the notation of section 2.2, where we have added the subscripts S and S' to some of the morphisms for distinction purposes. It follows that $\varphi_{n,S'}^{\alpha}$ restricted to $\mathcal{X}_{L_n,S}$ coincides with $\varphi_{n,S}^{\alpha}$ (under the canonical inclusion of their codomains). A similar statement holds for the Dirichlet regulator map: the commutativity of

implies $\lambda_{n,S'}^{\beta}|_{\mathbb{C}_p\otimes\mathcal{O}^*_{L_{n,S}}} = \lambda_{n,S}^{\beta}$ (cf. definition 2.3.3), and this in turn

$$\lambda_{n,S'}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_{n,S'}^{\alpha})|_{\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}} = \lambda_{n,S}^{\beta} \circ (\mathbb{C}_p \otimes_{\mathbb{Z}_p} \varphi_{n,S}^{\alpha}).$$

We are essentially done, since $R_S^{\beta}(\alpha, \chi \otimes \rho)$ and $R_{S'}^{\beta}(\alpha, \chi \otimes \rho)$ are given by the determinant of the \mathbb{C}_p -linear vertical maps in the diagram

which commutes by the above. The first part of the proof shows that the horizontal arrows are equalities for almost all $\rho \in \mathcal{K}^{\alpha}_{S'}(\chi)$, and this concludes the argument.

The first main result of this subsection now follows easily:

Proposition 3.2.9. Setting *B*. Let $v_0 \notin S \cup T$ be a place of *K* and set $S' = S \cup \{v_0\}$. Then:

- i) For any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, L, S', T, \alpha, \beta)$ does.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S', T, \alpha, \beta)$ does.

Proof. Let $\varphi_{v_0(L_{\infty})} \in \mathcal{G}_{v_0}$ denote the Frobenius at $v_0(L_{\infty})$, and $[(1 - \varphi_{v_0(L_{\infty})}^{-1})] \in K_1(\mathcal{Q}(\mathcal{G}))$ the class of the 1-by-1 invertible matrix $(1 - \varphi_{v_0(L_{\infty})}^{-1}) \in \mathrm{GL}_1(\mathcal{Q}(\mathcal{G}))$. Here left invertibility follows from the proof of corollary 3.2.6, and right invertibility from a symmetric argument.

Fix first $\chi \in \operatorname{Irr}_p(\mathcal{G})$ and assume that $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Set $F' = F_{S,T,\chi}^{\alpha,\beta} \cdot \psi_{\chi}([(1 - \varphi_{v_0(L_{\infty})}^{-1})]) \in \mathcal{Q}^c(\Gamma_{\chi})^*.$

Recall that $\psi_{\chi} = \psi_{\chi \otimes \rho}$: $K_1(\mathcal{Q}(\mathcal{G})) \to \mathcal{Q}^c(\Gamma_{\chi})^* = \mathcal{Q}^c(\Gamma_{\chi \otimes \rho})^*$ for all ρ of type W. Let ρ be such a character and choose n such that $\chi \otimes \rho$ factors through \mathcal{G}_n . Equations (3.21) and (3.22) imply

$$ev_{\gamma_{\chi\otimes\rho}}(\psi_{\chi}([(1-\varphi_{v_0(L_{\infty})}^{-1})])) = \det(\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} ((1-\varphi_{v_0(L_{\infty})}^{-1})_r)_{\Gamma^{p^n}} \mid \operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\chi\otimes\rho}, \mathbb{Q}_p^c[\mathcal{G}_n])),$$

where $\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} ((1 - \varphi_{v_0(L_{\infty})}^{-1})_{\Gamma^{p^n}})_{\Gamma^{p^n}}$ coincides with the endomorphism of $\mathbb{Q}_p^c[\mathcal{G}_n]$ given by right multiplication by $1 - \varphi_{v_0(L_n)}^{-1}$. Denote by $M_{\chi \otimes \rho}(\varphi_{v_0(L_n)}^{-1}) \in M_{\chi(1)}(\mathbb{Q}_p^c)$ the image of $\varphi_{v_0(L_n)}^{-1}$ under the representation associated to $\chi \otimes \rho$. Then an easy verification shows that, under a suitable \mathbb{Q}_p^c -basis, post-composition with $(1 - \varphi_{v_0(L_n)}^{-1})_r$ acts on $\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\chi \otimes \rho}, \mathbb{Q}_p^c[\mathcal{G}_n])$ via the transpose $M_{\chi \otimes \rho}(\varphi_{v_0(L_n)}^{-1})^t$.

Thus, the above determinant coincides with $\det(\mathrm{Id} - M_{\chi \otimes \rho}(\varphi_{v_0(L_n)}^{-1})^t) = \det(\mathrm{Id} - M_{\check{\chi} \otimes \rho^{-1}}(\varphi_{v_0(L_n)}))$ and, as a consequence,

$$ev_{\gamma_{\chi\otimes\rho}}(\psi_{\chi}([(1-\varphi_{v_0(L_{\infty})}^{-1})])) = \det(1-\varphi_{v_0(L_n)} \mid V_{\check{\chi}\otimes\rho^{-1}}) = \beta^{-1}(\det(1-\varphi_{v_0(L_n)} \mid V_{\beta(\check{\chi}\otimes\rho^{-1})}))).$$
(3.35)

This provides us with the necessary information about leading coefficients, since lemma 3.2.7 shows that

$$L^*_{K,S',T}(\beta(\check{\chi}\otimes\rho^{-1}),0)=L^*_{K,S,T}(\beta(\check{\chi}\otimes\rho^{-1}),0)\cdot\det(1-\varphi_{v_0}\mid V_{\beta(\check{\chi}\otimes\rho^{-1})})$$

for almost all ρ . Here we are implicitly using definition (0.20), which applies to the half-plane $\operatorname{Re}(s) > 1$, at the point s = 0. The reason we can do so was already outlined after equation (3.5).

It now follows that the series quotient F' satisfies

$$\begin{split} ev_{\gamma_{\chi\otimes\rho}}(F') &= ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha,\beta}) \cdot ev_{\gamma_{\chi\otimes\rho}}(\psi_{\chi}([(1-\varphi_{v_0(L_{\infty})}^{-1})])) \\ &= \frac{\beta^{-1}(L_{K,S,T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^{\beta}(\alpha,\chi\otimes\rho)} \cdot ev_{\gamma_{\chi\otimes\rho}}(\psi_{\chi}([(1-\varphi_{v_0(L_{\infty})}^{-1})])) \\ &= \frac{\beta^{-1}(L_{K,S',T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^{\beta}(\alpha,\chi\otimes\rho)} \\ &= \frac{\beta^{-1}(L_{K,S',T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_{S'}^{\beta}(\alpha,\chi\otimes\rho)} \end{split}$$

for almost all $\rho \in \mathcal{K}^{\alpha}_{S'}(\chi) \subseteq \mathcal{K}^{\alpha}_{S}(\chi)$, where the first equation relies on remark 2.4.9 ii) together with (3.35), and the last one is lemma 3.2.8. In other words, $\mathrm{IC}(L_{\infty}/K, \chi, L, S', T, \alpha, \beta)$ holds.

An analogous argument shows the converse implication: if the element $F_{S',T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K,\chi,L,S',T,\alpha,\beta)$, then $F = F_{S',T,\chi}^{\alpha,\beta} \cdot \psi_{\chi}([(1-\varphi_{v_0(L_{\infty})}^{-1})])^{-1}$ has the interpolation property required by $\mathrm{IC}(L_{\infty}/K,\chi,L,S,T,\alpha,\beta)$.

Part ii) is a direct consequence of part i) and corollary 3.2.6, as $[(1 - \varphi_{v_0(L_\infty)}^{-1})] \in K_1(\mathcal{Q}(\mathcal{G}))$ does not depend on the choice of a character. Assume first $\operatorname{eMC}(L_\infty/K, L, S, T, \alpha, \beta)$ and let $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ be the zeta element predicted therein. By assumption, $\operatorname{IC}(L_\infty/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in \operatorname{Irr}_p(\mathcal{G})$, and hence so does $\operatorname{IC}(L_\infty/K, \chi, L, S', T, \alpha, \beta)$.

Consider $\zeta' = \zeta_{S,T}^{\alpha,\beta} \cdot [(1 - \varphi_{v_0(L_{\infty})}^{-1})]$, which satisfies

$$\begin{aligned} \partial(\zeta') &= \partial(\zeta_{S,T}^{\alpha,\beta}) + \partial([(1 - \varphi_{v_0(L_{\infty})}^{-1})]) \\ &= -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha}) + [\Lambda(\mathcal{G}), (1 - \varphi_{v_0(L_{\infty})}^{-1})_r, \Lambda(\mathcal{G})] \\ &= -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S',T}^{\bullet}, t^{\alpha}) \end{aligned}$$

by corollary 3.2.6. Together with the relation $F_{S',T,\chi}^{\alpha,\beta} = F_{S,T,\chi}^{\alpha,\beta} \cdot \psi_{\chi}([(1 - \varphi_{v_0(L_{\infty})}^{-1})])$ established in part i), this proves that ζ' satisfies $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$. The reverse implication follows from the choice $\zeta = \zeta_{S',T}^{\alpha,\beta} \cdot [(1 - \varphi_{v_0(L_{\infty})}^{-1})]^{-1}$ in the obvious notation.

We now turn our attention to the change of T. Much of the necessary work has already been done in the treatment of S, and the elements which do differ follow similar arguments. The following aspects distinguish the two cases:

- The unit module $E_{S,T} = \varprojlim_n \mathbb{Z}_p \otimes \mathcal{O}^*_{L_n,S,T}$ does become smaller as we enlarge T, so the homomorphism α needs to be modified now. The next lemma provides a natural way to fix this discrepancy.
- The Stark-Tate regulator $R_S^{\beta}(\alpha, \chi)$ is independent of T.
- Unlike L-factors, the δ -factors of Artin L-functions never vanish at s = 0, as shown at the beginning of the proof of proposition 3.1.1.

Lemma 3.2.10. Setting A. Let $v_0 \notin S \cup T$ be a place of K. If $\alpha \colon \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S,T \cup \{v_0\}}$ is an injective $\Lambda(\mathcal{G})$ -homomorphism with $\Lambda(\Gamma)$ -torsion cokernel, then the same is true of the composition $\tilde{\alpha} \colon \mathcal{Y}_{S_{\infty}} \stackrel{\alpha}{\hookrightarrow} E_{S,T \cup \{v_0\}} \hookrightarrow E_{S,T}$.

Proof. Let T' denote $T \cup \{v_0\}$. Taking inverse limits of the exact sequence

$$1 \to \mathcal{O}_{L_n,S,T'}^* \to \mathcal{O}_{L_n,S,T}^* \to \sum_{w_n \in \{v_0\}(L_n)} \kappa(w_n)^* \to Cl_{L_n,S,T'} \to Cl_{L_n,S,T} \to 1$$

along the cyclotomic tower L_{∞}/L gives rise to an exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to E_{S,T'} \to E_{S,T} \to \overline{\operatorname{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1) \to X_{T',S}^{cs} \to X_{T,S}^{cs} \to 0$$
(3.36)

by the same argument used on (1.12), where $\overline{\operatorname{Ind}}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p(1)$ is as in proposition 3.2.5. In particular, $\operatorname{coker}(E_{S,T'} \hookrightarrow E_{S,T}) \hookrightarrow \overline{\operatorname{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}}\mathbb{Z}_p(1)$ is $\Lambda(\Gamma)$ -torsion, as it has finite \mathbb{Z}_p -rank. Now apply the snake lemma to

It only remains to find a strictly perfect representative of the complex measuring the difference between $C^{\bullet}_{S,T}$ and $C^{\bullet}_{S,T'}$ in the sense of (3.27). This result is likely well known among experts.

Lemma 3.2.11. Setting A, α as in setting B. Let $v_0 \notin S_{ram}(L_{\infty}/K)$ be a place of K. Then there exists a quasi-isomorphism of complexes of $\Lambda(\mathcal{G})$ -modules

$$[\Lambda(\mathcal{G}) \xrightarrow{(1-\mathfrak{N}(v_0)\varphi_{v_0(L_{\infty})}^{-1})_r} \Lambda(\mathcal{G})] \to \overline{\mathrm{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1)[0],$$

where $\varphi_{v_0(L_{\infty})}$ is the Frobenius automorphism at $v_0(L_{\infty})$ and $(1 - \varphi_{v_0(L_{\infty})}^{-1} \mathfrak{N}(v_0))_r$ denotes multiplication by $1 - \mathfrak{N}(v_0)\varphi_{v_0(L_{\infty})}^{-1}$ on the right.

Proof. We will be concise, since the argument is very similar to that of corollary 3.2.6. Taking the inverse limit of the finite-level sequences

$$0 \to \Lambda(\mathcal{G}_{n,v_0}) \xrightarrow{(\varphi_{v_0(L_n)} - \mathfrak{N}(v_0))_r} \Lambda(\mathcal{G}_{n,v_0}) \to \Lambda(\mathcal{G}_{n,v_0}) / \langle \varphi_{v_0(L_n)} - \mathfrak{N}(v_0) \rangle \to 0$$

along the cyclotomic tower yields an exact sequence of $\Lambda(\mathcal{G}_{v_0})$ -modules. If $L_{v_0(L)}$ does not contain a primitive *p*-th root of unity (that is, $p \nmid 1 - \mathfrak{N}(v_0(L))$), the cokernel is trivial on all finite levels and hence in the limit. Otherwise, the limit of the cokernels is isomorphic to \mathbb{Z}_p with the \mathcal{G}_{v_0} -action $\varphi_{v_0(L_{\infty})}z = \mathfrak{N}(v_0)z$, which is precisely $\mathbb{Z}_p(1)$. Multiplying the arrow by the unit $\varphi_{v_0(L_n)}^{-1}$ and applying induction results in an exact sequence

$$0 \to \Lambda(\mathcal{G}) \xrightarrow{(1-\mathfrak{N}(v_0)\varphi_{v_0(L_n)}^{-1})_r} \Lambda(\mathcal{G}) \to \operatorname{Ind}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_{p/\langle \varphi_{v_0(L_n)}^{-1} - \mathfrak{N}(v_0) \rangle} \to 0.$$

from which the lemma follows.

This allows us to prove the independence of the choice of T:

Proposition 3.2.12. Setting A, β as in setting B. Let $v_0 \notin S \cup T$ be a place of K and set $T' = T \cup \{v_0\}$. Choose $\alpha \colon \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S,T'}$ as in setting B and define $\tilde{\alpha}$ as in lemma 3.2.10. Then:

- i) For any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \tilde{\alpha}, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, L, S, T', \alpha, \beta)$ does.
- ii) $eMC(L_{\infty}/K, L, S, T, \tilde{\alpha}, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S, T', \alpha, \beta)$ does.

Proof. We proceed along very similar lines to the proof of proposition 3.2.9 and omit many of the details. The element $1 - \Re(v_0)\varphi_{v_0(L_{\infty})}^{-1}$ is a unit in $\mathcal{Q}(\mathcal{G})$ for the same reasons as $1 - \varphi_{v_0(L_{\infty})}^{-1}$ before and we may therefore consider its class $[(1 - \Re(v_0)\varphi_{v_0(L_{\infty})}^{-1})] \in K_1(\mathcal{Q}(\mathcal{G})).$

In order to show i), we first study the difference between the *L*-values and regulators associated to T and T'. Given $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$, choose $n \in \mathbb{N}$ large enough that ς factors through \mathcal{G}_n . Since δ -factors never vanish at 0 (see for instance the proof of proposition 3.1.1), one has

$$L_{K,S,T'}^*(\varsigma,0) = L_{K,S,T}^*(\varsigma,0) \cdot \delta_{v_0}(\varsigma,0).$$

As for the regulators, suppose $n \ge n(S)$ and consider the commutative diagram

in the notation of section 2.2. After applying $\mathbb{C}_p \otimes_{\mathbb{Z}_p} -$, the last vertical arrow becomes an equality (in fact, all vertical arrows do) because its image has finite index in the codomain. It follows that $R_S^\beta(\alpha,\varsigma) = R_S^\beta(\tilde{\alpha},\varsigma)$.

Assume now that $F_{S,T,\chi}^{\tilde{\alpha},\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K,\chi,L,S,T,\tilde{\alpha},\beta)$ for some $\chi \in \mathrm{Irr}_p(\mathcal{G})$. Then

$$F' = F_{S,T,\chi}^{\tilde{\alpha},\beta} \cdot \psi_{\chi}([(1 - \mathfrak{N}(v_0)\varphi_{v_0(L_{\infty})}^{-1})]) \in \mathcal{Q}^c(\Gamma_{\chi})^*$$

satisfies $IC(L_{\infty}/K, \chi, L, S, T', \alpha, \beta)$ by the same argument as in proposition 3.2.9, which also yields the reverse implication.

We now address the equivalence of the equivariant Main Conjectures, which involves comparing the refined Euler characteristics of $C_{S,T}^{\bullet}$ and $C_{S,T'}^{\bullet}$ induced by the trivialisations $t^{\tilde{\alpha}}$ and t^{α} , respectively. The difference between the complexes themselves is given by the exact triangle (3.27). Applying theorem 5.7 from [BB05], we obtain

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T},t^{\tilde{\alpha}}) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T'},t^{\alpha}) + \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\overline{\mathrm{Ind}}^{\mathcal{G}}_{\mathcal{G}_{v_0}}\mathbb{Z}_p(1)[0],0)$$

this time. Note that 0 is a valid trivialisation for $\overline{\text{Ind}}_{\mathcal{G}_{v_0}}^{\mathcal{G}} \mathbb{Z}_p(1)[0]$, as its only non-trivial module is torsion. The second summand can be computed explicitly using the strictly perfect representative from lemma 3.2.11. This results in

$$\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T}, t^{\tilde{\alpha}}) = \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T'}, t^{\alpha}) + [\Lambda(\mathcal{G}), (1 - \mathfrak{N}(v_0)\varphi_{v_0(L_{\infty})}^{-1})_r, \Lambda(\mathcal{G})]$$
$$= \chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T'}, t^{\alpha}) + \partial([(1 - \mathfrak{N}(v_0)\varphi_{v_0(L_{\infty})}^{-1})]).$$

Suppose that $\zeta_{S,T}^{\tilde{\alpha},\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ satisfies $\operatorname{eMC}(L_{\infty}/K, L, S, T, \tilde{\alpha}, \beta)$. Then $\zeta' = \zeta_{S,T}^{\tilde{\alpha},\beta} \cdot [(1 - \mathfrak{N}(v_0)\varphi_{v_0}^{-1})]$ satisfies $\operatorname{eMC}(L_{\infty}/K, L, S, T', \alpha, \beta)$ by part i) and the above equation. The converse is proved by setting $\zeta = \zeta_{S,T'}^{\alpha,\beta} \cdot [(1 - \mathfrak{N}(v_0)\varphi_{v_0}^{-1})]^{-1}$.

This proposition, together with 3.2.9, shows that the Main Conjecture is unaffected by the enlargement of S or T one place at a time. Therefore, it is independent of the choice S and T altogether. As we have seen, the homomorphism α needs to be modified when T changes. The previous subsection showed that changes in α do not affect the conjecture, but one can also formulate the following result without relying on that fact:

Corollary 3.2.13. Setting A, β as in setting B. Let $\widetilde{S}, \widetilde{T}$ be another valid choice of the corresponding sets of places. In particular, so is $S \cap \widetilde{S}, T \cup \widetilde{T}$. Consider an injective $\Lambda(\mathcal{G})$ -homomorphism $\alpha_0 \colon \mathcal{Y}_{S_{\infty}} \hookrightarrow E_{S \cap \widetilde{S}, T \cup \widetilde{T}}$ with $\Lambda(\Gamma)$ -torsion cokernel and set

$$\alpha \colon \mathcal{Y}_{S_{\infty}} \stackrel{\alpha_0}{\hookrightarrow} E_{S \cap \widetilde{S}, T \cup \widetilde{T}} \hookrightarrow E_{S \cap \widetilde{S}, T} = E_{S, T} \quad and \quad \widetilde{\alpha} \colon \mathcal{Y}_{S_{\infty}} \stackrel{\alpha_0}{\hookrightarrow} E_{S \cap \widetilde{S}, T \cup \widetilde{T}} \hookrightarrow E_{S \cap \widetilde{S}, \widetilde{T}} = E_{\widetilde{S}, \widetilde{T}}$$

- (cf. lemma 3.2.3). Then:
 - i) For all $\chi \in \operatorname{Irr}_p(\mathcal{G})$, $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds if and only if $IC(L_{\infty}/K, \chi, L, \widetilde{S}, \widetilde{T}, \widetilde{\alpha}, \beta)$ does.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, \widetilde{S}, \widetilde{T}, \widetilde{\alpha}, \beta)$ does.

Proof. In part i), it suffices to show that both conjectures are equivalent (for each fixed χ) to $\mathrm{IC}(L_{\infty}/K, \chi, L, S \cap \widetilde{S}, T \cup \widetilde{T}, \alpha_0, \beta)$. The same is true of ii) and $\mathrm{eMC}(L_{\infty}/K, L, S \cap \widetilde{S}, T \cup \widetilde{T}, \alpha_0, \beta)$. This is achieved by propositions 3.2.9 and 3.2.12, since the embedding $E_{S \cap \widetilde{S}, T \cup \widetilde{T}} \hookrightarrow E_{S \cap \widetilde{S}, T}$ is the composition

$$E_{S\cap\widetilde{S},T\cup\widetilde{T}} \hookrightarrow E_{S\cap\widetilde{S},T\cup\widetilde{T}\setminus\{v_0\}} \hookrightarrow \cdots \hookrightarrow E_{S\cap\widetilde{S},T\cup\widetilde{T}\setminus\{v_0,\dots,v_n\}} = E_{S\cap\widetilde{S},T}$$

and analogously for $E_{S\cap \widetilde{S}, T\cup \widetilde{T}} \hookrightarrow E_{S\cap \widetilde{S}, \widetilde{T}}.$

This corollary, together with propositions 3.2.1 and 3.2.2, proves theorem 3.0.1: one may first change L without affecting any other parameters; then replace α by the homomorphism induced by α_0 in corollary 3.2.13, also without altering the remaining parameters; and finally modify S and T. Recall that Stark's conjecture $\operatorname{Stark}^{\mathrm{T}}(L/K, \chi, f, S, T)$ is independent of S and T as well, as explained in section 3.1.

3.3 Functoriality

It has now been established that, for a fixed extension L_{∞}/K , the Main Conjecture is independent of all choices of parameters except possibly β , and conjecturally of this last one too. The next natural step is to study changes in L_{∞} and K. As we shall see, the conjecture for $\operatorname{Gal}(L_{\infty}/K)$ will imply that for *smaller* groups - be it subgroups or quotients. The challenge lies in the impact modifying the Galois group has on the algebraic machinery developed so far. In the last subsection, we apply K-theoretic techniques to briefly address the converse question: whether one can deduce the Main Conjecture for a given extension by assuming it for sufficiently many smaller ones.

3.3.1 Change of L_{∞}

We first discuss the top field by considering $L'_{\infty}/L_{\infty}/K$. The new situation can be formalised as follows:

Setting C. The objects $p, L'_{\infty}/K, \Gamma', S$ and T are fixed as in setting A in the obvious notation. We set $\Gamma_K = \text{Gal}(K_{\infty}/K) = \overline{\langle \gamma_K \rangle}$ and $H' = \text{Gal}(L'_{\infty}/K_{\infty}) \leq_c \mathcal{G}' = \text{Gal}(L'_{\infty}/K)$. Additionally, we consider:

- A finite normal subgroup $\widetilde{H} \leq_c \mathcal{G}'$. In particular, one has $\widetilde{H} \leq H'$ and we can define $L_{\infty} = (L'_{\infty})^{\widetilde{H}} \supseteq K_{\infty}, H = \operatorname{Gal}(L_{\infty}/K_{\infty}) = H'/\widetilde{H}$ and $\mathcal{G} = \operatorname{Gal}(L_{\infty}/K) = \mathcal{G}'/\widetilde{H}$.
- The projection of $\Gamma' \subseteq \mathcal{G}'$ to \mathcal{G} , which we denote by Γ . Since $\Gamma' \cap \widetilde{H} \subseteq \Gamma' \cap H' = 1$, the group Γ' identifies with Γ , and $\Lambda(\Gamma) \subseteq \Lambda(\mathcal{G})$ with $\Lambda(\Gamma') \subseteq \Lambda(\mathcal{G}')$. It follows immediately that $\Gamma \subseteq Z(\mathcal{G})$ is open in \mathcal{G} and $\Gamma \cap H = 1$. Therefore, $L = (L_{\infty})^{\Gamma}$ is a finite Galois extension of K such that L_{∞} is the cyclotomic \mathbb{Z}_p -extension of L.

As in setting A, we use the notation L'_n and L_n for the layers of L'_{∞}/K and L_{∞}/K , respectively, and set $\mathcal{G}'_n = \operatorname{Gal}(L'_n/K)$ and $\mathcal{G}_n = \operatorname{Gal}(L_n/K)$. Note that, for all $n \in \mathbb{N}$, the field L'_n is a Galois extension of L_n such that $\operatorname{Gal}(L'_n/L_n) = \ker(\mathcal{G}'_n \twoheadrightarrow \mathcal{G}_n) \cong \widetilde{H}$. • The natural numbers n'(S) and n(S) given by

$$(\Gamma')^{p^{n'(S)}} = \Gamma' \cap \bigcap_{w'_{\infty} \in S_f(L'_{\infty})} \mathcal{G}'_w \quad \text{and} \quad \Gamma^{p^{n(S)}} = \Gamma \cap \bigcap_{w_{\infty} \in S_f(L_{\infty})} \mathcal{G}_w$$

This implies that $L_{n(S)}$ is the smallest layer of L_{∞}/L such that all finite places in S are nonsplit in $L_{\infty}/L_{n(S)}$, and analogously for n'(S). Although it will not play a role, it is easy to show that $n'(S) \ge n(S)$. As usual, we denote the decomposition groups $\mathcal{G}'_{v(L'_{\infty})}$ and $\mathcal{G}_{v(L_{\infty})}$ of the distinguished prolongations of v by \mathcal{G}_v and \mathcal{G}'_v , respectively.

With the above conventions, $(L'_{\infty}/K, \Gamma', S, T)$ and $(L_{\infty}/K, \Gamma, S, T)$ are two valid choices of the corresponding parameters in setting A. The following diagram illustrates the relations between the relevant fields:



Our first aim is to study the relation between the semisimple algebras $\mathcal{Q}^{c}(\mathcal{G}')$ and $\mathcal{Q}^{c}(\mathcal{G})$, and in particular the fields $\mathcal{Q}^{c}(\Gamma_{\chi})$ contained in them. In general, it is not true that a continuous surjection of profinite groups induces a ring homomorphism between the total rings of fractions of their Iwasawa algebras, but the properties of \mathcal{G}' and \mathcal{G} ensure this is the case in our situation. The following discussion makes repeated use of the obvious identification

$$\operatorname{Irr}_{p}(\mathcal{G}) \leftrightarrow \{ \chi \in \operatorname{Irr}_{p}(\mathcal{G}') \text{ such that } H \subseteq \operatorname{ker}(\chi) \}$$

$$\overline{\chi} \leftrightarrow \chi = \operatorname{inf}_{\mathcal{G}}^{\mathcal{G}'} \overline{\chi},$$
(3.37)

which restricts to an identification of type-W characters.

Lemma 3.3.1. Setting C. The following hold:

- i) The projection $\mathcal{G}' \twoheadrightarrow \mathcal{G}$ induces a canonical surjection of $\mathcal{Q}(\Gamma')$ -algebras $\varepsilon \colon \mathcal{Q}(\mathcal{G}') \twoheadrightarrow \mathcal{Q}(\mathcal{G})$ (where the codomain is regarded as such via $\mathcal{Q}(\Gamma') \xrightarrow{\sim} \mathcal{Q}(\Gamma) \hookrightarrow Z(\mathcal{Q}(\mathcal{G}))$) which extends to one of $\mathcal{Q}^c(\Gamma')$ -algebras $\varepsilon \colon \mathcal{Q}^c(\mathcal{G}') \twoheadrightarrow \mathcal{Q}^c(\mathcal{G})$.
- ii) Let $\chi \in \operatorname{Irr}_p(\mathcal{G}')$. Then

$$\varepsilon(e_{\chi}) = \begin{cases} e_{\overline{\chi}}, & \widetilde{H} \subseteq \ker(\chi) \\ 0, & otherwise, \end{cases}$$

where $\overline{\chi}$ denotes the projection of χ to \mathcal{G} and e_{χ} and $e_{\overline{\chi}}$ are defined as in proposition 2.4.3. Furthermore, the map ε from part i) restricts to an isomorphism $\mathcal{Q}^{c}(\mathcal{G}')e_{\chi} \cong \mathcal{Q}^{c}(\mathcal{G})e_{\overline{\chi}}$ whenever χ factors through \mathcal{G} as $\overline{\chi}$.

- iii) Let $\overline{\chi} \in \operatorname{Irr}_p(\mathcal{G})$ and set $\chi = \inf_{\mathcal{G}}^{\mathcal{G}'} \overline{\chi}$. Then $\varepsilon(\gamma_{\chi}) = \gamma_{\overline{\chi}}$ (cf. proposition 2.4.4) and ε induces a canonical isomorphism $\mathcal{Q}^c(\Gamma_{\chi}) \cong \mathcal{Q}^c(\Gamma_{\overline{\chi}})$.
- Proof. i) The augmentation sequence (0.1) associated to $\widetilde{H} \hookrightarrow \mathcal{G}' \twoheadrightarrow \mathcal{G}$ features the continuous surjection $\operatorname{aug}_{\widetilde{H}} \colon \Lambda(\mathcal{G}') \twoheadrightarrow \Lambda(\mathcal{G})$, which in this case restricts to an isomorphism $\Lambda(\Gamma') \cong \Lambda(\Gamma)$. Therefore, the $\Lambda(\Gamma')$ -algebra structure of $\Lambda(\mathcal{G}')$ is compatible with that of $\Lambda(\mathcal{G})$ as a $\Lambda(\Gamma)$ -algebra. Since $\mathcal{Q}(\mathcal{G}') = \mathcal{Q}(\Gamma') \otimes_{\Lambda(\Gamma')} \Lambda(\mathcal{G}')$ and $\mathcal{Q}(\mathcal{G}) = \mathcal{Q}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})$ by (2.6), the map $\varepsilon = \mathcal{Q}(\Gamma') \otimes_{\Lambda(\Gamma')} \operatorname{aug}_{\widetilde{H}}$ satisfies the first claim. The second one is then obtained by applying $\mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} -$.
- ii) This follows from the analogous result for group algebras over finite groups, since the primitive central idempotents of $\mathcal{Q}^{c}(\mathcal{G}')$ and $\mathcal{Q}^{c}(\mathcal{G})$ only depend on the restriction of characters to H. Recall that e_{χ} is defined as

$$e_{\chi} = \sum_{\substack{\eta \in \operatorname{Irr}_{\mathbb{Q}_p^c}(H')\\ \eta | \operatorname{res}_{H'}^{\mathcal{G}'} \chi}} e(\eta)$$

where $e(\eta)$ is the primitive central idempotent of $\mathbb{Q}_p^c[H']$ corresponding to η (cf. (0.14)). By (0.15), the projection ε (which restricts to $\mathbb{Q}_p^c[H'] \twoheadrightarrow \mathbb{Q}_p^c[H]$) sends $e(\eta)$ to $e(\overline{\eta})$ if η factors through H (as $\overline{\eta}$), and to 0 otherwise.

Suppose first that $\widetilde{H} \subseteq \ker(\chi)$ and let $\overline{\chi}$ be the projection of χ to \mathcal{G} . Then $\widetilde{H} \subseteq \ker(\operatorname{res}_{H'}^{\mathcal{G}'}\chi)$ and $\operatorname{res}_{H'}^{\mathcal{G}'}\chi = \operatorname{res}_{H'}^{\mathcal{G}'}\inf_{\mathcal{G}}^{\mathcal{G}'}\overline{\chi} = \inf_{H}^{H'}\operatorname{res}_{H}^{\mathcal{G}}\overline{\chi}$. In particular, an $\eta \in \operatorname{Irr}_{\mathbb{Q}_{p}^{c}}(H')$ divides $\operatorname{res}_{H'}^{\mathcal{G}'}\chi$ if and only if it factors through H and its projection $\overline{\eta}$ divides $\operatorname{res}_{H}^{\mathcal{G}}\overline{\chi}$. Here we are using the general fact that if a character factors through a quotient group, then all of its divisors do as well. Note also that a character of a quotient group is irreducible if and only if its inflation to the entire group is so. We therefore have:

$$\varepsilon(e_{\chi}) = \sum_{\substack{\eta \in \operatorname{Irr}_{\mathbb{Q}_{p}^{c}}(H')\\\eta \mid \operatorname{res}_{H'}^{\mathcal{G}'}\chi}} \varepsilon(e(\eta)) = \sum_{\substack{\eta \in \operatorname{Irr}_{\mathbb{Q}_{p}^{c}}(H')\\\eta \mid \operatorname{res}_{H'}^{\mathcal{G}'}\chi}} e(\overline{\eta}) = \sum_{\substack{\overline{\eta} \in \operatorname{Irr}_{\mathbb{Q}_{p}^{c}}(H)\\\eta \mid \operatorname{res}_{H'}^{\mathcal{G}}\chi}} \overline{\eta} \mid \operatorname{res}_{H}^{\mathcal{G}}\overline{\chi}}} \varepsilon(\overline{\eta}) = e_{\overline{\chi}}.$$

Conversely, assume that $\widetilde{H} \not\subseteq \ker(\chi)$ and hence $\widetilde{H} \not\subseteq \ker(\operatorname{res}_{H'}^{\mathcal{G}'}\chi)$. Then at least one irreducible constituent η of $\operatorname{res}_{H'}^{\mathcal{G}'}\chi$ satisfies $\widetilde{H} \not\subseteq \ker(\eta)$. But Clifford theory shows that all such divisors are conjugate to one another by elements of \mathcal{G}' , so \widetilde{H} cannot be contained in the kernel of any of them. Thus, one has

$$\varepsilon(e_{\chi}) = \sum_{\eta | \mathrm{res}_{H'}^{\mathcal{G}' \chi}} \varepsilon(e(\eta)) = 0,$$

which concludes the proof of the first claim.

Let now $\chi \in \operatorname{Irr}_p(\mathcal{G}')$ factor through \mathcal{G} as $\overline{\chi}$. By the above argument, ε restricts to a surjective ring homomorphism $\mathcal{Q}^c(\mathcal{G}')e_{\chi} \twoheadrightarrow \mathcal{Q}^c(\mathcal{G})e_{\overline{\chi}}$ (note that the unities of $\mathcal{Q}^c(\mathcal{G}')e_{\chi}$ and $\mathcal{Q}^c(\mathcal{G})e_{\overline{\chi}}$ are e_{χ} and $e_{\overline{\chi}}$ respectively). The kernel of a ring homomorphism is a two-sided ideal, but the only such ideals of $\mathcal{Q}^c(\mathcal{G}')e_{\chi}$ are $\{0\}$ and the entire ring by simplicity.

iii) Part ii) already implies that ε restricts to an isomorphism

$$\mathcal{Q}^{c}(\Gamma_{\chi}) = Z(\mathcal{Q}^{c}(\mathcal{G}')e_{\chi}) \cong Z(\mathcal{Q}^{c}(\mathcal{G})e_{\overline{\chi}}) = \mathcal{Q}^{c}(\Gamma_{\overline{\chi}})$$

by proposition 2.4.4. This does not a priori mean that Γ_{χ} is mapped to $\Gamma_{\overline{\chi}}$ (see for instance the proof of lemma 2.4.5), which we prove now. The generator γ_{χ} of $\Gamma_{\chi} \subseteq Z(\mathcal{Q}^c(\mathcal{G}')e_{\chi})^*$ is given by $\gamma_{\chi} = gc$ as in proposition 2.4.4 i). Here $g \in \mathcal{G}'$ projects to $\gamma_K^{w_{\chi}} \in \Gamma_K$, where w_{χ} coincides with the number of irreducible constituents η of $\operatorname{res}_{H'}^{\mathcal{G}'}\chi$. We have shown these correspond bijectively with the irreducible constituents $\overline{\eta}$ of $\operatorname{res}_{H}^{\mathcal{G}}\overline{\chi}$, and thus $w_{\chi} = w_{\overline{\chi}}$.

The element $\varepsilon(\gamma_{\chi}) = \varepsilon(g)\varepsilon(c) \in Z(\mathcal{Q}^{c}(\mathcal{G})e_{\overline{\chi}})$ satisfies the following:

- $\varepsilon(g)$ projects to $\gamma_K^{w_{\chi}} = \gamma_K^{w_{\overline{\chi}}}$ by the above.
- $\varepsilon(c) \in \varepsilon((\mathbb{Q}_p^c[H']e_{\chi})^*) = (\mathbb{Q}_p^c[H]e_{\overline{\chi}})^*$ (observe that ε restricts to a ring isomorphism $\mathbb{Q}_p^c[H']e_{\eta} \cong \mathbb{Q}_p^c[H]e_{\overline{\eta}}$ for all η which factor through H).
- $\varepsilon(\gamma_{\chi})$ acts trivially on $V_{\overline{\chi}}$ (the \mathbb{Q}_p^c -vector space afforded by $\overline{\chi}$), since so does γ_{χ} on V_{χ} .

By proposition 2.4.4, there exists only one $\gamma_{\overline{\chi}} \in Z(\mathcal{Q}^c(\mathcal{G})e_{\overline{\chi}})$ with those three properties, which must therefore coincide with $\varepsilon(\gamma_{\chi})$.

Remark 3.3.2. Although the elements γ_{χ} and $\gamma_{\overline{\chi}}$ depend on the choice of γ_K , the lemma shows that one always has $\varepsilon(\gamma_{\chi}) = \gamma_{\overline{\chi}}$ as long as the same γ_K is chosen in both cases. In fact, as already mentioned in remark 2.4.8, replacing γ_K by another generator γ_K^v (where $v \in \mathbb{Z}_p^*$ necessarily) results in the substitution of γ_{χ} by γ_{χ}^v and $\gamma_{\overline{\chi}}$ by $\gamma_{\overline{\chi}}^v$. In particular, $\gamma_{\chi} \mapsto \gamma_{\overline{\chi}}$ and $\gamma_{\chi}^v \mapsto \gamma_{\overline{\chi}}^v$ define the same map on $\mathcal{Q}^c(\Gamma_{\chi})$: the canonical isomorphism induced by ε .

The above result sheds light on the structural relation between $\mathcal{Q}^c(\mathcal{G}')$ and $\mathcal{Q}^c(\mathcal{G})$. If χ and ρ are irreducible Artin characters of \mathcal{G}' with ρ of type W, then $\chi \otimes \rho$ factors through \mathcal{G} if and only if χ does. Furthermore, characters of \mathcal{G}' which factor through \mathcal{G} are W-equivalent (cf. section 2.4) if and only if their projections to \mathcal{G} are. This allows one to interpret the lemma as follows: for a chosen

set of representatives $\Xi_{\mathcal{G}}$ of $\operatorname{Irr}_{p}(\mathcal{G})/\sim_{W}$, set $\Xi_{\mathcal{G}'} = {\inf_{\mathcal{G}}^{\mathcal{G}'} \overline{\chi} : \overline{\chi} \in \Xi_{\mathcal{G}}}$. Let $\widetilde{\Xi}_{\mathcal{G}'}$ be an arbitrary set of \sim_{W} -representatives of the irreducible Artin characters of \mathcal{G}' which do not factor through \mathcal{G} (well defined by the above argument). Then the diagram

$$\begin{array}{cccc} \mathcal{Q}^{c}(\mathcal{G}') & \stackrel{\sim}{\longrightarrow} & \prod_{\chi \in \widetilde{\Xi}_{\mathcal{G}'}} \mathcal{Q}^{c}(\mathcal{G}')e_{\chi} & \times & \prod_{\chi \in \Xi_{\mathcal{G}'}} \mathcal{Q}^{c}(\mathcal{G}')e_{\chi} & \longleftrightarrow & \prod_{\chi \in \Xi_{\mathcal{G}'}} \mathcal{Q}^{c}(\Gamma_{\chi}) \\ & \downarrow^{\varepsilon} & & & \downarrow^{\gamma_{\chi} \mapsto \gamma_{\overline{\chi}}} \\ & \mathcal{Q}^{c}(\mathcal{G}) & \stackrel{\sim}{\longrightarrow} & \prod_{\overline{\chi} \in \Xi_{\mathcal{G}}} \mathcal{Q}^{c}(\mathcal{G})e_{\overline{\chi}} & \longleftrightarrow & \prod_{\overline{\chi} \in \Xi_{\mathcal{G}}} \mathcal{Q}^{c}(\Gamma_{\overline{\chi}}) \end{array}$$

commutes, where the first map in each row is (2.22). Although this decomposition only exists after extending scalars to a large enough p-adic field, the element

$$e_{\widetilde{H}} = \sum_{\chi \in \Xi_{\mathcal{G}'}} e_{\chi} = \frac{1}{|\widetilde{H}|} N_{\widetilde{H}}$$

(where $N_{\widetilde{H}} = \sum_{\sigma \in \widetilde{H}} \sigma \in \mathbb{Z}[\widetilde{H}] \subseteq \mathbb{Z}[H']$ is the norm²⁰ of \widetilde{H}) is a central idempotent of $\mathcal{Q}(\mathcal{G}')$ which induces a decomposition

$$\mathcal{Q}(\mathcal{G}') \cong \mathcal{Q}(\mathcal{G}')e_{\widetilde{H}} \times \mathcal{Q}(\mathcal{G}')(1 - e_{\widetilde{H}})$$
(3.38)

of $\mathcal{Q}(\mathcal{G})$ as a product of rings. The restriction of ε to the first factor yields an isomorphism $\mathcal{Q}(\mathcal{G}')e_{\widetilde{H}} \cong \mathcal{Q}(\mathcal{G}).$

So far we have only addressed the elements in setting A. The two other necessary parameters for the Main Conjecture are the α and β from setting B, the second of which is manifestly unrelated to the rest. As for α , we can use the one choice for L'_{∞}/K to construct its counterpart for L_{∞}/K in a way which will make the Main Conjectures easy to compare:

Lemma 3.3.3. Setting C. The following hold:

i) Let $\mathcal{Y}_{S_{\infty}}$ and $E_{S,T}$ be the usual modules for the extension L_{∞}/K (cf. section 1.4), and $\mathcal{Y}'_{S_{\infty}}$ and $E'_{S,T}$ their analogues for L'_{∞}/K , that is,

$$\mathcal{Y}_{S_{\infty}} = \bigoplus_{v \in S_{\infty}} \operatorname{Ind}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathbb{Z}_{p}, \quad \mathcal{Y}_{S_{\infty}}' = \bigoplus_{v \in S_{\infty}} \operatorname{Ind}_{\mathcal{G}_{v}'}^{\mathcal{G}'} \mathbb{Z}_{p}$$

and

$$E_{S,T} = \varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}_{L_{n},S,T}^{*}, \quad E_{S,T}' = \varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}_{L_{n}',S,T}^{*}$$

Suppose $\alpha' \colon \mathcal{Y}'_{S_{\infty}} \hookrightarrow E'_{S,T}$ is an injective $\Lambda(\mathcal{G}')$ -homomorphism with torsion cokernel. Then the map

$$\alpha \colon \mathcal{Y}_{S_{\infty}} \xrightarrow{\sim} (\mathcal{Y}'_{S_{\infty}})_{\widetilde{H}} \xrightarrow{\alpha'_{\widetilde{H}}} (E'_{S,T})_{\widetilde{H}} \xrightarrow{N_{\widetilde{H}}} (E'_{S,T})^{\widetilde{H}} = E_{S,T}$$

is an injective $\Lambda(\mathcal{G})$ -homomorphism with torsion cokernel.

²⁰In this section, we will refer to the norm N_G of a finite group G, rather than its trace Tr_G - but the distinction is purely conventional. What is meant by this is the element $N_G = \sum_{\sigma \in G} \sigma \in \mathbb{Z}[G]$ of any ring where $\mathbb{Z}[G]$ can be naturally embedded. If m is an element of a G-module (that is, a $\mathbb{Z}[G]$ -module) M, then $N_G m$ is either $\prod_{\sigma \in G} (\sigma m)$ or $\sum_{\sigma \in G} (\sigma m)$ depending on whether M uses multiplicative or additive notation.

ii) Let $\overline{\chi} \in \operatorname{Irr}_p(\mathcal{G})$ and denote its inflation to \mathcal{G}' by χ . If α' and α are as in part i) and $\beta \colon \mathbb{C}_p \to \mathbb{C}$ is a field isomorphism, the Stark-Tate regulators

$$R_S^\beta(\alpha',\chi) = R_S^\beta(\alpha,\overline{\chi})$$

coincide.

Proof. i) We first verify some implicit claims in the definition of α . The canonical isomorphism $\mathcal{Y}_{S_{\infty}} \cong (\mathcal{Y}'_{S_{\infty}})_{\widetilde{H}}$ is proved in the exact same way as the first part of proposition 2.2.3. The norm of \widetilde{H} defines a homomorphism from coinvariants to invariants of any \widetilde{H} module, and in particular $N_{\widetilde{H}} \colon (E'_{S,T})_{\widetilde{H}} \to (E'_{S,T})^{\widetilde{H}}$. This map is Galois-equivariant and hence a $\Lambda(\mathcal{G})$ -homomorphism too. Lastly, one has

$$(E'_{S,T})^{\widetilde{H}} = (\varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}^{*}_{L'_{n},S,T})^{\widetilde{H}} = \varprojlim_{n} \mathbb{Z}_{p} \otimes (\mathcal{O}^{*}_{L'_{n},S,T})^{\widetilde{H}} = \varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}^{*}_{L_{n},S,T} = E_{S,T}.$$

It remains to show that α is injective and has torsion cokernel. The Tate cohomology groups

$$\hat{H}_0(\widetilde{H}, E'_{S,T}) = \ker(N_{\widetilde{H}}) \text{ and } \hat{H}^0(\widetilde{H}, E'_{S,T}) = \operatorname{coker}(N_{\widetilde{H}}),$$

are $|\widetilde{H}|$ -torsion (cf. [NSW20] propositions 1.2.6 and 1.6.1) and hence $\Lambda(\mathcal{G})$ -torsion. As for $\alpha'_{\widetilde{H}}$, we consider the usual long exact cohomology sequence

$$\dots \to H_1(\widetilde{H}, \operatorname{coker}(\alpha')) \to (\mathcal{Y}'_{S_{\infty}})_{\widetilde{H}} \xrightarrow{\alpha'_{\widetilde{H}}} (E'_{S,T})_{\widetilde{H}} \to \operatorname{coker}(\alpha')_{\widetilde{H}} \to 0$$

associated to $\mathcal{Y}'_{S_{\infty}} \xrightarrow{\alpha'} E'_{S,T} \twoheadrightarrow \operatorname{coker}(\alpha')$. The term $H_1(\widetilde{H}, \operatorname{coker}(\alpha')) = \hat{H}_1(\widetilde{H}, \operatorname{coker}(\alpha'))$, which surjects onto $\operatorname{ker}(\alpha'_{\widetilde{H}})$, is $|\widetilde{H}|$ -torsion for the same reasons as above. The cokernel of $\alpha'_{\widetilde{H}}$ coincides with $\operatorname{coker}(\alpha')_{\widetilde{H}}$, which is $\Lambda(\mathcal{G})$ -torsion (i.e. $\Lambda(\Gamma)$ -torsion) by assumption. It follows that α itself has torsion kernel and cokernel, which implies injectivity by proposition 2.0.1.

ii) The diagram

commutes because, by definition, the norm map $N_{\tilde{H}}$ on coinvariants is given by first choosing a lift to the original module (and it is independent of that choice). The composition of the bottom row is precisely α .

Let $n \in \mathbb{N}$ be larger than n(S) and n(S'), and large enough that $\overline{\chi}$ factors through \mathcal{G}_n (in particular, χ factors through \mathcal{G}'_n). Recall that the projection $\Lambda(\mathcal{G}') \to \Lambda(\mathcal{G})$ restricts to an isomorphism $\Lambda(\Gamma') \cong \Lambda(\Gamma)$. Therefore, taking $(\Gamma')^{p^n}$ - and Γ^{p^n} -coinvariants on the previous diagram yields a commutative square which fits into

The last arrow of the bottom row is given by

$$\mathbb{Z}_p \otimes \mathcal{O}_{L_n,S,T}^* \hookrightarrow \mathbb{C}_p \otimes \mathcal{O}_{L_n,S,T}^* \xrightarrow{\lambda_{n,S}^\beta} \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S},$$

where $\lambda_{n,S}^{\beta}$ denotes the *p*-adic Dirichlet regulator map (cf. definition 2.3.3), and analogously for the arrow above it. The commutativity of the rightmost square boils down to the general fact that $|N_{L'_n/L_n}(u)|_v = \prod_{w|v} |u|_w$ in the obvious notation. The remaining squares are manifestly commutative.

We now apply the functor $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}'_n]}(V_{\chi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} -)$ to the above diagram except for the last column, where we apply $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}'_n]}(V_{\chi}, -)$ instead. On the bottom row, these functors coincide with $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\overline{\chi}}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} -)$ and $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\overline{\chi}}, -)$, respectively. Both the first and last vertical arrows of the resulting diagram are the \mathbb{C}_p -linear homomorphism

$$\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}'_n]}(V_{\chi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L'_n, S}) \twoheadrightarrow \operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\overline{\chi}}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n, S})$$

(note that $\mathbb{C}_p[\mathcal{G}'_n]$ is semisimple), which is in fact an isomorphism by the same argument used at the beginning of the proof of lemma 2.3.5. This shows that the \mathbb{C}_p -determinants of the two rows of the new diagram coincide - but these are precisely the regulators $R_S^\beta(\alpha', \chi)$ and $R_S^\beta(\alpha, \overline{\chi})$.

The equality of regulators in the previous lemma covers the analytic side of the functoriality under change of L_{∞} . On the homological side, we will resort to the good behaviour of refined Euler characteristics with respect to extension of scalars (lemma A.0.1). It is therefore necessary to show that this is precisely the relation between the complexes constructed from L'_{∞}/K and L_{∞}/K . The key result to do so is due to Fukaya and Kato [FK06].

The following piece of notation will be useful: if $\Lambda(G)$ is the usual Iwasawa algebra of a profinite group G, we denote by $\iota: \Lambda(G) \to \Lambda(G)$ the continuous isomorphism of \mathbb{Z}_p -modules which sends an element $\sigma \in G \subseteq \Lambda(G)$ to σ^{-1} . Given a topological right $\Lambda(G)$ -module M, we denote by M^{ι} the topological left $\Lambda(G)$ -module consisting of the topological abelian group M endowed with the $\Lambda(G)$ -action

$$\lambda \cdot^{\iota} m = m \cdot \iota(\lambda) \tag{3.39}$$

for $m \in M$, $\lambda \in \Lambda(G)$.

Proposition 3.3.4. Setting C. Let $(\mathcal{C}'_{S,T})^{\bullet}$ and $\mathcal{C}^{\bullet}_{S,T}$ be the complexes given by definition 1.4.1 for L'_{∞}/K and L_{∞}/K , respectively. Denote by $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}}$ – the derived extension of scalars induced

by $\Lambda(\mathcal{G}') \twoheadrightarrow \Lambda(\mathcal{G})$. Then

$$\mathcal{C}^{ullet}_{S,T} \cong \Lambda(\mathcal{G}) \otimes^{\mathbb{L}}_{\Lambda(\mathcal{G}')} (\mathcal{C}'_{S,T})$$

in $\mathcal{D}(\Lambda(\mathcal{G}))$.

an

Proof. Consistently with previous notation, let M_S denote the maximal S-ramified extension of K, which coincides with that of L'_{∞} and any intermediate field. Set $G_S = \operatorname{Gal}(M_S/K)$ and $H_{L_n,S} = \operatorname{Gal}(M_S/L_n)$ for $n \in \mathbb{N}$, and analogously for $H_{L'_n,S}$. Note that $H_{L_n,S}$ and $H_{L'_n,S}$ are open and normal in G_S for all n.

By the same argument as in proposition 3.2.5, the complex $(\mathcal{C}'_{S,T})^{\bullet}$ lies in an exact triangle

$$(\mathcal{C}'_{S,T})^{\bullet} \to (\mathcal{C}'_{S,\varnothing})^{\bullet} \to \bigoplus_{v \in T} \overline{\operatorname{Ind}}_{\mathcal{G}'_{v}}^{\mathcal{G}'} \mathbb{Z}_{p}(1)[0] \to .$$

in the notation introduced therein. Since derived functors are triangulated, extension of scalars yields

$$\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} (\mathcal{C}'_{S,T})^{\bullet} \to \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} (\mathcal{C}'_{S,\varnothing})^{\bullet} \to \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} \bigoplus_{v \in T} \overline{\operatorname{Ind}}_{\mathcal{G}'_{v}}^{\mathcal{G}'} \mathbb{Z}_{p}(1)[0] \to$$
(3.40)

in $\mathcal{D}(\Lambda(\mathcal{G}))$. We consider the middle term first. As shown in (1.19), $(\mathcal{C}'_{S,\emptyset})^{\bullet}$ is isomorphic in $\mathcal{D}(\Lambda(\mathcal{G}'))$ to $\varprojlim_n R\Gamma(H_{L'_n,S}, \mathbb{Z}_p(1))[1]$. By Shapiro's lemma (see for instance [Lim12] lemma 5.2.3), we have

$$R\Gamma(H_{L'_n,S},\mathbb{Z}_p(1)) \cong R\Gamma(G_S,\mathbb{Z}_p[G_S/H_{L'_n,S}]^{\iota} \otimes_{\mathbb{Z}_p} \mathbb{Z}_p(1))$$

where $\mathbb{Z}_p[G_S/H_{L'_n,S}]$ is regarded as a right G_S -module so that $\mathbb{Z}_p[G_S/H_{L'_n,S}]^{\iota}$ is a left G_S -module. This endows $\mathbb{Z}_p[G_S/H_{L'_n,S}]^{\iota} \otimes_{\mathbb{Z}_p} \mathbb{Z}_p(1)$ with a left $\Lambda(G_S)$ -module structure via

$$\sigma(\lambda \otimes z) = (\sigma \cdot^{\iota} \lambda) \otimes \sigma z = (\lambda \sigma^{-1}) \otimes \sigma z$$

for $\sigma \in G_S$, under which it is clearly isomorphic to $\mathbb{Z}_p[\mathcal{G}'_n]^{\iota}(1)$. Therefore, one has

$$(\mathcal{C}'_{S,\varnothing})^{\bullet} \cong \varprojlim_{n} R\Gamma(G_{S}, \mathbb{Z}_{p}[\mathcal{G}'_{n}]^{\iota}(1))[1] \cong R\Gamma(G_{S}, \varprojlim_{n} \mathbb{Z}_{p}[\mathcal{G}'_{n}]^{\iota}(1))[1] = R\Gamma(G_{S}, \Lambda(\mathcal{G}')^{\iota}(1))[1]$$
(3.41)

in $\mathcal{D}(\Lambda(\mathcal{G}'))$, where the second isomorphism is [Lim12] lemma 5.2.3. By the same token, $\mathcal{C}^{\bullet}_{S,\emptyset}$ is isomorphic to $R\Gamma(G_S, \Lambda(\mathcal{G})^{\iota}(1))[1]$ in $\mathcal{D}(\Lambda(\mathcal{G}))$.

We now apply proposition 1.6.5 from [FK06] (case 1). The ring $\Lambda = \Lambda(\mathcal{G}')$ satisfies the necessary hypotheses by 1.4.1 and 1.4.2 therein, and $T = \Lambda(\mathcal{G}')^{\iota}(1)$ is a free Λ -module with a compatible continuous $G = G_S$ -action as above. The reason G satisfies condition (ii) in the reference was explained at the beginning of the proof of proposition 1.4.3. Condition (i) can be verified with some work using [NSW20] propositions 1.6.7, 8.6.10 and 10.11.3. The upshot is that

$$\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} R\Gamma(G_S, \Lambda(\mathcal{G}')^{\iota}(1))[1] \cong R\Gamma(G_S, \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} \Lambda(\mathcal{G}')^{\iota}(1))[1] \cong R\Gamma(G_S, \Lambda(\mathcal{G})^{\iota}(1))[1],$$

d thus $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} (\mathcal{C}'_{S,\varnothing})^{\bullet} \cong \mathcal{C}^{\bullet}_{S,\varnothing}.$

Let us now address the last term in (3.40). By the additivity of $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} -$, it suffices to determine $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} \overline{\operatorname{Ind}}_{\mathcal{G}'_{v}}^{\mathcal{G}'} \mathbb{Z}_{p}(1)[0]$ for $v \in T$. The strictly perfect representative of $\overline{\operatorname{Ind}}_{\mathcal{G}'_{v}}^{\mathcal{G}'} \mathbb{Z}_{p}(1)[0]$ obtained in lemma 3.2.11 doubles as a projective resolution to compute the derived tensor product:

$$\begin{split} \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} \overline{\operatorname{Ind}}_{\mathcal{G}'_{v}}^{\mathcal{G}'} \mathbb{Z}_{p}(1)[0] &\cong [\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{-1} \Lambda(\mathcal{G}') \xrightarrow{\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} (1 - \mathfrak{N}(v_{0})\varphi_{v_{0}(L'_{\infty})}^{-1})_{r}} \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{0} \Lambda(\mathcal{G}')] \\ &\cong [\Lambda(\mathcal{G}) \xrightarrow{(1 - \mathfrak{N}(v_{0})\varphi_{v_{0}(L_{\infty})}^{-1})_{r}} \Lambda(\mathcal{G})] \\ &\cong \overline{\operatorname{Ind}}_{\mathcal{G}_{v}}^{\mathcal{G}} \mathbb{Z}_{p}(1)[0]. \end{split}$$

The result now follows from the diagram of exact triangles



where the second square commutes because both horizontal arrows come from the same maps on the level of Galois modules.

The same technique allows us to prove a fact which was necessary for the construction of the finitelevel maps φ_n^{α} in section 2.2, but whose proof we deferred to a later time. It concerns passage to finite level in setting A, rather than the change of groups $\mathcal{G}' \twoheadrightarrow \mathcal{G}$ from setting C - the choice of placement here is motivated by the close similarity to the previous argument. Note that none of the objects involved in the above proposition rely on φ_n^{α} , and hence no circular dependencies arise.

Lemma 3.3.5. Setting A. For all $n \in \mathbb{N}$, one has

$$\mathcal{B}^{\bullet}_{L_n,S,T} \cong \Lambda(\mathcal{G}_n) \otimes^{\mathbb{L}}_{\Lambda(\mathcal{G})} \mathcal{C}^{\bullet}_{S,T}$$

in the derived category $\mathcal{D}(\Lambda(\mathcal{G}_n))$, where $\mathcal{B}^{\bullet}_{L_n,S,T}$ is defined as in 1.5.1.

Proof. It suffices to prove the result for $T = \emptyset$ for the same reasons as above. We again have

$$\Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{C}_{S,\varnothing}^{\bullet} \cong R\Gamma(G_S, \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})} \Lambda(\mathcal{G})^{\iota}(1))[1] \cong R\Gamma(H_{L_n,S}, \mathbb{Z}_p(1))[1]$$

in $\mathcal{D}(\Lambda(\mathcal{G}_n))$ by [FK06] proposition 1.6.5 (whose hypotheses are still satisfied) and Shapiro's lemma. Since $R\Gamma(H_{L_n,S}, \mathbb{Z}_p(1))[1]$ is isomorphic to $\mathcal{B}^{\bullet}_{L_n,S,T}$ by (1.18), we are done.

We can now prove the first functoriality result:

Proposition 3.3.6. Setting C. Let α' and α be as in lemma 3.3.3 and β as in setting B. Then:

- i) Given $\overline{\chi} \in \operatorname{Irr}_p(\mathcal{G})$, set $\chi = \inf_{\mathcal{G}}^{\mathcal{G}'} \overline{\chi} \in \operatorname{Irr}_p(\mathcal{G}')$. Then $IC(L'_{\infty}/K, \chi, L', S, T, \alpha', \beta)$ holds if and only if $IC(L_{\infty}/K, \overline{\chi}, L, S, T, \alpha, \beta)$ does.
- ii) $eMC(L'_{\infty}/K, L', S, T, \alpha', \beta)$ implies $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$.
- iii) $eMC^{u}(L'_{\infty}/K, L', S, T, \alpha', \beta)$ implies $eMC^{u}(L_{\infty}/K, L, S, T, \alpha, \beta)$.

Proof. Suppose $F_{S,T,\chi}^{\alpha',\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L'_{\infty}/K, \chi, L', S, T, \alpha', \beta)$ and let $F = \varepsilon(F_{S,T,\chi}^{\alpha',\beta}) \in \mathcal{Q}^c(\Gamma_{\overline{\chi}})^*$ (cf. lemma 3.3.1). Recall that we identify the type-W characters of \mathcal{G} and \mathcal{G}' by virtue of (3.37). Thus, for almost all $\rho \in \mathcal{K}_S^{\alpha'}(\chi) \cap \mathcal{K}_S^{\alpha}(\overline{\chi})$, one has

$$ev_{\gamma_{\overline{\chi}\otimes\rho}}(F) = ev_{\gamma_{\chi\otimes\rho}}(F_{S,T,\chi}^{\alpha',\beta}) = \frac{\beta^{-1}(L_{K,S,T}^*(\beta(\check{\chi}\otimes\rho^{-1}),0))}{R_S^\beta(\alpha',\chi\otimes\rho)}$$
$$= \frac{\beta^{-1}(L_{K,S,T}^*(\beta((\overline{\chi})\otimes\rho^{-1}),0))}{R_S^\beta(\alpha',\chi\otimes\rho)}$$
$$= \frac{\beta^{-1}(L_{K,S,T}^*(\beta((\overline{\chi})\otimes\rho^{-1}),0))}{R_S^\beta(\alpha,\overline{\chi}\otimes\rho)},$$

where the equalities are, in order: lemmas 3.3.1 iii) and 2.4.5; the interpolation property of $F_{S,T,\chi}^{\alpha',\beta}$; lemma 0.3.1 ii); and lemma 3.3.3 ii). This shows F satisfies $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Since ε is an isomorphism on χ -parts by lemma 3.3.1 ii), the same argument shows the converse.

In order to prove ii), assume $\operatorname{eMC}(L'_{\infty}/K, L', S, T, \alpha', \beta)$ holds and let $\zeta_{S,T}^{\alpha',\beta} \in K_1(\mathcal{Q}(\mathcal{G}'))$ be the element predicted therein. In particular, $\operatorname{IC}(L'_{\infty}/K, \chi, L', S, T, \alpha', \beta)$ holds for all $\chi \in \operatorname{Irr}_p(\mathcal{G}')$ and therefore so does $\operatorname{IC}(L_{\infty}/K, \overline{\chi}, L, S, T, \alpha, \beta)$ for all $\overline{\chi} \in \operatorname{Irr}_p(\mathcal{G})$ by i). This uniquely determines interpolating elements $F_{S,T,\chi}^{\alpha',\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ and $F_{S,T,\overline{\chi}}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\overline{\chi}})^*$.

Let $\varepsilon \colon \mathcal{Q}(\mathcal{G}') \twoheadrightarrow \mathcal{Q}(\mathcal{G})$ be the projection map from lemma 3.3.1. Then the diagram

commutes for any $\overline{\chi}$ as above and $\chi = \inf_{\mathcal{G}}^{\mathcal{G}'} \overline{\chi}$. Part i) now yields

$$\psi_{\overline{\chi}}(K_1(\varepsilon)(\zeta_{S,T}^{\alpha,\beta})) = \varepsilon(\psi_{\chi}(\zeta_{S,T}^{\alpha',\beta})) = F_{S,T,\overline{\chi}}^{\alpha,\beta},$$

and hence the element $\zeta_{S,T}^{\alpha,\beta} = K_1(\varepsilon)(\zeta_{S,T}^{\alpha',\beta})$ satisfies the analytic part of $\operatorname{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$. On the homological side, let $(\mathcal{C}'_{S,T})^{\bullet}$ and $\mathcal{C}_{S,T}^{\bullet}$ be the main complexes constructed from L'_{∞}/K and L_{∞}/K , respectively. The maps $R = \Lambda(\mathcal{G}') \xrightarrow{\rho=\varepsilon} \Lambda(\mathcal{G}) = R'$ and $S = \mathcal{Q}(\mathcal{G}') \xrightarrow{\sigma=\varepsilon} \mathcal{Q}(\mathcal{G}) = S'$ constitute an example of setup (0.29) together with the canonical inclusions $\Lambda(\mathcal{G}') \hookrightarrow \mathcal{Q}(\mathcal{G}')$ and $\Lambda(\mathcal{G}) \hookrightarrow \mathcal{Q}(\mathcal{G})$. As a consequence, diagram (0.30) implies

$$\partial(\zeta_{S,T}^{\alpha,\beta}) = K_0(\varepsilon,\varepsilon)(\partial(\zeta_{S,T}^{\alpha',\beta})) = K_0(\varepsilon,\varepsilon)(-\chi_{\Lambda(\mathcal{G}'),\mathcal{Q}(\mathcal{G}')}((\mathcal{C}'_{S,T})^{\bullet}, t^{\alpha'})) \in K_0(\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G}))$$

By lemma A.0.1 i), the the last term coincides with $-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} (\mathcal{C}'_{S,T})^{\bullet}, t')$ for a certain trivialisation t'. Proposition 3.3.4 provides an isomorphism $\varphi \colon \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')}^{\mathbb{L}} (\mathcal{C}'_{S,T})^{\bullet} \cong \mathcal{C}_{S,T}^{\bullet}$ in $\mathcal{D}(\Lambda(\mathcal{G}))$, and thus it only remains to show that t' becomes t^{α} under φ . To see this, recall (2.7) and consider the commutative diagram

where the first vertical arrow is induced by φ (note that $\Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} -$ is right exact) and $\mathcal{X}'_S, \mathcal{Y}'_{S_{\infty}}$ and $E'_{S,T}$ are the L'_{∞}/K -analogues of their L_{∞}/K counterparts. Apply now $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} -$ to both rows. The composition of the bottom row then becomes the trivialisation t^{α} by definition. On the top row, we use the fact that $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} (-)_{\widetilde{H}} = \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} \Lambda(\mathcal{G}) \otimes_{\Lambda(\mathcal{G}')} -$ is the same functor as (or more formally, canonically naturally isomorphic to) $\mathcal{Q}(\mathcal{G}) \otimes_{\mathcal{Q}(\mathcal{G}')} \mathcal{Q}(\mathcal{G}') \otimes_{\Lambda(\mathcal{G}')} -$, which turns that row into $\mathcal{Q}(\mathcal{G}) \otimes_{\mathcal{Q}(\mathcal{G}')} t^{\alpha'}$. But this is precisely the trivialisation t' constructed from $t = t^{\alpha'}$ in lemma A.0.1 i), which concludes the proof of ii).

Lastly, we address iii). By part ii) and remark 2.5.4 ii), it suffices to show that the injectivity of $K_1(\mathcal{Q}(\mathcal{G}')) \xrightarrow{\operatorname{nr}} Z(\mathcal{Q}(\mathcal{G}'))^*$ implies that of $K_1(\mathcal{Q}(\mathcal{G})) \xrightarrow{\operatorname{nr}} Z(\mathcal{Q}(\mathcal{G}))^*$. This follows immediately from the decomposition (3.38) and the compatibility of K_1 and the reduced norm with products:

$$\begin{array}{cccc} K_{1}(\mathcal{Q}(\mathcal{G}')) & \stackrel{\sim}{\longrightarrow} & K_{1}(\mathcal{Q}(\mathcal{G}')e_{\widetilde{H}}) \oplus K_{1}(\mathcal{Q}(\mathcal{G}')(1-e_{\widetilde{H}})) & \stackrel{\sim}{\longrightarrow} & K_{1}(\mathcal{Q}(\mathcal{G})) \oplus K_{1}(\mathcal{Q}(\mathcal{G}')(1-e_{\widetilde{H}})) \\ & & \downarrow^{\mathrm{nr}} & & \downarrow^{\mathrm{nr}\oplus\mathrm{nr}} \\ & & & I^{\mathrm{nr}\oplus\mathrm{nr}} & & \downarrow^{\mathrm{nr}\oplus\mathrm{nr}} \\ & & & Z(\mathcal{Q}(\mathcal{G}'))^{*} & \stackrel{\sim}{\longrightarrow} & Z(\mathcal{Q}(\mathcal{G}')e_{\widetilde{H}})^{*} \oplus & Z(\mathcal{Q}(\mathcal{G}')(1-e_{\widetilde{H}}))^{*} & \stackrel{\sim}{\longrightarrow} & Z(\mathcal{Q}(\mathcal{G}))^{*} \oplus & Z(\mathcal{Q}(\mathcal{G}')(1-e_{\widetilde{H}}))^{*} \end{array}$$

where the last horizontal arrow of each horizontal arrow is induced by functoriality on ε .

The decomposition $\mathcal{Q}(\mathcal{G}') \cong \mathcal{Q}(\mathcal{G}) \times \mathcal{Q}(\mathcal{G}')(1 - e_{\widetilde{H}})$ illustrates why we do not have equivalence of the Main Conjectures for L'_{∞}/K and L_{∞}/K , but rather implication in one direction. Any potential zeta element $\zeta_{S,T}^{\alpha,\beta}$ for L_{∞}/K has no information about the part of $\mathcal{Q}(\mathcal{G}')$ orthogonal to $e_{\widetilde{H}}$.

3.3.2 Change of K

We now address the situation of a tower of extensions $L_{\infty}/K'/K$, where K' and K are number fields and over both of which L_{∞} satisfies our usual hypotheses. This is equivalent to considering L_{∞}/K as in setting A and choosing an open (not necessarily normal) subgroup \mathcal{G}' of $\mathcal{G} = \operatorname{Gal}(L_{\infty}/K)$. We summarise the new situation as follows:

Setting D. The objects p, $L_{\infty}/K_{\infty}/K$, S and T are fixed as in setting A. We also denote $\Gamma_K = \text{Gal}(K_{\infty}/K) = \overline{\langle \gamma_K \rangle}$ and $H = \text{Gal}(L_{\infty}/K_{\infty}) \leq_c \mathcal{G} = \text{Gal}(L_{\infty}/K)$. Additionally, we consider:

• \mathcal{G}' , an arbitrary open subgroup of \mathcal{G} . Let $K' = L_{\infty}^{\mathcal{G}'}$, which is a finite (but not necessarily Galois) extension of K. Then L_{∞} is a Galois extension of the cyclotomic \mathbb{Z}_p -extension K'_{∞} of K' and $H' = \operatorname{Gal}(L_{\infty}/K'_{\infty}) = \mathcal{G}' \cap H$. Set $\Gamma_{K'} = \operatorname{Gal}(K'_{\infty}/K)$.

We now fix an open central subgroup $\widetilde{\Gamma} \leq_o \mathcal{G}$ isomorphic to \mathbb{Z}_p as in setting A and let $\Gamma = \widetilde{\Gamma} \cap \mathcal{G}'$. In particular, Γ is open and central in both \mathcal{G} and \mathcal{G}' . As usual, we set $L = L_{\infty}^{\Gamma}$ and define $L_n = L_{\infty}^{\Gamma^{p^n}}, \mathcal{G}_n = \operatorname{Gal}(L_n/K) = \mathcal{G}/\Gamma^{p^n}, \mathcal{G}'_n = \operatorname{Gal}(L_n/K') = \mathcal{G}'/\Gamma^{p^n} \leq \mathcal{G}_n$.

The restriction $\Gamma_{K'} \hookrightarrow \Gamma_K$ is injective (because $K'_{\infty} = K_{\infty}K'$) and its image is an open subgroup $\Gamma_K^{p^M} \leq_o \Gamma_K$. We define the topological generator $\gamma_{K'}$ of $\Gamma_{K'}$ as the only preimage of $\gamma_K^{p^M}$ under the restriction map.

• For each place w of K', a distinguished prolongation w^c to $(K')^c = \mathbb{Q}^c$. Then, for each place v of K, we choose a place v^c among $\{w^c : w \in \{v\}(K')\}$. This is only tangentially relevant, namely for the isomorphisms in (3.42) below.

With the above conventions, $(L_{\infty}/K, \Gamma, S, T)$ and $(L_{\infty}/K', \Gamma, S(K'), T(K'))$ are two valid choices of the corresponding parameters in setting A. The relations between the relevant fields and Galois groups are depicted in the diagram



The restriction of scalars induced by the embedding of topological rings $\Lambda(\mathcal{G}') \hookrightarrow \Lambda(\mathcal{G})$ allows the parameter α from setting B to remain unchanged when passing from K to K':

• The $\Lambda(\mathcal{G}')$ -module structure of

$$E_{S(K'),T(K')} = \varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}_{L_{n},S(K'),T(K)}^{*} = \varprojlim_{n} \mathbb{Z}_{p} \otimes \mathcal{O}_{L_{n},S,T}^{*} = E_{S,T}$$

is simply the restriction of its $\Lambda(\mathcal{G})$ -structure.

• There is an isomorphism of $\Lambda(\mathcal{G}')$ -modules

$$\mathcal{Y}_{S_{\infty}} = \varprojlim_{n} \mathcal{Y}_{S_{\infty},L_{n}} \cong \varprojlim_{n} \bigoplus_{w_{n} \in S_{\infty}(L_{n})} \mathbb{Z}_{p} \cdot w_{n}$$
(3.42)

$$= \varprojlim_{n} \bigoplus_{w_n \in S_{\infty}(K')(L_n)} \mathbb{Z}_p \cdot w_n \cong \varprojlim_{n} \mathcal{Y}_{S_{\infty}(K'),L_n} = \mathcal{Y}_{S_{\infty}(K')}.$$
(3.43)

Therefore, any $\Lambda(\mathcal{G})$ -homomorphism $\alpha \colon \mathcal{Y}_{S_{\infty}} \to E_{S,T}$ as in setting B constitutes a valid $\Lambda(\mathcal{G}')$ homomorphism $E_{S(K'),T(K')} \to \mathcal{Y}_{S_{\infty}(K')}$. Note that conditions on $\Lambda(\Gamma)$ -torsionness also remain
unchanged, as the same Γ has been chosen in \mathcal{G}' and \mathcal{G} . As usual, the isomorphism $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ is
independent of all other parameters.

A few words are in order regarding the relation between Artin characters of \mathcal{G} and those of \mathcal{G}' . We will repeatedly write expressions like $\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi'$ for $\chi' \in \operatorname{Irr}_p(\mathcal{G}')$ even though induction was only introduced for characters of finite groups in section 0.2. This simply refers to the character

$$\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi' = \operatorname{inf}_{\mathcal{G}_n}^{\mathcal{G}}\operatorname{ind}_{\mathcal{G}'_n}^{\mathcal{G}_n}\overline{\chi'}$$

of the representation

$$\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} V_{\chi'} = \operatorname{inf}_{\mathcal{G}_n}^{\mathcal{G}} \operatorname{ind}_{\mathcal{G}'_n}^{\mathcal{G}_n} V_{\overline{\chi'}},$$

where $\overline{\chi'}$ is the projection of χ' to any quotient \mathcal{G}'_n through which it factors. The result is independent of the choice of \mathcal{G}'_n and one has

$$\left(\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi'\right)(\sigma) = \sum_{\substack{\tau \in \mathcal{G}/\mathcal{G}' \\ \tau^{-1}\sigma\tau \in \mathcal{G}'}} \chi'(\tau^{-1}\sigma\tau)$$

with τ running over any family of coset representatives of \mathcal{G}/\mathcal{G}' . This is essentially a reformulation of (0.10) which accommodates both the finite and infinite cases. Similarly, scalar products and divisibility of Artin characters can be addressed by passing to any finite quotient over which they factor.

As for type-W characters, it is clear that the restriction $\operatorname{res}_{\mathcal{G}'}^{\mathcal{G}}\rho$ of such a character ρ of \mathcal{G} is again a type-W character of \mathcal{G}' . Conversely, given any type-W character ρ' of \mathcal{G}' , there exists a ρ as above such that $\rho' = \operatorname{res}_{\mathcal{G}'}^{\mathcal{G}}\rho$. To see this, note that such a ρ' corresponds uniquely to the choice of a p-power root of unity $\rho'(\gamma_{K'}) \in \mu_{p^{\infty}}$ and $\gamma_{K'}$ identifies with $\gamma_{K}^{p^{M}}$ as in setting D. One may therefore define ρ by mapping γ_{K} to a p^{M} -th root of $\rho'(\gamma_{K'})$.

Character induction is fundamental to the relation of the Interpolation Conjectures for \mathcal{G}' and \mathcal{G} . In this regard, we have the following fact, which is one of the reasons the interpolation property (2.28) is only claimed for almost all $\rho \in \mathcal{K}_{S}^{\alpha}(\chi)$: each $\chi \in \operatorname{Irr}_{p}(\mathcal{G})$ divides $\operatorname{ind}_{\mathcal{G}}^{\mathcal{G}}\chi'$ for at most finitely many $\chi' \in \operatorname{Irr}_{p}(\mathcal{G}')$. This follows immediately from Frobenius reciprocity (0.11), as

$$\chi \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi' \iff \langle \chi, \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi' \rangle_{\mathcal{G}} > 0 \iff \langle \operatorname{res}_{\mathcal{G}'}^{\mathcal{G}} \chi, \chi' \rangle_{\mathcal{G}'} > 0 \iff \chi' \mid \operatorname{res}_{\mathcal{G}'}^{\mathcal{G}} \chi, \qquad (3.44)$$

where the last condition is satisfied by finitely many χ' only.

Unlike in the previous subsection, there is no natural ring homomorphism $\Lambda(\mathcal{G}) \to \Lambda(\mathcal{G}')$. *K*-theory does provide us, however, with a restriction map on K_1 groups: if $\iota: \Lambda(\mathcal{G}') \hookrightarrow \Lambda(\mathcal{G})$ denotes the canonical embedding, which extends to $\mathcal{Q}(\mathcal{G}') \hookrightarrow \mathcal{Q}(\mathcal{G})$ and $\mathcal{Q}^c(\mathcal{G}') \hookrightarrow \mathcal{Q}^c(\mathcal{G})$, one can define a group homomorphism $K_1^{\text{res}}(\iota): K_1(\mathcal{Q}(\mathcal{G})) \to K_1(\mathcal{Q}(\mathcal{G}'))$ as in section 0.4. Together with some results from [RW04], this will be enough to prove the relation of the equivariant Main Conjectures for L_{∞}/K and L_{∞}/K' . However, directly relating the Interpolation Conjectures demands a slightly finer analysis (which also relies on [RW04]) and the construction of homomorphisms $\operatorname{res}_{\chi'}^{\chi}: \mathcal{Q}^c(\Gamma_{\chi})^* \to \mathcal{Q}^c(\Gamma_{\chi'})^*$ for $\chi \in \operatorname{Irr}_p(\mathcal{G}), \chi' \in \operatorname{Irr}_p(\mathcal{G}')$ in a compatible manner with $K_1^{\operatorname{res}}(\iota)$. Given such a χ , consider the injective ring homomorphism $j_{\chi}: Z(\mathcal{Q}^c(\mathcal{G})e_{\chi}) = \mathcal{Q}^c(\Gamma_{\chi}) \hookrightarrow \mathcal{Q}^c(\Gamma_K)$ induced by $\gamma_{\chi} \mapsto \gamma_K^{w_{\chi}}$, where w_{χ} is as in proposition 2.4.4 (compare with the map j_{χ}^F therein). These fields fit into the diagram

with $j_{\chi'}$ defined analogously to j_{χ} for χ' as above. Recall that these maps are closely related to our $\psi_{\chi} \colon K_1(\mathcal{Q}(\mathcal{G})) \to \mathcal{Q}^c(\Gamma_{\chi})^*$, as shown in (2.29).

Proposition 3.3.7. Setting *D*. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ and $\chi' \in \operatorname{Irr}_p(\mathcal{G}')$. Then the restriction map

$$\operatorname{res}_{\chi'}^{\chi} \colon \mathcal{Q}^{c}(\Gamma_{\chi})^{*} \to \mathcal{Q}^{c}(\Gamma_{\chi'})^{*}$$

$$z \mapsto j_{\chi'}^{-1} \left(\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} j_{\varsigma}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma \rangle} \right)$$
(3.46)

is a well-defined group homomorphism. Furthermore, $\operatorname{res}_{\chi'}^{\chi} = \operatorname{res}_{\widetilde{\chi}'}^{\widetilde{\chi}}$ for any $\widetilde{\chi} \sim_W \chi$ and $\widetilde{\chi}' \sim_W \chi'$, and the diagram

commutes, where χ and χ' run over $\operatorname{Irr}_p(\mathcal{G})/\sim_W$ and $\operatorname{Irr}_p(\mathcal{G}')/\sim_W$, respectively. For any $z \in \mathcal{Q}^c(\Gamma_{\chi})^*$, one has

$$ev_{\gamma_{\chi'}}\left(\operatorname{res}_{\chi'}^{\chi}(z)\right) = \prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} ev_{\gamma_{\varsigma}}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma \rangle}$$

if the factors on the right-hand side with non-zero exponent are all finite.

Before delving into the proof, we clarify that the last product is a shorthand for

$$\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim_{W}\chi\\\varsigma \mid \operatorname{ind}_{\mathcal{G}}^{\mathcal{G}}\chi'}} ev_{\gamma_{\varsigma}}(z)^{\langle \operatorname{ind}_{\mathcal{G}}^{\mathcal{G}}\chi',\varsigma\rangle}$$

i.e. factors with exponent 0 are entirely disregarded (in particular, the product is finite). In other words, we adopt the convention that $0^0 = \infty^0 = 1$, which will be in place in other similar products as well.

Proof. As in [RW04] theorem 7, we define $\operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K}))$ as the set of all maps $f \colon \operatorname{Irr}_{p}(\mathcal{G}) \to \mathcal{Q}^{c}(\Gamma_{K})$ such that $f(\varsigma \otimes \rho) = \rho^{\sharp}(f(\varsigma))$ for all $\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})$ and ρ of type W. Here ρ^{\sharp} is as in (2.25), namely the field automorphism of $\mathcal{Q}^{c}(\Gamma_{K})$ induced by $\rho^{\sharp}(\gamma_{K}) = \rho(\gamma_{K})\gamma_{K}$. The ring structure of $\mathcal{Q}^{c}(\Gamma_{K})$ induces a ring structure on $\operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K}))$, the unit group of which is precisely $\operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K}))^{*} = \operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K})^{*})$. The proof of the cited theorem shows that

$$\delta \colon Z(\mathcal{Q}^{c}(\mathcal{G}))^{*} \to \operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K})^{*})$$

$$z \mapsto [\varsigma \mapsto j_{\varsigma}(ze_{\varsigma})]$$
(3.47)

is a group isomorphism. In particular, given $f \in \operatorname{Map}^W(\operatorname{Irr}_p(\mathcal{G}), \mathcal{Q}^c(\Gamma_K)^*)$ and $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$, one has $f(\varsigma) \in j_{\varsigma}(\mathcal{Q}(\Gamma_{\varsigma})^*) \subseteq \mathcal{Q}^c(\Gamma_K)^*$. For any such f, the map

$$\operatorname{res}^{W}(f) \colon \operatorname{Irr}_{p}(\mathcal{G}') \to \mathcal{Q}^{c}(\Gamma_{K'})^{*}$$
$$\varsigma' \mapsto \prod_{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})} f(\varsigma)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\varsigma',\varsigma \rangle}$$

is a well-defined element of $\operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}'), \mathcal{Q}^{c}(\Gamma_{K'})^{*})$ under the vertical arrow in (3.45) by [RW04] lemma 9.

The central square of the diagram



where μ and μ' are defined by commutativity of the triangles they are part of (note that δ' is an isomorphism), commutes by the same result²¹. The dashed vertical arrow is the homomorphism $(\delta')^{-1} \operatorname{res}^W \delta$, and its restriction to χ - and χ' -parts (in the domain and codomain, respectively) is precisely the map $\operatorname{res}_{\chi'}^{\chi}$ from the statement. The fact that it is invariant under ρ -twist of χ and χ' is clear, since so are all elements of the diagram and the simple components of $Z(\mathcal{Q}^c(\mathcal{G}))^*$ and $Z(\mathcal{Q}^c(\mathcal{G}'))^*$. It can also be deduced directly from (3.46).

In order to determine the behaviour under twisted evaluation maps, we resort to the commutative diagram



for $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$. We have already shown why the top and bottom faces commute (see for instance the triangles in (2.26)), and the front face does for essentially the same reason: $\gamma_{\chi'}$ is mapped to 1 either way, and this determines the map uniquely. For any $z \in \mathcal{Q}^c(\Gamma_{\chi})^*$, one therefore has

$$\begin{split} ev_{\gamma_{\chi'}}\left(\operatorname{res}_{\chi'}^{\chi}(z)\right) &= ev_{\gamma_{\chi'}}\left(j_{\chi'}^{-1}\left(\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} j_{\varsigma}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma\rangle}\right)\right) \\ &= ev_{\gamma_{K'}}\left(\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} j_{\varsigma}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma\rangle}\right) \\ &= ev_{\gamma_{K}}\left(\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} j_{\varsigma}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma\rangle}\right) \\ &= \prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} ev_{\gamma_{K}}(j_{\varsigma}(z))^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma\rangle} \\ &= \prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \sim W\chi}} ev_{\gamma_{\varsigma}}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi',\varsigma\rangle}. \end{split}$$

²¹The lemma in question concerns $\operatorname{Hom}^*(\mathcal{R}_p(\mathcal{G}), \mathcal{Q}^c(\Gamma_K)^*)$, where $\mathcal{R}_p(\mathcal{G})$ denotes the ring of virtual Artin characters of \mathcal{G} (cf. section 0.2). The essential difference between Map^W and Hom^* is that the latter adds a an extra Galoisequivariance condition. However, the same proof shows the case of interest to us - one simply needs to replace P by a finitely generated $\mathcal{Q}^c(\mathcal{G})$ -module in the claim on p. 560 of [RW04], and the objects in the commutative triangle on p. 558 by their counterparts with scalars in \mathbb{Q}_p^c .

The only equalities which do not follow from the last diagram are the first one, which is simply the definition of $\operatorname{res}_{\chi'}^{\chi}$, and the fourth one, which follows from remark 2.4.2 ii).

One may be tempted to rewrite the product in (3.46) as

$$\prod_{\rho} j_{\varsigma}(z)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi', \chi \otimes \rho \rangle}$$

with ρ running over the type-W characters of \mathcal{G} . We point out that the two expressions will differ in general, since $\chi \otimes \rho$ and $\chi \otimes \tilde{\rho}$ may coincide even if ρ and $\tilde{\rho}$ do not (cf. remark 2.4.6 i)).

The previous proposition establishes a relation between the Interpolation Conjectures for \mathcal{G} and \mathcal{G}' , and this in turn between the corresponding equivariant Main Conjectures:

Proposition 3.3.8. Setting *D*. Let α and β be chosen as in setting *B* for the extension L_{∞}/K . In particular, they are valid choices for L_{∞}/K' as explained above. Then:

- i) Let $\chi' \in \operatorname{Irr}_p(\mathcal{G}')$. If $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in \operatorname{Irr}_p(\mathcal{G})$ dividing $\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi'$, then so does $IC(L_{\infty}/K', \chi', L, S(K'), T(K'), \alpha, \beta)$.
- ii) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ implies $eMC(L_{\infty}/K', L, S(K'), T(K'), \alpha, \beta)$.

Proof. For each irreducible $\chi \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi'$, let $F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ be the unique series quotient satisfying $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Choose a set $\Xi_{\chi'}$ of representatives of these χ modulo \sim_W (we do not claim that $\chi \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi'$ implies $\chi \otimes \rho \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}} \chi'$) and define

$$F_{\chi'} = \prod_{\chi \in \Xi_{\chi'}} \operatorname{res}_{\chi'}^{\chi}(F_{S,T,\chi}^{\alpha,\beta}) \in \mathcal{Q}^c(\Gamma_{\chi'})^*.$$

Given a character $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$ which is W-equivalent to some (necessarily unique) $\chi \in \Xi_{\chi'}$, set

$$F_{S,T,\varsigma}^{\alpha,\beta} = F_{S,T,\chi}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_\chi)^* = \mathcal{Q}^c(\Gamma_\varsigma)^*.$$

Note that if two W-equivalent characters χ and $\tilde{\chi}$ both divide $\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi'$, one has $F_{S,T,\chi}^{\alpha,\beta} = F_{S,T,\tilde{\chi}}^{\alpha,\beta}$ by proposition 2.5.2 so that the two definitions of $F_{S,T,\tilde{\chi}}^{\alpha,\beta}$ (one from the Interpolation Conjecture and one from the last equation) do not conflict. The same proposition shows that $F_{S,T,\varsigma}$ satisfies $\operatorname{IC}(L_{\infty}/K,\varsigma,L,S,T,\alpha,\beta)$.

Let ρ' be a character of \mathcal{G}' of type W and choose a type-W character ρ of \mathcal{G} such that $\operatorname{res}_{\mathcal{G}'}^{\mathcal{G}}\rho = \rho'$ as explained earlier. It is easy to see (for instance, using Frobenius reciprocity) that

$$\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi' \otimes \rho') = (\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}\chi') \otimes \rho.$$
(3.49)

Proposition 3.3.7 then yields

$$\begin{split} ev_{\gamma_{\chi'\otimes\rho'}}(F_{\chi'}) &= \prod_{\chi\in\Xi_{\chi'}} ev_{\gamma_{\chi'\otimes\rho'}}(\operatorname{res}_{\chi'}^{\chi}(F_{S,T,\chi}^{\alpha,\beta})) \\ &= \prod_{\chi\in\Xi_{\chi'}} ev_{\gamma_{\chi'\otimes\rho'}}(\operatorname{res}_{\chi'\otimes\rho'}^{\chi}(F_{S,T,\chi}^{\alpha,\beta})) \\ &= \prod_{\chi\in\Xi_{\chi'}} \prod_{\substack{\varsigma\in\operatorname{Irr}_{p}(\mathcal{G})\\\varsigma\sim_W\chi}} ev_{\gamma_{\varsigma}}(F_{S,T,\varsigma}^{\alpha,\beta})^{\langle\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho'),\varsigma\rangle} \\ &= \prod_{\substack{\varsigma\in\operatorname{Irr}_{p}(\mathcal{G})\\\varsigma\mid\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho')}} ev_{\gamma_{\varsigma}}(F_{S,T,\varsigma}^{\alpha,\beta})^{\langle\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho'),\varsigma\rangle} \end{split}$$

whenever all factors are finite, where the last equality follows from (3.49). Consider now the set

 $\Omega = \{\varsigma \in \operatorname{Irr}_p(\mathcal{G}) \text{ such that } \varsigma \sim_W \chi \text{ for some } \chi \in \Xi_{\chi'} \}.$

which coincides with

$$\{\varsigma \in \operatorname{Irr}_p(\mathcal{G}) \text{ such that } \varsigma \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi' \otimes \rho') \text{ for some } \rho' \text{ of type } W\}$$

by (3.49). Then

$$ev_{\gamma_{\varsigma}}(F_{S,T,\varsigma}^{\alpha,\beta}) = \frac{\beta^{-1}(L_{K,S,T}^{*}(\beta\check{\varsigma},0))}{R_{S}^{\beta}(\alpha,\varsigma)}$$

for almost all $\varsigma \in \Omega$. The finitely many ς where interpolation fails divide only finitely many irreducible characters of \mathcal{G}' by (3.44), which implies that

$$ev_{\gamma_{\chi'\otimes\rho'}}(F_{\chi'}) = \prod_{\substack{\varsigma\in\operatorname{Irr}_{p}(\mathcal{G})\\\varsigma\mid\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho')}} ev_{\gamma_{\varsigma}}(F_{S,T,\varsigma}^{\alpha,\beta})^{\langle\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho'),\varsigma\rangle} = \prod_{\substack{\varsigma\in\operatorname{Irr}_{p}(\mathcal{G})\\\varsigma\mid\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho')}} \left(\frac{\beta^{-1}(L_{K,S,T}^{*}(\beta\check{\varsigma},0))}{R_{S}^{\beta}(\alpha,\varsigma)}\right)^{\langle\operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi'\otimes\rho'),\varsigma\rangle}$$

for almost all type-W characters ρ' of \mathcal{G}' . Now we simply use the functoriality of the regulated special L-values, namely their behaviour with respect to character addition and induction. For the leading coefficient at 0, this is lemma 0.3.1 iii) and iv). We cannot directly speak of functoriality of the Stark-Tate regulator from section 2.3, since it was only defined for irreducible characters. However, its relation to that defined in [Tat84] and the functorial properties of the latter suffice to show

$$\prod_{\substack{\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})\\\varsigma \mid \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}}(\chi' \otimes \rho')}} \left(\frac{\beta^{-1}(L_{K,S,T}^{*}(\beta \check{\varsigma}, 0))}{R_{S}^{\beta}(\alpha, \varsigma)} \right)^{\langle \operatorname{ind}_{\mathcal{G}'}^{\mathcal{G}'}(\chi' \otimes \rho'),\varsigma \rangle} = \frac{\beta^{-1}(L_{K',S(K'),T(K')}^{*}(\beta(\check{\chi}' \otimes (\rho')^{-1}), 0))}{R_{S(K')}^{\beta}(\alpha, \chi' \otimes \rho')}.$$
(3.50)

More specifically, the equality of denominators is a consequence of equations 6.4 (1) and (2) on p. 29 of [Tat84] together with (3.9) above. Recall that Tate only considers S-modified Artin L-functions (i.e. with $T = \emptyset$), but the δ -factors at places in T obey the same formalism as already argued in the proof of proposition 3.1.1. This proves that $F_{\chi'}$ satisfies $IC(L_{\infty}/K', \chi', L, S(K'), T(K'), \alpha, \beta)$.

The implication of equivariant Main Conjectures follows readily. Suppose $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ satisfies $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$. In particular, for each irreducible Artin character χ of $\mathrm{Irr}_p(\mathcal{G})$, the series quotient $F_{S,T,\chi}^{\alpha,\beta} = \psi_{\chi}(\zeta_{S,T}^{\alpha,\beta}) \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. Define

$$\zeta' = K_1^{\operatorname{res}}(\iota)(\zeta_{S,T}^{\alpha,\beta}) \in K_1(\mathcal{Q}(\mathcal{G}')).$$

For a fixed $\chi' \in \operatorname{Irr}_p(\mathcal{G}')$, define $\Xi_{\chi'}$ as above and extend it to a set Ξ of representatives of $\operatorname{Irr}_p(\mathcal{G})/\sim_W$. Then by proposition 3.3.7, one has

$$\psi_{\chi'}(\zeta') = \psi_{\chi'}(K_1^{\operatorname{res}}(\iota)(\zeta_{S,T}^{\alpha,\beta})) = \prod_{\chi \in \Xi} \operatorname{res}_{\chi'}^{\chi}(\psi_{\chi}(\zeta_{S,T}^{\alpha,\beta})) = \prod_{\chi \in \Xi} \operatorname{res}_{\chi'}^{\chi}(F_{S,T,\chi}^{\alpha,\beta}) = \prod_{\chi \in \Xi_{\chi'}} \operatorname{res}_{\chi'}^{\chi}(F_{S,T,\chi}^{\alpha,\beta}),$$

where the last equality relies on the observation that, for $\chi \in \Xi \setminus \Xi_{\chi'}$, the map $\operatorname{res}_{\chi'}^{\chi}$ is identically 1. Note that by the proof of part i), the right-hand side is precisely the element $F_{S(K'),T(K'),\chi'}^{\alpha,\beta}$ satisfying $\operatorname{IC}(L_{\infty}/K',\chi',L,S(K'),T(K'),\alpha,\beta)$.

Moving on to the homological side, let

$$t^{\alpha} \colon \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{1}(\mathcal{C}^{\bullet}_{S,T}) \xrightarrow{\sim} \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} H^{0}(\mathcal{C}^{\bullet}_{S,T})$$

and

$$\tilde{t}^{\alpha} \colon \mathcal{Q}(\mathcal{G}') \otimes_{\Lambda(\mathcal{G}')} H^1(\mathcal{C}^{\bullet}_{S(K'),T(K')}) \xrightarrow{\sim} \mathcal{Q}(\mathcal{G}') \otimes_{\Lambda(\mathcal{G}')} H^0(\mathcal{C}^{\bullet}_{S(K'),T(K')})$$

denote the trivialisations induced by α as a homomorphism of $\Lambda(\mathcal{G})$ - and $\Lambda(\mathcal{G}')$ -modules, respectively, as in (2.7). Then $\tilde{t}^{\alpha} = t^{\alpha}|_{\mathcal{Q}(\mathcal{G}')}$ and $\mathcal{C}^{\bullet}_{S(K'),T(K')} = \mathcal{C}^{\bullet}_{S,T}|_{\Lambda(\mathcal{G}')}$ in the notation of lemma A.0.1 ii), which shows

$$\partial(\zeta') = \partial(K_1^{\text{res}}(\iota)(\zeta_{S,T}^{\alpha,\beta})) = K_0^{\text{res}}(\iota,\iota)(\partial(\zeta_{S,T}^{\alpha,\beta}))$$
$$= K_0^{\text{res}}(\iota,\iota)(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet},t^{\alpha}))$$
$$= -\chi_{\Lambda(\mathcal{G}'),\mathcal{Q}(\mathcal{G}')}(\mathcal{C}_{S,T}^{\bullet}|_{\Lambda(\mathcal{G}')},t^{\alpha}|_{\mathcal{Q}(\mathcal{G}')})$$
$$= -\chi_{\Lambda(\mathcal{G}'),\mathcal{Q}(\mathcal{G}')}(\mathcal{C}_{S(K'),T(K')}^{\bullet},\tilde{t}_{\alpha})$$

(for the second equality, see (0.31)). Therefore, ζ' satisfies $eMC(L_{\infty}/K', L, S(K'), T(K'), \alpha, \beta)$.

3.3.3 Reduction steps

We end this section, and by extension this chapter, by exploring the converse to the question treated in the preceding lines, namely: can the conjectures for L_{∞}/K be deduced from their counterparts for smaller extensions? The answer is affirmative if the latter are chosen appropriately, and we give two examples of such possible choices. The technique we use mirrors the argument in [JN20] section 10, the main difference being that the relation between the Interpolation Conjectures must be proved in our setting - as opposed to the totally real case the cited paper is concerned with, where *p*-adic *L*-functions interpolating *L*-values are known to exist.

Before proceeding with the outlined plan, we make a minor detour to state an immediate consequence of the two preceding functoriality results. Only in this corollary do we dispense with the explicit notation of all parameters of the conjectures, opting instead for the abbreviations $IC(L_{\infty}/K, \chi, \beta)$ and $eMC(L_{\infty}/K, \beta)$ introduced at the beginning of the present chapter (recall theorem 3.0.1).

Corollary 3.3.9. Let the choice of an odd prime p and an isomorphism $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ be fixed for all conjectures below.

- i) Suppose $IC(\widetilde{L}_{\infty}/\mathbb{Q}, \chi, \beta)$ holds for every choice of \widetilde{L}_{∞}/K as in setting A with $K = \mathbb{Q}$ and every $\chi \in Irr_p(Gal(\widetilde{L}_{\infty}/\mathbb{Q}))$. Then so does $IC(L'_{\infty}/K', \chi', \beta)$ for any valid choice of L'_{∞}/K' and any $\chi' \in Irr_p(Gal(L'_{\infty}/K'))$.
- ii) Suppose $eMC(\widetilde{L}_{\infty}/\mathbb{Q},\beta)$ holds for every choice of \widetilde{L}_{∞}/K as in setting A with $K = \mathbb{Q}$. Then so does $eMC(L'_{\infty}/K',\beta)$ for any valid choice of L'_{∞}/K' .

Proof. Let \mathbb{Q}_{∞} denote the cyclotomic \mathbb{Z}_p -extension of \mathbb{Q} . Given L'_{∞}/K' as in the statement, choose a number field $L' \supseteq K'$ whose cyclotomic \mathbb{Z}_p -extension coincides with L'_{∞} and define \widetilde{L} as the Galois closure of L'/\mathbb{Q} . Its cyclotomic \mathbb{Z}_p -extension \widetilde{L}_{∞} is Galois over \mathbb{Q} (as it coincides with $\widetilde{L}\mathbb{Q}_{\infty}$) and has finite degree over \mathbb{Q}_{∞} , which means $\widetilde{L}_{\infty}/\mathbb{Q}$ is a suitable extension for the formulation of the conjectures. Propositions 3.3.8 and 3.3.6 now yield

$$\operatorname{IC}(\widetilde{L}_{\infty}/\mathbb{Q},\chi,\beta) \text{ for all } \chi \in \operatorname{Irr}_p(\operatorname{Gal}(\widetilde{L}_{\infty}/\mathbb{Q}))$$

$$\Rightarrow \operatorname{IC}(\widetilde{L}_{\infty}/K',\widetilde{\chi},\beta) \text{ for all } \widetilde{\chi} \in \operatorname{Irr}_p(\operatorname{Gal}(\widetilde{L}_{\infty}/K'))$$

$$\Rightarrow \operatorname{IC}(L'_{\infty}/K',\chi',\beta) \text{ for all } \chi' \in \operatorname{Irr}_p(\operatorname{Gal}(L'_{\infty}/K'))$$

and

$$\operatorname{eMC}(\widetilde{L}_{\infty}/\mathbb{Q},\beta) \Rightarrow \operatorname{eMC}(\widetilde{L}_{\infty}/K',\beta) \Rightarrow \operatorname{eMC}(L'_{\infty}/K',\beta).$$

Returning to the main purpose of this subsection, the first step is to reduce the Interpolation Conjecture to the case of linear characters, which conspicuously hints at Brauer's induction theorem 0.2.1 and thus leads to the consideration of elementary groups. Since our treatment of Artin characters of \mathcal{G} often relies on factoring them through some finite layer, the following definition from [JN20] section 10 should not come as a surprise:

Definition 3.3.10. Let p be a prime and \mathcal{G} a profinite group which fits into a short exact sequence $H \hookrightarrow \mathcal{G} \twoheadrightarrow \overline{\Gamma}$ with H finite and $\overline{\Gamma} \cong \mathbb{Z}_p$. Given a prime l, we say \mathcal{G} is l-elementary if there exists an open central subgroup $\Gamma \cong \mathbb{Z}_p$ of \mathcal{G} such that \mathcal{G}/Γ is an l-elementary finite group (see for instance section 0.2 for a definition). More generally, \mathcal{G} is said to be elementary if it is l-elementary for at least one l.

In the notation of setting A, let χ be an Artin character of \mathcal{G} . Choose a finite quotient $\mathcal{G}_n = \mathcal{G}/\Gamma^{p^n}$ over which χ factors and denote the corresponding projection by $\overline{\chi}$. By Brauer's theorem, there exists a decomposition

$$\overline{\chi} = \sum_{i=1}^{n} z_i \operatorname{ind}_{\overline{\mathcal{H}}_i}^{\mathcal{G}_n} \overline{\lambda}_i,$$

where $z_i \in \mathbb{Z}$, $\overline{\mathcal{H}}_i$ is an elementary subgroup of \mathcal{G}_n and $\overline{\lambda}_i$ is a \mathbb{Q}_p^c -valued linear character of $\overline{\mathcal{H}}_i$. The preimage \mathcal{H}_i of $\overline{\mathcal{H}}_i$ under $\mathcal{G} \twoheadrightarrow \mathcal{G}_n$ is elementary in the sense defined above, and $\overline{\lambda}_i$ inflates to a linear character $\lambda_i = \inf_{\overline{\mathcal{H}}_i} \overline{\lambda}_i$ which fits into

$$\chi = \sum_{i=1}^{n} z_i \operatorname{ind}_{\mathcal{H}_i}^{\mathcal{G}} \lambda_i.$$
(3.51)

As explained in the previous subsection, a choice of parameters as in setting B for L_{∞}/K yields a natural set of parameters for $L_{\infty}/K^{\mathcal{H}_i}$ for each *i*.

Lemma 3.3.11. Setting B. Given $\chi \in \operatorname{Irr}_p(\mathcal{G})$, choose a decomposition $\chi = \sum_{i=1}^n z_i \operatorname{ind}_{\mathcal{H}_i}^{\mathcal{G}} \lambda_i$ as in (3.51). For each *i*, set $K^i = L_{\infty}^{\mathcal{H}_i}$ and $L^i = L_{\infty}^{\mathcal{H}_i \cap \Gamma}$. Then, if $IC(L_{\infty}/K^i, \lambda_i, L^i, S(K^i), T(K^i), \alpha, \beta)$ holds for all *i*, so does $IC(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$.

Proof. Let Γ_{K^i} denote the Galois group over K^i of its cyclotomic \mathbb{Z}_p -extension $K_{\infty}^i = K_{\infty}K^i$. Restriction of Galois automorphisms yields an embedding $\Gamma_{K^i} \hookrightarrow \Gamma_K$ with image $\Gamma_K^{p^{M_i}}$ for some $M_i \in \mathbb{N}$. The formulation of the Interpolation Conjecture for λ_i contains the implicit choice of a topological generator γ_{K^i} of Γ_{K^i} , which we choose to be the unique preimage of $\gamma_K^{p^{M_i}}$.

Consider the injective ring homomorphisms

$$\mathcal{Q}^{c}(\Gamma_{\lambda_{i}}) \xrightarrow{j_{\lambda_{i}}} \mathcal{Q}^{c}(\Gamma_{K^{i}}) \xrightarrow{j_{K^{i}}} \mathcal{Q}^{c}(\Gamma_{K}),$$

where j_{K^i} is induced by $\Gamma_{K^i} \hookrightarrow \Gamma_K$, and $\mathcal{Q}^c(\Gamma_{\lambda_i}) = Z(\mathcal{Q}^c(\mathcal{G}_i)e_{\lambda_i})$ and $j_{\lambda_i}: \gamma_{\lambda_i} \mapsto \gamma_{K^i}^{w_{\lambda_i}} = \gamma_{K^i}$ are as in section 2.4 (cf. proposition 2.4.4). Note that $w_{\lambda_i} = 1$ because λ_i is linear.

Let $F_i = F_{S(K^i),T(K^i),\lambda_i}^{\alpha,\beta} \in \mathcal{Q}^c(\Gamma_{\lambda_i})^*$ be the element predicted by $\mathrm{IC}(L_{\infty}/K^i,\lambda_i,L^i,S(K^i),T(K^i),\alpha,\beta)$. Recall that, for any type-W character ρ' of \mathcal{H}_i , one has $\mathcal{Q}^c(\Gamma_{\lambda_i}) = \mathcal{Q}^c(\Gamma_{\lambda_i\otimes\rho'}) \subseteq Z(\mathcal{Q}^c(\mathcal{H}_i)e_{\lambda_i\otimes\rho'})$ and, as shown in proposition 2.5.2, F_i also satisfies $\mathrm{IC}(L_{\infty}/K^i,\lambda_i\otimes\rho',L^i,S(K^i),T(K^i),\alpha,\beta)$.

Fix now a type-W character ρ of \mathcal{G} and set $\rho_i = \operatorname{res}_{\mathcal{H}_i}^{\mathcal{G}} \rho$ (which is a type-W character of \mathcal{H}_i) for each *i*. Define

$$F_{\rho} = \prod_{i=1}^{n} j_{K_i}(j_{\lambda_i \otimes \rho_i}(F_i^{z_i})) \in \mathcal{Q}^c(\Gamma_K)^*.$$

We first prove an interpolation property for this element. Namely, for any type-W character $\tilde{\rho}$ of \mathcal{G} , one has

$$F_{\rho}|_{T_{K}=\tilde{\rho}(\gamma_{K})-1} = \prod_{i=1}^{n} j_{K_{i}}(j_{\lambda_{i}\otimes\rho_{i}}(F_{i}^{z_{i}}))|_{T_{K}=\tilde{\rho}(\gamma_{K})-1} = \prod_{i=1}^{n} ev_{\gamma_{K}}(j_{K_{i}}(j_{\lambda_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}(F_{i}^{z_{i}}))) = \prod_{i=1}^{n} ev_{\gamma_{\lambda_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}}(F_{i})^{z_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}(F_{i})^{z_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}(F_{i})^{z_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}(F_{i})$$

whenever all factors in the last product are finite (this is used in the first and last equalities). Here $\tilde{\rho}_i = \operatorname{res}_{\mathcal{H}_i}^{\mathcal{G}} \tilde{\rho}$, the last equality is a property we have already encountered multiple times, and the second one follows from the commutativity of

(recall that ρ^{\sharp} is the automorphism of $\mathcal{Q}^{c}(\Gamma_{K})$ determined by mapping γ_{K} to $\rho(\gamma_{K})\gamma_{K}$, and analogously for $\rho_{i}^{\sharp} \colon \mathcal{Q}^{c}(\Gamma_{K_{i}}) \xrightarrow{\sim} \mathcal{Q}^{c}(\Gamma_{K_{i}}), \gamma_{K_{i}} \mapsto \rho_{i}(\gamma_{K_{i}})\gamma_{K_{i}}).$

In order to determine the above product, let \mathcal{K}_i be the set of all $\rho' \in \mathcal{K}^{\alpha}_{S(K^i)}(\lambda_i)$ such that

$$ev_{\gamma_{\lambda_i\otimes\rho_i\otimes\rho'}}(F_i) = \frac{\beta^{-1}(L^*_{K^i,S(K^i),T(K^i)}(\beta((\lambda_i\otimes\rho_i)\otimes(\rho')^{-1}),0))}{R^\beta_{S(K^i)}(\alpha,\lambda_i\otimes\rho_i\otimes\rho')},$$

which contains almost all type-W characters of \mathcal{H}_i by $\mathrm{IC}(L_{\infty}/K_i, \lambda_i \otimes \rho_i, L^i, S(K^i), T(K^i), \alpha, \beta)$. It follows that, for almost all type-W characters $\tilde{\rho}$ of \mathcal{G} , one has: $\mathrm{res}_{\mathcal{H}_i}^{\mathcal{G}}(\rho) \in \mathcal{K}_i$ for all *i*. In particular,

$$F_{\rho}|_{T_{K}=\tilde{\rho}(\gamma_{K})-1} = \prod_{i=1}^{n} ev_{\gamma_{\lambda_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i}}}(F_{i})^{z_{i}} = \prod_{i=1}^{n} \left(\frac{\beta^{-1}(L_{K^{i},S(K^{i}),T(K^{i})}(\beta((\lambda_{i}\otimes\rho_{i})^{*}\otimes\tilde{\rho}_{i}^{-1}),0))}{R_{S(K^{i})}^{\beta}(\alpha,\lambda_{i}\otimes\rho_{i}\otimes\tilde{\rho}_{i})} \right)^{z_{i}} = \frac{\beta^{-1}(L_{K,S,T}^{*}(\beta(\tilde{\chi}\otimes\rho^{-1}\otimes\tilde{\rho}^{-1}),0))}{R_{S}^{\beta}(\alpha,\chi\otimes\rho\otimes\tilde{\rho})}$$
(3.53)

for any such $\tilde{\rho}$, where the last equality is a consequence of

$$\chi \otimes \rho \otimes \tilde{\rho} = \sum_{i=1}^{n} z_i (\operatorname{ind}_{\mathcal{G}_i}^{\mathcal{G}} \lambda_i) \otimes \rho \otimes \tilde{\rho} = \sum_{i=1}^{n} z_i \operatorname{ind}_{\mathcal{G}_i}^{\mathcal{G}} (\lambda_i \otimes \rho_i \otimes \tilde{\rho}_i)$$

and an analogous argument to that for (3.50) (which is unaffected by the fact that the z_i may be negative).

We now prove that F_1 lies in the image of $\mathcal{Q}^c(\Gamma_{\chi})^* \xrightarrow{j_{\chi}} \mathcal{Q}^c(\Gamma_K)^*$. Suppose first that $\chi \otimes \rho = \chi \otimes \rho'$ for two type-*W*-characters ρ and ρ' of \mathcal{G} . Then F_{ρ} and $F_{\rho'}$ take the same value at infinitely many points of $B_1^{\mathbb{Q}_p^c}(0)$ by (3.53), since the regulated *L*-values for $\chi \otimes \rho \otimes \tilde{\rho}$ and $\chi \otimes \rho' \otimes \tilde{\rho}$ coincide for any $\tilde{\rho}$. Thus, lemma 2.4.10 implies $F_{\rho} = F_{\rho'}$. In particular, the map $f \colon \operatorname{Irr}_p(\mathcal{G}) \to \mathcal{Q}^c(\Gamma_K)^*$ given by

$$f(\varsigma) = \begin{cases} F_{\rho}, & \varsigma = \chi \otimes \rho \text{ for some } \rho \text{ of type } W\\ 1, & \text{otherwise} \end{cases}$$
(3.54)

is well defined.

We claim that $f \in \operatorname{Map}^{W}(\operatorname{Irr}_{p}(\mathcal{G}), \mathcal{Q}^{c}(\Gamma_{K})^{*})$ in the notation of the previous subsection (see the proof of 3.3.7), i.e. $f(\varsigma \otimes \rho) = \rho^{\sharp}(f(\varsigma))$ for all $\varsigma \in \operatorname{Irr}_{p}(\mathcal{G})$ and ρ of type W. We distinguish three cases:

- If $\varsigma \nsim_W \chi$ then $f(\varsigma \otimes \rho) = f(\varsigma) = 1$.
- If $\varsigma = \chi$, one has

$$\rho^{\sharp}(f(\chi)) = \prod_{i=1}^{n} \rho^{\sharp}(j_{K_{i}}(j_{\lambda_{i}}(F_{i}^{z_{i}}))) = \prod_{i=1}^{n} j_{K_{i}}(\rho_{i}^{\sharp}(j_{\lambda_{i}}(F_{i}^{z_{i}}))) = \prod_{i=1}^{n} j_{K_{i}}(j_{\lambda_{i}\otimes\rho_{i}}(F_{i}^{z_{i}})) = f(\chi \otimes \rho),$$

where the second and third equalities follow from an analogous diagram to (3.52).
• If $\varsigma = \chi \otimes \tilde{\rho}$ for some $\tilde{\rho}$ of type W, the previous case implies

$$f(\varsigma \otimes \rho) = f(\chi \otimes (\tilde{\rho} \otimes \rho)) = (\tilde{\rho} \otimes \rho)^{\sharp}(f(\chi)) = \rho^{\sharp}(\tilde{\rho}^{\sharp}(f(\chi))) = \rho^{\sharp}(f(\chi \otimes \tilde{\rho})) = \rho^{\sharp}(f(\varsigma)).$$

Therefore, (3.47) implies that $F_1 = f(\chi) = j_{\chi}(\tilde{F}_{\chi})$ for some $\tilde{F}_{\chi} \in \mathcal{Q}^c(\Gamma_{\chi})^* \subseteq Z(\mathcal{Q}^c(\mathcal{G})e_{\chi})^*$. Now the commutativity of

$$\begin{array}{cccc} \mathcal{Q}^{c}(\Gamma_{\chi}) & \xrightarrow{j_{\chi}} & \mathcal{Q}^{c}(\Gamma_{K}) & \xrightarrow{T_{K} = \tilde{\rho}(\gamma_{K}) - 1} & \mathbb{Q}_{p}^{c} \cup \{\infty\} \\ \\ & & & & \\ & & & \\ \mathcal{Q}^{c}(\Gamma_{\chi \otimes \tilde{\rho}}) & \xrightarrow{ev_{\gamma_{\chi \otimes \tilde{\rho}}}} & \mathbb{Q}_{p}^{c} \cup \{\infty\} \end{array}$$

for any $\tilde{\rho}$, together with (3.53), shows that $\tilde{F}_{\chi} \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$.

The following technical result will reduce the number of elementary subextensions which need to be considered in order to deduce the Interpolation Conjecture for characters of L_{∞}/K :

Lemma 3.3.12. Let p, \mathcal{G} and Γ be as in setting A. Then, given any $\chi \in \operatorname{Irr}_p(\mathcal{G})$, there exists a type-W character ρ of \mathcal{G} such that $\chi \otimes \rho$ factors through $\mathcal{G}_0 = \mathcal{G}/\Gamma$.

Proof. Let $\overline{\chi}$ be the projection of χ to a quotient $\mathcal{G}_n = \mathcal{G}/\Gamma^{p^n}$ through which it factors. Since the subgroup $\Gamma_n = \Gamma/\Gamma^{p^n}$ is central in \mathcal{G}_n , Clifford theory (cf. [CR81] theorem 11.1) implies $\operatorname{res}_{\Gamma_n}^{\mathcal{G}_n} \overline{\chi} = e\lambda$ for some $e \in \mathbb{N}$ and $\lambda \in \operatorname{Irr}_{\mathbb{Q}_p^c}(\Gamma_n)$. By the abelianity of Γ_n , λ is linear and $e = \overline{\chi}(1) = \chi(1)$. In particular, $\lambda(\overline{\gamma})^{p^n} = \lambda(\overline{\gamma^{p^n}}) = 1$ and so $\lambda(\overline{\gamma})$ is a p^n -th root of unity $\zeta \in \mathbb{Q}_p^c$. Let p^N be as in setting A, that is, the index in Γ_K of the image of Γ under the canonical projection $\mathcal{G} \to \Gamma_K$. Choose $\tilde{\zeta} \in \mathbb{Q}_p^c$ such that $\tilde{\zeta}^{p^N} = \zeta$ and define ρ by (lifting to \mathcal{G}) $\rho(\gamma_K) = \tilde{\zeta}^{-1}$. Then $\chi \otimes \rho$ has the desired property.

The two preceding lemmas, together with some algebraic results from [JN20], allow us to characterise the Interpolation and equivariant Main Conjectures for L_{∞}/K in terms of certain families of subextensions. In doing so, we follow closely the approach in section 12 of the cited article. The key difference is that no analogue of lemma 3.3.11 is needed there, since the existence of *p*-adic *L*-functions which play the role of our $F_{S,T,\chi}^{\alpha,\beta}$ is a known fact in the totally real case. Furthermore, it will be necessary for us to prove a certain Galois-invariance property of interpolating series quotients.

As explained in remark 2.5.4, the connecting homomorphism $\partial : K_1(\mathcal{Q}(\mathcal{G})) \to K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ is surjective for any \mathcal{G} as in setting A. By the localisation sequence (0.27), its kernel is the image of $K_1(\Lambda(\mathcal{G}))$ in $K_1(\mathcal{Q}(\mathcal{G}))$, which implies the existence of a homomorphism

$$\nu_{\mathcal{G}} \colon K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G})) \to Z(\mathcal{Q}(\mathcal{G}))^* / \operatorname{nr}(K_1(\Lambda(\mathcal{G})))$$
(3.55)

via the reduced norm.

Remark 3.3.13. It is clear (alternatively, see [JN20] lemma 10.13) that $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ can be reformulated as the combination of the claims:

- $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in \operatorname{Irr}_{p}(\mathcal{G})$.
- The product $F_{S,T,\mathcal{G}}^{\alpha,\beta} = \prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta} \in Z(\mathcal{Q}^c(\mathcal{G}))^*$ of the interpolating series quotients lies in $Z(\mathcal{Q}(\mathcal{G}))^*$ (recall at this point diagram (3.12)).
- One has $\nu_{\mathcal{G}}(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T},t^{\alpha})) = [F^{\alpha,\beta}_{S,T,\mathcal{G}}]$, where [-] denotes class modulo $\operatorname{nr}(K_1(\Lambda(\mathcal{G})))$.

We introduce two families of one-dimensional compact p-adic Lie groups: on one hand,

 $\mathcal{S} = \{\pi_0^{-1}(\mathcal{G}_0') : \mathcal{G}_0' \text{ an elementary subgroup of } \mathcal{G}_0\},\$

where π_0 denotes the canonical projection $\pi_0: \mathcal{G} \twoheadrightarrow \mathcal{G}_0 = \mathcal{G}/\Gamma = \operatorname{Gal}(L/K)$; and on the other,

 $\mathcal{E}_p = \{\mathcal{G}'/H' : \mathcal{G}' \text{ a } p \text{-elementary subgroup of } \mathcal{G} \text{ containing } \Gamma, H' \text{ a finite normal subgroup of } \mathcal{G}' \}.$

Here the term *elementary* refers to the classical notion for finite groups in the case of S, and to the notion from definition 3.3.10 in that of \mathcal{E}_p . Both families are finite, as Γ is open in \mathcal{G} and \mathcal{G} has only finitely many subgroups of finite order (they are all contained in H).

As described in the two preceding subsections, a choice of parameters for the formulation of the Main Conjecture for L_{∞}/K yields parameters for the subextensions defined by elements of S and \mathcal{E}_p . The choice of prolongations v^c of places of K to \mathbb{Q}^c plays no role, since, as already explained, different prolongations give rise to naturally isomorphic modules.

Proposition 3.3.14. Setting *B* with *p* odd. The following are equivalent:

- i) $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ holds.
- ii) $eMC(L_{\infty}/K', L', S(K'), T(K'), \alpha, \beta)$ holds for all $\mathcal{G}' \in \mathcal{S}$, where $K' = L_{\infty}^{\mathcal{G}'}$ and L' = LK'.
- iii) $eMC(L'_{\infty}/K', L', S(K'), T(K'), \alpha', \beta)$ holds for all $\mathcal{G}'/H' \in \mathcal{E}_p$, where $K' = L^{\mathcal{G}'}_{\infty}, L'_{\infty} = L^{H'}_{\infty}$ and L' and α' are constructed from the corresponding parameters for L_{∞}/K' as in setting C and lemma 3.3.3, respectively.

Proof. The fact that i) implies ii) and iii) is a direct consequence of propositions 3.3.8 and 3.3.6. In order to prove their respective converses, we rely on the reformulation from remark 3.3.13.

• <u>ii)</u> \Rightarrow <u>i)</u>: We first need to show that IC($L_{\infty}/K, \chi, L, S, T, \alpha, \beta$) holds for any given $\chi \in \operatorname{Irr}_p(\mathcal{G})$. By lemmas 2.5.2 and 3.3.12, we can assume χ factors through \mathcal{G}_0 . In particular, a Brauer decomposition $\chi = \sum_{i=1}^n z_i \operatorname{ind}_{\mathcal{H}_i}^{\mathcal{G}} \lambda_i$ as in (3.51) can be chosen such that $\Gamma \subseteq \ker(\lambda_i) \subseteq \mathcal{H}_i$ for all *i*. It follows that all \mathcal{H}_i belong to \mathcal{S} by construction, and thus lemma 3.3.11 immediately yields IC($L_{\infty}/K, \chi, L, S, T, \alpha, \beta$). The next step is to prove that $F_{S,T,\mathcal{G}}^{\alpha,\beta} = \prod_{\chi/\sim_W} F_{S,T,\chi}^{\alpha,\beta} \in Z(\mathcal{Q}^c(\mathcal{G}))^*$ lies in fact in $Z(\mathcal{Q}(\mathcal{G}))^*$, or equivalently by diagram (3.12), that it is $G_{\mathbb{Q}_p}$ -invariant. The proof of lemma 3.3.11 realises each $F_{S,T,\chi}^{\alpha,\beta}$ as the preimage of a certain map f_{χ} (f in the referenced proof) under the isomorphism $\delta \colon Z(\mathcal{Q}^c(\mathcal{G}))^* \xrightarrow{\sim} \operatorname{Map}^W(\operatorname{Irr}_p(\mathcal{G}), \mathcal{Q}^c(\Gamma_K)^*)$. This isomorphism is in fact $G_{\mathbb{Q}_p}$ equivariant if one endows its codomain with the natural left $G_{\mathbb{Q}_p}$ -action $(\sigma g)(\varsigma) = \sigma(g(\sigma^{-1}\varsigma))$ for $g \in \operatorname{Map}^W(\operatorname{Irr}_p(\mathcal{G}), \mathcal{Q}^c(\Gamma_K)^*)$, $\varsigma \in \operatorname{Irr}_p(\mathcal{G})$ (cf. [RW04] proof of theorem 8). Therefore, it suffices to show that $f_{\mathcal{G}} = \prod_{\chi/\sim_W} f_{\chi}$ is itself invariant under that action. But this is a direct consequence of its definition (3.54) and the equivariant Main Conjecture for the $\mathcal{G}' \in \mathcal{S}$: given $\sigma \in G_{\mathbb{Q}_p}$ and $\chi \in \operatorname{Irr}_p(\mathcal{G})$ with Brauer decomposition as before (where we can again assume $\Gamma \subseteq \mathcal{H}_i \in \mathcal{S}$), one has $\sigma^{-1}\chi = \sum_{i=1}^n z_i \operatorname{ind}_{\mathcal{H}_i}^{\mathcal{G}} \sigma^{-1}\lambda_i$ and hence

$$\begin{aligned} (\sigma f_{\mathcal{G}})(\chi) &= \sigma(f_{\mathcal{G}}(\sigma^{-1}\chi)) = \sigma(f_{\sigma^{-1}\chi}(\sigma^{-1}\chi)) = \sigma\left(\prod_{i=1}^{n} j_{K^{i}}(j_{\sigma^{-1}\lambda_{i}}((F_{S(K^{i}),T(K^{i}),\sigma^{-1}\lambda_{i}}^{\alpha,\beta})^{z_{i}})))\right) \\ &= \prod_{i=1}^{n} j_{K^{i}}(\sigma(j_{\sigma^{-1}\lambda_{i}}((F_{S(K^{i}),T(K^{i}),\sigma^{-1}\lambda_{i}}^{\alpha,\beta})^{z_{i}})))) \\ &= \prod_{i=1}^{n} j_{K^{i}}(j_{\lambda_{i}}((F_{S(K^{i}),T(K^{i}),\lambda_{i}}^{\alpha,\beta})^{z_{i}}))) \\ &= f_{\mathcal{G}}(\chi), \end{aligned}$$

where $K^i = L^{\mathcal{H}_i}_{\infty}$ and the next-to-last equality follows from $F^{\alpha,\beta}_{S(K^i),T(K^i),\lambda_i} = \sigma(F^{\alpha,\beta}_{S(K^i),T(K^i),\sigma^{-1}\lambda_i})$, which is in turn implied by $\operatorname{eMC}(L_{\infty}/K^i, LK^i, S(K^i), T(K^i), \alpha, \beta)$ (see for instance the proof of corollary 3.1.6). The only subtlety lies in the third equality, as it is not guaranteed that the decomposition $\sigma^{-1}\chi = \sum_{i=1}^n z_i \operatorname{ind}_{\mathcal{H}_i}^{\mathcal{G}} \sigma^{-1}\lambda_i$ was the one used to *construct* $f_{\sigma^{-1}\chi}$. However, the cited lemma shows that it is at least a *valid* one for the construction, and the uniqueness of the interpolating elements (cf. remark 2.5.1 ii)) ensures that all choices of a decomposition lead to the same map $f_{\sigma^{-1}\chi}$. The same goes for the last equality. Since the value of $\sigma f_{\mathcal{G}}$ at χ determines its values at all ρ -twists of χ , applying the above argument to a suitable set of representatives of $\operatorname{Irr}_p(\mathcal{G})/\sim_W$ yields $\sigma f_{\mathcal{G}} = f_{\mathcal{G}}$, as required.

We now move on to the equivariant part of the proof. For each $\mathcal{G}' \in \mathcal{S}$, the canonical embeddings $\iota_{\mathcal{G}'} \colon \Lambda(\mathcal{G}') \hookrightarrow \Lambda(\mathcal{G})$ and $\iota_{\mathcal{G}'} \colon \mathcal{Q}(\mathcal{G}') \hookrightarrow \mathcal{Q}(\mathcal{G})$ induce restriction maps on K-theory as explained in section 0.4. Furthermore, the homomorphism res^W from the proof of proposition 3.3.7 gives rise to an arrow $\iota_{Z,\mathcal{G}'}$ making the diagram

commute (cf. [JN20] (3.9)), which in particular factors as

$$[\iota_{Z,\mathcal{G}'}]: Z(\mathcal{Q}(\mathcal{G}))^* / \operatorname{nr}(K_1(\Lambda(\mathcal{G}))) \to Z(\mathcal{Q}(\mathcal{G}'))^* / \operatorname{nr}(K_1(\Lambda(\mathcal{G}')))$$

Now (0.31) implies the commutativity of

with $\nu_{\mathcal{G}'}$ as in (3.55). The proof of proposition 3.3.8, together with (3.48), shows that

$$[\iota_{Z,\mathcal{G}'}]([F_{S,T,\mathcal{G}}^{\alpha,\beta}]) = [F_{S,T,\mathcal{G}'}^{\alpha,\beta}]$$

where $F_{S,T,\mathcal{G}'}^{\alpha,\beta} = \prod_{\chi' \in \operatorname{Irr}_p(\mathcal{G}')/\sim_W} F_{S(K'),T(K'),\chi'}^{\alpha,\beta} \in Z(\mathcal{Q}(\mathcal{G}'))^* \subseteq Z(\mathcal{Q}^c(\mathcal{G}'))^*$. The cornerstone of the present argument is the injectivity of the arrow $\prod_{\mathcal{G}'} [\iota_{Z,\mathcal{G}'}]$ in the last diagram, which is a direct application of [JN20] corollary 10.12 (this is the reason we require $p \neq 2$). Since the left vertical arrow sends $-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}, t^{\alpha})$ to

$$\prod_{\mathcal{G}'\in\mathcal{S}} K_0^{\operatorname{res}}(\iota_{\mathcal{G}'},\iota_{\mathcal{G}'})(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T},t^{\alpha})) = \prod_{\mathcal{G}'\in\mathcal{S}} -\chi_{\Lambda(\mathcal{G}'),\mathcal{Q}(\mathcal{G}')}(\mathcal{C}^{\bullet}_{S(K'),T(K')},t^{\alpha}|_{\mathcal{Q}(\mathcal{G}')})$$

by the proof of 3.3.8, $\nu_{\mathcal{G}}(-\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet},t^{\alpha}))$ is the unique preimage of $\prod_{\mathcal{G}'\in\mathcal{S}}[F_{S,T,\mathcal{G}'}^{\alpha,\beta}]$ under $\prod_{\mathcal{G}'\in\mathcal{S}}[\iota_{Z,\mathcal{G}'}]$, i.e. $[F_{S,T,\mathcal{G}}^{\alpha,\beta}]$.

• $\underline{\text{iii}} \Rightarrow \underline{\text{i}}$: The proof that $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for a given $\chi \in \operatorname{Irr}_p(\mathcal{G})$ is essentially the same as above, the only difference being that the \mathcal{H}_i appearing in the Brauer decomposition of χ may not belong to \mathcal{E}_p . Instead, we reason as follows: each λ_i is a linear character of \mathcal{H}_i (as before, we can assume $\Gamma \subseteq \ker(\lambda_i) \subseteq \mathcal{H}_i$), and hence establishes an isomorphism between $\mathcal{H}_i/\ker(\lambda_i)$ and a finite subgroup of $(\mathbb{Q}_p^c)^*$, which is necessarily cyclic. Set $K^i = L_{\infty}^{\mathcal{H}_i}$ and $\widetilde{K}^i = L_{\infty}^{\ker(\lambda_i)}$, so $K \subseteq K^i \subseteq \widetilde{K}^i \subseteq L$.

Let $\widetilde{K}_{\infty}^{i}$ be the cyclotomic \mathbb{Z}_{p} -extension of \widetilde{K}^{i} and denote $\operatorname{Gal}(L_{\infty}/\widetilde{K}_{\infty}^{i}) = H \cap \ker(\lambda_{i})$ by H_{i} . Then H_{i} is a normal subgroup of \mathcal{H}_{i} and λ_{i} factors through the abelian quotient $\mathcal{H}_{i}/H_{i} = \operatorname{Gal}(\widetilde{K}_{\infty}^{i}/K^{i})$ as a character $\widetilde{\lambda}_{i}$. This quotient has a (central) open subgroup $\operatorname{Gal}(\widetilde{K}_{\infty}^{i}/\widetilde{K}^{i}) \cong \mathbb{Z}_{p}$ with cyclic factor group $\operatorname{Gal}(\widetilde{K}^{i}/K^{i}) \cong \mathcal{H}_{i}/\ker(\lambda_{i})$ and is therefore pelementary in the sense of definition 3.3.10. In particular, $\mathcal{H}_{i}/H_{i} \in \mathcal{E}_{p}$ and the Interpolation Conjecture holds for $\widetilde{\lambda}_{i}$ and thus also for λ_{i} by proposition 3.3.6 i). As a result, so does $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. The fact that $F_{S,T,\mathcal{G}}^{\alpha,\beta} = \prod_{\chi/\sim_{W}} F_{S,T,\chi}^{\alpha,\beta} \in Z(\mathcal{Q}^{c}(\mathcal{G}))^{*}$ is $\mathcal{G}_{\mathbb{Q}_{p}}$ invariant follows from the analogous property for each \mathcal{H}_{i} , which is in turn a consequence of all arrows of

being $G_{\mathbb{Q}_p}$ -equivariant and the equivariant Main Conjecture for $\mathcal{H}_i/H_i \in \mathcal{E}_p$. Recall that the bottom arrow is an isomorphism on the components corresponding to characters which factor

through H_i , and identically 1 elsewhere (cf. lemma 3.3.1 iii)). Note that, given $\sigma \in G_{\mathbb{Q}_p}$ and $\chi' \in \operatorname{Irr}_p(\mathcal{H}_i), \chi'$ factors through H_i if and only if $\sigma \chi'$ does.

The equivariant part of the proof is completely analogous to that for ii) \Rightarrow i), with the caveat that the restriction maps (on K-theory, centres and characters) associated to $\mathcal{G}' \hookrightarrow \mathcal{G}$ now have to be composed with the projection maps arising from $\pi_{\mathcal{G}'/H'} \colon \mathcal{G}' \twoheadrightarrow \mathcal{G}'/H'$ for $\mathcal{G}'/H' \in \mathcal{E}_p$ (these were studied in subsection 3.3.1). The full argument can be found in [JN20], specifically in theorem 10.14 and the results leading up to it. We limit ourselves to pointing out that the key commutative diagram is

where the products in the middle row run over \mathcal{G}' for each $\mathcal{G}'/H' \in \mathcal{E}_p$ and the composition of the vertical arrows on the right is now injective by [JN20] theorem 10.5.

Remark 3.3.15. The proof also shows that, in the notation of the statement of the proposition, the following are equivalent:

- i) $\operatorname{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$ holds for all $\chi \in \operatorname{Irr}_{p}(\mathcal{G})$.
- ii) IC $(L_{\infty}/K', \lambda', L', S(K'), T(K'), \alpha, \beta)$ holds for all $\mathcal{G}' \in \mathcal{S}$ and all linear $\lambda' \in \operatorname{Irr}_p(\mathcal{G}')$.
- iii) $\operatorname{IC}(L'_{\infty}/K', \tilde{\lambda}, L', S(K'), T(K'), \alpha', \beta)$ holds for all $\mathcal{G}'/H' \in \mathcal{E}_p$ and all linear $\tilde{\lambda} \in \operatorname{Irr}_p(\mathcal{G}'/H')$.

Chapter 4

Compatibility with existing conjectures and new cases

As mentioned in the preceding chapters, setting A is, to a certain extent, a generalisation of that of some existing Main Conjectures. Most notably, $\mathcal{G} = \operatorname{Gal}(L_{\infty}/K)$ is not required to be abelian, or L_{∞} totally real. In this final chapter, we make that claim precise by discussing the Main Conjectures of Burns, Kurihara and Sano; and of Ritter and Weiss. Although more details will be given at the beginning of each section, the conclusion is the following:

- In the abelian case, our conjecture recovers a slight modification of that in [BKS17]. This allows us to deduce the validity of $eMC(L_{\infty}/\mathbb{Q},\beta)$ whenever L_{∞}/\mathbb{Q} is abelian.
- For a suitably defined CM extension L_{∞}/K , the minus part of our conjecture recovers that in [RW04] for the maximal totally real subextension L_{∞}^+/K . This implies that the minus part of $eMC(L_{\infty}/K,\beta)$ holds (for extensions of that form) in the so-called $\mu = 0$ case.

While not difficult, the proofs of these these equivalences necessitate some recalls on objects such as determinant functors and Ext-modules of Iwasawa modules. We strive to do so in a moderately self-contained manner.

We point out that the Main Conjecture from chapter 2 is by no means the most general conjecture in Iwasawa theory of number fields in every possible regard. For instance, it is limited to the rank-one cyclotomic case - which suggests a natural direction for further research.

4.1 The conjecture of Burns, Kurihara and Sano

The aim of this section is to prove that the Main Conjecture from chapter 2 is essentially equivalent to that put forward by Burns, Kurihara and Sano in 2017 in the cases where both can be formulated. Specifically, we are alluding to conjecture 3.1 in [BKS17], the setup of which was partially introduced in section 1.5. As explained in said section, the main difference is that the cited article is restricted to the case where L_{∞}/K is an abelian extension, and in turn does not require K_{∞} to be the cyclotomic \mathbb{Z}_p -extension of K. In order to compare the two conjectures, therefore, we will be working in setting B (sometimes simply A) under the additional assumption that \mathcal{G} is abelian. By "essentially equivalent" we mean that a claim about all characters must be replaced by one about almost all characters - in line with our Main Conjecture.

Remark 4.1.1. The commutativity of \mathcal{G} , and therefore of $\Lambda(\mathcal{G})$, has profound implications. For instance, it means that $\mathcal{Q}(\mathcal{G})$ is a commutative semisimple Artinian ring and therefore a finite product of fields by Wedderburn's theorem. In particular, the reduced norm becomes an isomorphism $\operatorname{nr}: K_1(\mathcal{Q}(\mathcal{G})) \xrightarrow{\sim} \mathcal{Q}(\mathcal{G})^*$, which renders the equivariant Main Conjectures with and without uniqueness equivalent (cf. remark 2.5.4 i)). Another consequence is that $\mathcal{G} \cong \Gamma_K \times H$ and all irreducible Artin characters of \mathcal{G} are linear.

Recall that theorem 1.5.4 also holds without assuming the abelianity of \mathcal{G} and states the crucial fact that the complexes employed by Burns, Kurihara and Sano and ourselves are isomorphic in the derived category $\mathcal{D}(\Lambda(\mathcal{G}))$. The relevant invariants of complexes (determinant functors and refined Euler characteristics) are unaffected by such isomorphisms, and thus it suffices to consider our usual main complex $\mathcal{C}^{\bullet}_{S,T}$ moving forward. We need to introduce its finite-level counterpart: in the notation of setting A, given $n \in \mathbb{N}$, we define the complex of $\Lambda(\mathcal{G}_n)$ -modules

$$\mathcal{C}^{\bullet}_{L_n,S,T} = \Lambda(\mathcal{G}_n) \otimes^{\mathbb{L}}_{\Lambda(\mathcal{G})} \mathcal{C}^{\bullet}_{S,T}.$$

By lemma 3.3.5, $C^{\bullet}_{L_n,S,T}$ is isomorphic in $\mathcal{D}(\Lambda(\mathcal{G}_n))$ to the complex introduced in [BKS17] p. 1535, denoted by $\mathcal{B}^{\bullet}_{L_n,S,T}$ in section 1.5 above. As stated in said article, $\mathcal{C}^{\bullet}_{S,T} = \varprojlim_n \mathcal{C}^{\bullet}_{L_n,S,T}$ (cf. also the proof of theorem 1.5.4) and $\mathcal{C}^{\bullet}_{L_n,S,T}$ is a perfect complex which is acyclic outside degrees 0 and 1 and satisfies:

- $H^0(\mathcal{C}^{\bullet}_{L_n,S,T}) \cong \mathbb{Z}_p \otimes \mathcal{O}^*_{L_n,S,T}.$
- $H^1(\mathcal{C}^{\bullet}_{L_n,S,T})$ fits into the short exact sequence of $\Lambda(\mathcal{G}_n)$ -modules

$$0 \to Cl_{L_n,S,T}(p) \to H^1(\mathcal{C}^{\bullet}_{L_n,S,T}) \to \mathcal{X}_{L_n,S} \to 0,$$
(4.1)

where $Cl_{L_n,S,T}(p)$ is the *p*-part of the (S,T)-ray class group $Cl_{L_n,S,T}$ of L_n (see the proof of proposition 1.4.4 for a definition).

Compare this with the cohomology of $\mathcal{C}_{S,T}^{\bullet}$ established in theorem 1.4.6. Since $Cl_{L_n,S,T}$ is finite, sequence (4.1) induces an isomorphism $\mathbb{C}_p \otimes_{\mathbb{Z}_p} H^1(\mathcal{C}_{L_n,S,T}^{\bullet}) \cong \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S}$.

The conjecture of Burns, Kurihara and Sano is formulated in terms of determinant functors of these complexes. A brief introduction to these functors and some of their relevant properties can be found in appendix B. We now explain how to express $\text{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p}^{\mathbb{L}} \mathcal{C}_{L_n,S,T}^{\bullet})$ in terms of $\text{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} H^0(\mathcal{C}_{L_n,S,T}^{\bullet}))$ and $\text{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} H^1(\mathcal{C}_{L_n,S,T}^{\bullet}))$. For simplicity, assume that $\mathcal{C}_{S,T}^{\bullet}$ has a strictly perfect representative of the form

$$\mathcal{P}^{\bullet} = [\overset{0}{\mathcal{P}^{0}} \xrightarrow{1} \overset{1}{\rightarrow} \overset{1}{\mathcal{P}^{1}}] \tag{4.2}$$

in $\mathcal{D}(\Lambda(\mathcal{G}))$ with \mathcal{P}^0 and \mathcal{P}^1 free. This is always the case if the set T is chosen large enough that $\mathcal{O}_{L_n,S,T}^*$ is \mathbb{Z} -torsion free for all n (cf. [BKS17] p. 1538 or [BH21] equation (12)). Since neither conjecture depends on T (cf. subsection 3.2.3 and [BKS17] p. 1541), this assumption is harmless. In particular, $\mathcal{C}_{L_n,S,T}^{\bullet}$ has a strictly perfect representative $\mathcal{P}_n^{\bullet} = [\mathcal{P}_n^0 \to \mathcal{P}_n^1]$ in $\mathcal{D}(\Lambda(\mathcal{G}_n))$ with \mathcal{P}_n^0 and \mathcal{P}_n^1 free.

Split the four-term exact sequence induced by (4.2) into two short ones:

$$H^{0}(\mathcal{C}^{\bullet}_{L_{n},S,T}) \longleftrightarrow \mathcal{P}^{0}_{n} \xrightarrow{} \mathcal{P}^{1}_{n} \xrightarrow{} H^{1}(\mathcal{C}^{\bullet}_{L_{n},S,T})$$

After applying $\mathbb{C}_p \otimes_{\mathbb{Z}_p} -$, the semisimplicity of $\mathbb{C}_p[\mathcal{G}_n]$ ensures the existence of splittings

which in turn induce isomorphisms

$$\operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} H^0(\mathcal{C}^{\bullet}_{L_n,S,T})) \otimes_{\mathbb{C}_p[\mathcal{G}_n]} \operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} W) \xrightarrow{\sim} \operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{P}^0_n)$$
(4.3)

and

$$\operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} W) \otimes_{\mathbb{C}_p[\mathcal{G}_n]} \operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} H^1(\mathcal{C}^{\bullet}_{L_n,S,T})) \xrightarrow{\sim} \operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{P}^1_n)$$
(4.4)

as in (B.4). This allows for the following definition from [BKS17] paragraph 3B1:

Definition 4.1.2. Setting A with \mathcal{G} abelian. Let $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$ be a ring isomorphism. Assume T is large enough that $\mathcal{C}_{S,T}^{\bullet}$ has a strictly perfect representative \mathcal{P}^{\bullet} of the form (4.2).

Given $\chi \in \operatorname{Irr}_p(\mathcal{G})$, choose $n \in \mathbb{N}$ such that χ factorises over \mathcal{G}_n . Set $\mathcal{P}_n^{\bullet} = \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{P}^{\bullet}$, which is a strictly perfect representative of $\mathcal{C}_{L_n,S,T}^{\bullet}$. We define the \mathbb{Z}_p -homomorphism λ_{χ} as the composition

$$\begin{split} \lambda_{\chi} \colon \mathrm{Det}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}) &= \mathrm{Det}_{\Lambda(\mathcal{G})}(\mathcal{P}^{\bullet}) \\ &\to \mathrm{Det}_{\Lambda(\mathcal{G}_{n})}(\mathcal{P}_{n}^{\bullet}) \\ &\hookrightarrow \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}}^{\mathbb{L}} \mathcal{P}_{n}^{\bullet}) \\ &= \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{P}_{n}^{0}) \otimes_{\mathbb{C}_{p}[\mathcal{G}_{n}]} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{P}_{n}^{1})^{-1} \\ &\stackrel{\sim}{\to} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} H^{0}(\mathcal{C}_{L_{n},S,T}^{\bullet})) \otimes_{\mathbb{C}_{p}[\mathcal{G}_{n}]} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} H^{1}(\mathcal{C}_{L_{n},S,T}^{\bullet}))^{-1} \\ &\stackrel{\sim}{\to} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes \mathcal{O}_{L_{n},S,T}^{*}) \otimes_{\mathbb{C}_{p}[\mathcal{G}_{n}]} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n},S})^{-1} \\ &\stackrel{\sim}{\to} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n},S}) \otimes_{\mathbb{C}_{p}[\mathcal{G}_{n}]} \mathrm{Det}_{\mathbb{C}_{p}[\mathcal{G}_{n}]}(\mathbb{C}_{p} \otimes_{\mathbb{Z}_{p}} \mathcal{X}_{L_{n},S})^{-1} \\ &\stackrel{\sim}{\to} \mathbb{C}_{p}[\mathcal{G}_{n}] \\ & \twoheadrightarrow \mathbb{C}_{p}. \end{split}$$

Here the first two arrows are induced by extension of scalars (cf. appendix B), whereas the fourth one is given by (4.3), (4.4) and evaluation $\operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p\otimes_{\mathbb{Z}_p}W)\otimes_{\mathbb{C}_p[\mathcal{G}_n]}\operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\mathbb{C}_p\otimes_{\mathbb{Z}_p}W)^{-1} \xrightarrow{\sim} \mathbb{C}_p[\mathcal{G}_n].$ The last three arrows are, in order, the determinant $\operatorname{Det}_{\mathbb{C}_p[\mathcal{G}_n]}(\lambda_{n,S}^\beta)$ of the *p*-adic Dirichlet regulator $\lambda_{n,S}^\beta$ (definition 2.3.3), the evaluation map and the homomorphism induced by χ (i.e. the projection to $e(\chi)\mathbb{C}_p[\mathcal{G}_n] \cong \mathbb{C}_p$).

- Remark 4.1.3. i) The definition in [BKS17] passes to the finite level $G_{\chi} = \text{Gal}(L_{\infty}^{\ker \chi}/K)$ instead of \mathcal{G}_n . However, the resulting map λ_{χ} is identical by virtue of the surjection $\mathbb{C}_p[\mathcal{G}_n] \twoheadrightarrow \mathbb{C}_p[G_{\chi}]$. In particular, it is independent of the choice of n.
 - ii) Instead of irreducible (hence linear) *p*-adic Artin characters of \mathcal{G} , the article considers linear \mathbb{C} -valued characters with open kernel the collection of which is denoted by $\hat{\mathcal{G}}$. Note that β induces a group isomorphism $\operatorname{Irr}_p(\mathcal{G}) \xrightarrow{\sim} \hat{\mathcal{G}}, \chi \mapsto \beta \chi$, and therefore everything can be formulated in terms of $\operatorname{Irr}_p(\mathcal{G})$ instead.
- iii) The homomorphism λ_{χ} is known to be independent of the choice of a strictly perfect representative \mathcal{P}^{\bullet} of $\mathcal{C}_{S,T}^{\bullet}$, as well as splittings σ_0 and σ_1 inducing sequences (4.3) and (4.4), by the properties of determinant functors (see for instance [CKV93] p. 71). However, it is not independent of the choice of $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}_p$ - and hence neither is the conjecture in [BKS17].
- iv) The condition on T is not necessary for the definition: the same map can be constructed using an arbitrary representative \mathcal{P}^{\bullet} (this is indeed the case in [BKS17], although the explicit construction is not given). This assumption, which will be dropped in the statement of the conjecture below, appears above only to simplify the description of the maps. Recall that both conjectures are independent of T.

We are now in a position to state conjecture 3.1 from $[BKS17]^{22}$. However, in order to prove equivalence with our conjecture (in the abelian case), it is necessary to apply a minor tweak to the former. We refer to this as the **modified BKS conjecture**:

Conjecture (BKS'($L_{\infty}/K, L, S, T, \beta$)). Setting A with \mathcal{G} abelian.

There exists a generator²³ $\mathcal{L}_{S,T}$ of the $\Lambda(\mathcal{G})$ -module $\operatorname{Det}_{\Lambda(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T})$ such that, for almost all $\psi \in \operatorname{Irr}_p(\mathcal{G})$, one has $\lambda_{\psi}(\mathcal{L}_{S,T}) = \beta^{-1}(L^*_{K,S,T}(\beta\psi^{-1}, 0)).$

Remark 4.1.4. Although perhaps not entirely evident, we claim that the above conjecture is indeed equivalent to the outcome of replacing "every $\psi \in \hat{\mathcal{G}}_{\chi}$ " by "almost every $\psi \in \hat{\mathcal{G}}_{\chi}$ " in [BKS17] conjecture 3.1. We recall the following definitions therein, which will not be used beyond the scope

²²The cited, published article contains the following erratum: the \hat{G}_{χ} in the statement should read $\hat{\mathcal{G}}_{\chi}$. This has been confirmed by the authors and is in fact correct in available versions of the article prior to publication.

²³Despite the notation, which is taken from [BKS17], this element is unrelated to the $\mathcal{L}_{S,T}(\chi, f) \in \mathbb{C}_p$ from Stark's conjecture Stark^T($L/K, \chi, f, S, T$).

of this remark: for $\psi \in \operatorname{Irr}_p(\mathcal{G})$, let

$$r_{\beta\psi,S} = \begin{cases} |\{v \in S \colon \mathcal{G}_v \subseteq \ker(\psi)\}|, & \psi \neq \mathbb{1} \\ |S| - 1, & \psi = \mathbb{1}. \end{cases}$$

This is precisely the order of vanishing of the Artin *L*-function $L_{K,S,T}(\beta\psi, s)$ at s = 0 (i.e. $r_{\beta\psi}(S)$ in the notation of section 0.3) - see for instance [BKS17] subsection 2A. It is immediate from the definition that $r_{\beta\psi,S} = r_{\beta\psi,S}$.

Given a character $\eta \in \operatorname{Irr}_p(H) = \operatorname{Hom}(H, \mathbb{C}_p^*)$ (in the article: a $\chi \in \hat{\Delta}$), let $\mathcal{G}_\eta = \mathcal{G}/\ker(\eta)$ (where $\ker(\eta)$ is really a subgroup of H). We regard the canonical embedding $\operatorname{Irr}_p(\mathcal{G}_\eta) \hookrightarrow \operatorname{Irr}_p(\mathcal{G})$ as an inclusion. Set

$$r_{\eta} = |\{v \in S \colon \mathcal{G}_v \subseteq \ker(\eta)\}|,$$

which is the number of places in S which split completely in $L_{\infty}^{\ker(\eta)}/K$. Since K_{∞}/K is the cyclotomic \mathbb{Z}_p -extension, that can only be the case for archimedean places.

Assume first BKS' $(L_{\infty}/K, L, S, T, \beta)$. Let $\eta \in \operatorname{Irr}_{p}(H)$. By assumption, one has²⁴

$$\lambda_{\psi}(\mathcal{L}_{S,T}) = \beta^{-1}(L^*_{K,S,T}(\beta\psi^{-1}, 0))$$

for almost all $\psi \in \operatorname{Irr}_p(\mathcal{G})$. In particular, the same holds for almost all $\psi \in \operatorname{Irr}_p(\mathcal{G}_\eta) \subset \operatorname{Irr}_p(\mathcal{G})$ satisfying $r_\eta = r_{\beta\psi,S}$ (or any other condition). This shows one implication.

The converse is subtler because of the way characters are grouped in terms of $\operatorname{Irr}_p(\mathcal{G}_\eta)$ in [BKS17] conjecture 3.1. We first point out that it is not necessarily true that, for a given $\eta \in \operatorname{Irr}_p(H)$, one has $r_\eta = r_{\beta\psi,S}$ for almost all $\psi \in \operatorname{Irr}_p(\mathcal{G}_\eta)$. For instance, if $p \neq 2, L = \mathbb{Q}(\zeta_p), K = L^+$ (the maximal totally real subfield of L), S is arbitrary and η is the non-trivial character of $H \cong \mathbb{Z}/2\mathbb{Z}$, then $r_\eta = 0$ but any $\psi \in \operatorname{Irr}_p(\mathcal{G}_\eta) = \operatorname{Irr}_p(\mathcal{G})$ which is trivial on H will have $r_{\beta\psi,S} > 0$ (as H contains the decomposition groups of all archimedean places) and there are infinitely many such ψ .

However, the following does hold: given $\eta \in \operatorname{Irr}_p(H)$, the equality $r_\eta = r_{\beta\psi,S}$ is satisfied for almost all $\psi \in \operatorname{Irr}_p(\mathcal{G})$ such that $\operatorname{res}_H^{\mathcal{G}} \psi = \eta$ (this latter condition is stricter than $\psi \in \operatorname{Irr}_p(\mathcal{G}_\eta) \subseteq \operatorname{Irr}_p(\mathcal{G})$, which just amounts to $\ker(\eta) \subseteq \ker(\psi)$). To see why, observe that if $\psi \in \operatorname{Irr}_p(\mathcal{G})$ restricts to η , one has

$$\{v \in S_{\infty} \colon \mathcal{G}_{v} \subseteq \ker(\psi)\} = \{v \in S_{\infty} \colon \mathcal{G}_{v} \subseteq \ker(\eta)\},\$$

as all archimedean places have finite decomposition group and H contains all elements of \mathcal{G} of finite order. Therefore, any potential difference between $r_{\beta\psi,S}$ and r_{η} must come from the non-archimedean places, that is,

 $r_{\beta\psi,S} \neq r_{\eta} \quad \Longleftrightarrow \quad \{v \in S_f \colon \mathcal{G}_v \subseteq \ker(\psi)\} \neq \emptyset.$

But $\mathcal{G}_v \subseteq \ker(\psi)$ implies $\bigcap_{v \in S_f} \mathcal{G}_v \subseteq \ker(\psi)$, which is only satisfied by finitely many ψ (precisely those which factor over the finite quotient $\mathcal{G} / \bigcap_{v \in S_f} \mathcal{G}_v$).

²⁴In the notation of the article, $L_{K,S,T}^{(r)}(\beta\psi^{-1},0)$ is the coefficient of s^r in the series expansion around 0, rather than the value of the *r*-th derivative at 0 (see p. 1533). In particular, $L_{K,S,T}^{(r_{\beta\psi,S})}(\beta\psi^{-1},0) = L_{K,S,T}^*(\beta\psi^{-1},0)$.

Having established this, the remaining implication follows easily: assume [BKS17] conjecture 3.1 with "every $\psi \in \hat{\mathcal{G}}_{\eta}$ " replaced by "almost every $\psi \in \hat{\mathcal{G}}_{\eta}$ ". The set $\operatorname{Irr}_{p}(\mathcal{G})$ is the disjoint union of the equivalence classes of the relation $\psi \sim \psi' \Leftrightarrow \operatorname{res}_{H}^{\mathcal{G}} \psi = \operatorname{res}_{H}^{\mathcal{G}} \psi'$ (although unnecessary, we know this is precisely the *W*-twist equivalence relation \sim_{W}). Let C_{η} be the the class represented by $\eta \in \operatorname{Irr}_{p}(\mathcal{G})$, which in particular contains $\operatorname{Irr}_{p}(\mathcal{G}_{\eta})$. The assumption implies $\lambda_{\psi}(\mathcal{L}_{S,T}) = \beta^{-1}(L_{K,S,T}^{*}(\beta\psi^{-1}, 0))$ for all $\psi \in \operatorname{Irr}_{p}(\mathcal{G}_{\eta})$ satisfying $r_{\eta} = r_{\beta\psi,S}$, which we have shown is the case for almost all $\psi \in C_{\eta}$. Now $\operatorname{BKS'}(L_{\infty}/K, L, S, T, \beta)$ follows from the fact that there are only finitely many classes C_{η} . \Box

In preparation for the central result of this section, we prove two technical results. The first one concerns determinant functors and (somewhat in disguise) refined Euler characteristics:

Proposition 4.1.5. Let $\varphi \colon R \hookrightarrow S$ be an injective homomorphism of commutative rings such such that S is a flat right R-module (via φ) and semisimple Artinian. Suppose given a strictly perfect (cochain) complex \mathcal{P}^{\bullet} of R-modules and an isomorphism of S-modules $t \colon S \otimes_R \mathcal{P}^{\text{even}} \xrightarrow{\sim} S \otimes_R \mathcal{P}^{\text{odd}}$ (in the notation of the beginning of section 2.1). Consider the composition

where arrows are defined as in appendix B. Then, for any generator g of $\text{Det}_R(\mathcal{P}^{\bullet})$ as an R-module (if it exists), one has $\varphi_t(g) \in S^* \subseteq S$.

i) Assume that $[\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}]$ lies in the image of the connecting map $\partial \colon K_1(S) \to K_0(R, S)$ (cf. (0.27)). Then any g as above satisfies

$$\partial(\varphi_t(g)) = [\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}], \qquad (4.6)$$

where $\varphi_t(g) \in S^*$ is regarded as an element of $K_1(S)$ via the canonical injection $S^* \hookrightarrow K_1(S)$ (cf. [CR87] (40.27)).

ii) Assume that $[\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}] = \partial(x)$ for some $x \in K_1(S)$. Then there exists a generator g of $\text{Det}_R(\mathcal{P}^{\bullet})$ as an R-module such $\varphi_t(g) = \operatorname{nr}(x) \in S^*$.

Proof. The first claim is immediate. If g generates $\text{Det}_R(\mathcal{P}^{\bullet})$ over R, then $1 \otimes g$ is a generator of $S \otimes_R \text{Det}_R(\mathcal{P}^{\bullet}) = \text{Det}_S(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$ over S. The remaining maps defining φ_t are isomorphisms of S-modules, and thus $\varphi_t(g) \in S$ generates S as an S-module - that is, it is a unit.

Recall that, since S is commutative semisimple, the reduced norm $\operatorname{nr}: K_1(S) \to S^*$ is an isomorphism (cf. (0.32)) with obvious inverse $s \mapsto [(s)]$.

i) We can assume without loss of generality that $\mathcal{P}^{\text{even}}$ and \mathcal{P}^{odd} are free: since $[\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}]$ lies in im(∂), its image $[\mathcal{P}^{\text{even}}] - [\mathcal{P}^{\text{odd}}]$ in $K_0(R)$ is zero by exactness of (0.27). This implies there exists a finitely generated projective *R*-module *N* such that $\mathcal{P}^{\text{even}} \oplus N \cong \mathcal{P}^{\text{odd}} \oplus N$, and we can assume this sum to be free. Now define

$$\widetilde{\mathcal{P}}^{\bullet} = \mathcal{P}^{\bullet} \oplus [\overset{0}{N} \overset{\mathrm{Id}}{\longrightarrow} \overset{1}{N}].$$

Then the relations of $K_0(R, S)$ immediately imply $[\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}] = [\widetilde{\mathcal{P}}^{\text{even}}, t \oplus \text{Id}, \widetilde{\mathcal{P}}^{\text{odd}}]$, and the isomorphisms

$$\operatorname{Det}_R(\widetilde{\mathcal{P}}^{\operatorname{even}}) = \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}} \oplus N) \xrightarrow{\sim} \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}}) \otimes_R \operatorname{Det}_R(N)$$

 $\operatorname{Det}_R(\widetilde{\mathcal{P}}^{\operatorname{odd}}) \xrightarrow{\sim} \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}}) \otimes_R \operatorname{Det}_R(N)$ (analogously) and $ev \colon \operatorname{Det}_R(N) \otimes_R \operatorname{Det}_R(N)^{-1} \xrightarrow{\sim} R$ induce an *R*-isomorphism *f* making the diagram

$$\begin{array}{ccc} \operatorname{Det}_{R}(\mathcal{P}^{\bullet}) & \stackrel{\varphi_{t}}{\longrightarrow} S \\ & & & & & \\ & & & & \\ & & & & \\ \operatorname{Det}_{R}(\widetilde{\mathcal{P}}^{\bullet}) & \stackrel{\varphi_{t \oplus \operatorname{Id}}}{\longrightarrow} S \end{array}$$

commute. Hence it suffices to show that $\partial(\varphi_{t \oplus \mathrm{Id}}(f(g))) = [\widetilde{\mathcal{P}}^{\mathrm{even}}, t \oplus \mathrm{Id}, \widetilde{\mathcal{P}}^{\mathrm{odd}}]$, which allows us to assume $\mathcal{P}^{\mathrm{even}}$ and $\mathcal{P}^{\mathrm{odd}}$ to be free from now on. In particular, $\mathrm{Det}_R(\mathcal{P}^{\mathrm{even}})$ and $\mathrm{Det}_R(\mathcal{P}^{\mathrm{odd}})$ are free *R*-modules of rank 1.

In order to prove (4.6), we perform some simplification steps. Firstly, the theory of determinant functors shows that $[\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}] = [\text{Det}_R(\mathcal{P}^{\text{even}}), \text{Det}_S(t), \text{Det}_R(\mathcal{P}^{\text{odd}})]$. Choose and fix an isomorphism of S-modules ι : $\text{Det}_S(S \otimes_R \mathcal{P}^{\text{even}}) \xrightarrow{\sim} S$. The commutative diagram

where horizontal arrows denote multiplication by the scalar $\varphi_t(g) \in S$, implies that

$$[\operatorname{Det}_R(\mathcal{P}^{\operatorname{even}}), \varphi_t(g), \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}})] = [R, \varphi_t(g), R] = \partial(\varphi_t(g))$$

in $K_0(R, S)$ (see relation i) in the definition of $K_0(R, S)$ from section 0.4). This reduces the proof of (4.6) to showing that

$$[\operatorname{Det}_{R}(\mathcal{P}^{\operatorname{even}}), \varphi_{t}(g), \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{even}})] = [\operatorname{Det}_{R}(\mathcal{P}^{\operatorname{even}}), \operatorname{Det}_{S}(t), \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{odd}})],$$

or equivalently,

$$[\operatorname{Det}_R(\mathcal{P}^{\operatorname{even}}), \operatorname{Det}_S(t) \circ \varphi_t(g)^{-1}, \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}})] = 0.$$

Equation (0.26) implies it suffices in turn to prove

$$\varphi_t(g)^{-1} \cdot \operatorname{Det}_S(t)(1 \otimes \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}})) = 1 \otimes \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}}) \subseteq S \otimes \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}}), \qquad (4.7)$$

as this would induce an isomorphism of R-modules

$$\operatorname{Det}_R(\mathcal{P}^{\operatorname{even}}) \cong 1 \otimes \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}}) \xrightarrow{\varphi_t(g)^{-1} \cdot \operatorname{Det}_S(t)} 1 \otimes \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}}) \cong \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}})$$

by the injectivity of $R \hookrightarrow S$.

Fix an arbitrary generator g^{odd} of $\text{Det}_R(\mathcal{P}^{\text{odd}})$ as an *R*-module and consider the generator $g^{\text{odd},*}$ of $\text{Det}_R(\mathcal{P}^{\text{odd}})^{-1} = \text{Hom}_R(\text{Det}_R(\mathcal{P}^{\text{odd}}), R)$ uniquely determined by $g^{\text{odd},*}(g^{\text{odd}}) = 1$ as

described in lemma B.0.1. Then $g = g^{\text{even}} \otimes g^{\text{odd},*}$ for some generator g^{even} of $\text{Det}_R(\mathcal{P}^{\text{even}})$. To see this, choose any generator \tilde{g}^{even} of $\text{Det}_R(\mathcal{P}^{\text{even}})$ and note that, since both g and $\tilde{g}^{\text{even}} \otimes g^{\text{odd},*}$ generate $\text{Det}_R(\mathcal{P}^{\bullet})$, they must differ by a unit in R.

The image of g in $\operatorname{Det}_S(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}) = (S \otimes_R \operatorname{Det}_R(\mathcal{P}^{\operatorname{even}})) \otimes_S (S \otimes_R \operatorname{Det}_R(\mathcal{P}^{\operatorname{odd}})^{-1})$ is simply $(1 \otimes g^{\operatorname{even}}) \otimes (1 \otimes g^{\operatorname{odd},*})$, which the vertical arrow in (4.5) sends to

$$\operatorname{Det}_{S}(t)(1 \otimes g^{\operatorname{even}}) \otimes (1 \otimes g^{\operatorname{odd},*}) \in (S \otimes_{R} \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{odd}})) \otimes_{S} (S \otimes_{R} \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{odd}})^{-1}).$$

But $\text{Det}_S(t)(1 \otimes g^{\text{even}})$ is equal to $(s \otimes g^{\text{odd}})$ for some $s \in S$, in which case one has

$$\varphi_t(g) = ev(\operatorname{Det}_S(t)(1 \otimes g^{\operatorname{even}}) \otimes (1 \otimes g^{\operatorname{odd},*})) = s$$

by definition of the dual generator $g^{\text{odd},*}$. Since g^{even} and g^{odd} generate $\text{Det}_R(\mathcal{P}^{\text{even}})$ and $\text{Det}_R(\mathcal{P}^{\text{odd}})$ over R respectively and $\text{Det}_S(t)$ is S-(and hence R-)equivariant, (4.7) follows.

ii) By the same argument as above, we can assume without loss of generality that $\mathcal{P}^{\text{even}}$ and \mathcal{P}^{odd} are free *R*-modules. In particular, $\text{Det}_R(\mathcal{P}^{\bullet})$ is free of rank 1 and we may choose a generator g'. Part i) implies $\partial(\varphi_t(g')) = [\mathcal{P}^{\text{even}}, t, \mathcal{P}^{\text{odd}}] = \partial(x)$.

By the exact sequence of K-theory (0.27), there exists a $y \in K_1(R)$ such that $x = \varphi_t(g')K_1(\varphi)(y)$. Then $g = \det(y)g'$ is a generator of $\operatorname{Det}_R(\mathcal{P}^{\bullet})$ which satisfies

$$\varphi_t(g) = \det(y)\varphi_t(g') = \operatorname{nr}(K_1(\varphi)(y)\varphi_t(g')) = \operatorname{nr}(x).$$

- Remark 4.1.6. i) The connecting homomorphism $\partial: K_1(\mathcal{Q}(\mathcal{G})) \to K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ is surjective in our case of interest, as explained in remark 2.5.4 ii). Therefore, the hypothesis in part i) of the proposition is automatically satisfied.
- ii) The only reason for the flatness assumption is to ensure $S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$ coincides with the result of applying $S \otimes_R -$ to \mathcal{P}^{\bullet} degree-wise, which allows us to state the proposition in a convenient form for our purposes. However, the proof yields a completely analogous result without the flatness requirement after replacing $\operatorname{Det}_R(\mathcal{P}^{\bullet}) \hookrightarrow \operatorname{Det}_S(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$ by

$$\operatorname{Det}_R(M) \otimes_R \operatorname{Det}_R(N)^{-1} \hookrightarrow \operatorname{Det}_S(S \otimes_R M) \otimes_S \operatorname{Det}_S(S \otimes_R N)^{-1},$$

where M and N are two finitely generated projective R-modules playing the role of $\mathcal{P}^{\text{even}}$ and \mathcal{P}^{odd} , respectively.

The second preparatory result, which is rather specific to our setting, will be instrumental in relating the maps in our Main Conjecture to the λ_{χ} from definition 4.1.2 used by Burns, Kurihara and Sano:

Lemma 4.1.7. Setting A with \mathcal{G} abelian. Let M and N be free $\Lambda(\mathcal{G})$ -modules of the same rank and $\tau: M \to N$ a homomorphism between them such that $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} \tau$ is a $\mathcal{Q}(\mathcal{G})$ -isomorphism. For a given $\chi \in \operatorname{Irr}_p(\mathcal{G})$, choose $n \in \mathbb{N}$ large enough that χ factorises through \mathcal{G}_n . Denote by λ_l and λ_r the composition of the left and right columns of the diagram

respectively. Then, for any generator g of $\text{Det}_{\Lambda(\mathcal{G})}(M) \otimes_{\Lambda(\mathcal{G})} \text{Det}_{\Lambda(\mathcal{G})}(N)^{-1}$ over $\Lambda(\mathcal{G})$, one has $\lambda_r(g) \in \mathcal{Q}(\mathcal{G})^*$ and

$$\lambda_l(g) = ev_{\gamma_{\chi}}(\psi_{\chi}(\operatorname{nr}^{-1}(\lambda_r(g)))),$$

where the maps $\mathcal{Q}(\mathcal{G})^* \xrightarrow{\operatorname{nr}^{-1}} K_1(\mathcal{Q}(\mathcal{G})) \xrightarrow{\psi_{\chi}} \mathcal{Q}^c(\Gamma_{\chi})^* \xrightarrow{ev_{\gamma_{\chi}}} \mathbb{Q}_p^c \cup \{\infty\}$ are given by remark 4.1.1 and definitions 2.5.3 and 2.4.1.

Proof. Recall the notation $-^*$ for dual generators of determinant modules introduced in lemma B.0.1. Since $\text{Det}_{\Lambda(\mathcal{G})}(M)$ and $\text{Det}_{\Lambda(\mathcal{G})}(N)^{-1}$ are free rank-one $\Lambda(\mathcal{G})$ -modules, any g as in the statement is necessarily of the form $g = \mu \otimes \nu^*$ for some generators μ and ν of $\text{Det}_{\Lambda(\mathcal{G})}(M)$ and $\text{Det}_{\Lambda(\mathcal{G})}(N)$, respectively. The two bottom arrows in the right column of the diagram are $\mathcal{Q}(\mathcal{G})$ -isomorphisms, and therefore g is mapped under λ_r to a $\mathcal{Q}(\mathcal{G})$ -generator of $\mathcal{Q}(\mathcal{G})$, that is, a unit $\lambda_r(g) \in \mathcal{Q}(\mathcal{G})^*$.

Consider the ring homomorphism $\overline{-} \colon \Lambda(\mathcal{G}) \to \Lambda(\mathcal{G}_n) \hookrightarrow \mathbb{Q}_p^c[\mathcal{G}_n]$, which induces a homomorphism of $\Lambda(\mathcal{G})$ -modules $\overline{-} \colon T \to \mathbb{Q}_p^c[\mathcal{G}_n] \otimes_{\Lambda(\mathcal{G})} T = \mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} T_{\Gamma^{p^n}}$ for any $\Lambda(\mathcal{G})$ -module T. The elements $\overline{\mu}$ and $\overline{\nu^*} = \overline{\nu}^*$ generate $\operatorname{Det}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} M_{\Gamma^{p^n}})$ and $\operatorname{Det}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} N_{\Gamma^{p^n}})^{-1}$ over $\mathbb{Q}_p^c[\mathcal{G}_n]$, respectively (see for instance (B.2)). Similarly, $1 \otimes \mu$ is a $\mathcal{Q}(\mathcal{G})$ -generator of $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} \operatorname{Det}_{\Lambda(\mathcal{G})}(M)$ (which coincides with $\operatorname{Det}_{\mathcal{Q}(\mathcal{G})}(\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} M)$), and analogously for $(1 \otimes \nu)^* = 1 \otimes \nu^*$.

Since ν generates $\operatorname{Det}_{\Lambda(\mathcal{G})}(N)$, there exists an $x \in \Lambda(\mathcal{G})$ such that $\operatorname{Det}_{\Lambda(\mathcal{G})}(\tau)(\mu) = x\nu$. It follows that $\operatorname{Det}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(\mathbb{Q}_p^c \otimes_{\mathbb{Z}_p} \tau_{\Gamma^{p^n}})(\mu) = \overline{x\nu}$ and $\operatorname{Det}_{\mathcal{Q}(\mathcal{G})}(\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} \tau)(1 \otimes \mu) = x(1 \otimes \nu) = x \otimes \nu$. Therefore, the image of $g = \mu \otimes \nu^*$ in $\mathbb{Q}_p^c[\mathcal{G}_n]$ (that is, under λ_l minus the last arrow) is $ev(\overline{x\nu} \otimes \overline{\nu}^*) = \overline{x}$, whereas $\lambda_r(g) = \lambda_r(\mu \otimes \nu^*) = ev(x(1 \otimes \overline{\nu}) \otimes (1 \otimes \overline{\nu})^*) = x \in \mathcal{Q}(\mathcal{G})$.

Finally, we apply (3.21) and (3.22) to $P = \Lambda(\mathcal{G})$ and f = x (the endomorphism of $\Lambda(\mathcal{G})$ given by left multiplication by x) to conclude that

$$\lambda_l(g) = \chi(\overline{x}) = \det(\operatorname{Hom}_{\mathbb{Q}_p^c[\mathcal{G}_n]}(V_{\chi}, \overline{f})) = ev_{\gamma_{\chi}}(\psi_{\chi}([\mathcal{Q}(\mathcal{G}), \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} f])) = ev_{\gamma_{\chi}}(\psi_{\chi}(\operatorname{nr}^{-1}(\lambda_r(g)))),$$

as desired.

We are now in a position to prove the main result of this section:

Theorem 4.1.8. Setting B with \mathcal{G} abelian. The modified BKS conjecture $BKS'(L_{\infty}/K, L, S, T, \beta)$ holds if and only if $eMC(L_{\infty}/K, L, S, T, \alpha, \beta)$ does.

Proof. In this proof, we denote $\text{Det}_R(-)$ for a commutative ring R by $D_R(-)$. In the diagrams and displayed equations, we will simplify the notation of $\Lambda(\mathcal{G}), \Lambda(\mathcal{G}_n), \mathcal{Q}(\mathcal{G})$ and $\mathcal{Q}^c(\mathcal{G})$ to $\Lambda, \Lambda_n, \mathcal{Q}$ and \mathcal{Q}^c respectively. In diagrams, we will furthermore remove the subscript R from \otimes_R when the involved modules are of the form $\text{Det}_R(M)$ for some R-module M.

As explained after equation (4.2), we may assume $\mathcal{C}_{S,T}^{\bullet}$ has a representative in $\mathcal{D}(\Lambda(\mathcal{G}))$ of the form $\mathcal{P}^{\bullet} = [\mathcal{P}^0 \to \mathcal{P}^1]$ (in degrees 0 and 1), with both \mathcal{P}^i free, by replacing the original set T with a suitable one, which does not affect the validity of either conjecture. We set $\mathcal{P}_n^{\bullet} = \Lambda(\mathcal{G}_n) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{P}^{\bullet}$.

The first step is to bridge some differences in the formulation of both conjectures. Let $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfy the kernel condition (KC). Choose an $n \in \mathbb{N}$ such that χ factors over $\mathcal{G}_n = \mathcal{G}/\Gamma^{p^n}$ and let $\chi, \mathbb{C}_p - \text{ denote } e(\chi) \cdot \mathbb{C}_p \otimes_{\mathbb{Z}_p} -$, where $e(\chi) \in \mathbb{Q}_p^c[\mathcal{G}_n] \subseteq \mathbb{C}_p[\mathcal{G}_n]$ is the primitive central idempotent corresponding to (the projection of) χ . Then one has the following commutative diagram:



Here the bottom arrow is to be understood as multiplication by the regulator $R_S^{\beta}(\alpha, \chi) \in \mathbb{C}_p^*$. Commutativity follows from the fact that the right column is defined by essentially the same maps as λ_{χ} (cf. definition 4.1.2, including the implicit choice of splittings σ_0 and σ_1), the difference being that this time we take χ -parts *before* the last step (which does not affect the resulting composition), and we choose as the isomorphism

$$D_{\mathbb{C}_p[\mathcal{G}_n]}(_{\chi,\mathbb{C}_p}H^{\operatorname{even}}(\mathcal{C}^{\bullet}_{L_n,S,T})) \xrightarrow{\sim} D_{\mathbb{C}_p[\mathcal{G}_n]}(_{\chi,\mathbb{C}_p}H^{\operatorname{odd}}(\mathcal{C}^{\bullet}_{L_n,S,T}))$$

the map $\mathbb{D}_{\mathbb{C}_p[\mathcal{G}_n]}((\chi,\mathbb{C}_p\varphi_n^{\alpha})^{-1})$ rather than $\mathbb{D}_{\mathbb{C}_p[\mathcal{G}_n]}(\chi,\mathbb{C}_p\lambda_S^{\beta})$. Since the Stark-Tate regulator $R_S^{\beta}(\alpha,\chi)$ is precisely the determinant of the \mathbb{C}_p -action of $\chi,\mathbb{C}_p\lambda_S^{\beta}\circ(\chi,\mathbb{C}_p\varphi_n^{\alpha})$ on $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\chi},H^{\operatorname{odd}}(\mathcal{C}_{L_n,S,T}^{\bullet}))$,

and determinant functors take endomorphisms of free modules to actual matrix determinants (cf. (B.3)), a simple computation shows the diagram commutes indeed. This relies crucially on the fact that $\chi(1) = 1$ by virtue of the abelianity of \mathcal{G} .

Consider now the right column of said diagram. Since it does not feature the Dirichlet regulator map, all arrows and objects after $D_{\Lambda}(\mathcal{C}_{S,T}^{\bullet})$ can be replaced by their analogues with scalars in \mathbb{Q}_{p}^{c} instead of \mathbb{C}_{p} . To do so, we also replace, $\chi_{,\mathbb{C}_{p}}$ by $\chi^{-} = e(\chi) \cdot \mathbb{Q}_{p}^{c} \otimes_{\mathbb{Z}_{p}} -$. Here we tacitly use the fact that splittings exist over $\mathbb{Q}_{p}^{c}[\mathcal{G}_{n}]$ by semisimplicity (the resulting map is independent of the choice of splittings, similarly to remark 4.1.3). Let $\tilde{\lambda}_{\chi} \colon D_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}) \to \mathbb{Q}_{p}^{c}$ denote the composition of the entire resulting column.

In order to relate the finite and infinite levels, we would like to shift the complex so that φ_n^{α} does not need to be inverted (the trivialisation t^{α} is a map from odd to even degree, too). This can be done as follows: since $\mathcal{P}^{\text{even}}$ and \mathcal{P}^{odd} are free $\Lambda(\mathcal{G})$ -modules by assumption, $D_{\Lambda(\mathcal{G})}(\mathcal{P}^{\text{even}})$ and $D_{\Lambda(\mathcal{G})}(\mathcal{P}^{\text{odd}})$ are free of rank 1. Let g^{even} and g^{odd} be arbitrary respective generators. In particular, $g^{\text{odd},*}$ is a generator of $D_{\Lambda(\mathcal{G})}(\mathcal{P}^{\text{odd}})^{-1}$ in the notation of lemma B.0.1, and $g^{\text{even}} \otimes g^{\text{odd},*}$ generates $D_{\Lambda(\mathcal{G})}(\mathcal{P}^{\bullet}) = D_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet})$. This yields an isomorphism

$$\iota_{g^{\text{even}}} \otimes \iota_{g^{\text{odd},*}} \colon D_{\Lambda}(\mathcal{C}_{S,T}^{\bullet}) = D_{\Lambda}(\mathcal{P}^{\text{even}}) \otimes D_{\Lambda}(\mathcal{P}^{\text{odd}})^{-1} \xrightarrow{\sim} D_{\Lambda}(\mathcal{P}^{\text{even}})^{-1} \otimes D_{\Lambda}(\mathcal{P}^{\text{odd}}) = D_{\Lambda}(\mathcal{C}_{S,T}^{\bullet}[1]),$$
(4.8)

with $\iota_{g^{\text{even}}}$ and $\iota_{g^{\text{odd},*}}$ given by (B.5). Lemma B.0.2 then shows that the diagram

commutes, where the bottom arrow again denotes multiplication by the square of the inverse of $\tilde{\lambda}_{\chi}(g^{\text{even}} \otimes g^{\text{odd},*}) \in (\mathbb{Q}_p^c)^*$. This relies on the fact that the arrow $\chi \colon \mathbb{Q}_p^c[\mathcal{G}_n] \to \mathbb{Q}_p^c$ is a ring homomorphism.

The last necessary simplification in order to relate the finite- and infinite-level maps is to replace the above morphisms in cohomology by maps on \mathcal{P}^{\bullet} . A convenient way to do so is to find a map on integral level $\Lambda(\mathcal{G})$ which induces both the finite-level splittings used in the construction of λ_{χ} (definition 4.1.2); and the infinite-level splittings used in the construction of the refined Euler characteristic (cf. (2.2)). This presents some difficulties, which are however not new: we encountered essentially the same issue in the proof of proposition 3.2.2 (where α itself might not be invertible), and we shall use the same workaround to resolve it. Namely, let π_0 be the arrow in $\mathcal{P}^{\bullet} = [\mathcal{P}^0 \to \mathcal{P}^1]$, $W = \operatorname{im}(\pi_0) \subseteq \mathcal{P}^1$ its image, and π_1 the surjection $\pi_1 \colon \mathcal{P}^1 \twoheadrightarrow H^1(\mathcal{P}^{\bullet}) = H^1(\mathcal{C}^{\bullet}_{S,T})$. Chose splittings σ_0 and σ_1 for the complex $\mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{P}^{\bullet}$ as in (2.2):

$$\mathcal{Q} \otimes_{\Lambda} H^{0}(\mathcal{P}^{\bullet}) \longleftrightarrow \mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{0} \xrightarrow{\mathcal{Q} \otimes_{\Lambda} \pi_{0}} \mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{1} \xrightarrow{\mathcal{Q} \otimes_{\Lambda} \pi_{1}} \mathcal{Q} \otimes_{\Lambda} H^{1}(\mathcal{P}^{\bullet})$$

$$(4.10)$$

We can scale σ_0 and σ_1 by regular elements elements $d_0, d_1 \in \Lambda(\Gamma) \subseteq \Lambda(\mathcal{G})$ to obtain $\Lambda(\mathcal{G})$ homomorphisms $d_0\sigma_0 \colon W \to \mathcal{P}^0$ and $d_1\sigma_1 \colon H^1(\mathcal{P}^{\bullet}) \to \mathcal{P}^1$ (recall, for instance, equation (2.1)). By lemma 2.2.1, there exists an $n_0 \in \mathbb{N}$ such that multiplication by d_0d_1 (or, equivalently, its projection $\overline{d_0d_1} \in \Lambda(\mathcal{G}_m)$) is an automorphism of $e(\chi)\mathbb{Q}_p^c[\mathcal{G}_m]$ for all $m \geq n_0$ (simply take $M = \Lambda(\mathcal{G})/\langle d_0d_1 \rangle$ regarded as a $\Lambda(\Gamma)$ -module) as long as χ doesn't factor over \mathcal{G}_{n_0} . This is essentially the same argument as in (3.20).

Fix d_0, d_1 and n_0 as above, which do not depend on the choice of a character χ . Assume χ factors through \mathcal{G}_n but not \mathcal{G}_{n_0} , and it still satisfies (KC). Let δ_0 and δ_1 be the images of d_0 and d_1 under $\Lambda(\mathcal{G}) \twoheadrightarrow \Lambda(\mathcal{G}_n) \hookrightarrow \mathbb{Q}_p^c[\mathcal{G}_n] \xrightarrow{\chi} \mathbb{Q}_p^c$ respectively, both of which are non-zero. Then $\delta_0^{-1} \cdot \chi(d_0\sigma_0)_{\Gamma^{p^n}}$ and $\delta_1^{-1} \cdot \chi(d_1\sigma_1)_{\Gamma^{p^n}}$ are splittings for the finite-level sequences

$$\chi H^{0}(\mathcal{P}^{\bullet})_{\Gamma^{p^{n}}} \longleftrightarrow \chi \mathcal{P}^{0}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{n}}}} \chi \mathcal{P}^{1}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{1})_{\Gamma^{p^{n}}}} \chi H^{1}(\mathcal{P}^{\bullet})_{\Gamma^{p^{n}}}$$

$$\chi \mathcal{P}^{0}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{n}}}} \chi \mathcal{P}^{1}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{n}}}} \chi \mathcal{P}^{1}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{n}}}} \chi \mathcal{P}^{1}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{n}}}}} \chi \mathcal{P}^{1}_{\Gamma^{p^{n}}} \xrightarrow{\chi(\pi_{0})_{\Gamma^{p^{$$

Note that $\mathcal{P}^{0}_{\Gamma^{p^{n}}}$ and $\mathcal{P}^{1}_{\Gamma^{p^{n}}}$ coincide with \mathcal{P}^{0}_{n} and \mathcal{P}^{1}_{n} respectively, and $_{\chi}H^{0}(\mathcal{P}^{\bullet})_{\Gamma^{p^{n}}} = _{\chi}\mathcal{E}_{L_{n},S,T}$ is precisely $_{\chi}\mathbb{Z}_{p} \otimes_{\mathbb{Z}} \mathcal{O}^{*}_{L_{n},S,T} = _{\chi}H^{0}(\mathcal{P}^{\bullet}_{n})$ by proposition 2.2.8 and the preceding results. The injectivity of $_{\chi}W_{\Gamma^{p^{n}}} \hookrightarrow _{\chi}\mathcal{P}^{1}_{\Gamma^{p^{n}}}$ then follows from a \mathbb{Q}^{c}_{p} -dimension argument.

We can now define our desired integral map

$$\tau \colon \mathcal{P}^{\mathrm{odd}} = \mathcal{P}^1 \to H^{\mathrm{odd}}(\mathcal{P}^{\bullet}) \oplus W \xrightarrow{t^{\circ}_{\iota} \oplus \mathrm{Id}} H^{\mathrm{even}}(\mathcal{P}^{\bullet}) \oplus W \to \mathcal{P}^0 = \mathcal{P}^{\mathrm{even}},$$

where t_{ι}^{α} is the integral trivialisation defined in (2.7), and the first and last arrows are given by $d_1\pi_1 \oplus (d_1 \operatorname{Id} - (d_1\sigma_1)\pi_1)$ and $(d_0\sigma_0) \oplus d_0 \operatorname{Id}$. These are simply the standard maps arising from a split short exact sequence, only scaled by d_0 and d_1 to make them integral (which is why we write $(d_1\sigma_1)$ and $(d_0\sigma_0)$ in brackets).

Denote by λ_{χ} the composite map given by the second column of (4.9). Since the choice of splittings implicit in the third vertical arrow is irrelevant, we can assume they coincide with those in (4.11) on χ -parts, after which an easy verification shows the commutativity of the leftmost square in our last (and key) diagram



where $r = \operatorname{rank}_{\Lambda} \mathcal{P}^0 = \operatorname{rank}_{\Lambda} \mathcal{P}^1$. In this diagram (and only there), we have made the following further notational simplification: D_n and \mathcal{Q} - stand for $D_{\mathbb{Q}_p^c[\mathcal{G}_n]}$ and $\mathcal{Q} \otimes_{\Lambda}$ -, respectively. Commutativity of the bottom-right square follows from (3.21) and (3.22) (choosing f to be be multiplication by $(d_0d_1)^r$ on a free rank one $\Lambda(\mathcal{G})$ -module), and the rest is clear.

When taken together, the above diagrams draw a clear connection between the map λ_{χ} of Burns, Kurihara and Sano and our evaluation maps $ev_{\gamma_{\chi}}$. In combination with the two previous lemmas, this suffices to conclude the proof. We only need to consider characters χ satisfying the following condition:

$$\Gamma^{p^{\max(n_0, n(S, \alpha))}} \not\subseteq \ker(\chi). \tag{KC'}$$

This is a slight strengthening of (KC) (i.e. $\Gamma^{p^{n(S,\alpha)}} \not\subseteq \ker(\chi)$) which still holds for almost all $\chi \in \operatorname{Irr}_{p}(\mathcal{G})$.

Assume first BKS' $(L_{\infty}/K, L, S, T, \beta)$ and let $\mathcal{L}_{S,T}$ be the element predicted therein. As a generator of $\mathcal{D}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}) = \mathcal{D}_{\Lambda(\mathcal{G})}(\mathcal{P}^{\text{even}}) \otimes_{\Lambda(\mathcal{G})} \mathcal{D}_{\Lambda(\mathcal{G})}(\mathcal{P}^{\text{odd}})^{-1}$, it admits an expression $\mathcal{L}_{S,T} = g^{\text{even}} \otimes g^{\text{odd},*}$ as above. Set $\widetilde{\mathcal{L}}_{S,T} = (\iota_{g^{\text{even}}} \otimes \iota_{g^{\text{odd},*}})(\mathcal{L}_{S,T}) \in \mathcal{D}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}[1])$ and define $\widetilde{\zeta}$ as the image of $\widetilde{\mathcal{L}}_{S,T}$ under the composition

$$D_{\Lambda}(\mathcal{C}^{\bullet}_{S,T}[1]) \hookrightarrow D_{\mathcal{Q}}(\mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{\text{even}})^{-1} \otimes D_{\mathcal{Q}}(\mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{\text{odd}}) \to D_{\mathcal{Q}}(\mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{\text{even}})^{-1} \otimes D_{\mathcal{Q}}(\mathcal{Q} \otimes_{\Lambda} \mathcal{P}^{\text{even}}) \xrightarrow{ev} \mathcal{Q}$$

where the middle arrow is $D_{\mathcal{Q}(\mathcal{G})}((d_0d_1)^{-1} \cdot \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} \tau)$. This amounts to the top row and first vertical arrows in the fourth column of the last diagram above. Proposition 4.1.5 i) now shows that $\tilde{\zeta}$ lies in lies in $\mathcal{Q}(\mathcal{G})^*$, which we identify with $K_1(\mathcal{Q}(\mathcal{G}))$ via the reduced norm; and its inverse, which we denote by $\zeta \in K_1(\mathcal{Q}(\mathcal{G}))$, satisfies

$$\partial(\zeta) = -\partial(\widetilde{\zeta}) = -[\mathcal{P}^{\text{odd}}, (d_0d_1)^{-1} \cdot \mathcal{Q} \otimes_{\Lambda} \tau, \mathcal{P}^{\text{even}}]$$

(beware the difference in shift between $\mathcal{P}^{\bullet}[1]$ here and \mathcal{P}^{\bullet} in the proposition). But $(d_0d_1)^{-1} \cdot \mathcal{Q} \otimes_{\Lambda} \tau$ is precisely the map constructed from the trivialisation $t^{\alpha} = \mathcal{Q}(\mathcal{G}) \otimes_{\Lambda(\mathcal{G})} t^{\alpha}_{\iota}$ and the splittings σ_0 and σ_1 from (4.10). In other words, $\partial(\zeta) = -\chi_{\Lambda(\mathcal{G}),\mathcal{Q}(\mathcal{G})}(\mathcal{C}^{\bullet}_{S,T}, t^{\alpha})$ is the inverse of the refined Euler characteristic of the trivialised complex $(\mathcal{C}^{\bullet}_{S,T}, t^{\alpha})$. On the analytic side, we have

$$\begin{aligned} ev_{\gamma_{\chi}}(\psi_{\chi}(\zeta)) &\doteq ev_{\gamma_{\chi}}(\psi_{\chi}(\widetilde{\zeta}))^{-1} \\ &= \dot{\lambda}_{\chi}(\widetilde{\mathcal{L}}_{S,T})^{-1} \\ &= \tilde{\lambda}_{\chi}(\mathcal{L}_{S,T}) \\ &= R_{S}^{\beta}(\chi,\alpha)^{-1} \cdot \lambda_{\chi}(\mathcal{L}_{S,T}) \doteq \frac{\beta^{-1}(L_{K,S,T}^{*}(\beta\check{\chi},0))}{R_{S}^{\beta}(\alpha,\chi)}, \end{aligned}$$

where \doteq denotes that the equality holds for almost all $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfying (KC'). The equalities follow from, in order: multiplicativity of $ev_{\gamma_{\chi}}$ (this holds whenever the evaluation is neither 0 nor ∞ , which is true a fortiori for almost all χ by the remaining equalities); the last diagram above together with lemma 4.1.7; diagram (4.9); diagram (4.1); and the assumption BKS' $(L_{\infty}/K, L, S, T, \beta)$.

Since ζ does not depend on χ and the map ψ_{χ} is invariant under ρ -twists (see the explanation before (2.29)), the element $F_{S,T,\chi}^{\alpha,\beta} = \psi_{\chi}(\zeta) \in \mathcal{Q}^c(\Gamma_{\chi})^*$ satisfies $\mathrm{IC}(L_{\infty}/K, \chi, L, S, T, \alpha, \beta)$. It follows that ζ itself satisfies $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$.

The converse is proved very similarly. Namely, suppose the zeta element $\zeta_{S,T}^{\alpha,\beta} \in K_1(\mathcal{Q}(\mathcal{G}))$ satisfies $\mathrm{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$. Then proposition 4.1.5 ii) yields a generator $\widetilde{\mathcal{L}}_{S,T} \in \mathrm{D}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet}[1])$ which is mapped to $\zeta_{S,T}^{\alpha,\beta}$ under the top row and rightmost column of the last diagram. The preimage $\mathcal{L}_{S,T}$ of $\widetilde{\mathcal{L}}_{S,T}$ under the isomorphism $\iota_{g^{\mathrm{even}}} \otimes \iota_{g^{\mathrm{odd},*}}$ is a generator of $\mathrm{D}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S,T}^{\bullet})$ which can be easily verified to have the interpolation property in BKS' $(L_{\infty}/K, L, S, T, \beta)$ using the same argument as above.

Remark 4.1.9. It is a by-product of this theorem that our Main Conjecture is independent from the choice α , as proved separately in subsection 3.2.2 - indeed, some of the core ideas of both arguments coincide. Note, however, that the above result concerns the abelian case exclusively.

This settles the case "abelian over \mathbb{Q} " as well as, under some conditions, "abelian over imaginary quadratic":

Corollary 4.1.10. Setting B with \mathcal{G} abelian. Conjecture $eMC(L_{\infty}/\mathbb{Q}, L, S, T, \alpha, \beta)$ holds in the following cases:

- i) $K = \mathbb{Q}$ (so $p \neq 2$).
- ii) K is an imaginary quadratic field where p does not split, and the μ -invariant $\mu(X_S^{cs})$ vanishes (for the extension L_{∞}/L). Condition $\mu(X_S^{cs}) = 0$ is automatically satisfied if [L:K] is a power of p.

Proof. Conjecture 3.1 from [BKS17] is known in both cases, which implies $BKS'(L_{\infty}/\mathbb{Q}, L, S, T, \beta)$ by the above theorem. For part i), see the proof of [BKS17], which in turn refers to [BG03]. Remark 6.1 ii) in the latter points to the original work of Mazur and Wiles.

For part ii), see [BH21] theorem 5.5 and proposition 5.6.

Remark 4.1.11. As an application, we outline the proof of a new case of the Main Conjecture which relies on the above corollary and the functoriality results from section 3.3. The relevance of this example is that, to the best of our knowledge, it is not covered by existing formulations of a Main Conjecture - let alone proved.

Consider setting B with $K = \mathbb{Q}$ (so $p \neq 2$) and $\operatorname{Gal}(L/\mathbb{Q})$ of the form $A \rtimes \mathbb{Z}/2\mathbb{Z}$, where A is abelian of odd order and the non-trivial element of $\mathbb{Z}/2\mathbb{Z}$ acts on A by inversion. Such a group is said to be **generalised dihedral**. Assume furthermore that L^A is imaginary (quadratic) and satisfies the conditions of part ii) of the corollary.

By proposition 3.3.14, the Main Conjecture for L_{∞}/\mathbb{Q} can be deduced from that for L_{∞}/L^U for each elementary subgroup U of $\operatorname{Gal}(L/K)$. It is easy to verify that any such U satisfies at least one of the following:

- 1) $U \leq A$.
- 2) |U| = 2.

The Main Conjecture for L_{∞}/L^A holds by ii) of the corollary, which settles all subgroups of type 1) by proposition 3.3.8.

For U of type 2), it is convenient to recall the notion of so-called maximal-order Main Conjectures. These have the same structure as our equivariant Main Conjecture except for the fact that a preimage in $K_1(\mathcal{Q}(\mathcal{G}))$ of the refined Euler characteristic of the trivialised complex is claimed to have reduced norm congruent to the analytic object modulo not the image of $K_1(\Lambda(\mathcal{G}))$ (recall (3.55)), but rather all of $Z(\mathfrak{M}(\mathcal{G}))^*$, where $\Lambda(\mathcal{G}) \subseteq \mathfrak{M}(\mathcal{G}) \subseteq \mathcal{Q}(\mathcal{G})$ denotes a maximal $\Lambda(\Gamma)$ -order in $\mathcal{Q}(\mathcal{G})$ (note that the resulting claim is weaker). These conjectures are known to enjoy similar functoriality properties to the ones in chapter 3 and, more importantly, to decompose into χ -parts (with χ an irreducible character), since so does $\mathfrak{M}(\mathcal{G})$.

If U is of type 2), then one has $\operatorname{Gal}(L_{\infty}/L^U) = \Gamma \times U$ and $\Lambda(\Gamma \times U)$ is a maximal $\Lambda(\Gamma)$ -order in $\mathcal{Q}(\Gamma \times U)$. Therefore, it suffices to prove the maximal-order Main Conjecture for L_{∞}/\mathbb{Q} by functoriality - and for this, in turn, to prove the χ -part of said conjecture for each $\chi \in \operatorname{Irr}_p(\mathcal{G})$.

We distinguish two further cases. If χ is linear, it factors as a character of an abelian extension of \mathbb{Q} , in which case the full equivariant Main Conjecture is known by part i) of the corollary - and therefore so is (the χ -part of) the maximal-order conjecture.

If χ is not linear, then [JN19] theorem 9.3 shows that $\chi = \operatorname{ind}_{A}^{\operatorname{Gal}(L/K)}\chi'$ for some non-trivial irreducible character χ' of A. But then functoriality together with case 1) above yields the desired conjecture. This covers the remaining cases and therefore concludes the argument.

4.2 The conjecture of Ritter and Weiss

This last section studies the relation between our Main Conjecture and that formulated by Ritter and Weiss in [RW04], the first example of an equivariant Main Conjecture for arbitrary totally real base field and one of the main motivations for the present work. It was proved in the so-called " $\mu = 0$ case" by the same authors in a series of articles leading up to [RW11]. Independently from this, Kakde formulated and proved a Main Conjecture in [Kak13] which was shown to be equivalent to that of Ritter and Weiss by, separately, Nickel (cf. [Nic13]) and Venjakob (cf. [Ven13]).

A common feature of the aforementioned conjectures is their limitation to the totally real case. Through a duality argument, it will be shown that, whenever L is a CM field containing a primitive p-th root of unity, the *minus part* of our Main Conjecture for L_{∞}/K is equivalent to the conjecture of Ritter and Weiss for the maximal totally real subextension L_{∞}^+/K . The set of places T can be disregarded altogether, as it is not featured in [RW04]. The setting is as follows:

Setting E. We consider the same objects as in setting A with the exception of T. The following additional restrictions are imposed:

- K is totally real. In particular, p is odd.
- L is a CM field containing a primitive p-th root of unity. The former amounts to the existence of a totally real number field $L^+ \subseteq L$ such that $[L:L^+] = 2$. This L^+ is necessarily a Galois extension of K, as is its own cyclotomic \mathbb{Z}_p -extension L^+_{∞} . We set $\mathcal{G}^+ = \operatorname{Gal}(L^+_{\infty}/K)$.

The Galois group of L/L^+ is generated by complex conjugation, which has order 2 and will be denoted by τ here. It identifies naturally with the generator of $\operatorname{Gal}(L_{\infty}/L_{\infty}^+)$, which allows us to write $\mathcal{G}^+ = \mathcal{G}/\langle \tau \rangle$.

We point out that the α and β from setting B are not necessary here for reasons which will become apparent soon.

Since τ is central in \mathcal{G} and has order 2, every $\chi \in \operatorname{Irr}_p(\mathcal{G})$ satisfies either $\chi(\tau) = \chi(1)$ or $\chi(\tau) = -\chi(1)$. We say χ is **totally even** in the former case, and **totally odd** in the latter. We denote the set of totally even Artin characters of \mathcal{G} by $\operatorname{Irr}_p(\mathcal{G})^+$, which is in bijection with $\operatorname{Irr}_p(\mathcal{G}^+)$; and that of totally odd ones, by $\operatorname{Irr}_p(\mathcal{G})^-$. The subgroup $H = \operatorname{Gal}(L_{\infty}/K_{\infty})$ contains τ , and therefore the W-equivalence relation \sim_W introduced in section 2.2 restricts separately to $\operatorname{Irr}_p(\mathcal{G})^+$ and $\operatorname{Irr}_p(\mathcal{G})^-$.

The primitive central idempotents $e^+ = (1 + \tau)/2$ and $e^- = (1 - \tau)/2$ yield a ring decomposition $\Lambda(\mathcal{G}) = e^+\Lambda(\mathcal{G}) \times e^-\Lambda(\mathcal{G})$, and hence also for $\mathcal{Q}(\mathcal{G})$ and $\mathcal{Q}^c(\mathcal{G})$. We denote the **plus part** of a $\Lambda(\mathcal{G})$ -module M by $M^+ = e^+M$, and analogously for its **minus part** M^- . An immediate computation shows that M^+ and M^- consist precisely of the elements of M upon which τ acts as the identity and as -1, respectively. Taking plus parts is an exact functor (whether one chooses the target category to be that of left modules over $\Lambda(\mathcal{G})$ or $\Lambda(\mathcal{G})^+$), as is taking minus parts. Note that these functors identify naturally with $\Lambda(\mathcal{G})^+ \otimes_{\Lambda(\mathcal{G})} -$ and $\Lambda(\mathcal{G})^- \otimes_{\Lambda(\mathcal{G})} -$, respectively. The canonical augmentation map $\Lambda(\mathcal{G}) \to \Lambda(\mathcal{G}^+)$ restricts to a ring isomorphism on $\Lambda(\mathcal{G})^+$ and to the zero map on $\Lambda(\mathcal{G})^-$.

The reduced norm and structure of $Z(\mathcal{Q}^c(\mathcal{G}))$ also decompose into plus and minus parts:



This corresponds essentially to the discussion in the first part of subsection 3.3.1 for the particular case $\tilde{H} = \langle \tau \rangle$. Recall in particular lemma 3.3.1 and the first few lines following remark 3.3.2. Similarly, the relative K_0 group decomposes canonically as

$$K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G})) = K_0(\Lambda(\mathcal{G})^+, \mathcal{Q}(\mathcal{G})^+) \times K_0(\Lambda(\mathcal{G})^-, \mathcal{Q}(\mathcal{G})^-).$$

Although it will be stated in precise terms shortly, the shape of the "Main Conjecture on minus parts" should be apparent from the above discussion. There is no need, however, for an analogue of the Interpolation Conjecture: it follows from the well-known existence of p-adic L-functions and a recent result of Ellerbrock and Nickel (cf. [EN22]). We briefly recall these concepts now.

The *p*-adic cyclotomic character of \mathcal{G} (introduced in more generality in section 0.1) is the homomorphism

$$\chi_{\text{cyc}} \colon \mathcal{G} = \text{Gal}(L_{\infty}/K) \twoheadrightarrow \text{Gal}(K(\mu_{p^{\infty}})/K) \hookrightarrow \mathbb{Z}_p^* = \mathbb{Z}_p^{(1)} \times \mu_{p-1},$$

where $\mathbb{Z}_p^{(1)} = 1 + p\mathbb{Z}_p$ and μ_{p-1} denote the principal units and the (p-1)-th roots of unity in \mathbb{Z}_p , respectively. Since complex conjugation acts on roots of unity as inversion, one has $\chi_{\text{cyc}}(\tau) = -1$. The projection of χ_{cyc} to $\mathbb{Z}_p^{(1)}$ factors as a character of $\text{Gal}(K_{\infty}/K) = \Gamma_K = \overline{\langle \gamma_K \rangle}$ and is often denoted by κ . The projection of χ_{cyc} to μ_{p-1} factors as a character of $\text{Gal}(K(\zeta_p)/K)$ and its usual notation is ω : the **Teichmüller character**.

Let $\chi \in \operatorname{Irr}_p(\mathcal{G})^+$ be a totally even character of \mathcal{G} . Then there exists a unique *p*-adic meromorphic function $L_{p,S}(\chi, -) \colon \mathbb{Z}_p \to \mathbb{C}_p$ such that, for all integers r > 1, one has

$$L_{p,K,S}(\chi, 1-r) = L_{K,S}(\chi\omega^{-r}, 1-r).$$
(4.12)

The right-hand side should be interpreted as $\beta^{-1}(L_{K,S}(\beta(\chi\omega^{-r}), 1-r))$ for an arbitrarily chosen $\beta \colon \mathbb{C}_p \xrightarrow{\sim} \mathbb{C}$. This notation, which is standard, reflects the classical fact - due to Siegel [Sie70] - that the outcome is independent of that choice. The function $L_{p,K,S}(\chi, -)$ is known as the *S*-truncated *p*-adic *L*-function attached to χ . Its construction in the abelian case (i.e. for linear χ) was given, independently, by Pierrette Cassou-Noguès in [Cas79]; Deligne and Ribet in [DR80]; and Barsky in [Bar78]. Interpolation (4.12) at 0 (that is, for r = 1), which is the point of interest to us, was shown to hold later on.

Greenberg extended the definition of $L_{p,K,S}$ beyond the abelian case by means of Brauer induction (cf. [Gre83]). Note that interpolation at 0 does not follow automatically from the linear case due to the potential vanishing of *L*-values in the denominator. However, Ellerbrock and Nickel have shown it to hold through the study of bounds on denominators of Stickelberger elements along cyclotomic towers. More precisely, (4.12) holds for r = 1 and arbitrary $\chi \in \operatorname{Irr}_p(\mathcal{G})^+$ by [EN22] theorem 3.

A crucial fact about *p*-adic *L*-functions is their relation to series quotients. Namely, in the above notation, there exists a $\Phi_{p,K,S,\chi} \in \mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T]])$ such that, for all $s \in \mathbb{Z}_p$, one has

$$L_{p,K,S}(\chi, 1-s) = \Phi_{p,K,S,\chi}(\kappa(\gamma_K)^s - 1).$$

These series quotients satisfy the usual functoriality properties and exhibit the following behaviour under W-twists: for any $\rho \in \operatorname{Irr}_p(\mathcal{G})$ of type W, there is an equality

$$\Phi_{p,K,S,\chi \otimes \rho}(T) = \Phi_{p,K,S,\chi}(\rho(\gamma_K)(T+1) - 1)$$
(4.13)

(see for instance [RW04] p. 563). It is therefore natural to identify T with $T_K = \gamma_K - 1$ and thus regard $\Phi_{p,K,S,\chi}$ as an element of $\mathbb{Q}_p^c \otimes_{\mathbb{Q}_p} \operatorname{Frac}(\mathbb{Z}_p[[T_K]]) = \mathcal{Q}^c(\Gamma_K)$, since then the above equation becomes

$$\Phi_{p,K,S,\chi\otimes\rho} = \rho^{\sharp}(\Phi_{p,K,S,\chi})$$

in the notation of section 2.4.

In order to relate these power series to the Interpolation Conjecture on minus parts, we introduce the continuous \mathbb{Z}_p -algebra automorphism $t_{\text{cyc}} \colon \Lambda(\mathcal{G}) \to \Lambda(\mathcal{G})$ uniquely determined by $t_{\text{cyc}}(g) = \chi_{\text{cyc}}(g)g$ for all $g \in \mathcal{G}$. This is a particular case of the t^s_{cyc} from [JN19] section 6.1. It is easy to verify it induces field automorphisms on $\mathcal{Q}(\mathcal{G})$ and $\mathcal{Q}^c(\mathcal{G})$, and therefore group automorphisms on $K_1(\mathcal{Q}(\mathcal{G}))$, $K_1(\mathcal{Q}^c(\mathcal{G}))$ and $Z(\mathcal{Q}^c(\mathcal{G}))^*$ - all of which we denote by t_{cyc} as well. Note that t_{cyc} swaps plus and minus parts, as $t_{\text{cyc}}(e^+) = e^-$ and $t_{\text{cyc}}(e^-) = e^+$.

Recall also the involution ι of $\Lambda(\mathcal{G})$ given by $g \mapsto g^{-1}$ introduced immediately before proposition 3.3.4, which induces a map $\iota: Z(\mathcal{Q}^c(\mathcal{G}))^* \to Z(\mathcal{Q}^c(\mathcal{G}))^*$. Since the dual $\check{\chi}$ of a character $\chi \in \operatorname{Irr}_p(\mathcal{G})$ is precisely given by $\check{\chi}(g) = \chi(g^{-1})$, a simple argument shows that $\iota(\gamma_{\chi}) = \gamma_{\check{\chi}} \in Z(\mathcal{Q}^c(\mathcal{G}))^*$.

The following is a direct application of some results in [JN19]:

Lemma 4.2.1. Setting *E*. There exists a unique element $F_{S,\mathcal{G}}^+ \in Z(\mathcal{Q}(\mathcal{G})^+)^* \subseteq Z(\mathcal{Q}(\mathcal{G}))^*$ such that, for any $\chi \in \operatorname{Irr}_p(\mathcal{G})^+$, one has

$$j_{\chi}(F_{S,\mathcal{G}}^+) = \Phi_{p,K,S,\chi} \in \mathcal{Q}^c(\Gamma_K)^*$$

where $j_{\chi} \colon Z(\mathcal{Q}^c(\mathcal{G})) \to \mathcal{Q}^c(\Gamma_K)^*$ denotes the map j_{χ}^c from (2.29).

For each $\psi \in \operatorname{Irr}_p(\mathcal{G})^-$, let $F_{S,\psi} = \iota(t_{\operatorname{cyc}}(F_{S,\mathcal{G}}^+))e_{\psi} \in \mathcal{Q}^c(\Gamma_{\psi})^* \subseteq Z(\mathcal{Q}^c(\mathcal{G})^-)^* \subseteq Z(\mathcal{Q}^c(\mathcal{G}))^*$. Then one has

$$ev_{\gamma_{\psi}}(F_{S,\psi}) = L_{K,S}(\check{\psi}, 0) \tag{4.14}$$

for all such ψ , where the notation of the right-hand is as in (4.12).

Proof. The first claim takes place entirely on plus parts and a brief proof can be found in [Nic13] p. 1231. The argument is straightforward: the twisting property (4.13) together with the Map^Wdescription (3.47) of $Z(\mathcal{Q}^c(\mathcal{G}^+))^* \cong Z(\mathcal{Q}^c(\mathcal{G})^+)^*$ immediately yield a unique $F_{S,\mathcal{G}}^+ \in Z(\mathcal{Q}^c(\mathcal{G})^+)^*$ with the asserted property. The fact that $F_{S,\mathcal{G}}^+$ actually lies in $Z(\mathcal{Q}(\mathcal{G})^+)^*$, i.e. it is $G_{\mathbb{Q}_p}$ -invariant (cf. (3.12)), is a direct consequence of behaviour of the $G_{\mathbb{Q}_p}$ -action on $\Phi_{p,K,S,\chi}$ (see for instance [RW04] p. 563 property (5) and p. 558 equation (*)).

Now the second claim follows from

$$ev_{\gamma_{\psi}}(F_{S,\psi}) = ev_{\gamma_{K}}(j_{\psi}(\iota(t_{cyc}(F_{S,\mathcal{G}}^{+}))))$$
$$= ev_{\gamma_{K}}(j_{\psi}(t_{cyc}(F_{S,\mathcal{G}}^{+})))$$
$$= ev_{\gamma_{K}}(j_{\psi\omega}^{1}(F_{S,\mathcal{G}}^{+}))$$
$$= L_{p,S}(\check{\psi}\omega, 1-1)$$
$$= L_{K,S}(\check{\psi}\omega\omega^{-1}, 0)$$
$$= L_{K,S}(\check{\psi}, 0).$$

The map $j^1_{\tilde{\psi}\omega}$ is defined in [JN19] section 6.1. It will not appear again, so we shall dispense with its definition. The two equalities involving it (third and fourth) are lemma 6.1 and equation (6.4) in the cited article, whereas the remaining equalities are clear from the preceding discussion. Note at this point that $\tilde{\psi}\omega$ is even, as the Teichmüller character ω is odd.

Remark 4.2.2. i) Let ρ be a type-W character of \mathcal{G} . Then $F_{S,\psi}$ and $F_{S,\psi\otimes\rho}$ coincide, since so do e_{ψ} and $e_{\psi\otimes\rho}$.

ii) For almost all $\psi \in \operatorname{Irr}_p(\mathcal{G})^-$, the term $L_{K,S}(\check{\psi}, 0)$ is precisely the regulated leading coefficient in the right-hand side of the interpolation property (2.28) (when $\psi = \chi \otimes \rho$). The reason for the apparent lack of β was explained after (4.12). As $L_{K,S_{\infty}}(\check{\psi}, s)$ never vanishes at s = 0 (for ψ totally odd), the leading coefficient $L_{K,S}^*(\check{\psi}, 0)$ coincides with the value $L_{K,S}(\check{\psi}, 0)$ whenever the Euler factors introduced by passing from $L_{K,S_{\infty}}$ to $L_{K,S}$ are non-zero, which is indeed the case for almost all ψ (this follows from a similar argument to lemma 3.2.7). By virtue of lemma 0.3.1 v), $L_{K,S}(\check{\psi}, 0) \neq 0$ implies that the \mathbb{C}_p -vector space $\operatorname{Hom}_{\mathbb{C}_p[\mathcal{G}_n]}(V_{\psi}, \mathbb{C}_p \otimes_{\mathbb{Z}_p} \mathcal{X}_{L_n,S})$) featured in the definition of the regulator (definition 2.3.4) has dimension 0. In particular, $R_S^{\beta}(\alpha, \psi) = 1$ for any valid choice of α and β .

iii) Although not necessary for our purposes, the $G_{\mathbb{Q}_p}$ -invariance of $\prod_{\psi \in \operatorname{Irr}_p(\mathcal{G})^-/\sim_W} F_{S,\psi} \in Z(\mathcal{Q}^c(\mathcal{G})^-)^*$ follows immediately from that of $F_{S,\mathcal{G}}^+$. In other words, the former lies in $Z(\mathcal{Q}(\mathcal{G})^-)^*$.

Together with points i) and ii) above, the lemma immediately shows that $\mathrm{IC}(L_{\infty}/K, \psi, L, S, \emptyset, \alpha, \beta)$ holds for any $\psi \in \mathrm{Irr}_p(\mathcal{G})^-$ and any (irrelevant) choice of L, α and β (one may also add a set T by 3.2.13). This covers the analytic side of our current pursuits. Before delving into the algebraic side, however, we introduce a piece of notation and present the conjecture of Ritter and Weiss.

Let \widetilde{R} be an integral domain and $\widetilde{S} = \operatorname{Frac}(\widetilde{R})$ its field of fractions. Suppose given an \widetilde{R} -order R inside an \widetilde{S} -algebra²⁵ $S = \widetilde{S} \otimes_{\widetilde{R}} R$. Assume furthermore that S is semisimple. Then the canonical embedding $R \hookrightarrow S$ satisfies the conditions of the φ in section 2.1 - that is, S is flat as an R-module. A perfect complex \mathcal{C}^{\bullet} of R-modules with \widetilde{R} -torsion cohomology admits only one trivialisation over S, namely the zero map. Accordingly, we set

$$[\mathcal{C}^{\bullet}] = \chi_{R,S}(\mathcal{C}^{\bullet}, 0) \in K_0(R, S), \tag{4.15}$$

where the right-hand side denotes the refined Euler characteristic defined in (2.3). If M is an R-torsion R-module of finite projective dimension over R, then M[0] is a perfect complex (isomorphic in $\mathcal{D}(R)$ to any finite projective resolution of M) and one may define

$$[M] = [M[0]] = \chi_{R,S}(M[0], 0) \in K_0(R, S).$$

It can be shown that, for any short exact sequence of complexes $\mathcal{C}_1^{\bullet} \hookrightarrow \mathcal{C}_2^{\bullet} \twoheadrightarrow \mathcal{C}_3^{\bullet}$ with each \mathcal{C}_i^{\bullet} as \mathcal{C}^{\bullet} above, one has $[\mathcal{C}_2^{\bullet}] = [\mathcal{C}_1^{\bullet}] + [\mathcal{C}_3^{\bullet}]$ in $K_0(R, S)$. Note that this is a particular instance of the additivity of refined Euler characteristics which appeared in subsection 3.2.3. An analogous relation holds for any short exact sequence of R-modules $M_1 \hookrightarrow M_2 \twoheadrightarrow M_3$ with each M_i as M above.

Let us specialise to our case of interest $\widetilde{R} = \Lambda(\Gamma)$ in the notation of setting E. The relevant $\Lambda(\Gamma)$ orders are $\Lambda(\mathcal{G})$ and its plus and minus parts. In a few lines, the localisations of these objects at
height-one prime ideals of $\Lambda(\Gamma)$ will be considered as well. Let $\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})$ denote the global complex
constructed in section 1.1 for the specific extension L^{+}_{∞}/K , which was shown to be perfect in
proposition 1.4.3. Its cohomology (determined in proposition 1.1.1) is indeed $\Lambda(\Gamma)$ -torsion: this is
clear for $H^{1}(\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})) = \mathbb{Z}_{p}$, and it follows from theorem 1.0.3 (the weak Leopoldt conjecture) and
the fact that L^{+} is totally real for $H^{0}(\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})) = X_{S}(\mathcal{G}^{+})$ by [NSW20] 10.3.22. Therefore, $\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})$ defines a class $[\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})] \in K_{0}(\Lambda(\mathcal{G}^{+}), \mathcal{Q}(\mathcal{G}^{+}))$ as explained above. This accounts for the lack of a

 $^{^{25}}$ There should be no ambiguity between the ring S and the set of places S. In this section, the former (whose notation is carried over from section 2.1) appears only in the current paragraph, and the latter only outside of it.

trivialisation, or an analogous map to α (cf. setting B), in [RW04]. In the context of setting E, the Main Conjecture of Ritter and Weiss is the following assertion:

Conjecture (RW $(L_{\infty}^+/K, S)$). Setting E.

There exists an element $\zeta_S^+ \in K_1(\mathcal{Q}(\mathcal{G}^+))$ such that $\partial(\zeta_S^+) = [\mathcal{T}_S^{\bullet}(\mathcal{G}^+)] \in K_0(\Lambda(\mathcal{G}^+), \mathcal{Q}(\mathcal{G}^+))$ and $\operatorname{nr}(\zeta_S^+) = F_{S,\mathcal{G}}^+ \in Z(\mathcal{Q}(\mathcal{G})^+)^*$, where $F_{S,\mathcal{G}}^+$ is as in lemma 4.2.1.

- Remark 4.2.3. i) The original conjecture in [RW04] p. 564 is formulated in a slightly more general setting than the above. Namely, it is only required that L_{∞}^+ be the cyclotomic \mathbb{Z}_p -extension of a totally real number field K' containing K under no assumption that K' is of the form L^+ . However, the possibility of a direct comparison with our Main Conjecture seems unclear in that generality, as it will shortly become apparent that the argument relating the two relies on the notion of plus and minus parts, as well as the cyclotomic character and Tate twists. We will briefly touch on this topic again after theorem 4.2.5.
- ii) The assertion in $\operatorname{RW}(L_{\infty}^+/K, S)$ is not identical to, but rather an exact reformulation of, that in [RW04]. This follows from the argument in [Nic13] p. 1231. Essentially, our $F_{S,\mathcal{G}}^+ \in Z(\mathcal{Q}(\mathcal{G})^+)^*$ corresponds to the $L_{k,S} \in \operatorname{Hom}^*(R_p(\mathcal{G}^+), \mathcal{Q}^c(\Gamma_K)^*)$ of Ritter and Weiss by theorem 8 in their article; and the $\widetilde{\mathcal{O}}_S$ therein coincides with $[\mathcal{T}_S^{\bullet}(\mathcal{G}^+)]$ by equation (1.17) and [Nic13] theorem 2.4.
- iii) The conjecture in [RW04] includes a uniqueness quantifier on ζ_S^+ (such that $\operatorname{nr}(\zeta_S^+) = F_{S,\mathcal{G}}^+$). The relation between the resulting claim and $\operatorname{RW}(L_{\infty}^+/K, S)$ mirrors that between the equivariant Main Conjectures with and without uniqueness from section 2.5. Since this has already been discussed (see for instance remark 2.5.4 i)), we limit ourselves to the version formulated above.

Our aim is to prove the equivalence of $\operatorname{eMC}(L_{\infty}/K, L, S, T, \alpha, \beta)$ on minus parts and $\operatorname{RW}(L_{\infty}^+/K, S)$. Consider first the minus part of the main complex $\mathcal{C}_{S,\varnothing}^{\bullet}$ constructed in section 1.4 for the extension L_{∞}/K and $T = \emptyset$. More formally, this refers to the perfect complex of $\Lambda(\mathcal{G})^-$ -modules

$$\mathcal{C}_{S}^{\bullet,-} = \Lambda(\mathcal{G})^{-} \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} \mathcal{C}_{S,\varnothing}^{\bullet}$$

By the exactness of the minus-parts functor, $\Lambda(\mathcal{G})^- \otimes_{\Lambda(\mathcal{G})}^{\mathbb{L}} -$ amounts to degree-wise $\Lambda(\mathcal{G})^- \otimes_{\Lambda(\mathcal{G})}$ and there are equalities $H^i(\mathcal{C}_S^{\bullet,-}) = H^i(\mathcal{C}_{S,\emptyset}^{\bullet})^-$ at all degrees *i*. Theorem 1.4.6 therefore implies $H^0(\mathcal{C}_S^{\bullet,-}) \cong E_{S,\emptyset}^- = E_S^-$, as well as the existence of a short exact sequence

$$0 \to X_S^{cs,-} \to H^1(\mathcal{C}_S^{\bullet,-}) \to \mathcal{X}_S^- \to 0.$$
(4.16)

The term E_S^- can be determined using [NSW20] theorem 11.3.11 ii). Set $\tilde{\mathcal{G}} = \operatorname{Gal}(L_{\infty}/L^+)$. Since $e^- \in \Lambda(\tilde{\mathcal{G}})$, it is enough to consider the module structure of E_S over this Iwasawa algebra (this is necessary due to the setup of section XI§3 in the reference). One then has an isomorphism $E_S \cong \mathbb{Z}_p(1) \oplus (\operatorname{Ind}_{\langle \tau \rangle}^{\tilde{\mathcal{G}}} \mathbb{Z}_p)^{|S_{\infty}(L^+)|}$. The summand $\mathbb{Z}_p(1)$ is the inverse limit of the *p*-power roots of unity along L_{∞}/L , and lies in E_S^- because complex conjugation τ acts on it as -1. The second summand vanishes on minus parts, as

$$e^{-\mathrm{Ind}_{\langle \tau \rangle}^{\widetilde{\mathcal{G}}}} \mathbb{Z}_p = e^{-} \Lambda(\widetilde{\mathcal{G}}) \widehat{\otimes}_{\Lambda(\langle \tau \rangle)} \mathbb{Z}_p = \Lambda(\widetilde{\mathcal{G}}) \widehat{\otimes}_{\Lambda(\langle \tau \rangle)} e^{-} \mathbb{Z}_p = 0$$

Hence, $H^0(\mathcal{C}_S^{\bullet,-}) \cong E_S^- \cong \mathbb{Z}_p(1)$ (also as a $\Lambda(\mathcal{G})$ -module) is $\Lambda(\Gamma)$ -torsion. As for sequence (4.16), one has

$$\mathcal{X}_S^- = \mathcal{Y}_S^- = \mathcal{Y}_{S_f}^-$$

by the same argument as above. Both $X_S^{cs,-}$ and $\mathcal{X}_S^- = \mathcal{Y}_{S_f}^-$ are $\Lambda(\Gamma)$ -torsion as well: the former is so even before taking minus parts, as explained shortly after (2.7), whereas the latter has finite \mathbb{Z}_p -rank.

The upshot is that the main complex $\mathcal{C}_{S,\emptyset}^{\bullet}$ defines a class $[\mathcal{C}_{S}^{\bullet,-}] \in K_{0}(\Lambda(\mathcal{G})^{-}, \mathcal{Q}(\mathcal{G})^{-}) \subseteq K_{0}(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ on minus parts without the need for a map α . In light of this, as well as remark 4.2.2, the minus part of $\operatorname{eMC}(L_{\infty}/K, L, S, \emptyset, \alpha, \beta)$ is precisely the following assertion, which we unoriginally refer to as the **equivariant Main Conjecture on minus parts**:

Conjecture (eMC⁻ $(L_{\infty}/K, S)$). Setting E.

There exists an element $\zeta_S^- \in K_1(\mathcal{Q}(\mathcal{G})^-)$ such that $\partial(\zeta_S^-) = -[\mathcal{C}_S^{\bullet,-}] \in K_0(\Lambda(\mathcal{G})^-, \mathcal{Q}(\mathcal{G})^-)$ and, for any $\chi \in \operatorname{Irr}_p(\mathcal{G})^-$, one has $\psi_{\chi}(\zeta_S^-) = F_{S,\chi} \in \mathcal{Q}^c(\Gamma_{\chi})^*$, where $F_{S,\chi}$ is as in lemma 4.2.1 (with χ replaced by ψ).

The fundamental algebraic result which will allow us to relate $\operatorname{RW}(L_{\infty}^+/K, S)$ to $\operatorname{eMC}^-(L_{\infty}/K, S)$ is (essentially) [JN20] theorem A.8. Intuitively, it states that a relation of the form $\partial(x) = y$, $\operatorname{nr}(x) = z$, where $x \in K_1(\mathcal{Q}(\mathcal{G}))$, is preserved (meaning it still holds for a possibly different x) if one replaces yby a similar enough $y' \in K_0(\Lambda(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$. The y of interest to us is $[\mathcal{T}_S^{\bullet}(\mathcal{G}^+)]$, but some brief recalls are in order before y' can be introduced. Our reference is [NSW20] section V§4.

Given a finitely generated left $\Lambda(\mathcal{G})$ -module M and $i \in \mathbb{N}$, we set $E^i(M) = \operatorname{Ext}_{\Lambda(\mathcal{G})}^i(M, \Lambda(\mathcal{G}))$, that is, the *i*-th right derived functor of $\operatorname{Hom}_{\Lambda(\mathcal{G})}(-, \Lambda(\mathcal{G}))$ applied to M. The left $\Lambda(\mathcal{G})$ -action on $\operatorname{Hom}_{\Lambda(\mathcal{G})}(N, \Lambda(\mathcal{G}))$ (for arbitrary N) given by $(\lambda f)(n) = f(n)\iota(\lambda)$, where ι is the involution of $\Lambda(\mathcal{G})$ introduced before (3.39), endows $E^i(M)$ with a left $\Lambda(\mathcal{G})$ -module structure. Some basic properties of these Ext-groups include the following:

- i) There are $\Lambda(\Gamma)$ -isomorphisms $\operatorname{Ext}^{i}_{\Lambda(\mathcal{G})}(M, \Lambda(\mathcal{G})) \cong \operatorname{Ext}^{i}_{\Lambda(\Gamma)}(M, \Lambda(\Gamma))$ for each $i \in \mathbb{N}$ (see [NSW20] proposition 5.4.17 and recall that $\Lambda(\mathcal{G})$ is Noetherian).
- ii) $E^0(M) = \operatorname{Hom}_{\Lambda(\mathcal{G})}(M, \Lambda(\mathcal{G}))$, which is trivial if M is $\Lambda(\Gamma)$ -torsion by i).
- iii) If M is $\Lambda(\Gamma)$ -torsion, then so is $E^1(M)$. If, in addition, $\mu(M) = 0$ (as defined after theorem 0.1.3), then $E^1(M) \cong \operatorname{Hom}_{\mathbb{Z}_p}(M, \mathbb{Z}_p)$ (cf. [NSW20] corollary 5.5.7).
- iv) $E^i(M) = 0$ for all $i \ge 3$, since the global dimension of $\Lambda(\Gamma)$ is 2.

Returning to our complex of interest, it is possible to show that $\mathcal{C}^{\bullet}_{S,\emptyset}$ has a representative in $\mathcal{D}(\Lambda(\mathcal{G}))$ of the form $[\stackrel{0}{A} \rightarrow \stackrel{1}{B}]$ with B projective and $\mathrm{pd}_{\Lambda(\mathcal{G})} A \leq 1$. Passing to minus parts, this implies the existence of an exact sequence

$$0 \to \mathbb{Z}_p(1) \to A^- \to B^- \to H^1(\mathcal{C}_S^{\bullet,-}) \to 0,$$

where the middle arrow represents $\mathcal{C}_{S}^{\bullet,-}$ in $\mathcal{D}(\Lambda(\mathcal{G})^{-})$ (and also in $\mathcal{D}(\Lambda(\mathcal{G}))$). Now a straightforward homological computation shows that

$$H^{i}(R\operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-},\Lambda(\mathcal{G}))) = \begin{cases} E^{1}(H^{1}(\mathcal{C}_{S}^{\bullet,-})), & i = 0\\ E^{1}(\mathbb{Z}_{p}(1)) = \mathbb{Z}_{p}(-1), & i = 1\\ 0, & \text{otherwise.} \end{cases}$$

The correct complex y' to compare to $\mathcal{T}_{S}^{\bullet}(\mathcal{G}^{+})$ is the Tate twist $R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-}, \Lambda(\mathcal{G}))(1)$. Here and below, the notation -(r) on a complex of modules over $\Lambda(\mathcal{G})$ (or a related ring) formally refers to $\mathbb{Z}_{p}(r) \otimes_{\mathbb{Z}_{p}}^{\mathbb{L}}$ –. This amounts to applying $\mathbb{Z}_{p}(r) \otimes_{\mathbb{Z}_{p}}$ – (i.e. taking the usual *r*-th Tate twist) on each degree, and it commutes with cohomology.

The last preparatory step is to show that the two aforementioned complexes are indeed similar enough - in a sense which is made precise by localisation. Given a height-one prime ideal \mathfrak{p} of $\Lambda(\Gamma)$, let $\Lambda_{\mathfrak{p}}(\Gamma)$ denote the corresponding localisation and set $\Lambda_{\mathfrak{p}}(\mathcal{G}) = \Lambda_{\mathfrak{p}}(\Gamma) \otimes_{\Lambda(\Gamma)} \Lambda(\mathcal{G})$. Localisation at \mathfrak{p} constitutes a exact functor from the category of left $\Lambda(\mathcal{G})$ -modules to that of $\Lambda_{\mathfrak{p}}(\mathcal{G})$ -modules and induces a functor on the corresponding categories of cochain complexes in the usual way.

Proposition 4.2.4. Setting *E*. For each height-one prime ideal \mathfrak{p} of $\Lambda(\Gamma)$, one has

$$[R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-}, \Lambda(\mathcal{G}))(1)_{\mathfrak{p}}] = [\mathcal{T}_{S}^{\bullet}(\mathcal{G}^{+})_{\mathfrak{p}}]$$

in $K_0(\Lambda_{\mathfrak{p}}(\mathcal{G}), \mathcal{Q}(\mathcal{G})).$

Proof. We first point out that the classes of the complexes in the statement are well defined in the sense of (4.15). Indeed, $\Lambda_{\mathfrak{p}}(\mathcal{G})$ is a $\Lambda_{\mathfrak{p}}(\Gamma)$ -order in the $\operatorname{Frac}(\Lambda_{\mathfrak{p}}(\Gamma)) = \mathcal{Q}(\Gamma)$ -algebra $\mathcal{Q}(\mathcal{G})$, and both complexes have $\Lambda_{\mathfrak{p}}(\Gamma)$ -torsion cohomology - they do even before localising by the preceding discussion. Here we are tacitly using the fact that E^1 preserves torsionness.

Consider the short exact sequence (4.16), whose last non-zero term was shown to coincide with $\mathcal{Y}_{S_f}^-$. In the associated long exact Ext-sequence

$$\cdots \to E^0(X_S^{cs,-}) \to E^1(\mathcal{Y}_{S_f}) \to E^1(H^1(\mathcal{C}_S^{\bullet,-})) \to E^1(X_S^{cs,-}) \to E^2(\mathcal{Y}_{S_f}) \to \cdots,$$

the terms $E^0(X_S^{cs,-})$ and $E^2(\mathcal{Y}_{S_f})$ vanish. The former does by property ii) of E^i above, whereas in the case of the latter, vanishing follows immediately from the length-one projective resolution of $\operatorname{Ind}_{\mathcal{G}_v}^{\mathcal{G}}\mathbb{Z}_p$ constructed in the proof of corollary 3.2.6. Thus, we obtain a short exact sequence on the E^1 -terms.

In order to apply Kummer duality, $E^1(X_S^{cs,-})$ must be replaced by $E^1(X_{nr})$, which can be achieved as follows: start at the five-term exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to E_{S_{\infty}} \to E_S \to \varprojlim_n w_n \in S_f(L_n) \mathbb{Z}_p \cdot w_n \to X_{S_{\infty}}^{cs} \to X_S^{cs} \to 0,$$

which is constructed in a completely analogous manner to (3.25) (note that $X_{S_{\infty}}^{cs}$ is precisely X_{nr}). As explained in remark 3.2.4, the non-*p*-adic places are irrelevant in the middle term, which therefore becomes \mathcal{Y}_{S_p} . Since $E_{S_{\infty}}^- = E_S^- = \mathbb{Z}_p(1)$ (apply [NSW20] theorem 11.3.11 as before), the sequence $\mathcal{Y}_{S_p}^- \hookrightarrow X_{nr}^- \twoheadrightarrow X_S^{cs,-}$ is exact. Using the same Ext-sequence argument as above results in a new short exact sequence on E^1 -terms which, when spliced with the previous one, yields

$$0 \to E^{1}(\mathcal{Y}_{S_{f}}^{-}) \to E^{1}(H^{1}(\mathcal{C}_{S}^{\bullet,-})) \to E^{1}(X_{nr}^{-}) \twoheadrightarrow E^{1}(\mathcal{Y}_{S_{p}}^{-}) \to 0.$$

$$(4.17)$$

Less effort is needed on plus parts. Let $(S_f \setminus S_p)^{p+}$ denote the set of $v \in S_f \setminus S_p$ such that L^+ contains a primitive *p*-th root of unity locally at any (equivalently, all) prolongations of v. Then there exists a short exact sequence of $\Lambda(\mathcal{G})$ -modules

$$0 \to \bigoplus_{v \in (S_f \setminus S_p)^{p_+}} \operatorname{Ind}_{\mathcal{G}_v^+}^{\mathcal{G}^+}(\mathbb{Z}_p(1)) \to X_S(\mathcal{G}^+) \to X_{S_\infty \cup S_p}(\mathcal{G}^+) \to 0$$
(4.18)

by [NSW20] theorem 11.3.5. A simple computation shows that

$$\bigoplus_{v \in (S_f \setminus S_p)^{p+}} \operatorname{Ind}_{\mathcal{G}_v^+}^{\mathcal{G}^+}(\mathbb{Z}_p(1)) = \mathcal{Y}_{(S_f \setminus S_p)^{p+}}^-(1).$$

Furthermore, $\mathcal{Y}_{\{v\}}^-$ vanishes for any $v \in S_f \setminus S_p$ which does not lie in $(S_f \setminus S_p)^{p+}$: for any such vand any arbitrarily chosen prolongation of w of v to L, the extension L/L^+ is not trivial locally at w (L does contain ζ_p), and therefore $\tau \in \operatorname{Gal}(L/L^+)_w$. As a consequence, one may replace the first non-zero term of (4.18) by $\mathcal{Y}_{S_f \setminus S_p}^-(1)$.

The relation between (4.17) and (4.18) comes in the form of *Kummer duality*, which provides a pseudo-isomorphism

$$X_{S_{\infty}\cup S_p}(\mathcal{G}^+)(-1) \xrightarrow{\approx} E^1(X_{nr}^-)$$

(see [NSW20] corollary 11.4.4 or, more in line with our notation, [Sha] corollary 3.4.8). In order to prove the claim in the proposition, we distinguish two cases:

• If \mathfrak{p} is the singular prime $p\Lambda(\Gamma)$: Any $\Lambda(\mathcal{G})$ -module which is finitely generated over \mathbb{Z}_p vanishes after localisation at \mathfrak{p} by the structure theorem 0.1.3. This is the case for the degree-1 cohomology modules of both $R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_S^{\bullet,-}, \Lambda(\mathcal{G}))(1)$ and $\mathcal{T}_S^{\bullet}(\mathcal{G}^+)$. In particular, one has

$$[R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-}, \Lambda(\mathcal{G}))(1)_{\mathfrak{p}}] = [H^{0}(R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-}, \Lambda(\mathcal{G}))(1)_{\mathfrak{p}})]$$
$$= [H^{0}(R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_{S}^{\bullet,-}, \Lambda(\mathcal{G})))(1)_{\mathfrak{p}}]$$
$$= [E^{1}(H^{1}(\mathcal{C}_{S}^{\bullet,-}))(1)_{\mathfrak{p}}]$$

and $[\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+})_{\mathfrak{p}}] = [H^{0}(\mathcal{T}^{\bullet}_{S}(\mathcal{G}^{+}))_{\mathfrak{p}}] = [X_{S}(\mathcal{G}^{+})_{\mathfrak{p}}]$ in $K_{0}(\Lambda_{\mathfrak{p}}(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$. Note that the degree-0 cohomology modules have a fortiori finite projective dimension when localised at \mathfrak{p} , as localisation preserves perfectness.

Localise now (4.17) and (4.18) at \mathfrak{p} , which causes the terms of the form \mathcal{Y}_{-} and $E^{1}(\mathcal{Y}_{-})$ to vanish by their finite-generatedness over \mathbb{Z}_{p} . This results in

$$[E^1(H^1(\mathcal{C}_S^{\bullet,-}))(1)_{\mathfrak{p}}] = [E^1(X_{nr}^-)(1)_{\mathfrak{p}}] = [X_{S_{\infty}\cup S_p}(\mathcal{G}^+)_{\mathfrak{p}}] = [X_S(\mathcal{G}^+)_{\mathfrak{p}}],$$

the middle equality being Kummer duality, which concludes the proof of the singular case.

• If \mathfrak{p} is a regular (that is, not the singular) prime: The localisation $\Lambda_{\mathfrak{p}}(\mathcal{G})$ is known to have the following property: any finitely generated $\Lambda_{\mathfrak{p}}(\mathcal{G})$ -module has finite projective dimension (cf. [Swa68]). In particular, the class in $K_0(\Lambda_{\mathfrak{p}}(\mathcal{G}), \mathcal{Q}(\mathcal{G}))$ of a perfect complex with $\Lambda(\Gamma)_{\mathfrak{p}}$ -torsion cohomology can be computed as an alternating sum of the classes of its cohomology modules. Using this fact, together with (4.17) and (4.18), Kummer duality and the obvious short exact sequence $\mathcal{Y}^-_{S_f \setminus S_p} \hookrightarrow \mathcal{Y}^-_{S_f} \twoheadrightarrow \mathcal{Y}^-_{S_p}$, yields the desired result.

Now the main result of this section follows easily:

Theorem 4.2.5. Setting E. Conjecture $eMC^{-}(L_{\infty}/K, S)$ holds if and only if $RW(L_{\infty}^{+}/K, S)$ does.

Proof. Suppose first that $\zeta_S^+ \in K_1(\mathcal{Q}(\mathcal{G}^+)) \subseteq K_1(\mathcal{Q}(\mathcal{G}))$ satisfies $\operatorname{RW}(L_{\infty}^+/K, S)$, so $\partial(\zeta_S^+) = [\mathcal{T}_S^{\bullet}(\mathcal{G}^+)]$ and $\operatorname{nr}(\zeta_S^+) = F_{S,\mathcal{G}}^+$. Apply proposition 4.2.4 above and [JN20] theorem A.8 to conclude that there exists a $\tilde{\zeta}_S$ such that $\partial(\tilde{\zeta}_S) = [R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}_S^{\bullet,-}, \Lambda(\mathcal{G}))(1)]$ and $\operatorname{nr}(\tilde{\zeta}_S) = F_{S,\mathcal{G}}^+$. We stress that hypothesis (ii) of the cited theorem may not be satisfied, but its proof shows that hypotheses (i) and (ii) can be replaced by the property in proposition 4.2.4.

Set $\zeta_S^- = \iota(t_{\text{cyc}}(\tilde{\zeta}_S)^t)$, where t denotes the transpose. For all $\chi \in \text{Irr}_p(\mathcal{G})^-$, one has

$$\psi_{\chi}(\zeta_S^-) = \operatorname{nr}(\zeta_S^-)e_{\chi} = \iota(t_{\operatorname{cyc}}(F_{S,\mathcal{G}}^+))e_{\chi} = F_{S,\psi}$$

(see lemma 4.2.1 for the last equality).

On the homological side, it is well known that $\partial(t_{\rm cyc}(x)) = \partial(x)(-1)$ for any $x \in K_1(\mathcal{Q}(\mathcal{G}))$; and if, furthermore, $\partial(x) = [\mathcal{C}^{\bullet}]$ for a perfect $\Lambda(\mathcal{G})$ -complex with torsion cohomology, then one has $\partial(\iota(x^t)) = -[R \operatorname{Hom}_{\Lambda(\mathcal{G})}(\mathcal{C}^{\bullet}, \Lambda(\mathcal{G}))]$. It follows that $\partial(\zeta_S^-) = -[\mathcal{C}_S^{\bullet,-}]$, and hence ζ_S^- satisfies $\operatorname{eMC}^-(L_{\infty}/K, S)$.

A completely symmetric argument proves the converse.

As explained in remark 4.2.3 i), setting E does not cover all cases for which the original conjecture of Ritter and Weiss can be formulated. However, if $\text{RW}(L_{\infty}^+/K, S)$ holds in general, then so does their original conjecture by the functoriality properties of the latter. The above theorem shows that this is the case if $\text{eMC}^-(L_{\infty}/K, S)$ holds in full generality - which would in turn be implied by the equivariant Main Conjecture from section 2.5.

We conclude with some immediate consequences of the last theorem:

Corollary 4.2.6. Setting *E*. Conjecture $eMC^{-}(L_{\infty}/K, S)$ holds in the following cases:

- i) The μ -invariant of $X_S(\mathcal{G}^+)$ vanishes. For instance, this is always the case if L^+ is an abelian extension of \mathbb{Q} .
- ii) \mathcal{G} has an abelian Sylow p-subgroup.

Proof. In the case $\mu(X_S(\mathcal{G}^+)) = 0$, $\operatorname{RW}(L_{\infty}^+/K, S)$ is known by [RW11] or [Kak13]. The second statement in part i) is the celebrated Ferrero-Washington theorem from [FW79]. Part ii) follows from corollary 1.2 in [JN20].

Recall at this point that the vanishing of the μ -invariant for the cyclotomic \mathbb{Z}_p -extension of any number field was conjectured by Iwasawa.

Appendix A

Functoriality of refined Euler characteristics

We show that refined Euler characteristics (as defined in section 2.1) behave well with respect to derived extension and restriction of scalars. This is probably known to experts, as the proof amounts to a formal verification, but the author is not aware of any reference for the precise formulation required above. Given a ring homomorphism $f: R \to S$, we denote the corresponding extensionand restriction-of-scalars functors by $S \otimes_R -: \mathfrak{M}_R^l \to \mathfrak{M}_S^l$ and $-|_R: \mathfrak{M}_S^l \to \mathfrak{M}_R^l$, respectively, where \mathfrak{M}_R^l denotes the category of left *R*-modules and analogously for \mathfrak{M}_S^l . Recall the notation $K_0(\rho, \sigma): K_0(R, S) \to K_0(R', S')$ and $K_0^{\text{res}}(\rho, \sigma): K_0(R', S') \to K_0(R, S)$ from diagrams (0.30) and (0.31).

Lemma A.0.1. Let

$$\begin{array}{ccc} R & \stackrel{\varphi}{\longrightarrow} S \\ \downarrow^{\rho} & \qquad \downarrow^{\sigma} \\ R' & \stackrel{\varphi'}{\longrightarrow} S' \end{array}$$

be a commutative diagram of ring homomorphisms. Suppose that S and S' are semisimple, S is flat as a right R-module via φ and S' is flat as a right R'-module via φ' . Then:

i) Let \mathcal{C}^{\bullet} be a perfect complex of left R-modules and t a trivialisation, that is, an isomorphism $t: S \otimes_R H^{\text{odd}}(\mathcal{C}^{\bullet}) \xrightarrow{\sim} S \otimes_R H^{\text{even}}(\mathcal{C}^{\bullet})$. Then t induces a trivialisation

$$t' \colon S' \otimes_{R'} H^{\mathrm{odd}}(R' \otimes_{R}^{\mathbb{L}} \mathcal{C}^{\bullet}) \to S' \otimes_{R'} H^{\mathrm{even}}(R' \otimes_{R}^{\mathbb{L}} \mathcal{C}^{\bullet})$$

such that

$$K_0(\rho,\sigma)(\chi_{R,S}(\mathcal{C}^{\bullet},t)) = \chi_{R',S'}(R' \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet},t')$$

ii) Assume that R' is a finitely generated projective R-module via ρ . Let \mathcal{C}^{\bullet} be a perfect complex of left R'-modules and $t: S' \otimes_{R'} H^{\text{odd}}(\mathcal{C}^{\bullet}) \to S' \otimes_{R'} H^{\text{even}}(\mathcal{C}^{\bullet})$ a trivialisation. Then $\mathcal{C}^{\bullet}|_{R}$ is a

perfect complex of R-modules, $t|_{S}$ is a trivialisation for it, and one has

$$K_0^{\operatorname{res}}(\rho,\sigma)(\chi_{R',S'}(\mathcal{C}^{\bullet},t)) = \chi_{R,S}(\mathcal{C}^{\bullet}|_{R},t|_{S})$$

- *Proof.* i) We assume first that C^{\bullet} is strictly perfect and denote it by \mathcal{P}^{\bullet} to avoid confusion with the general case. The following facts will be relevant:
 - 1) $(R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^i = R' \otimes_R \mathcal{P}^i$, where the left-hand side denotes the *i*-th cochain module of $R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$. In other words, the complex $R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$ is the result of applying $R' \otimes_R -$ to \mathcal{P}^{\bullet} degree-wise (and also to the differentials). This is an immediate consequence of \mathcal{P}^i being projective for all *i*. Since $R' \otimes_R -$ sends finitely generated projective *R*-modules to finitely generated projective R'-modules, $R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$ is strictly perfect as well. In particular, $(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}))^i = S' \otimes_{R'} (R' \otimes_R \mathcal{P}^i).$
 - 2) For the same reason as above, $(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^i = S \otimes_R \mathcal{P}^i$. The semisimplicity of S therefore implies

$$(S' \otimes_{S}^{\mathbb{L}} (S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}))^{i} = S' \otimes_{S}^{\mathbb{L}} (S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})^{i} = S' \otimes_{S} (S \otimes_{R} \mathcal{P}^{i}).$$

3) The functors $S' \otimes_S (S \otimes_R -)$ and $S' \otimes_{R'} (R' \otimes_R -)$ are canonically naturally isomorphic by the commutative diagram above, so we may identify them. Using 1) and 2) yields

$$S' \otimes_{S}^{\mathbb{L}} (S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) = S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}).$$

4) $S \otimes_R -$, $S' \otimes_{R'} -$ and $S' \otimes_S -$ are additive and exact and hence commute with taking cocycles, coboundaries and cohomology when applied to a complex degree-wise.

Note that we do not require R' to be flat over R, and hence $H^i(R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$ might not coincide with $R' \otimes_R H^i(\mathcal{P}^{\bullet})$. Some care is therefore needed in the order the subsequent operations.

We first define the new trivialisation t'. By virtue of 1) - 4), there are canonical isomorphisms

$$S' \otimes_S H^i(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}) \cong H^i(S' \otimes_S^{\mathbb{L}} (S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})) = H^i(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})) \cong S' \otimes_{R'} H^i(R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$$
(A.1)

for each $i \in \mathbb{Z}$. We let t' be unique arrow making the diagram

commute, where the vertical isomorphisms come from (A.1). Since t is an isomorphism, so are $S' \otimes_S t$ and t' in the obvious categories.

The refined Euler characteristic $\chi_{R,S}(\mathcal{C}^{\bullet}, t)$ relies on a choice of splittings of certain short exact sequences (cf. (2.2) and preceding lines). Those choices induce analogous splittings for the complex $S' \otimes_{R'} (R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$ in a natural way: for each $i \in \mathbb{Z}$, let δ_i be a splitting for the sequence $B^i(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}) \hookrightarrow Z^i(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}) \twoheadrightarrow H^i(S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$. Then the S'-homomorphism δ'_i induced by $S' \otimes_S \delta_i$ and the last two vertical arrows of

$$B^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) \longleftrightarrow Z^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) \xrightarrow{\pi_{i}} H^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})$$

$$\downarrow^{S' \otimes_{S} \mathrm{Id}} \downarrow^{S' \otimes_{S} \mathrm{Id}} \downarrow^{S' \otimes_{S} \mathrm{Id}} \downarrow^{S' \otimes_{S} \mathrm{Id}}$$

$$S' \otimes_{S} B^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) \longleftrightarrow S' \otimes_{S} Z^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) \xrightarrow{S' \otimes_{S} \pi_{i}} S' \otimes_{S} H^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})$$

$$\downarrow^{\wr} \qquad \downarrow^{\wr} \qquad \downarrow^{\wr} \qquad \downarrow^{\wr} \qquad \downarrow^{\wr}$$

$$B^{i}(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})) \longleftrightarrow Z^{i}(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})) \xrightarrow{\pi_{i}'} H^{i}(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}))$$

(where the isomorphisms are analogous to (A.1)) is a splitting of the bottom row. Likewise, we use the splitting τ_i of

$$Z^{i}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}) \longleftrightarrow (S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})^{i} \xrightarrow{\longrightarrow} B^{i+1}(S \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})$$

to construct one for

$$Z^{i}(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})) \longleftrightarrow (S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet}))^{i} \longrightarrow B^{i+1}(S' \otimes_{R'}^{\mathbb{L}} (R' \otimes_{R}^{\mathbb{L}} \mathcal{P}^{\bullet})).$$

If $\varphi_t : S \otimes_R \mathcal{P}^{\text{odd}} \xrightarrow{\sim} S \otimes_R \mathcal{P}^{\text{even}}$ and $\varphi_{t'} : S \otimes_{R'} (R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^{\text{odd}} \xrightarrow{\sim} S' \otimes_{R'} (R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^{\text{even}}$ are as in (2.2), the diagram

can be easily checked to commute, and hence

$$\begin{aligned} K_0(\rho,\sigma)(\chi_{R,S}(\mathcal{P}^{\bullet},t)) = & K_0(\rho,\sigma)([\mathcal{P}^{\mathrm{odd}},\varphi_t,\mathcal{P}^{\mathrm{even}}]) \\ = & [R'\otimes_R \mathcal{P}^{\mathrm{odd}}, S'\otimes_S \varphi_t, R'\otimes_R \mathcal{P}^{\mathrm{even}}] \\ = & [(R'\otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^{\mathrm{odd}},\varphi_{t'}, (R'\otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})^{\mathrm{even}}] = \chi_{R',S'}(R'\otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet},t'). \end{aligned}$$

We now treat the general case of an arbitrary perfect complex \mathcal{C}^{\bullet} trivialised by some t as in the statement. Let \mathcal{P}^{\bullet} be a strictly perfect representative of \mathcal{C}^{\bullet} and $q: \mathcal{C}^{\bullet} \xrightarrow{\sim} \mathcal{P}^{\bullet}$ an isomorphism in

the derived category $\mathcal{D}(R)$. This induces isomorphisms on cohomology $q_H^i \colon H^i(\mathcal{C}^{\bullet}) \xrightarrow{\sim} H^i(\mathcal{P}^{\bullet})$ for all $i \in \mathbb{Z}$, as well as an isomorphism $R' \otimes_R^{\mathbb{L}} q \colon R' \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet} \xrightarrow{\sim} R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$ in $\mathcal{D}(R')$ (so $R' \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet}$ is perfect). In particular, there exist R'-isomorphisms

$$(R' \otimes_R^{\mathbb{L}} q)_H^i \colon H^i(R' \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet}) \xrightarrow{\sim} H^i(R' \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet})$$

in each degree i (as mentioned above, one should not expect $(R' \otimes_R^{\mathbb{L}} q)_H^i = R' \otimes_R q_H^i)$. Consider the diagram

In the top square, the three solid arrows are isomorphisms and we define the dashed one by commutativity. The middle square is (A.2) (with its central square omitted). As for the bottom square, the two vertical arrows are isomorphisms and the bottom one is again determined by commutativity. This concludes the proof, since it implies

$$\chi_{R,S}(\mathcal{C}^{\bullet},t) = \chi_{R,S}(\mathcal{P}^{\bullet},\tilde{t}) = \chi_{R',S'}(\mathcal{P}^{\bullet},\tilde{t}') = \chi_{R',S'}(\mathcal{C}^{\bullet},t'),$$

the middle equality being case treated above.

- ii) The argument for restriction of scalars is simpler than that for extension. The functor $-|_R$ induced by φ
 - is additive and exact.
 - sends finitely generated (resp. projective) R'-modules to finitely generated (resp. projective) R-modules.
 - sends quasi-isomorphisms to quasi-isomorphisms²⁶.
 - commutes with cohomology, coboundaries and cocycles.

The analogous statements for $-|_{S}$ (via σ) hold too. By the commutativity of the diagram in the statement, we can identify the functors $S \otimes_{R} (-|_{R}) = (S' \otimes_{R'} -)|_{S} : \mathfrak{M}_{R'}^{l} \to \mathfrak{M}_{S}^{l}$ (more precisely, they are canonically naturally isomorphic). This explains why $t|_{S}$ is a trivialisation for $\mathcal{C}^{\bullet}|_{R}$.

²⁶It is not necessary to consider a left derived functor as we did with $R' \otimes_R^{\mathbb{L}} -:$ one can directly restrict scalars in the derived category. It is of note, however, that $\mathcal{C}^{\bullet}|_R$ might have a larger isomorphism class in $\mathcal{D}(R)$ than \mathcal{C}^{\bullet} did in $\mathcal{D}(R')$, since there may exist quasi-isomorphisms which are *R*-equivariant, but not *R'*-equivariant.

By the same formal argument as in part i), we can assume \mathcal{C}^{\bullet} is strictly perfect, say \mathcal{P}^{\bullet} . The splittings used to define $\varphi_t \colon S' \otimes_{R'} \mathcal{P}^{\text{odd}} \to S' \otimes_{R'} \mathcal{P}^{\text{even}}$ induce splittings for the for their counterparts associated to $\mathcal{P}^{\bullet}|_{R}$. This results in an S-isomorphism

$$\varphi_{(t|_S)} \colon S \otimes_R (\mathcal{P}^{\bullet}|_R)^{\mathrm{odd}} \xrightarrow{\sim} S \otimes_R (\mathcal{P}^{\bullet}|_R)^{\mathrm{even}}$$

which coincides with $(\varphi_t)|_S$. Therefore,

$$\begin{split} K_0^{\text{res}}(\rho,\sigma)(\chi_{R',S'}(\mathcal{P}^{\bullet},t)) &= K_0^{\text{res}}(\rho,\sigma)([\mathcal{P}^{\text{odd}},\varphi_t,\mathcal{P}^{\text{even}}]) \\ &= [\mathcal{P}^{\text{odd}}|_R,(\varphi_t)|_S,\mathcal{P}^{\text{even}}|_R] \\ &= [(\mathcal{P}^{\bullet}|_R)^{\text{odd}},\varphi_{(t|_S)},(\mathcal{P}^{\bullet}|_R)^{\text{even}}] = \chi_{R,S}(\mathcal{P}^{\bullet}|_R,t|_S). \end{split}$$
Appendix B

Determinant functors

We recall the notion of determinant functors in our setting of interest and prove two simple facts about their duals. The original, more general definition of these functors is due to Knudsen and Mumford (cf. [KM76]), but our needs are limited to the case treated in [Wei13] p. 21 ff. Another good reference is [CKV93] p. 70 ff.

Let a commutative ring R be given which decomposes as a direct product of rings $R = \prod_{i=1}^{n} R_i$. If we denote the unit element of R_i by e_i , which we identify with its natural preimage in R, one has $1 = \sum_{i=1}^{n} e_i$ and each e_i is an idempotent of R. Every R-module M decomposes as $M = \bigoplus_{i=1}^{n} e_i M$, with $e_i M$ inheriting a natural R_i -module structure.

A prime ideal \mathfrak{p} of R is necessarily of the form $\mathfrak{p} = R_1 \times \cdots \times R_{i-1} \times \mathfrak{p}_i \times R_{i+1} \cdots \times R_n$ for some i and some prime ideal \mathfrak{p}_i of R_i , which yields an identification of $\operatorname{Spec}(R)$ with the disjoint union $\bigsqcup_{i=1}^n \operatorname{Spec}(R_i)$. For any R-module M, the localisation $M_{\mathfrak{p}}$ is canonically isomorphic to $(e_i M)_{\mathfrak{p}_i}$ as an $R_{\mathfrak{p}} \cong (R_i)_{\mathfrak{p}_i}$ -module.

Let R be an arbitrary commutative ring and P a finitely generated projective R-module. Then the rank function

rank_P: Spec(R)
$$\rightarrow \mathbb{N}$$

 $\mathfrak{p} \mapsto \operatorname{rank}_{R_\mathfrak{p}} P_\mathfrak{p}$

is locally constant (cf. [Wei13] p. 21) when the domain is endowed with the usual Zariski topology. By the compactness of $\operatorname{Spec}(R)$, there exists a decomposition $R = \prod_{i=1}^{n} R_i$ such that $\operatorname{rank}_{P|_{\operatorname{Spec}(R_i)}}$ is constant. We define the **determinant** of P as the R-module

$$\operatorname{Det}_{R}(P) = \bigoplus_{i=1}^{n} \bigwedge_{R_{i}}^{\operatorname{rank}_{P,i}} e_{i}P,$$

where e_i is as before, \bigwedge is the usual exterior power and $\operatorname{rank}_{P,i}$ denotes the constant value of rank_P at $\operatorname{Spec}(R_i)$. Recall that, for any R_i -module M, the zero-th exterior power $\bigwedge_{R_i}^0 M$ is R_i by convention. Different valid decompositions of R result in isomorphic determinant modules, and

hence $\text{Det}_R(P)$ is well defined. A concise way to rephrase this is: $\text{Det}_R(P)$ is locally defined as the highest exterior power. In order to avoid sign issues, one should regard $\text{Det}_R(P)$ as a graded invertible module (see the comment at the beginning of [BKS16] section 3.2) with the grading being given by the rank function. We omit this grading from the notation at no risk of ambiguity, as it is univocally determined by the module P itself.

Since the r-th power of a rank-r free module is free of rank 1 for any $r \in \mathbb{N}$, it is easy to verify that $\text{Det}_R(P)$ is projective and locally free of rank 1 - in other words, an invertible R-module. This implies that, letting $\text{Det}_R(P)^{-1} = \text{Hom}_R(\text{Det}_R(P), R)$, the R-linear evaluation map

$$ev \colon \operatorname{Det}_R(P) \otimes_R \operatorname{Det}_R(P)^{-1} \to R$$

 $m \otimes f \mapsto f(m)$

is an isomorphism.

Suppose now that P and Q are two finitely generated projective R-modules which are locally of the same rank (an important example is the case $P \cong Q$), and choose a decomposition $R = \prod_{i=1}^{n} R_i$ such that rank_P and rank_Q coincide on Spec(R_i) for all i. Let P_i denote e_iP , and analogously for Q_i . Then, given an R-homomorphism $f: P \to Q$, which we identify with the sum $f = \bigoplus_{i=1}^{n} f_i$ of the R-(or R_i -)homomorphisms $f_i: P_i \to Q_i$, we define

$$\operatorname{Det}(f_i) \colon \bigwedge_{R_i}^{\operatorname{rank}_{P,i}} P_i \to \bigwedge_{Q_i}^{\operatorname{rank}_{Q,i}} Q_i$$
$$\bigwedge_{j=1}^{\operatorname{rank}_{P,i}} m_j \mapsto \bigwedge_{j=1}^{\operatorname{rank}_{Q,i}} f_i(m_j)$$

(extended by R-linearity) and

$$\operatorname{Det}_R(f) = \bigoplus_{i=1}^n \operatorname{Det}(f_i) \colon \operatorname{Det}(Q) \to \operatorname{Det}(P).$$
 (B.1)

This construction satisfies the following:

- i) $\text{Det}_R(f)$ is a homomorphism of *R*-modules and it is independent of the chosen decomposition of *R*.
- ii) It is functorial: $\text{Det}_R(\text{Id}) = \text{Id}$ and $\text{Det}_R(g \circ f) = \text{Det}_R(g) \circ \text{Det}_R(f)$. In particular, it sends isomorphisms to isomorphisms and inverses to inverses.

The motivation behind the name of determinant functors is the following: suppose P is a free Rmodule of finite rank r and fix a basis m^1, \ldots, m^r . In particular, the map rank P is identically r on
all of Spec(R) and

$$\mu = \bigwedge_{j=1}^{r} m^j \tag{B.2}$$

is a generator of $\text{Det}_R(P) = \bigwedge_R^r P$. It can then be shown that, for any endomorphism $f \in \text{End}_R(P)$, the diagram

$$\begin{array}{cccc}
R & & \xrightarrow{\operatorname{det}(M_f)} & R \\
\downarrow \mu & & \downarrow \mu \\
\operatorname{Det}_R(P) & \xrightarrow{\operatorname{Det}_R(f)} & \operatorname{Det}_R(P)
\end{array} \tag{B.3}$$

commutes, where μ denotes the isomorphism $s \mapsto s\mu$ and $\det(M_f)$ is multiplication by the determinant of the matrix of f in any basis (cf. [Wei13] p. 21). Note that the vertical arrows can a fortiori be replaced by any arbitrary R-homomorphism.

Determinant functors behave well with respect to extension of scalars: a ring homomorphism $R \to S$ (with R and S commutative) induces a natural isomorphism $S \otimes_R \text{Det}_R(P) \cong \text{Det}_S(S \otimes_R P)$. They are furthermore additive in short exact sequences in the sense that $P' \stackrel{\iota}{\hookrightarrow} P \twoheadrightarrow P''$ yields an isomorphism

$$\operatorname{Det}_R(P') \otimes_R \operatorname{Det}_R(P'') \cong \operatorname{Det}_R(P)$$
 (B.4)

which is locally defined by mapping $(\bigwedge_{i=1}^{n} x_i) \otimes (\bigwedge_{j=1}^{m} y_j)$ to $(\bigwedge_{i=1}^{n} \iota(x_i)) \wedge (\bigwedge_{j=1}^{m} \sigma(y_j))$, where σ is an arbitrarily chosen splitting of $P \to P''$.

The above notion can be extended to perfect complexes as follows: given a strictly perfect complex \mathcal{P}^{\bullet} of *R*-modules, the **determinant of** \mathcal{P}^{\bullet} is the module

$$\operatorname{Det}_{R}(\mathcal{P}^{\bullet}) = \bigotimes_{i \in \mathbb{Z}} \operatorname{Det}_{R}(\mathcal{P}^{i})^{(-1)^{i}} = \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{even}}) \otimes_{R} \operatorname{Det}_{R}(\mathcal{P}^{\operatorname{odd}})^{-1},$$

where the tensor product in the middle term is over R. If \mathcal{C}^{\bullet} is now a perfect complex of R-modules, Det_R(\mathcal{C}^{\bullet}) is defined as Det_R(\mathcal{P}^{\bullet}) for a strictly perfect representative \mathcal{P}^{\bullet} of \mathcal{C}^{\bullet} . This is in fact well defined (up to canonical isomorphism) regardless of the choice of \mathcal{P}^{\bullet} (cf. [CKV93] p.71).

This behaves well with respect to derived extension of scalars: consider \mathcal{C}^{\bullet} and \mathcal{P}^{\bullet} as above and a homomorphism of commutative rings $R \to S$. Then $S \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet}$ is represented by $S \otimes_R^{\mathbb{L}} \mathcal{P}^{\bullet}$, which is simply the result of applying $S \otimes_R -$ to \mathcal{P}^{\bullet} degree-wise. Therefore, extension of scalars on determinant modules induces a natural map $\operatorname{Det}_R(\mathcal{C}^{\bullet}) \to \operatorname{Det}_S(S \otimes_R^{\mathbb{L}} \mathcal{C}^{\bullet})$.

We conclude this appendix with two small lemmas used in section 4.1:

Lemma B.0.1. Let R be a commutative ring and $M = \langle m \rangle$ a cyclic invertible R module. Then there exists a unique $m^* \in M^{-1} = \operatorname{Hom}_R(M, R)$ such that $m^*(m) = 1$. Furthermore, any $f \in M^{-1}$ satisfies $f = f(m)m^*$, and therefore m^* generates M^{-1} as an R-module.

Proof. By the invertibility of M, the R-module homomorphism $ev: M \otimes_R M^{-1} \to R$ which sends $x \otimes f$ to f(x) is an isomorphism. Let $\sum_{i=1}^n x_i \otimes f_i$ be the (unique) preimage of $1 \in R$ via ev. Since m is a generator of M, for each i there exists a scalar $r_i \in R$ such that $r_i m = x_i$. Hence $\sum_{i=1}^n x_i \otimes f_i = m \otimes \sum_{i=1}^n r_i f_i$. In particular,

$$1 = ev\left(\sum_{i=1}^{n} x_i \otimes f_i\right) = ev\left(m \otimes \sum_{i=1}^{n} r_i f_i\right) = \left(\sum_{i=1}^{n} r_i f_i\right)(m).$$

This shows existence of m^* . As for uniqueness, assume $f, g \in \text{Hom}_R(M, R)$ satisfy f(m) = g(m) = 1. Given any $x \in M$, there exists a scalar $r \in R$ with rm = x, and therefore f(x) = rf(m) = r = g(x). Hence, f = g.

For the last claim, we note that $f(m) = (f(m)m^*)(m)$ and, by the same argument as above, this implies $f = f(m)m^*$.

In view of this lemma, the choice of a generator m of an invertible module M (if it exists) yields an R-isomorphism

$$\iota_m \colon M \xrightarrow{\sim} M^{-1} \tag{B.5}$$
$$m \mapsto m^*.$$

Note that ι_m does depend on the choice of m: if m' is a different generator, say m' = rm for some $r \in R^*$, then an immediate computation shows that $\iota_m = r^2 \iota_{m'}$.

Any invertible R-module M is automatically reflexive. That is, the canonical R-homomorphism

$$M \to (M^{-1})^{-1} = \operatorname{Hom}_R(\operatorname{Hom}_R(M, R), R)$$

 $x \mapsto [f \mapsto f(x)]$

is an isomorphism. This follows from the fact that it holds for free modules, and hence in particular for the localisation of M at any prime ideal; and a map which becomes an isomorphism after localisation at every maximal ideal is an isomorphism globally too (cf. [Eis95] corollary 2.8). We often identify M and $(M^{-1})^{-1}$, in light of which one has $(m^*)^* = m$ and $\iota_{m^*} = \iota_m^{-1}$ for any generator m of M (if it exists).

The following application of this duality is relevant to our endeavours:

Lemma B.0.2. Let R be a commutative ring and $f: M \xrightarrow{\sim} N$ an isomorphism of finitely generated projective R-modules (which are therefore locally of the same rank). Suppose $\text{Det}_R(M)$ and $\text{Det}_R(N)$ are generated over R by m and n, respectively. Consider the homomorphisms α and β defined by the rows of

$$\alpha \colon \operatorname{Det}_{R}(M) \otimes_{R} \operatorname{Det}_{R}(N)^{-1} \xrightarrow{\operatorname{Det}_{R}(f) \otimes \operatorname{Id}} \operatorname{Det}_{R}(N) \otimes_{R} \operatorname{Det}_{R}(N)^{-1} \xrightarrow{ev} R$$

$$\downarrow^{i_{m} \otimes i_{n^{*}}} \qquad (B.6)$$

$$\beta \colon \operatorname{Det}_{R}(M)^{-1} \otimes_{R} \operatorname{Det}_{R}(N) \xrightarrow{\operatorname{Id} \otimes \operatorname{Det}_{R}(f^{-1})} \operatorname{Det}_{R}(M)^{-1} \otimes_{R} \operatorname{Det}_{R}(M) \xrightarrow{ev} R$$

Then $\beta((i_m \otimes i_{n^*})(m \otimes n^*)) = \alpha(m \otimes n^*)^{-1} \in \mathbb{R}^*.$

Proof. This is a straightforward computation. Since α is an *R*-isomorphism, it maps the generator $m \otimes n^*$ of $\text{Det}_R(M) \otimes_R \text{Det}_R(N)^{-1}$ to a unit $\alpha(m \otimes n^*) \in R^*$. By definition, one has

$$(i_m \otimes i_{n^*})(m \otimes n^*) = m^* \otimes n \in \operatorname{Det}_R(M)^{-1} \otimes_R \operatorname{Det}_R(N)$$

(note that we are implicitly using the duality $\iota_{n^*} = \iota_n^{-1}$ explained above).

The element n generates $\text{Det}_R(N)$, and hence there exists an $r \in R$ such that $\text{Det}_R(f)(m) = rn$ In particular, $\text{Det}_R(f^{-1})(n) = \text{Det}_R(f)^{-1}(n) = r^{-1}m$. Thus, on one hand,

$$\alpha(m \otimes n^*) = ev(\operatorname{Det}_R(f)(m) \otimes n^*) = r \cdot ev(n \otimes n^*) = r;$$

and on the other,

$$\beta((i_m \otimes i_{n^*})(m \otimes n^*)) = \beta(m^* \otimes n) = ev(m^* \otimes \operatorname{Det}_R(f^{-1})(n)) = r^{-1} \cdot ev(m^* \otimes m) = r^{-1},$$

as desired.

All arrows in (B.6) are isomorphisms, and therefore the diagram can be completed uniquely into a commutative square through a vertical arrow $R \to R$. The lemma shows this arrow is multiplication by r^{-2} , where $r = \alpha(m \otimes n^*) \in R^*$.

Bibliography

- [Bar78] Daniel Barsky. "Fonctions zeta p-adiques d'une classe de rayon des corps de nombres totalement réels". In: Groupe d'Étude d'Analyse Ultramétrique (5e année: 1977/78). Secrétariat Math., Paris, 1978, Exp. No. 16, 23.
- [BB05] Manuel Breuning and David Burns. "Additivity of Euler characteristics in relative algebraic K-groups". In: Homology Homotopy Appl. 7.3 (2005), pp. 11–36.
- [BF01] D. Burns and M. Flach. "Tamagawa numbers for motives with (non-commutative) coefficients". In: *Doc. Math.* 6 (2001), pp. 501–570.
- [BF98] David Burns and Matthias Flach. "On Galois structure invariants associated to Tate motives". In: Amer. J. Math. 120.6 (1998), pp. 1343–1397.
- [BG03] D. Burns and C. Greither. "On the equivariant Tamagawa number conjecture for Tate motives". In: *Invent. Math.* 153.2 (2003), pp. 303–359.
- [BH21] Dominik Bullach and Martin Hofer. The equivariant Tamagawa Number Conjecture for abelian extensions of imaginary quadratic fields. 2021. DOI: 10.48550/ARXIV.2102.
 01545. URL: https://arxiv.org/abs/2102.01545.
- [BKS16] David Burns, Masato Kurihara, and Takamichi Sano. "On zeta elements for \mathbb{G}_m ". In: Doc. Math. 21 (2016), pp. 555–626.
- [BKS17] David Burns, Masato Kurihara, and Takamichi Sano. "On Iwasawa theory, zeta elements for \mathbb{G}_m , and the equivariant Tamagawa number conjecture". In: Algebra & Number Theory 11.7 (2017), pp. 1527–1571.
- [Ble06] W. Bley. "On the equivariant Tamagawa number conjecture for Abelian extensions of a quadratic imaginary field". In: *Doc. Math.* 11 (2006), pp. 73–118.
- [Bur11] David Burns. "On derivatives of Artin *L*-series". In: *Invent. Math.* 186.2 (2011), pp. 291– 371.

- [Cas79] Pierrette Cassou-Noguès. "Valeurs aux entiers négatifs des fonctions zêta et fonctions zêta *p*-adiques". In: *Invent. Math.* 51.1 (1979), pp. 29–59.
- [CKV93] J.-L. Colliot-Thélène, K. Kato, and P. Vojta. Arithmetic algebraic geometry. Vol. 1553. Lecture Notes in Mathematics. Lectures from the Second C.I.M.E. Session held in Trento, June 24–July 2, 1991, Edited by E. Ballico. Springer-Verlag, Berlin, 1993, pp. vi+223.
- [CR81] Charles W. Curtis and Irving Reiner. Methods of representation theory. Vol. I. Pure and Applied Mathematics. John Wiley & Sons, Inc., New York, 1981.
- [CR87] Charles W. Curtis and Irving Reiner. *Methods of representation theory. Vol. II.* Pure and Applied Mathematics (New York). John Wiley & Sons, Inc., New York, 1987.
- [DK20] Samit Dasgupta and Mahesh Kakde. On the Brumer-Stark Conjecture. To appear in: Ann. of Math. 2020. DOI: 10.48550/ARXIV.2010.00657. URL: https://arxiv.org/ abs/2010.00657.
- [DK21] Samit Dasgupta and Mahesh Kakde. Brumer-Stark Units and Hilbert's 12th Problem.
 2021. DOI: 10.48550/ARXIV.2103.02516. URL: https://arxiv.org/abs/2103.02516.
- [DR80] Pierre Deligne and Kenneth A. Ribet. "Values of abelian *L*-functions at negative integers over totally real fields". In: *Invent. Math.* 59.3 (1980), pp. 227–286.
- [Eis95] David Eisenbud. Commutative Algebra. with a View Toward Algebraic Geometry. Vol. 150.Graduate Texts in Mathematics. Springer-Verlag New York, 1995.
- [EN22] Nils Ellerbrock and Andreas Nickel. Integrality of Stickelberger elements and annihilation of natural Galois modules. 2022. DOI: 10.48550/ARXIV.2203.12945. URL: https: //arxiv.org/abs/2203.12945.
- [FK06] Takako Fukaya and Kazuya Kato. "A formulation of conjectures on p-adic zeta functions in non-commutative Iwasawa theory". In: Proceedings of the St. Petersburg Mathematical Society. Vol. XII. Vol. 219. Amer. Math. Soc. Transl. Ser. 2. Amer. Math. Soc., Providence, RI, 2006, pp. 1–85.
- [FW79] Bruce Ferrero and Lawrence C. Washington. "The Iwasawa invariant μ_p vanishes for abelian number fields". In: Ann. of Math. (2) 109.2 (1979), pp. 377–395.
- [Gre83] Ralph Greenberg. "On *p*-adic Artin *L*-functions". In: Nagoya Math. J. 89 (1983), pp. 77– 87.

[Jan89]	Uwe Jannsen. "Iwasawa modules up to isomorphism". In: Algebraic number theory. Vol. 17. Adv. Stud. Pure Math. Academic Press, Boston, MA, 1989, pp. 171–207.
[JN19]	Henri Johnston and Andreas Nickel. "On the non-abelian Brumer-Stark conjecture and the equivariant Iwasawa main conjecture". In: <i>Math. Z.</i> 292.3-4 (2019), pp. 1233–1267.
[JN20]	Henri Johnston and Andreas Nickel. An unconditional proof of the abelian equivariant Iwasawa main conjecture and applications. 2020. DOI: 10.48550/ARXIV.2010.03186. URL: https://arxiv.org/abs/2010.03186.
[Kak13]	Mahesh Kakde. "The main conjecture of Iwasawa theory for totally real fields". In: <i>Invent. Math.</i> 193.3 (2013), pp. 539–626.
[KM76]	Finn Faye Knudsen and David Mumford. "The projectivity of the moduli space of stable curves. I. Preliminaries on "det" and "Div"". In: <i>Math. Scand.</i> 39.1 (1976), pp. 19–55.
[Kuz72]	L. V. Kuz'min. "The Tate module of algebraic number fields". In: <i>Izv. Akad. Nauk SSSR Ser. Mat.</i> 36 (1972), pp. 267–327.
[Lim12]	Meng Fai Lim. "Poitou-Tate duality over extensions of global fields". In: J. Number Theory 132.11 (2012), pp. 2636–2672.
[LMFDB]	The LMFDB Collaboration. The L-functions and Modular Forms Database. http://www.lmfdb.org. 2021.
[LS13]	Meng Fai Lim and Romyar T. Sharifi. "Nekovář duality over <i>p</i> -adic Lie extensions of global fields". In: <i>Doc. Math.</i> 18 (2013), pp. 621–678.
[Mer94]	Alexander S. Merkurjev. "Algebraic K-theory and Galois cohomology". In: First European Congress of Mathematics, Vol. II (Paris, 1992). Vol. 120. Progr. Math. Birkhäuser, Basel, 1994, pp. 243–255.
[MN15]	A. Movahhedi and T. Nguyen Quang Do. "On universal norms and the first layers of \mathbb{Z}_p -extensions of a number field". In: <i>Math. Ann.</i> 362.3-4 (2015), pp. 817–838.
[MW84]	B. Mazur and A. Wiles. "Class fields of abelian extensions of Q". In: Invent. Math. 76.2 (1984), pp. 179–330.
[Nek06]	Jan Nekovář. "Selmer complexes". In: Astérisque 310 (2006), pp. viii+559.
[Neu99]	Jürgen Neukirch. Algebraic number theory. Vol. 322. Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin, 1999.

[Nic10]	Andreas Nickel. "Non-commutative Fitting invariants and annihilation of class groups". In: J. Algebra 323.10 (2010), pp. 2756–2778.
[Nic13]	Andreas Nickel. "Equivariant Iwasawa theory and non-abelian Stark-type conjectures". In: <i>Proc. Lond. Math. Soc. (3)</i> 106.6 (2013), pp. 1223–1247.
[Nic14]	Andreas Nickel. "A conductor formula for completed group algebras". In: <i>Doc. Math.</i> 19 (2014), pp. 601–627.
[Nic20]	Andreas Nickel. "A generalization of a theorem of Swan with applications to Iwasawa theory". In: <i>Canad. J. Math.</i> 72.3 (2020), pp. 656–675.
[NSW20]	Jürgen Neukirch, Alexander Schmidt, and Kay Wingberg. <i>Cohomology of number fields</i> . Second edition, corrected version 2.3. Springer-Verlag, 2020.
[PRS11]	Cristian Popescu, Karl Rubin, and Alice Silverberg, eds. Arithmetic of L-functions. Vol. 18. IAS/Park City Mathematics Series. Lectures from the Graduate Summer School held in Park City, UT, June 29–July 17, 2009. American Mathematical Society, Provi- dence, RI; Institute for Advanced Study (IAS), Princeton, NJ, 2011.
[Rub91]	Karl Rubin. "The "main conjectures" of Iwasawa theory for imaginary quadratic fields". In: <i>Invent. Math.</i> 103.1 (1991), pp. 25–68.
[RW04]	Jürgen Ritter and Alfred Weiss. "Toward equivariant Iwasawa theory. II". In: Indag. Math. (N.S.) 15.4 (2004), pp. 549–572.
[RW11]	Jürgen Ritter and Alfred Weiss. "On the "main conjecture" of equivariant Iwasawa theory". In: J. Amer. Math. Soc. 24.4 (2011), pp. 1015–1050.
[RW96]	Jürgen Ritter and Alfred Weiss. "A Tate sequence for global units". In: <i>Compositio Math.</i> 102.2 (1996), pp. 147–178.
[Sha]	Romyar Sharifi. "Iwasawa theory". Source: https://www.math.ucla.edu/~sharifi/ iwasawa.pdf. Online notes.
[Sie70]	Carl Ludwig Siegel. "Über die Fourierschen Koeffizienten von Modulformen". In: Nachr. Akad. Wiss. Göttingen MathPhys. Kl. II 1970 (1970), pp. 15–56.
[Swa68]	Richard G. Swan. <i>Algebraic K-theory</i> . Vol. 76. Lecture Notes in Mathematics. Springer-Verlag, Berlin-New York, 1968.
[Tat67]	John T. Tate. " <i>p</i> -divisible groups". In: <i>Proc. Conf. Local Fields (Driebergen, 1966)</i> . Springer, Berlin, 1967, pp. 158–183.

[Tat84]	John T. Tate. Les conjectures de Stark sur les fonctions L d'Artin en $s = 0$. Vol. 47. Progress in Mathematics. Birkhäuser Boston, Inc., Boston, MA, 1984.
[Ven]	Otmar Venjakob. "Characteristic elements in noncommutative Iwasawa theory". Source: https://www.mathi.uni-heidelberg.de/~venjakob/papervenjakob/habil.pdf. Habilitation thesis.
[Ven13]	Otmar Venjakob. "On the work of Ritter and Weiss in comparison with Kakde's approach". In: <i>Noncommutative Iwasawa main conjectures over totally real fields</i> . Vol. 29. Springer Proc. Math. Stat. Springer, Heidelberg, 2013, pp. 159–182.
[Was]	Lawrence C. Washington. <i>Introduction to cyclotomic fields</i> . Second. Vol. 83. Graduate Texts in Mathematics. Springer-Verlag, New York.
[Wei13]	Charles A. Weibel. <i>The K-book.</i> Vol. 145. Graduate Studies in Mathematics. An intro- duction to algebraic <i>K</i> -theory. American Mathematical Society, Providence, RI, 2013.
[Wil90]	A. Wiles. "The Iwasawa conjecture for totally real fields". In: Ann. of Math. (2) 131.3 (1990), pp. 493–540.
[Wit13]	Malte Witte, "On a localisation sequence for the K-theory of skew power series rings".

[Wit13] Malte Witte. "On a localisation sequence for the K-theory of skew power series rings". In: J. K-Theory 11.1 (2013), pp. 125–154.

Index

Artin character, 52 Artin L-series, 14 augmentation map, 15augmentation ideal, 3 augmentation map, 3boundary homomorphism, 19 character, 9Artin, 52 cyclotomic, 6, 154 of type W, 53 realisation, 13 Teichmüller, 155 totally even, 153totally odd, 153 vritual, 11 W-equivalent, 53 coinvariants, 3compact induction, 5complete tensor product, 5complex global, 28 local, 31main, 33perfect, 33 strictly perfect, 33 connecting homomorphism, 19 cyclotomic character, 6, 154determinant functor

on homomorphisms, 170 on modules, 169

on perfect complexes, 171 Dirichlet regulator map, 15 p-adic, 65 Dirichlet's unit theorem, 15 elementary finite group, 11 equivariant Main Conjecture, 78 of Burns, Kurihara and Sano, 141 of Ritter and Weiss, 158 on minus parts, 159 with uniqueness, 78 Euler factor, 14 exact sequence of K-theory, 19 finite-level map, 60Frobenius reciprocity, 11 global complex, 28integral trivialisation, 51 Interpolation Conjecture, 76 invariants, 3 invariants-coinvariants exact sequence, 8 Iwasawa algebra, 3 Iwasawa module, 3 elementary, 8of coinvariants, 3of invariants, 3pseudo-null, 7 K-group, 17–19 kernel condition, 62

L-series, 14 leading coefficient, 16 local complex, 31

```
localisation sequence, 20
main complex, 33
Main Conjecture, see equivariant Main Conjec-
        ture
minus part, 154
number field, 2
p-adic L-function, 155
p-elementary
    finite group, 11
    one-dimensional p-adic Lie group, 130
p-adic field, 2
perfect complex, 33
plus part, 154
Pontryagin duality, 4
primitive central idempotent
    e(\chi), 12
    e_{\chi}, 69
pseudo-isomorphism, 7
realisation of a character, 13
reduced norm, 21
refined Euler characteristic, 49, 51
regulated special value, 63
regulator, 66
representation, 9
series quotient, 50
Stark's conjecture, 82
Stark-Tate regulator, 66
strictly perfect complex, 33
Tate twist, 6
trivialisation, 49, 51
    integral, 51
twisted evaluation map, 72
type-W character, 53
universal norm, 59
weak Leopoldt conjecture, 26
Weierstrass polynomial, 7
```